

Provided by the author(s) and University of Galway in accordance with publisher policies. Please cite the published version when available.

Title	Brewery spent grain cascade process: protein recovery and improved downstream fermentation for simultaneous volatile fatty acid production and recovery
Author(s)	Castilla-Archilla, Juan
Publication Date	2022-03-23
Publisher	NUI Galway
Item record	http://hdl.handle.net/10379/17056

Downloaded 2024-04-28T16:54:13Z

Some rights reserved. For more information, please see the item record link above.





Brewery spent grain cascade process: protein recovery and improved downstream fermentation for simultaneous volatile fatty acid production and recovery

A thesis submitted to the National University of Ireland as fulfilment of the requirements for the degree of Doctor of Philosophy

by

Juan Castilla-Archilla

Research Supervisor: Prof. Piet N. L. Lens

School of Natural Sciences

November 2021

Declaration of authorship

I, Juan Castilla-Archilla, declare that this thesis has been generated by me and presents my original work and results unless stated otherwise. This work was not been used to obtain any other degree at the National University of Ireland, or elsewhere.

Juan Castilla-Archilla

For my wife, Mónica, who decided to share this journey of life with me. My daughter, Alejandra, I hope to be able to teach you all the good things I can learn from this life.

For my parents, Juan and Amalia, who are still teaching me to work hard and never give up without forgetting the important things of life. My sister, Ana, I have the luck to be sharing this life with you since my first memory.

&

For my grandparents, Miguel and Ana, who witnessed how I started my PhD and I'm sure you are proud of me. Your kindness and love fill the heart of everyone who was lucky to be around you. My grandmother Rosa, who was unique and strong despite the difficulties of life.

I could not have become who I am without any of you

"Lo mejor será que bailemos" "¿y que nos juzguen por locos sr conejo?" "¿usted conoce algún cuerdo feliz?" – Bailemos pues

Alone, we go faster. Together, we go further

Acknowledgments

First of all I would like to express my sincere gratitude to my promoter Prof Piet N. L. Lens for providing the opportunity to develop a project together under his supervision. You gave me a lot of freedom to develop this project but also valuable guidance and encouragement when needed. All the discussions and advice allowed me to evolve and become a better researcher.

I have many people to thank for what was supposed to be a year of adventure in Ireland, and it has turned already seven years. Dr Dermot Hughes, to inspire me and give me courage to persuade a PhD, but also to share many of your thoughts and life experience. Prof Vincent O'Flaherty and Therese Mahony, to trust me and give me such a wonderful opportunities during the DPTC, working with many industries and partners.

I will always be grateful for the opportunity moving to Ireland in 2015 to work with Anna Piterinna, Owenroe Lemass, Pádraic Ó hUiginn and Bart Bonsall in the TCBB Resource. All the people I met back then in the MEL lab, Ciara, Corine, Aoife... and all the people in the department Fabiana, Estefania, Octavi. Special thanks to my GRC: Dr Florence Abram, Dr Owen Clifford and Dr Maria Tuohy, all the meetings, comments and suggestions not only to improve this work but also to enjoy this journey are unevaluable. Further, I will never be able to show all the gratitude to all the staff member of the department, Ann Smith, Michael Coughlan, Maurice Martyn.

All the help given by Prof Gerald Wall and Violeta Sanchez i Nogue for helping me to develop the application for a visiting research to USA that it was not possible to carry out, I really hope to do it in the near future. I have to thank all the people who helped me to develop some part of my work, Dr Stefano Papirio, Dr Camilla Thorn, Dr Maria Tuohy and Dr Maria Cermeno. I have to show my deep gratitude to Dr Yifeng Zhang at DTU (Denmark) and all his group for hosting me, especially to Danfei Zheng. As well, to Dr Borja Valverde for helping me to be able to organise this visiting research. The help of Dublin Zoo, especially to the zookeepers and Sandra Molloy.

All the people from IETSBIO³, where I have been not only with colleague, but also with friends. Peyman, far all the talks and dinners. Yaxue, for all the cooking lessons. Rachel and Anna Patricia, for all your support and advice at the beginning of my PhD. Carlos, for all the good experiences shared together. George for the discussion about the possible future development of my project. Borja, Leah and Manu for all the assistance for analysis and ordering. Simon, thank you so much, writing the thesis in such a short period, it would have been impossible without your help. Also to Armando, Fede, Simone, Ana Maria, Sonia, James *et al.*, it was a good experience to share this time with all of you. Prof Maria Reis to become my promoter during a master thesis in the Universidade Nova de Lisboa (Lisbon, Portugal), and thanks to Ana, Tania Palmeiro and Nicola Frisson for helping to "survive" the first three months there. My former professors at the Universidad de Almería (Almeria, Spain) Jose Antonio Sánchez Pérez, Gabriel Acién, Emilio Molina, Jose María Fernández Sevilla, Jose Luis Casas and Mari Carmen Cerón for trying to teach me all about Chemical Engineering. Tania Mazzuca and Celeste Brindley to take care of the bunch of frighten students we were with 18 years to open any valve of a rectification column. Thank you so much to Mr Antonio Cambil, CEO of Sistemas de Calor S.L., he truly helped me to develop myself professionally and personally.

Last but not least, Rosalía Martínez, I didn't know where to put you as you have been part of many of these experiences in my life. This is finally done! Cannot wait to see you here too!

List of content

C	hapter 1	1 General introduction	1
	1.1	General introduction	2
	1.2	Biorefinery definition	4
	1.2.1	Plant refinery session	4
	1.2.2	International energy agency, working group bioenergy	4
	1.2.3	Green biorefinery	4
	1.2.4	Circular economy	5
	1.3	Potential feedstock for an Irish biorefinery	6
	1.3.1	Industrial sectors and current disposal of the biomass	7
	1.3.2	Biomass selection for this PhD thesis	8
	1.4	Thesis scope	9
	1.5	References1	0
C	hapter 2	2 Review and statement of the thesis1	4
	2.1	Chemical platform1	5
	2.1.1	Ethanol production1	5
	2.1.1.1	Market development in USA1	6
	2.1.1.2	Valorisation of agricultural surplus: Bazancourt-Pomacle ethanol and biorefinery1	6
	2.1.1.3	Disseminated ethanol production from current wastes: St1 Biofuels1	7
	2.1.1.4	Problems and failures during scaling-up1	7
	2.1.2	Bioplastics	9
	2.1.3	Succinic acid2	1
	2.1.4	Biogas production from complex/mixed wastes2	1
	2.2	Anaerobic digestion2	2
	2.2.1	Hydrolysis2	3
	2.2.2	Acidogenesis2	3
	2.2.3	Acetogenesis2	4
	2.2.4	Methanogenesis2	4
	2.3	Volatile fatty acids2	4
	2.3.1	Production approach2	5
	2.3.1.1	Simultaneous hydrolysis and acidogenesis2	6
	2.3.1.2	2 Two-step process	7
	2.3.1.3	Volatile fatty acid recovery2	8
	2.4	Lignocellulosic feedstock2	9
	2.5	Brewery spent grain	1

2.6	References	33
Chapter 3	Biodegradation of brewery spent grain by mixed microbial cultures from anaero	obic
sludge an	d animal dungs	49
Abstrac		50
3.1	Introduction	51
3.2	Material and methods	53
3.2.1	Feedstock	53
3.2.2	Source of inoculum	53
3.2.3	Biodegradation activity test	53
3.2.4	Chemical analysis	54
3.2.5	Calculations	55
3.2.6	Nucleic acid extractions and sequencing analysis	56
3.3	Results	56
3.3.1	Biodegradation performance	56
3.3.1.1	Granular sludge	56
3.3.1.2	Crushed granular sludge	58
3.3.1.3	Giraffe manure	59
3.3.1.4	Rumen fluid	60
3.3.1.5	Rhinoceros manure	61
3.3.1.6	Tiger manure	61
3.3.2	Microbial community analysis of the inocula	62
3.4	Discussion	63
3.4.1	Volatile solids destruction	63
3.4.2	Gas production	64
3.4.3	Volatile fatty acid production	66
3.4.4	Potential inhibition	66
3.5	Conclusions	69
3.6	References	69
Chapter 4	Thermal diluted acid hydrolysis of brewery spent grain and downstream hydrol	lysate
acidogeni	c fermentation	74
4.1	Introduction	76
4.2	Material and methods	78
4.2.1	Feedstock and inoculum	78
4.2.2	Diluted acid hydrolysis of brewery spent grain	78
4.2.3	Batch fermentation	79
4.2.4	Physico-chemical analysis	80

	4.2.5	Calculations	81
	4.3 I	Results	82
	4.3.1	Efficiency of diluted acid hydrolysis of brewery spent grain	82
	4.3.2	Acidogenic fermentation of brewery spent grain hydrolysate	84
	4.3.2.1	Initial conditions	84
	4.3.2.2	Acidogenic fermentation at uncontrolled pH	84
	4.3.2.3	Acidogenic fermentation at controlled pH (4.50, 5.0, 6.0 and 8.0)	86
	4.3.2.4	Nitrogen compounds, sulfate and soluble chemical oxygen demand	88
	4.4 [Discussion	88
	4.4.1	Operational conditions during the acidogenic fermentation	88
	4.4.2	Total carbohydrates and monosaccharides consumption	90
	4.4.3	Carbon compounds conversion	90
	4.4.4	Conversion of nitrogen and sulfur compounds	92
	4.4.5	Hydrolysis efficiency	93
	4.5 0	Conclusions	96
	4.6 I	References	97
C	hapter 5	Continuous volatile fatty acid production from brewery spent grain hydrolysate	using an
e	xpanded	granular sludge bed reactor	
	Abstrac	t	104
	5.1 I	Introduction	105
	5.2 I	Material and methods	106
	5.2.1	Feedstock and inoculum	106
	5.2.2	Experimental set-up	107
	5.2.3	Experimental design	108
	5.2.4	Analytical methods	109
	5.2.5	Calculations	109
	5.3 I	Results	110
	5.3.1	Total carbohydrates fermentation during the acidogenic fermentation	110
	5.3.2	Biomass behaviour	112
	5.3.2.1	Flocculent sludge (R1)	112
	5.3.2.2	Intact granular sludge	114
	5.3.3	Gas production	115
	5.3.4	Sulfate and nitrogen compounds	116
	5.4 I	Discussion	118
	5.4.1	Conversion of carbon compounds	

5.4.2 sludgo	Performance and selection pressure of the flocculent (R1) and whole gr	anular (R2)
5 d 2	Agregation and diaggregation of the biomass	
5.4.3	Aggregation and disaggregation of the biomass	
5.5		
5.0 Chamban (
Chapter 6	Brewery spent grain cascade process for recovery of carbonydrates, prot	21ns and lignin 128
Abstra	ct	
6.1	Introduction	
6.2	Material and methods	
6.2.1	Feedstock	
6.2.2	Hydrolysis process	
6.2.2.1	Sulfuric and hydrochloric acid comparison	
6.2.2.2	Pretreatment optimization	
6.2.2.3	Enzymatic hydrolysis	
6.2.2.4	Solid and liquid fractionation	
6.2.3	Physico-chemical analysis	
6.2.4	Calculation	
6.3	Results	
6.3.1	Comparison using hydrochloric and sulfuric acid	
6.3.2	Response surface methodology	
6.3.3	Small enzymatic hydrolysis	
6.3.4	Large enzymatic hydrolysis	141
6.4	Discussion	
6.4.1	Effect of pretreatment	
6.4.2	Enzymatic hydrolysis	
6.4.3	Solid fraction	
6.5	Conclusion	
6.6	References	
Chapter 7	Continuous volatile fatty acid production and recovery at high organic co	ncentration
using an u	upflow granular sludge bed reactor	155
7.1	Introduction	
7.2	Material and methods	158
7.2.1	Feedstock	158
7.2.1.1	Electrochemical cell characterisation with synthetic broth	158
7.2.1.2	Acidogenic fermentation	

7.2.2	Inoculum	158	
7.2.3	Experimental design and operation	158	
7.2.3.1	Electrochemical cell characterisation with synthetic broth	158	
7.2.3.2	Acidogenic fermentation and volatile acid recovery	159	
7.2.4	Analytical methods	161	
7.2.5	Calculations	162	
7.3 F	esults	162	
7.3.1	Electrochemical cell characterisation with synthetic broth	162	
7.3.2	Continuous volatile fatty acid production and recovery	165	
7.3.2.1	Reactor performance	165	
7.3.2.2	Electrochemical cell	168	
7.4 [Discussion	170	
7.4.1	Fermentation performance	170	
7.4.2	Volatile fatty acids	172	
7.5 0	onclusion	174	
7.6 F	eferences	175	
Chapter 8	General discussion and future perspectives	178	
8.1 k	ey findings	179	
8.2 0	ore findings and future recommendations		
8.3 F	uture developments		
8.3.1	Brewery spent grain biorefinery		
8.3.2	Simultaneous volatile fatty acid production and recovery		
8.4 F	eferences		
Suppleme	ntary information Chapter 3		
S3.1 M	Nethods		
S3.1. 1	Nucleic acid extractions		
\$3.1.2	Sequence analysis		
S3.2 M	Aicrobial community analyses of the inocula	190	
S3.2. 1	Comparison of the microbial composition of the inocula	190	
\$3.2.2	Presence vs activity of OTUs in each inocula (Day 21)	192	
\$3.2.3	S3.2.3 Differences in microbial community structure and relationships with biodegradation		
data.		193	
\$3.3. 1	Volatile solids destruction	194	
S3.3 F	eferences	195	

List of figures

Figure 2.2 Anaerobic carbon degradation of complex wastes until CH₄ and CO₂. 1: hydrolytic microorganisms, 2: primary fermenting bacteria, 3: primary fermenting bacteria, 4: secondary-fermenting (syntrophic) bacteria, 5: homoacetogenic bacteria: 6: acetate-cleaving methanogens; 7: hydrogen-oxidizing methanogens (adapted from: Schink, 1997; Kleerebezem et al., 2015)......23

Figure 3.4 Volatile fatty acid profile and total soluble chemical oxygen demand. a) Granular sludge, b) Crushed granular sludge, c) Giraffe manure, d) rumen fluid, e) rhino manure, and f) tiger manure ..60

Figure 5.3 Total concentration of the fermentation products (acetic acid, propionic, butyric acid and ethanol) for R1 and R2 during the continuous acidogenic fermentation of brewery spent grain hydrolysate in expanded granular sludge bed reactors. Note that R2 started 15 days after R1. The changes for the operational conditions were applied simultaneously for both reactors (see **Table 5.1**)

Supplementary Figure 1 Relative abundances of OTUs assigned to the level of phylum, in each of the 6 inocula used for BSG degradation. Nine bars are shown for each inocula, where the first three are

List of tables

Table 1.1 Biorefinery and biorefining definition from different authors broadly used in the actual industry6 Table 3.1 pH value for all the different conditions at the beginning, lowest value achieved and end of Table 3.2 Main parameters for the solid biodegradation activity test for each of the different Table 3.3 Relative abundances of archaea in the DNA fraction (days 0 and 21) and in the cDNA fraction Table 4.1 Operating conditions used for the hydrolysis of brewery spent grain as well as the composition of carbohydrates extracted and efficiency of the hydrolysis process expressed as Table 4.2 Total carbohydrates efficiency recovery, gram of total carbohydrates per gram of brewery spent grain used (dried weight), for the thermal diluted acid hydrolysis of brewery spent grain carried out at 121°C for 20 minutes under different sulphuric acid and brewery spent grain concentrations84 Table 4.3 Glucose, xylose and arabinose concentration during the acidogenic fermentation along the

Table 5.1 Hydraulic retention time and organic loading rate for each of the different phases during thecontinuous acidogenic fermentation of brewery spent grain hydrolysate using in an expanded granularsludge bed reactor109

 Table 5.2 Optimal conditions for volatile fatty acids production in different high rate reactors119

Table 6.1 Parameters used in the Response Surface Methodology for the optimization of the thermaldiluted hydrochloric acid hydrolysis as pretreatment prior to enzyme addition132

Table 6.3 Increase in carbohydrate concentration after 24 hours of enzymatic hydrolysis for fourdifferent pretreatment conditions138

Table 6.5 Liquid characterisation for each of the fractions produced during the fractionation anddifferent streams for brewery spent grain cascade process144

Table 6.6 Solid fraction characterisation during the fractionation and different streams for breweryspent grain cascade process144

Table 7.1 Hydraulic retention time and glucose feeding concentration for each of the phases, with andwithout the electrochemical cell coupled to the recycling line of the reactor. HRT: hydraulic retentiontime; EC: electrochemical cell161

Table 7.3 Main average parameters during the simultaneous production and recovery of the volatile

 fatty acids using an expanded granular sludge bed reactor coupled with an electrochemical cell ... 167

List of publications and chapter contributions

Publication	Author contribution
Chapter 2: Castilla-Archilla J., O'Flaherty V., Lens P. N. L. (2019) "Biorefineries: Industrial Innovation and Tendencies" In: JR. Bastidas-Oyanedel, J. S. Schmidt, (Eds.), Biorefinery: Integrated Sustainable Processes for Biomass Conversion to Biomaterials, Biofuels, and Fertilizers, Springer, Switzerland, pp 3-35	Castilla-Archilla J. : conceptualization, design, literature revision, writing - original draft and manuscript. O'Flaherty V. : writing - review & editing. Lens P. N. L. : funding acquisition, supervision, writing - review & editing
Chapter 3: Castilla-Archilla J., Thorn C., Pau S., Lens P. N. L. (2022) "Screening for suitable mixed microbial consortia from anaerobic sludge and animal dungs for biodegradation of brewery spent grain" Submitted to Biomass and Bioenergy (major revisions)	Castilla-Archilla J. : conceptualization, methodology, investigation, interpretation, writing – original draft. Thorn C. : methodology, interpretation, writing – review & editing. Pau S. : investigation. Lens P. N. L. : funding acquisition, methodology, supervision, writing - review & editing.
*The work in this chapter formed part of a collaboration where any microbial analysis work (lab work, bioinformatics and data representation) were completed by Dr Camilla Thorn	
Chapter 4: Castilla-Archilla, J., Papirio, S., and Lens, P. N. L. (2021) "Two step process for volatile fatty acid production from brewery spent grain: hydrolysis and direct acidogenic fermentation using anaerobic granular sludge" Process Biochemistry 100, 272–283. doi: 10.1016/j.procbio.2020.10.011	Castilla-Archilla J. : conceptualization, methodology, investigation, interpretation, writing - original draft. Papirio S. : methodology, validation, supervision, writing - review & editing. Lens P. N. L. : funding acquisition, methodology, supervision, writing - review & editing.
Chapter 5: Castilla-Archilla J., Heiberger J., Mills S., Hilbig J., Collins G., Lens P. N. L. (2021) "Continuous Volatile Fatty Acid Production From Acid Brewery Spent Grain Leachate in Expanded Granular Sludge Bed Reactors" Frontiers in Sustainable Food Systems 5, 129. DOI=10.3389/fsufs.2021.664944	Castilla-Archilla J.: conceptualization, methodology, investigation, interpretation and writing – original draft and manuscript. Heiberger J. : investigation and interpretation. Mills S. : methodology, interpretation and writing - draft review. Hilbig J. : investigation. Collins G.: methodology and writing - supervision.

xiii

Lens P. N. L.: Funding acquisition, Methodology,

Supervision, Writing - review & editing.

Abstract

This PhD thesis aimed to develop sustainable volatile fatty acid (VFA) production using brewery spent grain (BSG) as lignocellulosic feedstock, with an overall production in Ireland above 170,000 tonnes. VFAs were produced by using partially inhibited anaerobic digestion (AD) to avoid the decomposition of fermented compounds into biogas. AD is the conversion of organic matter to biogas by four different steps: hydrolysis, acidogenesis, acetogenesis and methanogenesis. VFAs are produced during the acidogenesis and acetogenesis step. Thus, the control of the operational conditions to avoid methanogenesis leads to the accumulation of VFAs. Firstly, two different approaches were followed for VFA production: i) the simultaneous hydrolysis and fermentation, and ii) a two-step process using a pretreatment hydrolysis to obtain a liquid rich in carbohydrates for downstream fermentation as a second step. A second part was developed to recover additional added value compounds (other than VFAs, i.e. protein) from the BSG during the hydrolysis step, and optimisation of the downstream process by the simultaneous production and recovery of the VFAs.

In the first experimental chapter (*Chapter 3*), different inocula were tested for their ability to degrade BSG. The highest solids degradation was achieved by the inocula which also presented the highest gas production. On the other hand, the inocula that were able to accumulate fermented compounds presented an inefficient solid degradation despite presenting a large hydrolytic community. This led to the hypothesis that possible inhibition of the hydrolysis process was caused by the accumulation of the VFAs. Further study of the AD process of lignocellulosic materials for chemical production is required to determine this possible inhibitory effect if methane is not the end product.

The second experimental chapter of this thesis covered a two-step process, first the hydrolysis pretreatment of BSG, and second, the fermentation of the liquid hydrolysate. This approach relied on the production of a mixed liquor of simple carbohydrates that could be easily fermented by microbial communities, as the hydrolysis step is not required. Different combinations of sulfuric acid and BSG concentrations were tested for the release of carbohydrates into the liquor fraction. The optimal conditions corresponded to 7 % BSG w/w (dry matter), 1.5 % of sulfuric acid v/v at 121 °C for 20 min.

The efficiency of this combination resulted in the extraction of 0.44 g of carbohydrates per g of BSG. The obtained hydrolysate was used for the acidogenic fermentation in batch reactors under different pH conditions (*Chapter 4*). The liquid hydrolysate was directly used without detoxification or chemical addition. However, it was diluted to a concentration of 15 g/L and the pH was increased up to the operational pH (7.0 and not further control, 4.5, 5.0, 6.0 and 8.0). The highest VFA concentration was achieved at pH 6.0, which corresponded to 17.6 (\pm 0.67) gCOD/L and VFAs contributed 93.0 % of the sCOD. However, the highest VFA production rate was obtained at pH 5.0 which corresponded to 0.79 gCOD/(L-h). The next step was the continuous production of VFAs using a high rate reactor to achieve a large volumetric production (*Chapter 5*). The highest achieved VFA production corresponded to 120.4 (\pm 15.0) mmol/L, which contributed approximately 83.4 (\pm 5.9) % of the soluble COD. However, VFA production was unstable as the concentration of acetic acid, propionic acid, butyric acid and ethanol fluctuated as major compounds. This fluctuation could be the result of metabolic fluxes caused by the stress responses of fermentative microorganisms to the continuous accumulation of fermented compounds. Thus, the simultaneous production and recovery of the VFA could contribute to alleviate such effects and increase the VFA selectivity.

Chapter 6 and *Chapter 7* of this PhD thesis developed an optimization during the hydrolysis step and the acidogenic fermentation to improve any future applications for the potential scale up of the two-step process. As BSG retains a considerable pool of high value compounds, a cascade process was developed by combining a chemical pretreatment with enzymatic hydrolysis (*Chapter 6*). The best combinations led to a liquid fraction rich in carbohydrates up to 39.9, 52.4 and 44.6 g/L of total carbohydrates. Besides the liquid fraction, two different solid fractions were obtained with great potential for food industrial applications due to the content of proteins and water retention capacity.

Finally, further optimisation for the VFA production and recovery was developed by coupling an electrochemical cell unit to the recycling line of an upflow anaerobic sludge bed reactor (UASB). In this case glucose was used as the sole carbon source. The glucose concentration was increased from 15 to

60 g/L. The highest VFA production achieved was above 30 g/L of VFAs. The electrochemical cell was able to enhance the VFA concentration up to 4 fold during the recovery process, although this was when operating at lower glucose concentrations. The UASB-electrochemical cell system showed additional advantages, such as the reduction of alkaline demand and favoured the production of caproic acid with higher value than acetic and butyric acids. The process requires further optimisation as the electrochemical cell was under dimensioned. Nevertheless, the process showed a high potential for the recovery of the VFAs.

In summary, this PhD thesis demonstrated the potential for using BSG as a feedstock in a biorefinery. The combination of different chemical and enzymatic treatments during initial processing allowed for the recovery of proteins and the release of high concentrations of carbohydrates into the liquid fraction. During downstream processing, this carbohydrate rich fraction led to a high concentration of VFAs, which in combination of low pH operational conditions, led to minimal methane production. Finally, a novel electrochemical cell was implemented, which demonstrated efficient VFA recovery and decreased the requirement for chemical (base) addition.

Funding

This research was supported by the Science Foundation Ireland (SFI) Research Professorship Programme Innovative *Energy Technologies for Biofuels, Bioenergy and a Sustainable Irish Bioeconomy* (IETSBIO³, award no. 15/RP/2763) and the Research Infrastructure research grant *Platform for Biofuel Analysis* (Grant Number 16/RI/3401). Part of the PhD thesis was developed as a visiting research supported by the Thomas Crawford Hayes Award Research & Travel Fund 2021.

Chapter 1

General introduction

Keywords: Biorefinery; circular economy; biomas disposal; biomass feedstock

1.1 General introduction

Industrial advancement during the last century has led to a more comfortable way of life. However, this has often been at the expense of the environment (Yang et al., 2018). At the beginning of the 20th century, water and fossil fuels were considered an unlimited resource and not enough attention was paid to waste treatment. This attitude resulted in the increase of the greenhouse gas emissions and deterioration of natural habitats, such as the vast plastic accumulation in the Pacific and Atlantic oceans. Fortunately, in recent years, sustainability has become more prominent, driven by heightened environmental awareness and decreasing reservoirs of fossil fuels and natural resources. For example, in Ireland the solid wastes, which were previously sent to landfills have been diverted to new waste disposal facilities such as incinerators (Davies, 2008; EPA, 2019). However, many of these wastes could be used to produce compounds with high added value. The next milestone in sustainability must be achieved by replacing fossil fuel based industries with wastes, which act as renewable biomass.

The biorefinery concept has emerged as an alternative to traditional petroleum refineries, where renewable by-products and wastes are used for production of high value compounds (*Figure 1.1*). In a biorefinery, renewable biomass is separated and fractionated into a range of products such as chemicals, biomaterials, or biofuels. Wastes are starting to be considered valuable entities and many researchers are focusing on acquiring a deep understanding of advanced processes to obtain high value products from different biowastes, not only from waste treatment (Karthikeyan et al., 2017). An example of the evolution from waste treatment to waste valorisation is the shift from aerobic to anaerobic digestion. In the past, aerobic digestion was the main process for wastewater treatment which results in the production of large volumes of activated sludge, which requires further downstream processing. Anaerobic digestion allows the conversion of organic matter into biogas, CO₂ and CH₄ with much less sludge production (Chetty and Pillay, 2015), making wastewater a resource for biogas production.



Figure 1.1 Range of products and chemicals that are obtained in a traditional petroleum-based refinery and in a biorefinery

The partial or total inhibition of methane production during anaerobic fermentation can lead to accumulation of some intermediate compounds such as hydrogen, lactate, or volatile fatty acids (VFAs) (Rosa et al., 2014; Cisneros-Perez et al., 2017). Different strategies have been followed at lab scale to promote hydrogen production, coupled to simultaneous accumulation of VFAs (Garcia-Aguirre et al., 2017). VFAs are short chain fatty acids (C2 to C6) with a broad range of industrial applications, however their current production relies on fossil fuel sources (Strazzera et al., 2018). The main problem with VFA production from renewable sources is the diluted concentration of VFAs in the fermented broth. This relegates VFA production as an intermediate step for production of other compounds that are easier to recover, e.g. polyhydroxyalkanoates (PHA) or lipids for biodiesel production (Albuquerque et al., 2011; Patel et al., 2018). Nevertheless, recent improvements in extraction systems may allow efficient extraction of VFAs as the final product, especially if a high VFA concentration can be obtained (Zhang and Angelidaki, 2015; Saboe et al., 2018; Wainaina et al., 2019). Thus, this enables the uncoupling the VFA production from fossil fuels and provides a trade opportunity in countries that rely on importing gas and oil. However, further investigations must be carried out to develop this process.

Conventional wastewater cannot be used to achieve a higher VFA concentration due to the low organic matter content. In addition, wastewater often contains inhibitory compounds for VFA production. For example, the raw dairy wastewater typically contains a large amount of fats that are inhibitory for microbial degradation (Chetty and Pillay, 2015). Thus, the current system for dairy wastewater treatment includes the removal of the fats prior to the anaerobic process for methane production, which decreases the organic content below 5 g/L (Chetty and Pillay, 2015; Trego et al., 2021), which is not sufficient for VFA production and recovery. Therefore it is necessary to find an alternative to traditional wastewater treatment. For example, solid by-products or a waste with low or no value that could be used as a source of carbohydrates. However, if this solid by-product/waste requires a pretreatment, the feasibility of the process could be compromised by a higher cost to obtain the final product than the value of the product itself.

The first milestone of the present work is to find an industrial solid waste, or low value by-product, as feedstock. The characteristics of this by-product/waste must meet some criteria such as large availability. It would also be beneficial if it contained additional compounds of interests other than carbohydrates, which could be recovered by a sustainable fractionation of the biomass in a biorefinery approach. For example, the hydrolysis of the solid feedstock by a combination of different physical, chemical and biological processes to obtain a liquid hydrolysate rich in carbohydrates with the simultaneous recovery of additional added value compounds (e.g. proteins and fibers). The recovery

of these added value compounds can partially counterbalance the costs of the hydrolysis process, such as energy use or enzymes.

1.2 Biorefinery definition

The biological processes for conversion of raw material into a final product have been largely exploited throughout history, e.g. the fermentation of milk to yogurt or barley to beer. Despite the widespread application of these different practices, it was not until the 17th century that the microorganisms responsible for these processes were observed and identified. This led to the development of more sophisticated technologies, with industrial microbiology taking off on a large scale for the production of pharmaceutical compounds, e.g. antibiotics, hormones and steroids. These innovations led to the start of the biotechnology industry, which relies on microbial activity to obtain high value products at industrial level. Nowadays, this sector plays a key role in sustainability and reducing reliance on fossil fuels.

1.2.1 Plant refinery session

The use of renewable biomass for chemical and energy production started at the end of the 19th century. However, this was rapidly displaced by the use of petrol, before increasing again at the beginning of the 1980s, when crops were suggested as renewable feedstock for the production of fuels and organic chemicals (Levy et al., 1981). This definition was expanded upon in the biomass refinery concept (Koukios 1985). The approach was adopted and further developed by the Plant Refinery Session in the early 90s, emphasizing the differences between the biofuel production facility and the biorefinery facility (Galletti, 1991). The biorefinery facility, beside energy, would produce industrial intermediates with a wide range of chemicals and materials (Wyman and Goodman, 1993).

1.2.2 International energy agency, working group bioenergy

Currently, the Bioenergy working group of the International Energy Agency (IEA) developed the biorefinery concept as the sustainable processing of biomass into a spectrum of marketable products (food, feed, materials and chemical) and energy (fuels, power and heat) (IEA Bioenergy, 2009). The IEA classified biorefineries based on four features: i) platforms, ii) products, iii) feedstocks, and iv) processes. Platforms were considered the principle feature as they are the link between raw and final product, and they can be used to link different biorefinery concepts (Cherubini et al., 2009). Regarding the feedstock, there were two main parts: i) dedicated crops and ii) residues, subsequently dividing each of them into different categories (Cherubini et al., 2010).

1.2.3 Green biorefinery

Additionally, the term of green biorefinery was introduced by Morais and Bogel-Lukasik (2013), based on the twelve principles of the green chemistry. These twelve principles were designed by

Anastas and Warner (1998) to tackle and reduce pollution in the chemical industry. They considered alternative processes and the use of less polluting materials to make a more environmental and sustainable system. For instance, the term "green" was established at the end of the 1980s to define the sustainable development for economic progress as Green Economy (Allen and Clouth, 2012). The use of this term has evolved in the last decade, related with a lower carbon dependency and more environmental friendly practices (Barbier, 2010).

1.2.4 Circular economy

The largest greenhouse gas emissions from material management are related with the extraction, processing and production stages (PACE, 2020). The development of the biorefinery should go handin-hand with the switch from the linear economy to the circular economy. In other words, moving away from the take-make-use-dispose practise (Lieder and Rashid 2016) and closing the loop, turning wastes into resources (Deselnicu et al. 2018). However, approximately only 9 % of the world operates in a circular economy system (PACE, 2020). To promote the circular economy, biorefineries could become a technological hub for collection, treatment and production of value added compounds from current wastes. Thus, reinforcing industrial sustainability and moving away from the use of fossil fuels (*Figure 1.2*)





There are alternative definitions for the biorefinery concept and classification. Some of them have been summarized in *Table 1.1*. However, based on the aforementioned statements, the present work has accepted the definition and classification of biorefineries given by the IEA working group Bioenergy. Although an important amendment must be made for further improvement of sustainability. The feedstock should not be split in dedicated crops and residues, instead it should be classified based on its composition and the different fractions that can be extracted for downstream valorisation. For example, the amount of proteins, lipids, cellulose, hemicellulose or the phenolic content.

 Table 1.1 Biorefinery and biorefining definition from different authors broadly used in the actual industry

Author	Year	Definition
Koukios	1985	A biomass refinery is an integrated complex of industrial processes that separate the major components of bioresources and convert them to fuels, food, feed, fiber, fertilizer and chemicals
Wyman and Goodman	1993	A biorefinery could produce a wide range of chemicals and materials through microbial conversion of renewable resources
Kamm et al.	2008	Biorefining involves the transfer of efficiency and logic from the fossil-based chemistry and substance conversion industry as well as energy production to the biomass industry
Biopol	2008	Biorefinery is the sustainable processing of biomass into a spectrum of marketable products
Biorefinery Euroview	2008	Biorefineries are integrated bio-based industries, using a variety of different technologies to produce chemicals, biofuels, food and feed ingredients, biomaterials and power from biomass raw materials
IEA Bioenergy Task 42	2009	Biorefining is the sustainable processing of biomass into a spectrum of bio-based products and energy
Morais and Bogel-Lukasik	2013	Green biorefinery is based on the ideal processes that use a non-polluting material as a feedstock, there is no production of any contaminant and solvents are avoided

1.3 Potential feedstock for an Irish biorefinery

Many different wastes have been broadly studied for the development of the biorefinery approach, from industrial to household wastes. Even though food wastes from households, restaurants or similar settings represent a continuous source of organic feedstock, their wide heterogeneous and changing composition could affect the stability of the biorefinery processes (Cristobal, 2019). On the other hand, some of the agricultural and food processing wastes present a more homogeneous fraction, but these products could be more affected by seasonality (Fava et al., 2015; Cristobal et al., 2019). Thus, the present work was looking for an industrial waste produced throughout the year, with consistent characteristics. Another key factor is the composition of the material, which should contain high value compounds to facilitate a more straight forward valorisation in a shorter term. This section aims to evaluate the different industrial sectors that contribute to the categories in the EPA Waste Classification that could be used as source of renewable feedstock for a biorefinery (EPA 2019).

1.3.1 Industrial sectors and current disposal of the biomass

In Ireland, the production of solid wastes in 2016 was above 47 million tonnes (Eurostat, 2020). In 2018, the Environmental Protection Agency (EPA) updated the EPA Waste Classification which includes a total of 843 different types of wastes, split into 10 different categories with several subdivisions based on their characteristics (EPA 2019). At least 37 of these categories could be easily used as a renewable biorefinery feedstock. Most of these wastes are sent for incineration or co-incineration (1,177,175 tonnes), landfilling (570,000 tonnes), composting and biostabilization (602,700 tonnes); and gas production via anaerobic digestion (110,000 tonnes). The number of landfills in Ireland has decreased in the last decade at the same time that the number of incineration and co-incineration plants have increased (EPA 2020). This was caused by a change in waste management practices in the mid 1990s due to the requirement for more sustainability and lower solid waste production (Davies 2008). Despite this improvement in waste management, industrialisation in the last decade has resulted in the increase of the greenhouse gas emissions in Ireland (Jensen, 2021). Thus, not only waste management must be improved, but also the valorisation of current wastes as alternative to fossil fuels.

The goal of this PhD thesis is to find a well-established industry in Ireland that generates a stream of solid wastes or low value by-products, in enough quantity to be used for chemical and energy production. Some of the most important industrial sectors from the EPA classification (2019) that could be used as a source of feedstock for biorefinery purposes can be summarised in three key sectors:

- 1. Meat and livestock: beef, pig meat, sheep meat and poultry
- 2. **Prepared food:** chocolate confectionary, frozen prepared-food, cooked meats and meat ingredients, and bakery products and snacks
- 3. Beverages: whiskey, cream liqueur, beer and cider

Industries in these fields such as dairy, brewing or meat industries have a large contribution to the Irish economy and have increased their exports and value over the last decade (*Figure 1.3*).



Figure 1.3 Main exports from the Irish industry; processing food (a), livestock (b, c and d), and beverage (e and f)

Most solid wastes from the different industries are mixed and treated equally, for example nonhazardous organic material from the dairy, brewery or food confectionary industries (EPA, 2020). This results in a loss of opportunities for the current wastes that contain valuable compounds. Some of these products could even be considered as a by-product instead of wastes, with uses in the chemical, cosmetic, or food industries. Some examples are the production of proteins using dairy wastes, potato wastes or brewery wastes.

1.3.2 Biomass selection for this PhD thesis

During the development of this PhD thesis, some different industrial waste samples were collected for their valorisation. This includes rumen fluid, known as belly grass, which is a mix of the content from the four stomachs of cows, this material was sourced from a slaughterhouse in County Mayo (Ireland). The non-used dough was collected from an industrial pizza confectioning plant in County Longford (Ireland). However, because of the large content of oil and a discontinuous production this waste was discarded to be studied in this PhD, despite of being rich in carbohydrates. Brewery spent grain was collected from a plant located in County Galway (Ireland). Brewery spent grain was selected as the feedstock for this PhD thesis. This was decided based on the large availability and constant composition of the brewery spent grain throughout the year.

The main goal of this PhD thesis was **i)** the fractionation and processing of brewery spent grain into different streams, including a liquid fraction rich in carbohydrate and two solid streams rich in lignin and proteins and **ii)** the production and simultaneous recovery of the VFAs from the carbohydrate rich liquid fraction.

1.4 Thesis scope

The core of this PhD thesis is the sustainable production and recovery of VFAs from brewery spent grain. Thus, contributing to shifting the view of anaerobic digestion from a waste treatment system to a more sophisticated tool for chemical and energy production. However, the conditions mentioned above, such as the requirement of a liquid hydrolysate rich in carbohydrates, will have a cost that the production of the VFAs as the sole final product could not compensate. Implementing the biorefinery approach using brewery spent grain as feedstock could counterbalance the cost of the initial processing. The production of a different range of compounds such as proteins, lignin, VFAs, hydrogen, carbon dioxide and methane could also make up some of the costs.

Chapter 2 is a review of some examples of bioerefineries using lignocellulosic feedstock. In addition, Chapter 2 provides an overview of the anaerobic digestion process and how this can be used for volatile fatty acid production, including current approaches for the use and recovery of volatile fatty acids. Two different approaches for VFA production are reviewed, i) the simultaneous degradation and fermentation process, and ii) a two-step process focus on hydrolysis and downstream fermentation of the liquid hydrolysate. As well, the current treatments for lignocellulosic materials as feedstock to provide added value compounds is overviewed. Finally, the main composition and characteristics of the brewery spent grain with current use and valorisation are presented.

Chapter 3 describes simultaneous hydrolysis and fermentation of brewery spent grain as a lignocellulosic feedstock using sacrificial tests. Different sources of inocula were used to find a suitable microbial consortium for chemical and energy production. This includes industrial sludge, cow rumen fluid, and animal faeces.

Chapter 4 describes a two-step process for the production of the volatile fatty acids. First, a thermal diluted sulfuric acid hydrolysis was carried by a combination of different acid and brewery spent grain

concentrations. The liquor produced was fermented using anaerobic granular sludge under different pH values in a batch reactor system. Finally, this approach was chosen for further development along the thesis due to the content of high value compounds in BSG to be recovered during the first step.

Chapter 5 followed up the batch approach developed during Chapter 4 for the VFA production. The pretreatment conditions developed at chapter 4 were used to obtain the liquid hydrolysate. In this case, the fermentation was carried out using a continuous expanded granular sludge bed reactor. The system was operated at pH 5.90 (± 0.05) at different organic loading rates and inhibition of methanogenesis was achieved using specific operational conditions.

Chapter 6 focused on optimising the hydrolysis of brewery spent grain using a cascade process for the high-value product recovery, lignin valorisation, and simultaneous release of the carbohydrates into the liquid fraction. First, a combination of different parameters (acid, temperature and time) for the pretreatment was assessed, followed by enzymatic hydrolysis. The release of carbohydrates, and the effect of the pretreatment on the protein and lignin extraction were evaluated.

Chapter 7 aimed to enhance the production of the volatile fatty acids using a synthetic high strength feedstock with carbohydrate concentrations up to 60 g/L. A similar system than in Chapter 5 was used, however in this case an electrochemical cell was used for the recovery of the volatile fatty acids, alleviating their possible toxic effect to the bioreactor biomass.

Chapter 8 summarises the principle conclusions of this work, including a diagram with the whole proposed process for the brewery spent grain biorefining. This could be used as foundation for scaling up the process and practical implication of a biorefinery for production of proteins, lignin, volatile fatty acids and biogas.

1.5 References

- Albuquerque, M. G. E., V. Martino, E. Pollet, L. Averous, and M. A. M. Reis (2011) "Mixed culture polyhydroxyalkanoate (PHA) production from volatile fatty acid (VFA)-rich streams: Effect of substrate composition and feeding regime on PHA productivity, composition and properties." *Journal of Biotechnology* 151 (1): 66-76. https://doi.org/10.1016/j.jbiotec.2010.10.070
- Allen, and Clouth (2012) A Guidebook to the Green Economy. In *Issue 1: Green Economy, Green Growth, and Low-Carbon Development history, definitions and a guide to recent publications.*
- Anastas, P. T., and J. C. Warner (1998) *Green chemistry: theory and practice*. New York: Oxford University Press.
- Barbier, Edward(2010) "How is the Global Green New Deal going?" *Nature* 464 (7290): 832-833. https://doi.org/10.1038/464832a

- Biopol (2009) Assessment of BIOrefinery concepts and the implications for agricultural and forestry POLicy https://edepot.wur.nl/173825
- Biorefinery Euroview (2009) Current situation and potential of the biorefinery concept in the EU: strategic framework and guidelines for its development https://cordis.europa.eu/project/id/44275/reporting/es
- IEA Bioenergy (2009) "Biorefineries: adding value to the sustainable utilisation of biomass".
- Cherubini, F.(2010) "The biorefinery concept: using biomass instead of oil for producing energy and chemicals." *Energy Convers Manag* 51. https://doi.org/10.1016/j.enconman.2010.01.015.
- Cherubini, F., Jungmeier G., Wellisch M., Willke T., Skiadas I., Ree R. V., Jong E. ed. (2009) *Toward a* common classification approach for biorefinery systems. Vol. 3.
- Chetty S., Pillay K. (2015) "Application of the DIY carbon footprint calculator to a wastewater treatment works." *Water SA* 41: 263-272. http://www.scielo.org.za/scielo.php?script=sci_arttext&pid=S1816-79502015000200013&nrm=iso.
- Cisneros-Pérez C., Etchebehere C., Celis L. B., Carrillo-Reyes J., Alatriste-Mondragón F., Razo-Flores E. (2017) "Effect of inoculum pretreatment on the microbial community structure and its performance during dark fermentation using anaerobic fluidized-bed reactors." *International Journal of Hydrogen Energy* 42 (15): 9589-9599. https://doi.org/https://doi.org/10.1016/j.ijhydene.2017.03.157.
- Cristóbal J., Caldeira C., Corrado S., Sala S. (2018) "Techno-economic and profitability analysis of food waste biorefineries at European level." *Bioresource Technology* 259: 244-252. https://doi.org/https://doi.org/10.1016/j.biortech.2018.03.016.
- Davies A. R. (2008) "Civil society activism and waste management in Ireland: The Carranstown antiincineration campaign." *Land Use Policy* 25 (2): 161-172. https://doi.org/https://doi.org/10.1016/j.landusepol.2007.04.002.
- Deselnicu D., Militaru G., Deselnicu V., ZĂInescu G., L. A. (2018) *Towards a Circular Economy– a Zero Waste Programme for Europe*.
- EPA (2019) Waste Classification. edited by Ireland Environmental Protection Agency.
- EPA (2020) Waste Infrastructure in Ireland. EPA Website.
- EUROSTAT (2020) European Statistical Recovery Dashboard.
- Fava F., Totaro G., Diels D., Reis M., Duarte J., Carioca O. B., Poggi-Varaldo H. M., Ferreira B. S. (2015)
 "Biowaste biorefinery in Europe: opportunities and research & development needs." New Biotechnology 32 (1): 100-108. http://dx.doi.org/10.1016/j.nbt.2013.11.003.
- Rosa F., Rúbia P., Santos S. C., Silva E. L. (2014) "Different ratios of carbon sources in the fermentation of cheese whey and glucose as substrates for hydrogen and ethanol production in continuous reactors." *International Journal of Hydrogen Energy* 39 (3): 1288-1296. https://doi.org/10.1016/j.ijhydene.2013.11.011.

- Galletti, G.C. (1991) Production and Utilization of Lignocellulosics: Plant refinery and breeding, analysis, feeding to herbivores, and economic aspects. Springer Netherlands.
- Garcia-Aguirre J., Aymerich E., González-Mtnez. J de G, Esteban-Gutiérrez M. (2017) "Selective VFA production potential from organic waste streams: Assessing temperature and pH influence." *Bioresource Technology* 244 (Part 1): 1081-1088. https://doi.org/https://doi.org/10.1016/j.biortech.2017.07.187.
- Jensen L. (2021) Climate action in Ireland. In *Latest state of play*. European Parliamentary Research Service.
- Kamm B., Gruber P. R., Kamm M. (2008) Biorefineries Industrial Processes and Products. Vol. 5.
- Karthikeyan O. P., Mehariya S., Wong J. W. C. (2017) "Bio-refining of food waste for fuel and value
products."*Procedia*136:14-21.https://doi.org/https://doi.org/10.1016/j.egypro.2017.10.253.
- Koukios E. G. (1985) "Biomass refining: a non-waste approach." In *Economics of ecosystems management*, edited by D. O. Hall, N. Myers and N. S. Margaris, 233-244. Dordrecht: Springer Netherlands.
- Levy P. F., Sanderson J. E., Kispert R. G., Wise D. L. (1981) "Biorefining of biomass to liquid fuels and organic chemicals." *Enzyme and Microbial Technology* 3 (3): 207-215. https://doi.org/https://doi.org/10.1016/0141-0229(81)90087-9.
- Lieder M., Rashid A. (2016) "Towards circular economy implementation: a comprehensive review in context of manufacturing industry." *Journal of Cleaner Production* 115: 36-51. https://doi.org/https://doi.org/10.1016/j.jclepro.2015.12.042.
- Morais A. R. C., Bogel-Lukasik R. (2013) "Green chemistry and the biorefinery concept." *Sustainable Chemical Processes* 1 (1): 18. https://doi.org/10.1186/2043-7129-1-18.

PACE (2020) The circularity gap report.

- Patel A., Mikes F., Bühler S., Matsakas L. (2018) "Valorization of Brewers' Spent Grain for the Production of Lipids by Oleaginous Yeast." *Molecules (Basel, Switzerland)* 23 (12): 3052. https://doi.org/10.3390/molecules23123052. https://pubmed.ncbi.nlm.nih.gov/30469531.
- Saboe P. O., Manker L. P., Michener W. E., Peterson D. J., Brandner D. G., Deutch S. P., Kumar M., Cywar R. M., Beckham G. T., Karp E. M. (2018) "In situ recovery of bio-based carboxylic acids." *Green Chemistry* 20 (8): 1791-1804. https://doi.org/10.1039/C7GC03747C.
- Strazzera G., Battista F., Garcia N. H., Frison N., Bolzonella D. (2018) "Volatile fatty acids production from food wastes for biorefinery platforms: A review." *Journal of Environmental Management* 226: 278-288. https://doi.org/https://doi.org/10.1016/j.jenvman.2018.08.039.
- Trego A. C., Holohan C. B., Keating C., Graham A., O'Connor S., Gerardo M., Hughes D., Ijaz U. Z., O'Flaherty V. (2021) "First proof of concept for full-scale, direct, low-temperature anaerobic treatment of municipal wastewater." *Bioresource Technology* 341: 125786. https://doi.org/https://doi.org/10.1016/j.biortech.2021.125786.
- UNDESA (2012) A Guidebook to the Green Economy. In *Issue 1: Green Economy, Green Growth, and Low-Carbon Development - history, definitions and a guide to recent publications.*
- Wainaina S., Parchami M., Mahboubi A., Horváth I. S., Taherzadeh M. J. (2019) "Food waste-derived volatile fatty acids platform using an immersed membrane bioreactor." *Bioresource Technology* 274: 329-334. https://doi.org/10.1016/j.biortech.2018.11.104.
- Wyman C. E., Goodman B. J. (1993) "Biotechnology for production of fuels, chemicals, and materials from biomass." *Applied Biochemistry and Biotechnology* 39 (1): 41. https://doi.org/10.1007/BF02918976
- Yang N., Zhang Z., Xue B., Ma J., Chen X., Lu C. (2018) "Economic Growth and Pollution Emission in China: Structural Path Analysis." Sustainability 10 (7). https://doi.org/10.3390/su10072569.
- Zhang Y., Angelidaki I. (2015) "Bioelectrochemical recovery of waste-derived volatile fatty acids and production of hydrogen and alkali." Water Research 81 (Supplement C): 188-195. https://doi.org/https://doi.org/10.1016/j.watres.2015.05.058.

Chapter 2

Review and statement of the thesis

A modified version of this chapter has been publish as:

Castilla-Archilla J., O'Flaherty V., Lens P. N. L. (2019) "Biorefineries: Industrial Innovation and Tendencies" In: J.-R. Bastidas-Oyanedel, J. S. Schmidt, (Eds.), Biorefinery: Integrated Sustainable Processes for Biomass Conversion to Biomaterials, Biofuels, and Fertilizers, Springer, Switzerland, pp 3-35

Keywords: Biorefinery; chemical platform; chemical building blocks; ethanol; bioplastics; biogas; anaerobic digestion; volatile fatty acids; lignocellulose; carbohydrates; protein; lignin

2.1 Chemical platform

The development of the biorefinery concept mirrors that of the petrochemical refinery, as the initial goal of petrochemical processes was to produce fuels but it progressed to include the production of many chemical and pharmaceutical products (Jones 2006). Similarly, the objective of the biorefinery is to develop a new biochemical platform which relies on renewable biomass rather than fossil fuels (Werpy et al., 2004). The ability to utilise renewable biomass, including wastes, to produce chemical compounds would reduce reliance on fossil fuels and at the same time provide a commercial opportunity for countries without their own fossil fuel resources.

In 2004 the U.S Department of Energy developed a list of basic compounds which could replace fossil fuels. This list, known as the chemical building blocks, identified 12 of the most promising chemicals from over 300 molecules. These twelve compounds include: aspartic acid, succinic acid, glucaric acid, itaconic acid, 3-Hydroxybutyrolactone, sorbitol, 2,5 – Furand dicarboxylic acid, 3-Hydroxypropionic acid, glutamic acid, levulonic acid, glycerol and xylitol (Werpy et al., 2004). In 2010 additional criteria were used to update the list, focused on the industrial development of the compounds and excluding molecules with low extend for new derivatives or with low research activities. Newly added compounds included, lactic acid, as precursor of PLA; and ethanol, a precursor of chemical compounds such as ethylene (Bozell and Petersen, 2010). Similarly, the project E4tech analysed the industrial market in UK to identify the most notable chemicals based on the market attractiveness and industrial uses (E4tech, 2017). They identified 30 compounds, with lactic acid the most attractive to be produced in the UK.

A lot of effort and investment has been put into the production of these compounds. However only some successful cases achieved industrial level production. For example, 3-hydroxipropionic acid has been widely investigated, but no further announcement of industrial production has been made (Novozymes 2014; Sims 2012). In other cases, the technological process belongs to a single company or the technology is not mature enough. For example, 1,3–propanediol and bio-esobuteno, developed by Dupont & Lyle Bioproducts and Global Bioenergies (DuPont, 2018). Some other processes have been scaled to industrial levels and are being used by different companies including the production of bioethanol, lactic and polylactic acid, succinic acid, and biogas.

2.1.1 Ethanol production

Bio-ethanol production from crops has only increased since the year 2000, despite the initial process being developed at the end of the 19th century. This is because cheap, abundant petroleum took over in the 20th century. The current change back to renewable fuels was triggered by three main

reasons: i) decrease in fossil fuel sources, ii) dependency on third countries; and iii) increasing air pollution.

2.1.1.1 Market development in USA

The petrol embargo in the summer of 1973 led to the battle for energy security and the development of the Solar Energy Research Institute (SERI), which was operational by 1977 (NREL, 2002). At the time, increasing pollution problems were addressed by some states by enforcing the use of more oxygenated compounds in gasoline to improve combustion and reduce contamination (Vinuesa et al., 2003). In the 1990s, the use of ethanol rather than fossil fuel products was promoted by a tax exemption (Duffield et al., 2015).

In 1991, the SERI was given National Laboratory Status and its name was changed to the National Renewable Energy Laboratory (NREL) to integrate all renewable energy technologies. This research centre implemented the Alternative Fuels Utilization Programme, which involved different stakeholders (i.e. government agencies, industry, universities, and coalitions), to develop competitive renewable energy technologies at commercial level (Bailey and Colucci 1997). As a result, from 1980 to 2000 ethanol production cost dropped down from \$ 1,056.69 to 371.00 per cubic meter (NREL, 2002).

Finally, further progress was made in the beginning of the 21st century, when California enforced the use of bioethanol instead of fossil fuels. This was followed by other states, which made a large contribution for the development of the bioethanol industry (Duffield et al., 2015). Nowadays, the USA is the biggest producer of bioethanol with more than half of the worldwide production using different renewable sources, such as corn, sorghum, sugar wastes, beverage wastes, wheat straw and lignocellulosic materials (RFA, 2018).

2.1.1.2 Valorisation of agricultural surplus: Bazancourt-Pomacle ethanol and biorefinery

At the beginning of the 20th Century, the Bazancourt-Pomacle biorefinery site was developed and included a dedicated crop field for sugar beet, with a distillery plant. In 1950, the distillery was converted into a sugar mill factory and it was converted into a cooperative, which eventually evolved with other sites to emerge as Cristal Union. At the end of the 1980s the overproduction of sugar beet resulted in the development of a research centre for its alternative use and valorisation. At the beginning of the 1990s, a factory for production of starch and glucose was added and the Abro-Industrie Recherches et Développements (ARD) research centre was developed for further valorisation of the sugar beet, wheat cereal and alfalfa (Schieb et al., 2014). As a result of the French Governments industrial policy for competiveness clusters in 2005, the Cristanol plant was developed

for production of ethanol from sugar beet and cereal on the site in 2007. In addition, the Futurol project opened a pilot facility for production of ethanol using lignocellulosic materials as feedstock.

Currently, the site of the Bazancourt-Pomacle has evolved into a biorefinery hub using wheat, sugar beet and alfalfa, for production of sugar, starch, ethanol, liquid CO₂, surfactant and cosmetic ingredients. The current capacity of the Bazancourt-Pomacle facility is above 235,000 cubic meters per year (Schieb et al., 2014). In addition, through the financial support of local authorities the plant developed a research centre within multiple colleges and universities. This allowed for the development of biotechnological processes on industrial scales, such as the Bioamber facility plant (Schieb et al., 2014; Stadler and Chauvet, 2018). This resulted in the construction of a demo platform for additional projects, such as the development of isobutene by Global Bioenergies (Global Bioenergies, 2015).

2.1.1.3 Disseminated ethanol production from current wastes: St1 Biofuels

The Finnish company St1 Biofuels Oy, currently owned by St1, started as a joint venture in 2006 between St1 and VTT Technical Research Centre of Finland. This project was a cooperation between different stakeholders, with companies from different industrial clusters and sectors of innovation, partially supported by the Finnish Government (Mäkinen and Leppälahti 2009). First, the Etanolix technology was developed for production of ethanol using different wastes, such as bakery and sweet wastes. Afterwards, additional technologies were developed using lignocellulosic material and municipal and retail wastes, such as Cellunolix and Bionolix, respectively. Ethanol production started in 2007 with small plants in areas close to the sites where wastes were produced. These plants produced ethanol at a final concentration of 85 %. This diluted ethanol was subsequently transported to a central hub for the hydration process, resulting in a final ethanol concentration of 99.8 %.

This design, of small production sites close to the original waste production avoided transporting large amounts of the raw solid material and reduced transport costs. Thereafter, bigger additional plants were developed within different industries to use their current wastes as the main sources, offering other advantages than only production of bioethanol (Novozymes, 2015). For example, one of the plants is located beside an enzyme production plant owned by DuPont, which allows the use of enzymes for the process, reducing costs. Another plant was built at the side of the Hartwall brewery, for simultaneous production of animal feed as co-product. Further plants were developed using municipal and retail wastes, coupled with biogas, electricity and fertilizer production.

2.1.1.4 Problems and failures during scaling-up

Although the production of ethanol from lignocellulosic materials has been possible at industrial scale, some plants have failed as their processes were economically unfeasible due to different factors.

Chapter 2

The first commercial-scale cellulosic ethanol production facility was developed in 2012 in Italy with the Proesa technology by Beta Renewables using giant reed as feedstock (Nickel, 2014). This technology was exported to countries, including USA, China and Brazil. Nevertheless, after a couple of years the company announced its shutdown for economic reasons (Patel et al., 2019). There were several problems related to the source of feedstock and mechanical issues during the up-stream process when using technologies typically applied to other lignocellulosic feedstock. For example, the higher difficulty of treating corn stove instead of wood chips (Slupska and Bushong 2019), or cane leaves rich in silica that caused greater damage than wheat straw during the pretreatment step (Marques 2018). This failure was attributed to a lack of expertise in agronomic and agricultural engineering specifically relating to transport, storage and pretreatment of the biomass source (Dale 2018).

GranBio (Brazil) implemented the Beta Renewables technology for ethanol production using sugarcane as lignocellulosic feedstock. The technology was supposed to hydrolyse the material within 19 hours, however it took between 48 to 90 hours (Marques, 2018). This, and other problems related with the plant production made it unfeasible at industrial scales. Subsequently, GranBio has developed their own yeast stream to improve the process (GranBio, 2015) and acquired different licenses fpr technology which improve process efficiency (GranBio, 2020); in combination with the use of a more efficient hybrid sugarcane (GranBio, 2018).

In 2009, POET developed a pilot plant as part of the Liberty Project for ethanol production using corn cob as feedstock (POET, 2009). The plant was expected to be operational by 2011, however difficulties related to the pretreatment process held ethanol production up until 2017 (Kennedy, 2017), increasing the cost of the plant and the delaying commercialization of the technology (Slupska and Bushong, 2019). Eventually by 2019 as a results of a loss of competitiveness for cellulosic biofuels with the decrease of the volume obligations in USA, POET-DSM announced that the production of ethanol would be paused. Then, all the efforts was shifted towards research and development for the improvement of the technology (POET, 2009).

The production of ethanol from microalgae was a trending topic at the beginning of the 21st Century. Many companies developed different technologies, such as engineered microorganisms, for the production of the ethanol (Ziolkiwska and Jadwiga, 2014; de Farias Silva and Bertucco, 2016). These approaches promised to achieve much higher quantities and yields than the current ethanol production technologies can produce (Khanra et al., 2018). Moreover, during upscaling to industrial plants, the production and the recovery yield were much lower than expected, making the process economically unfavourable (Lane, 2015). As a result, many of these companies closed or shifted to

conversion of ethanol or biofuels to more valuable compounds such as cosmetic supplements, nutraceuticals, pigments, human and animal feed (Wesoff, 2017).

2.1.2 Bioplastics

Nowadays the term of bioplastic is poorly defined, but in general it represents plastics originating from renewable sources which are fully biodegradable (Li, K., and Juliane 2009). Different studies have investigated bioplastic production, and identified some of the best candidates that fit this description such as PLA and polyhydroxyalkanoates (PHA) (Albuquerque et al., 2007; Braunegg et al., 1998; Shen et al., 2009; Siracusa and Ingrao, 2016). These bioplastics are both polyesters with potential for use in biomedical and food packing, which are synthesized through the fermentation of renewable resources by microorganisms.

Industrial production of PHA started in 2010 with an estimated production of 50,000 tons of commercial grade PHA named Mirel, developed by Metabolix and Archer Daniels Midland Company under the name Telles. However, in 2012 the closure of the plant was announced due to uncertainty around the projected capital and production (DiGregorio, 2009; Laird, 2012; Metabolix 2012). Although some other attempts have been announced (Laird, 2018; Bio-on 2015a), no further industrial production of PHA has been done so far.

The commercial production of PLA is well developed, with several markets including, chemical production for cosmetics, or automotive industry or the food industry. A joint venture between Cargill and Dow started the production of PLA, using corn as a feedstock in 2001 under the trade name NatureWorks (Larson, 2010). Subsequently, Cargill acquired whole control of the company and changed the name to NatureWorks, doubling the production capability to 140,000 tonnes per year.

Sulzer Chemtech opened a pilot plant in Switzerland in 2011. The total production capacity of this plant was 1,000 tonnes of PLA. This plant was equipped with their own technology which relied in a new cost effective polymerization process (Corbion, 2008). The aim of the plant was to show customers the polymerization process on a one-to-one basis before acquiring this technology. Indeed, one of the developments of this company led to an improvement in the heat resistance of the current bioplastics and make them more competitive in price, employing a cost-efficient polymerization process using lactide monomers to produce high-quality PLA. This new PLA product endures temperatures of up to 180°C, making it possible to use this material in the automotive, electronic and textile industries. This technology was implemented by NatureWorks in 2013 to upgrade the quality of the product and the process up to 150,000 tons/year (*Figure 2.1*) (NatureWorks, 2012).





In 2007, a new competitor joined the PLA market, known as Futerro, a joint venture between Total Petrochemicals and Galactic, the former being a producer of lactic acid. They developed a production plant in Belgium in 2010 with a production capacity of 2,000 tonnes/year (Laird, 2011). This joint venture ended in 2016, when Galactic took full ownership of Futerro. At the same time, Total Petrochemicals started a new joint venture with Corbion to develop a new production plant in Thailand. This plant owned by Total Carbion PLA also uses Sulzer's polymerization technology and Corbion's technology. Initially, they developed a pilot plant with a capacity of 1,000 tonnes per year for further optimization prior scaling up to industrial level (Corbion, 2008). Finally, a plant producing 75,000 tonnes of PLA per year was opened in 2019 (Morão and de Bie, 2019).

Since the break-up of the joint venture, Futerro developed the Loopla technology. This technology allows the reconversion of used PLA into lactic acid and subsequent production into new PLA material (Laird, 2011). In 2018, Futerrro, Sulzer and TechnicFmc developed the PLAnet[™] initiative. The aim of the company is to develop, commercialize and support the integral development of PLA facilities of

up to 100,000 tonnes per year (Sulzer, 2019). This includes the production of lactic acid from biomass, purification of lactide and polymerization to obtain PLA.

2.1.3 Succinic acid

DNG Green Technology was formed in 2008 as a spin off from Diversified Natural Products. The new company retained all intellectual property, join ventures, employees and contracts related with to succinic acid production (Runge, 2014). In the same year, they developed a joint venture named Bio-Amber with Abro-Industrie Recherches et Développements (ARD), which resulted in the development and construction of the BioDémo platform in Bazancourt-Pomacle, beside the sugar mill owned by Cristal Union and the ethanol production plant mentioned in section (2.1.1.2) (Runge, 2014; Stadler and Chauvet 2018). This plant initially produced 2,000 cubic tonnes of succinic acid using modified yeast which was developed by the U.S. DOE under the Alternative Feedstocks Program (Ebert, 2007). This bacteria achieved the theoretical maximum conversion of glucose into succinic acid at lab scale, however its implementation at industrial scale resulted in a lower conversion (Carlson et al., 2016). Some other additional problems were related with the cost of base chemicals to prevent decreasing of the pH and resultant salt precipitation during the recovery phase (Hoeven, 2016). These issues were solved at the pilot facility by changing the bacteria into a yeast strain developed by Cargill, operating at a lower pH that also simplified the downstream process (Rokem, 2020). Before finishing the development of the pilot facility, in 2010, DNP Green Technology acquired the whole control of the joint venture, changing its name to Bioamber Inc (Runge, 2014).

Bioamber established a new joint venture with Mitsue & Co in 2011 to develop an industrial plant for succinic acid production in Sarnia (Canada) with an initial capacity of 17,000 tonnes, and potential of up to 35,000 tonnes. Beside succinic acid, the plant aimed for the production of 23,000 tonnes of 1,4-butanediol per year. This joint venture planned the construction of another plant in Thailand by 2012 and a third one in North America or Brazil (Jagger, 2011). However, by 2016 another company, Mater-Biotech (Italy), claimed to open the first commercial biobased 1,4-butanedioal plant in 2016 (Novamont, 2016). In 2017 after Bioamber showed the increasing number of sales, the company bought all the shares to Mitsue and gained the whole control of the company (Songer, 2017). In 2018 Bioamber had liquidity constraints and finally declared bankruptcy (Guzman, 2018). Finally the plant was acquired by LCY Biosciences, who has completed recently the preparations of the plant to resume the production of succinic acid (LCYGroup, 2020).

2.1.4 Biogas production from complex/mixed wastes

Biogas production is traditionally just a tool for waste processing and cost reduction, instead of fuel optimization. For example, in the dairy industry the driving force of using anaerobic instead of aerobic

technologies was to avoid the aeration system and a significant reduction of the biomass production that must be treated downstream (Chetty and Pillay, 2015). However, the number of plants in Europe that are producing biomethane as renewable gas has increased in the period 2018 – 2020 from 483 to 729 plants (EBA, 2020). The countries with the largest number of biomethane plants are Germany, France and the United Kingdom, corresponding to 232, 131 and 80 plants, respectively. In contrast, only one plant in Ireland is producing renewable biogas (EBA, 2020).

An Irish company, Green generation uses wastes from different industries, including prepared industrial food waste, such as dough; lignocellulosic material; animal fats and pig and cow slurry for the production of biogas. This biogas is upgraded within the plant, which allows its injection to the gas grid since 2020 (EBA, 2020). The remaining fraction after the anaerobic digestion is used as fertilizer. Another example of biogas production is Dairygold, a large dairy company in Mitchelstown (Ireland) that reuses the biogas produced in their own plant. This company has built the largest above-ground reactor for treatment of dairy wastewater. Besides their own wastes, this reactor allows the treatment of many wastewaters from industries in the surrounding area.

Incorporating the upcycling process of the biogas to obtain higher additional value compounds could increment the competitiveness of the biorefineries by increasing the range of final products (Hagman et al., 2018). For example, the CO₂ after biogas upgrading, and the liquid fraction after the anaerobic digestion could be used for microalgae production (Jebali et al., 2018). Furthermore, existing biogas plants have been proposed to become a core element of biorefineries (Andersen et al., 2018). Indeed, the anaerobic digestion process could be tuning for the production of chemical compounds if methane production is inhibited.

2.2 Anaerobic digestion

Anaerobic digestion is the process whereby raw material is converted to methane and carbon dioxide by microorganisms in absence of oxygen. These microorganisms co-operate and interact to break down complex organic polymers into smaller molecules, and finally to biogas, mostly composed of methane and CO₂ (Azman et al., 2015). This process mainly consists of four different steps: i) hydrolysis, ii) acidogenesis, iii) acetogenesis and iv) methanogenesis (*Figure 2.2*) (Schink, 1997; Monlau et al., 2011; Kleerebezem et al., 2015).



Figure 2.2 Anaerobic carbon degradation of complex wastes until CH_4 and CO_2 . 1: hydrolytic microorganisms, 2: primary fermenting bacteria, 3: primary fermenting bacteria, 4: secondary-fermenting (syntrophic) bacteria, 5: homoacetogenic bacteria: 6: acetate-cleaving methanogens; 7: hydrogen-oxidizing methanogens (adapted from: Schink, 1997; Kleerebezem et al., 2015)

2.2.1 Hydrolysis

Organic material is decomposed into soluble molecules by hydrolytic microorganisms, such as hydrolytic bacteria, fungi or cellulose-degrading bacteria (Liang et al., 2014). For example, the degradation of proteins to amino acids, lipids to long chain fatty acids, or cellulose and hemicellulose to disaccharides and monosaccharides, e.g. cellobiose, maltose, glucose, xylose or arabinose. However, the accumulation of some of these hydrolysed compounds, such as humic acids or reducing sugars, could inhibit hydrolysis (Andrić et al., 2009; Azman et al., 2015).

2.2.2 Acidogenesis

The decomposition and fermentation of soluble organic substances to short chain compounds is known as acidogenesis. These compounds include lactic acid, ethanol or volatile fatty acids (VFAs) (Nzeteu et al., 2018; Teixeira et al., 2020). The production of the different middle compounds is influenced by the starting substrate. Thus, a change in the substrate composition could shift the production of intermediate compounds. In addition, operational conditions and other factors alter

process functions and could trigger the accumulation of the middle fermented compound (Itoh et al., 2012). If these compounds exceed the inhibitory threshold concentration, it could affect the next steps of the anaerobic digestion process.

2.2.3 Acetogenesis

Bacteria further degrade intermediate compounds to acetate and hydrogen, which are used by methanogenic archaea in the final step. Methanogens can only use acetate, hydrogen, carbon dioxide and one carbon compounds, such as formic acid, to produce methane and carbon dioxide (Schink, 1997). However, if hydrogen is not consumed and the hydrogen partial pressure increases in the system, acetate production becomes thermodynamically unfavourable (Siriwongrungson et al., 2007). Thus, there is an obligate syntrophic relationship between the acetogens and methanogens, whereby methanogens keep hydrogen levels low allowing acetate production to proceed (Schink, 1997; Conrad, 1999).

2.2.4 Methanogenesis

Acetoclastic methanogens use the acetate and hydrogenotrophic methanogens use carbon dioxide and hydrogen for the methane production (Schmidt and Ahring, 1991). Therefore, hydrogenotrophic methanogens are in competition with the homoacetogens for hydrogen availability (Kotsyurbenko et al., 2001). Under normal circumstances, methanogenic hydrogen oxidation is favoured as it is energy favourable (Schink, 1997). Although, there are some situations where homoacetogens are more active, for example, in slightly acidic conditions or at lower temperatures (Schink, 1997, Kotsyurbenko et al., 2001).

Some other situations could result in reduced or completely inhibited methane production. For example, low pH, increased organic loading rate or increased ammonium concentration (Shi et al., 2017; Infantes et al., 2011). This could lead to accumulation of fermented compounds instead of methane, for example, succinic acid, lactic acid or volatile fatty acids. Lactic and succinic acid are currently produced at industrial scale from renewable sources, however in the case of the volatile fatty acids their production relies on fossil sources.

2.3 Volatile fatty acids

Volatile fatty acids are short chain fatty acids (C2 to C6) with a large range of applications and their current industrial production relies on fossil fuel sources (Strazzera et al., 2018). VFAs can be used for the synthesis of a broad range of different chemicals at industrial level (Kim et al., 2018). Many studies have focused on the simultaneous production of VFAs and hydrogen (Garcia-Aguirre et al. 2017; Zhu and Yang 2004; Jiang et al. 2009), however most of these consider hydrogen as the main product and are not focused on optimization of VFA production (Infantes et al., 2011; Kargi et al., 2012; Ghimire et al., 2012; Ghimir

al., 2015). For this last process focused on hydrogen, many researchers use the term of dark fermentation (Cisneros-Perez, 2017). Other researchers focusing on the production of VFAs use the term of acidogenic fermentation (Ramos-Suarez et al., 2021).

The main problem for VFA production at large scales is that their recovery is difficult at the low concentrations obtained during the anaerobic fermentation (Kim et al., 2018), meaning that VFAs are often considered intermediate compounds instead of a final product. Consequently, the main use of the VFAs to date has been for the production of other chemicals, which are easier to recover such biopolymers, lipids or larger carboxylic compounds (Albuquerque et al., 2011; Agler et al., 2014). There are different approaches for VFA production. The accumulation of some chemicals in the broth could inhibit the last steps of the anaerobic fermentation process, for example the increase of the ammonia concentration which led to the collapse of the methane production and inhibit methanogenesis such as, temperature, pH and organic loading rate (Kargi, Eren, and Ozmihci 2012). Biomass pretreatment is also often used to inhibit methanogens, for example heat pretreatment, pH shocks or wash-out (Cisneros-Perez et al., 2017, Pugazhendhi *et al.*, 2017). However, biomass pretreatment can also have a negative impact on homoacetogens, which could decrease the range of the VFA produced (Zhang *et al.*, 2007; Larriba *et al.*, 2016; Rodríguez-Valderrama *et al.*, 2019; Menezes and Silva, 2019; Pugazhendhi *et al.*, 2017).

2.3.1 Production approach

The use of some operational conditions or accumulation of inhibitors could change the final product of the AD process from methane to VFAs (Agler *et al.*, 2013; Lu *et al.*, 2015; Chapleur et al., 2016). However, VFAs can have a self-inhibitory effect if their concentration in the broth is high (Kaur et al., 2020). Therefore, the approach followed during this thesis to prevent methane production was the use of a high organic loading rate that triggered VFA accumulation. This was achieved by using a lignocellulosic biomass as a feedstock to obtain a high concentration of organic compounds. However, the increase of concentration of the VFAs could affect the hydrolysis, acidogenesis and acetogenesis (Siegert and Banks, 2005; Jankowska et al., 2017). Indeed the accumulation of the VFAs or other fermented compounds can trigger the metabolic fluxes resulting in the shift of different fermented compounds over time (Ciranna *et al.*, 2014; Regueira *et al.*, 2018; Saady, 2013), which should be studied for the stability of the process. Also, a similar effect for shifting the end products has been reported when combining and changing different types of wastes overtime (van Aarle et al., 2015).

pH and temperature are key in the production of the VFAs. For example, operating the system under thermophilic conditions improve hydrogen production, meanwhile changing the temperature

to mesophilic conditions resulted in a decrease of hydrogen and increase of the VFA production (Kargi et al., 2011). Lowering temperature was previously demonstrated to shift the primary fermentation product from acetic to butyric acid (Infantes et al., 2011). Operating the system at higher pH resulted in a decrease of butyric acid concentration (Uke and Stentiford, 2013), and an increase in propionic acid production instead of acetic acid (Infantes et al., 2011).

To understand and evaluate the effect of the VFA accumulation and process fermentation, two different approaches were investigated during the development of this thesis. The first was a focus on the simultaneous hydrolysis and acidogenic fermentation of lignocellulosic feedstock (*Chapter 3*). The second approach was based on a two-step process, first hydrolysis was carried out to obtain liquid hydrolysate rich in carbohydrates (*Chapter 4*) then, this liquid hydrolysate was used for VFA production (*Chapter 5*) and further studies was carried by the simultaneous VFA production and recovery (*Chapter 7*).

2.3.1.1 Simultaneous hydrolysis and acidogenesis

The simultaneous degradation and fermentation for VFA production involves the feasibility of treating the feedstock in a single step, resulting in a lower energy demand and the minimal control requirements for simultaneous hydrolysis and production of the fermented compounds. In this case, a long hydraulic retention time (HRT) (10 to 49 days) is required in order to degrade the solid fraction (Li et al., 2014, Nzeteu et al., 2018; Texeira et al., 2020).

Besides physical and chemical treatment, other techniques used to improve anaerobic digestion of lignocellulosic feedstock include, codigestion (Ngan et al., 2020), and bioaugmentation with hydrolytic microorganisms (Martin-Ryals, et al., 2015). Codigestion is widely applied for methane production from complex feedstock, for example the use of lignocellulose or feedstock with a high lipid content as the sole carbon source could result in an unbalance in the anaerobic digestion process (Ngan et al., 2020). This approach is used at industrial level for the degradation of complex mixtures of feedstock (Angelidaki and Ellegaard, 2003). Nevertheless, some lignocellulosic feedstock is still too recalcitrant for microorganisms to break down and additional pretreatment is required (Singh and Kumar, 2019). As an alternative, bioaugmentation can be used to improve the native hydrolytic community for degradation of cellulose and hemicellulose by introducing microorganisms with a better hydrolytic activity (Haruta et al., 2002; Aydin et al., 2017; Tsapekos et al., 2017). This has been successfully used in batch reactors (Peng et al., 2014; Baba et al., 2017). However, some difficulties can occur in continuous reactors when the introduced microorganisms do no develop or stabilised population over time (Tsapekos et al., 2017; Li et al., 2020).

Some of the most common reactors for simultaneous hydrolysis and fermentation are semicontinuous stirred tank reactors (Li et al., 2014), or leaching bed reactors (Cysneiros et al., 2012; Uke and Stentiford, 2013; Li et al., 2017; Nzeteu et al., 2018; Texeira et al., 2020). In all of them, one of the major problems is the accumulation of the VFAs. This, could affect the hydrolysis of even simple molecules such as glucose (Siegert and Banks, 2005). Leaching bed reactors have been shown to be effective when operating with food waste as feedstock (Want et al., 2014; Nzeteu et al., 2018; Texeira et al., 2020). However, food waste is readily degradable and it is possible to obtain a larger amount of VFAs then when using more complex wastes, such as cellulosic feedstock (Cysneiros et al., 2012; Li et al., 2017; Texeira et al., 2020). Therefore, the screening for a suitable source of inocula with an active hydrolytic activity could improve the performance for the biodegradation of lignocellulose and simultaneous production of VFAs.

2.3.1.2 Two-step process

The main advantages of using a two-step process are a higher control of the operational conditions and a much lower hydraulic retention time. For simultaneous hydrolysis and acidogenesis the total time could be up to 49 days in a two-step system, whereas the liquid fermentation could be carried within hours. The most typical reactor configuration for the production of VFAs in the liquid phase are continuous stirring tank reactor (CSTR) (Bengtsson et al., 2008; Shen et al., 2014; Palomo-Briones et al., 2017), up-flow anaerobic sludge bed (UASB) reactors (Mota et al., 2018; Rodriguez-Valderrama et al., 2019), fixed bed reactors (Zhang et al., 2008; Pugazhendhi et al., 2017; Gorgec and Karapinar, 2019), packed bed reactors (Domingos et al., 2017) and expanded granular sludge bed (EGSB) reactors (de Menezes and Silva, 2019).

Two-step processes have been largely applied for hydrogen production, with volatile fatty acids considered as a by-product. Some studies have even used the resultant VFAs for the production of methane and did not consider the recovery or use of VFAs for any upgrading process (Nkongndem et al., 2015). As well, some of these studies have been more focusing on hydrogen production rather than VFAs production, pretreating the biomass in order to decrease not only the methanogens, but also homoacetogens that could consume the hydrogen in favour of VFA production (Carrillo-Reyes et al., 2012, Antonopoulou et al., 2016, Carrillo-Reyes et al., 2016).

Reactor configuration and the aggregation state of the mixed microbial culture, for example granular or flocculent sludge, have a strong influence on the microbial selection pressure by the operational conditions of the system (Cisneros-Pérez et al., 2015; Atasoy et al., 2019). For example, using the same feedstock and same pretreatment, an anaerobic fluidized-bed reactor with an up-flow velocity of 13 m/h led to a higher hydrogen and lower VFA production than an EGSB reactor an up-

flow velocity of 3 m/h (Cisneros-Pérez et al., 2015; Cisneros-Pérez et al., 2017). In order to increase hydrogen yield and volumetric production, short HRTs have been beneficial and resulted in butyric acid as the major product due to the wash-out of homoacetogens (Pugazhendhi et al., 2017; Gorgec and Karapinar, 2019; Rodriguez-Valderrama et al., 2019; De Menezes and Silva, 2019). Indeed, the use of larger HRTs can result in a more complex mixture of VFAs than a short HRT (Zhang et al., 2007). Even though inoculum pretreatment has a great impact on the initial hydrogen and VFA production, the operational parameters of the reactor have a more extended influence on VFA production (Cisneros-Pérez et al., 2015).

2.3.1.3 Volatile fatty acid recovery

Many different technologies have been evaluated for the recovery of VFAs, including liquid-liquid extraction (Reyhanitash et al., 2016; Rocha et al., 2017; Saboe et al., 2018), filtration (Zacharof and Lovitt, 2015), electrodialysis (Jones et al., 2015; Scoma et al., 2016; Tao et al., 2016; Bak et al., 2019), and microbial electrochemical cells (Zhang and Angelidaki, 2015). In addition, the emergence of new membranes in the market is attractive as their cost is decreasing and they are able to achieve a higher selectivity towards VFAs (Cohen-Tanugi et al., 2016).

The use of different compounds as extractants, additives such as amines, esters or organophosphourus compound, in the diluent during the liquid-liquid VFA recovery can enhance the process efficiency by up to 85 % (Kaur et al., 2020). The recovery of a larger broth of VFAs that contains caproic acid would result more efficient when using organic solvents due to the increasing alkyl length of the VFA (Rocha et al., 2017). Therefore, promoting the elongation of short volatile fatty acids into longer VFAs, for example acetic or butyric to caproic acid, could be beneficial to achieve a higher recovery efficiency. A two-step process would allow the recovery of the VFAs from the aqueous fraction using heavy solvents and extracts, followed by distillation of the VFA containing fraction (Saboe et al., 2018). Some parameters that affect the equilibrium between the VFAs in the aqueous and the organic liquid fraction are pH, acid concentration and ratio between permeate and retentate (Rocha et al., 2017; Saboe et al., 2018). An alternative method would be to promote the elongation for caproic acid production, which is easier to recover (Xu et al., 2020). The increase of the caproic acid concentration above 10 g/L in its undissociated form would result in the formation of a second hydrophobic phase allowing the easier recovery of the caproic acid (Xu et al., 2015).

Electrodyalysis combines a membrane separation system with electrochemical interaction to separate cations and anions by their electrical potential differences (Bak et al., 2019). Baroi et al. (2015) recovered butyric acid, which contained a low concentrations of acetic acid, using a pure culture in reverse electro enhanced dialysis technology, followed by the distillation of the butyric and

acetic acid using octanol. This purification step represents 35 % of the total operation costs (Baroi et al., 2017). Some important parameters for VFA recovery using electrodyalysis are the applied voltage, the volumetric ratio of the broth that contains VFAs, and the liquid fraction for the VFA recovery (Bak et al., 2019). Similar approaches have been used for lactic acid recovery and purification (González et al., 2008; Komesu et al., 2017). Zhang and Angelidaki (2015) used a microbial electrodialysis cell, by combining electrodialysis with microbial electrochemical systems. Thus, the required potential could be obtained by the oxidation of the organic matter and a lower electricity input was thus required. An additional advantages of this system was the production of hydrogen and alkali, which could contribute to the neutralization of the pH during VFA production and decreasing chemical requirements for the process (Zhang and Angelidaki, 2015).

2.4 Lignocellulosic feedstock

Lignocellulose represents the most abundant biomass which could act as a source of carbohydrates for bio-industrial purposes. However, due to its complex and recalcitrant structure, where cellulose and hemicellulose are interwoven within lignin, microorganisms cannot degrade and use it (*Figure 2.3*) (Ayyachamy et al., 2013). Cellulose is a linear semi-crystalline biopolymer formed by D-glucose units, hemicellulose is a heterogeneous polysaccharide formed by xylose, arabinose (pentoses), glucose, galactose and mannose (hexose); while lignin is an aromatic biopolymer (Ravindran and Jaiswal, 2016). Properties and the ratio of cellulose, hemicellulose and lignin vary between different types of lignocellulosic biomass (Roy et al., 2020).



Figure 2.3 Lignocellulose structure before and after pretreatment to break down the lignin and cellulose crystallinity to facilitate the degradation of the microorganisms for production of simple carbohydrates (adapted from Mosier et al., 2005). \triangleleft enzymes, \blacksquare lignin, \triangleleft cellulose, \Re hemicellulose, and \blacklozenge proteins

Only a small group of microorganisms have the ability to break down the lignin (Sanchez, 2009). Thus, to make this material susceptible for microbial degradation it is important to break the lignin outer layer and crystalline structure of the cellulose (Mosier et al., 2005; López-Linares et al., 2020). Physical processing of the material does not largely contribute to improve enzymatic digestibility and is quite limited when compared with chemical pretreatment methods (Niemi et al., 2012; Niemi et al., Chapter 2

2013). The use of acid method is quite extended, with sulfuric acid used the most (Wagner et al., 2021). However, this approach could also convert the carbohydrates into compounds which inhibit downstream fermentation processes, such as furfural or 5- hydroxymethylfurfural (Yoon et al., 2014; Tizazu and Moholkar, 2017). Alkali hydrolysis has also been studied as pretreatment for the breakdown of the lignocellulose (Ravindran et al., 2018) or organosolvent pretreatment targeting lignin degradation, having both a more limited effect on cellulose degradation than acid pretreatment (Patel et al., 2018; Oliva et al., 2020). Additionally, the use of ionic liquids is increasing as an alternative to traditional organic solvents due to lower vapour pressure and being more environmental friendly (Liu and Chen, 2006). Ionic liquids are organics salts composed of organic cations and an organic or inorganic anion (Karimi et al., 2013). Based on the nature of the ionic liquid, this could dissolve the lignin structure to facilitate enzymatic hydrolysis (Ma et al., 2016). In addition, the ionic liquid could be recycled for sequential pretreatments without losing its potential (Xu et al., 2017).

The different combined parameters for the chemical hydrolysis (temperature, time and acid concentration) can be integrated in a single value known as combined severity factor. This factor is based on the Arrhenius equation, but it fits the multiphasic behaviour for lignocellulose degradation with an initial high rate of lignin loss of pseudo-first-order, followed by the bulk-phase delignification, and a very slow final dissolution (Chum et al., 1990). This factor was widely evaluated for the effectiveness of carbohydrate production (Carvalheiro et al., 2004), or as pretreatment to enhance the enzymatic hydrolysis (Zoulikha et al., 2015).

One parameter used to determine the degradation of the lignocellulose structure is the water retention capacity. This represents the ratio between the amount of water retained and the weight of the solid fraction, which is affected by the composition of the material, the particle size and pore size (Robertson et al., 2000; Raghavendra et al., 2004, Saw et al., 2008; Lara-Vazquez et al., 2014). Based on the severity of the pretreatment and the change of the composition, the ability of the lignocellulosic material for water retention could increase or decrease. For example, dissolving fats in the lignocellulosic matrix increases the water retention capacity (Raghavendra et al., 2004), whereas the destruction of fibre results in a loss of water retention capacity (Célino et al., 2014). Therefore, the hydration dynamic can contribute to understanding the effectiveness of the applied treatment (Sanchez et al., 2019) and could predict the status of the lignocellulosic feedstock prior to enzyme addition.

Many studies focused on different pretreatment methods for the recovery of high-value compounds, such as fibres, proteins or phenolic compounds from lignocellulosic material, instead of carbohydrates (Martínez et al., 2012; Valadez-Carmon et al., 2018). The valorisation of current

lignocellulose residues into more valuable products would bring new opportunities for industrial development in the biorefinery field. For example, the combination of mechanical homogenization, thermal hydrolysis and chemical precipitation for recovery and purification of pectin using cocoa waste as feedstock (Muñoz-Almagro et al, 2019). This biorefinery approach using cocoa wastes would use a series of different steps to extract polyphenols, antioxidant compounds and use the remaining fraction for chemical and biogas production (Martínez et al. 2012; Valadez-Carmona et al. 2018; Handojo et al., 2019). Some examples are the recovery of high value compounds such as pectin and phenolic compounds from pomegranate peels with production of bioethanol using the remaining solid fraction (Talekar et al., 2018), or the recovery of different compounds from citrus peel wastes such as essential oils, pectin, succinic acid and fertilizer (Marin et al., 2007; Patsalou et al., 2020).

Certainly, some of these products have a higher value than carbohydrates and their recovery should be targeted during the production of carbohydrates. For example, as explained in section **2.1.1.4** of the current chapter, the use of microalgae for carbohydrate and ethanol production was considered a failure, however, shifting ethanol to higher value compounds made the process feasible (Wesoff, 2017). Thus, hydrolysis should be focused on the fractionation and recovery of the different high value compounds from lignocellulosic feedstock and resultant residues can be used for carbohydrate or biogas production. This approach is known as cascade process, which becomes an important part of the biorefinery approach developed during the current PhD thesis. Brewery spent grain (BSG) is a by-product produced in large volumes, with low or no value from beer preparation with a stable composition, rich in proteins, carbohydrates and lignin, with a constant supply. This makes brewery spent grain an optimal candidate for a biorefinery using a cascade process for recovery of the different high valuable compounds.

2.5 Brewery spent grain

The basic ingredients for beer production are water, barley, hops and yeast. Barley is processed by briefly submerging it in a tank with water and placing it in a germination vessel to promote the natural release of some of the carbohydrates, such as the starch contained in the endosperm (*Figure 2.4*). Before germination is complete, the grains are dried to a final water content between 4 - 5 % to avoid the consumption of the carbohydrates available inside the grain. This is known as malted barley and is then ground, mixed with water and boiled to obtain a sweet liquor (Arendt and Zannini, 2013). This sweet liquid, known as the wort, is recovered for downstream processing, and the residual solid fraction left with none or low value is known as brewery spent grain. The latter is the most abundant solid waste from the brewery industry with, approximately, around 15 – 20 kg of BSG produced per hectolitre of beer (Mussatto *et al.*, 2006). Just in Ireland, in 2019, the production of beer was above

8.2 million cubic meters (IBA, 2020). Thus, over 170,000 tonnes of this brewery spent grain was produced.

Brewery spent grain has a dry weight content of around 23 - 31 % of proteins, 19 - 26 % lignin, 17 - 22 % hemicellulose and 25 - 31 % cellulose (Mussatto *et al.*, 2006; Vieira *et al.* 2014; Rommi *et al.* 2018). Some compounds of brewery spent grain have shown prebiotic and immunomodulatory activity. The antioxidant and phenolic content strongly depends on the variety of barley and the malting processes during the brewery process (Moreira et al., 2013).

The recovery of carbohydrates from brewery spent grain has been largely investigated to use them for production of different chemicals, including, for production of bioethanol (Wilkinson et al., 2014; Liguori et al., 2015; Rojas-Chamorro et al., 2020), biobutanol (Plaza et al., 2017; Giacobble et al., 2019; López-Linares et al., 2020), lactic acid (Mussatto et al., 2007; Pejin et al., 2018) or lipid (Patel et al., 2018). Some other researchers have focused on the recovery of bioactive compounds (Cermeño et al. 2019), phenolic extracts (McCarthy et al., 2012; Stefanello et al., 2018; Connolly et al., 2019) or protein extraction (Niemi et al., 2013; Connolly et al., 2013; Qin et al., 2018). However, there are few studies that evaluate the simultaneous recovery of carbohydrates and other fractions such as lignin or proteins (Niemi et al., 2013; Rommi et al., 2018). The protein and carbohydrate fractions are closely related, as the latter behave as barrier for enzymatic and mild chemical protein extraction (Niemi et al., 2013). The husk, composed mainly of cellulose with a small fraction of lignin, and the pericarp act as protective cover for the grain (Jaeger et al., 2021). Inside, the grain the aleurone that surrounds the endosperm costumes of a protein rich fraction (*Figure 2.4*). As well, additional protein are interwoven with hemicellulose forming additional cell walls such as the scutellum (Arendt and Zannini, 2013).



Figure 2.4 Structure of the barley kernel a) sectioned transversely, b) detailed husk, and c) starchy endosperm (Arendt and Zannini, 2013). Permission for using the image from Elsevier

The recovery of different fractions from brewery spent grain, including a protein rich solid fraction, a phenolic rich solid fraction and a carbohydrate rich liquid fraction could contribute to the development of a biorefinery. Furthermore, if a cascade process could be carried out within the brewery, even the protein and phenolic fraction could be used as biospecialities for food ingredients, or as source of human and animal dietary products and prebiotic compounds (Cermeño et al., 2019). As the mixed carbohydrates in remaining liquid fraction does not have high value, this can be used in a downstream fermentation for production of chemicals, such as the volatile fatty acids.

2.6 References

- Agler M. T., Spirito C. M., Usack J. G., Werner J. J., Angenent L. T. (2013) "Development of a highly specific and productive process for n-caproic acid production: applying lessons from methanogenic microbiomes." Water Science and Technology 69 (1): 62-68. https://doi.org/10.2166/wst.2013.549.
- Albuquerque M. G. E., Eiroa M., M. Torres M., Nunes B. R., Reis M. A. M. (2007) "Strategies for the development of a side stream process for polyhydroxyalkanoate (PHA) production from sugar cane molasses." Journal of Biotechnology 130 (4): 411-421. https://doi.org/10.1016/j.jbiotech.2007.05.011.
- Albuquerque M. G. E., Martino V., Pollet E., Averous L., Reis M. A. M. (2011) "Mixed culture polyhydroxyalkanoate (PHA) production from volatile fatty acid (VFA)-rich streams: Effect of substrate composition and feeding regime on PHA productivity, composition and properties." Journal of Biotechnology 151 (1): 66-76. https://doi.org/10.1016/j.jbiotec.2010.10.070.

- Andersen L., Lamp A., Dieckmann C., Baetge S., Schmidt L.-M., Kaltschmitt M. (2018) "Biogas plants as key units of biorefinery concepts: Options and their assessment." Journal of Biotechnology 283: 130-139. https://doi.org/10.1016/j.jbiotec.2018.07.041.
- Andrić P., Meyer A. S., Jensen P. A., Dam-Johansen K. (2009) "Effect and Modeling of Glucose Inhibition and In Situ Glucose Removal During Enzymatic Hydrolysis of Pretreated Wheat Straw." Applied Biochemistry and Biotechnology 160 (1): 280. https://doi.org/10.1007/s12010-008-8512-9
- Angelidaki I., Ellegaard L. (2003) "Codigestion of manure and organic wastes in centralized biogas plants." Applied Biochemistry and Biotechnology 109 (1): 95-105. https://doi.org/10.1385/ABAB:109:1-3:95.
- Antonopoulou, Georgia, Dimitrios Vayenas, and Gerasimos Lyberatos. 2016. "Ethanol and hydrogen production from sunflower straw: The effect of pretreatment on the whole slurry fermentation." Biochemical Engineering Journal 116 (Supplement C): 65-74. https://doi.org/10.1016/j.bej.2016.06.014.
- Arendt, Elke, and Emanuele Zannini. 2013. "Barley." 155-201e.
- Atasoy M., Eyice O., Schnürer A., Cetecioglu Z. (2019) "Volatile fatty acids production via mixed culture fermentation: Revealing the link between pH, inoculum type and bacterial composition." Bioresource Technology 292: 121889. https://doi.org/10.1016/j.biortech.2019.121889.
- Aydin, Sevcan, Elif Yıldırım, Orhan Ince, and Bahar Ince. 2017. "Rumen anaerobic fungi create new opportunities for enhanced methane production from microalgae biomass." Algal Research 23 (Supplement C): 150-160. https://doi.org/10.1016/j.algal.2016.12.016.
- Ayyachamy M., Cliffe F. E., Coyne J. M., Collier J., Tuohy M. G. (2013) "Lignin: untapped biopolymers in biomass conversion technologies." Biomass Conversion and Biorefinery 3 (3): 255-269. https://doi.org/10.1007/s13399-013-0084-4.
- Azman S., Khadem A. F., van Lier J. B., Zeeman G., Plugge C. M. (2015) "Presence and Role of Anaerobic Hydrolytic Microbes in Conversion of Lignocellulosic Biomass for Biogas Production." Critical Reviews in Environmental Science and Technology 45 (23): 2523-2564. https://doi.org/10.1080/10643389.2015.1053727.
- Baba Y., Matsuki Y., Mori Y., Suyama Y., Tada C., Fukuda Y., Saito M., Nakai Y. (2017) "Pretreatment of lignocellulosic biomass by cattle rumen fluid for methane production: Bacterial flora and enzyme activity analysis." Journal of Bioscience and Bioengineering 123 (4): 489-496. https://doi.org/10.1016/j.jbiosc.2016.11.008.
- Bailey, Brent, and Chris Colucci. 1997. "Research and development on the utilization of alcohol fuels at the National Renewable Energy Laboratory." Renewable Energy 10 (2): 273-278. https://doi.org/10.1016/0960-1481(96)00077-8.
- Bak C., Yun Y.-M., Kim J.-H., Kang S. (2019) "Electrodialytic separation of volatile fatty acids from hydrogen fermented food wastes." International Journal of Hydrogen Energy 44 (6): 3356-3362. https://doi.org/10.1016/j.ijhydene.2018.07.134.
- Baroi G. N., Gavala H. N., Westermann P., Skiadas I. V. (2017) "Fermentative production of butyric acid from wheat straw: Economic evaluation." Industrial Crops and Products 104 (Supplement C): 68-80. https://doi.org/https://doi.org/10.1016/j.indcrop.2017.04.008.

- Baroi G. N., Skiadas I. V., Westermann P., Gavala H. N. (2015) "Continuous Fermentation of Wheat Straw Hydrolysate by Clostridium tyrobutyricum with In-Situ Acids Removal." Waste and Biomass Valorization 6 (3): 317-326. https://doi.org/10.1007/s12649-015-9348-5.
- Bengtsson, Simon, Jakob Hallquist, Alan Werker, and Thomas Welander. 2008. "Acidogenic fermentation of industrial wastewaters: Effects of chemostat retention time and pH on volatile fatty acids production." Biochemical Engineering Journal 40 (3): 492-499. https://doi.org/http://dx.doi.org/10.1016/j.bej.2008.02.004.
- Bio-on, 2015, "World first facility for producing Bioplastic from biodiesel co-product glycerol, to be realised in Italy. Industrial licence granted," http://www.bio-on.it/immagini/comunicati-finanziari/CS_45_BIO-ON_22_12_2015_ENG.pdf.
- Bozell J., Petersen G.R. (2010) Technology development for the production of biobased products from biorefinery carbohydrates—the US Department of Energy's "Top 10" revisited, Green Chemistry 12(4) 525-728. https://doi.org/10.1039/B922014C
- Braunegg, Gerhart, Gilles Lefebvre, and Klaus F. Genser. 1998. "Polyhydroxyalkanoates, biopolyesters from renewable resources: Physiological and engineering aspects." Journal of Biotechnology 65 (2–3): 127-161. https://doi.org/http://dx.doi.org/10.1016/S0168-1656(98)00126-6
- Carlson, Alfred, Bill Coggio, Kit Lau, Christopher Mercogliano, and Jim Millis. 2016. "Industrial Production of Succinic Acid." In Chemicals and Fuels from Bio-Based Building Blocks, 173-190.
- Carrillo-Reyes, Julián, Lourdes B. Celis, Felipe Alatriste-Mondragón, and Elías Razo-Flores. 2012. "Different start-up strategies to enhance biohydrogen production from cheese whey in UASB reactors." International Journal of Hydrogen Energy 37 (7): 5591-5601. https://doi.org/10.1016/j.ijhydene.2012.01.004.
- Carrillo-Reyes, Julián, Miguel Angel Cortés-Carmona, Christian Daniela Bárcenas-Ruiz, and Elías Razo-Flores. 2016. "Cell wash-out enrichment increases the stability and performance of biohydrogen producing packed-bed reactors and the community transition along the operation time." Renewable Energy 97 (Supplement C): 266-273. https://doi.org/10.1016/j.renene.2016.05.082.
- Carvalheiro F., Duarte L. C., Medeiros R., Gírio F. M. (2004) "Optimization of Brewery's spent grain dilute-acid hydrolysis for the production of pentose-rich culture media." Applied Biochemistry and Biotechnology 115 (1-3): 1059-1072. https://doi.org/10.1385/ABAB:115:1-3:1059.
- Célino, Amandine, Sylvain Freour, Frederic Jacquemin, and Pascal Casari. 2014. "The hygroscopic behavior of plant fibers: a review." Frontiers in Chemistry 1 (43). https://doi.org/10.3389/fchem.2013.00043.
- Cermeño M., Connolly A., O'Keeffe M. B., Flynn C., Alashi A. M., Aluko R. E., FitzGerald R. J. (2019) "Identification of bioactive peptides from brewers' spent grain and contribution of Leu/Ile to bioactive potency." Journal of Functional Foods 60: 103455. https://doi.org/10.1016/j.jff.2019.103455.
- Chapleur O., Madigou C., Civade R., Rodolphe Y., Mazéas L., Bouchez T. (2016) "Increasing concentrations of phenol progressively affect anaerobic digestion of cellulose and associated microbial communities." Biodegradation 27 (1): 15-27. https://doi.org/10.1007/s10532-015-9751-4.

- Chetty, Siva, and Kaverajen Pillay. 2015. "Application of the DIY carbon footprint calculator to a wastewater treatment works." Water SA 41: 263-272. http://www.scielo.org.za/scielo.php?script=sci_arttext&pid=S1816-79502015000200013&nrm=iso.
- Chum, Helena L., David K. Johnson, Stuart K. Black, and Ralph P. Overend. 1990. "Pretreatment-Catalyst effects and the combined severity parameter." Applied Biochemistry and Biotechnology 24 (1): 1. https://doi.org/10.1007/BF02920229.
- Ciranna, Alessandro, Roberto Ferrari, Ville Santala, and Matti Karp. 2014. "Inhibitory effects of substrate and soluble end products on biohydrogen production of the alkalithermophile Caloramator celer: Kinetic, metabolic and transcription analyses." International Journal of Hydrogen Energy 39 (12): 6391-6401. https://doi.org/10.1016/j.ijhydene.2014.02.047.
- Cisneros-Pérez C., Etchebehere C., Celis L. B., Carrillo-Reyes J., Alatriste-Mondragón F., Razo-Flores E. (2017) "Effect of inoculum pretreatment on the microbial community structure and its performance during dark fermentation using anaerobic fluidized-bed reactors." International Journal of Hydrogen Energy 42 (15): 9589-9599. https://doi.org/10.1016/j.ijhydene.2017.03.157.
- Cisneros-Pérez C., Carrillo-Reyes J., Celis L. B., Alatriste-Mondragón F., Etchebehere C., Razo-Flores E. (2015) "Inoculum pretreatment promotes differences in hydrogen production performance in EGSB reactors." International Journal of Hydrogen Energy 40 (19): 6329-6339. https://doi.org/10.1016/j.ijhydene.2015.03.048.
- Cohen-Tanugi, David, Li-Chiang Lin, and Jeffrey C. Grossman. 2016. "Multilayer Nanoporous Graphene Membranes for Water Desalination." Nano Letters 16 (2): 1027-1033. https://doi.org/10.1021/acs.nanolett.5b04089.
- Connolly A., Cermeño M., Crowley D., O'Callaghan Y., O'Brien N. M., FitzGerald R. J. (2019) "Characterisation of the in vitro bioactive properties of alkaline and enzyme extracted brewers' spent grain protein hydrolysates." Food Research International 121: 524-532. https://doi.org/10.1016/j.foodres.2018.12.008.
- Connolly A., Piggott C., FitzGerald R. (2013) "Characterisation of protein-rich isolates and antioxidative phenolic extracts from pale and black brewers' spent grain". Vol. 48.
- Conrad, R. 1999. "Contribution of hydrogen to methane production and control of hydrogen concentrations in methanogenic soils and sediments." FEMS Microbiology Ecology 28 (3): 193-202. https://doi.org/10.1111/j.1574-6941.1999.tb00575.x.
- Corbion, 2008, "PURAC-SULZER develop new polymerization technology for PLA".
- Cysneiros, Denise, Charles J. Banks, Sonia Heaven, and Kimon-Andreas G. Karatzas. 2012. "The effect of pH control and 'hydraulic flush' on hydrolysis and Volatile Fatty Acids (VFA) production and profile in anaerobic leach bed reactors digesting a high solids content substrate." Bioresource Technology 123 (Supplement C): 263-271. https://doi.org/10.1016/j.biortech.2012.06.060.
- Dale, Bruce. 2018. "Time to Rethink Cellulosic Biofuels?" Biofuels, Bioproducts and Biorefining 12 (1): 5-7. https://doi.org/10.1002/bbb.1856.
- Domingos, Joana M. B., Gonzalo A. Martinez, Alberto Scoma, Serena Fraraccio, Frederiek-Maarten Kerckhof, Nico Boon, Maria A. M. Reis, Fabio Fava, and Lorenzo Bertin. 2017. "Effect of

Operational Parameters in the Continuous Anaerobic Fermentation of Cheese Whey on Titers, Yields, Productivities, and Microbial Community Structures." ACS Sustainable Chemistry & Engineering 5 (2): 1400-1407. https://doi.org/10.1021/acssuschemeng.6b01901.

- Duffield, James, Robert Johansson, and Seth Meyer. 2015. U.S. Ethanol: An Examination of Policy, Production, Use, Distribution, and Market Interactions.
- DuPont. 2018. DuPont Tate & Lyle expands biobased propanediol production. Biomass Magazine.
- E4tech. 2017. UK Top Bio based Chemicals Opportunities.
- EBA. 2020. The 'European Biomethane Map 2020' shows a 51% increase of biomethane plants in Europe in two years.
- Ebert, Jessica. 2007. The Quest to Commercialize Biobased Succinic Acid. Biomass Magazine.
- Erwin, Vink, and Davies Steve. 2015. "Life Cycle Inventory and Impact Assessment Data for 2014 Ingeo™ Polylactide Production." Industrial Biotechnology 11 (3). https://doi.org/https://doi.org/10.1089/ind.2015.0003.
- de Farias Silva, Carlos Eduardo, and Alberto Bertucco. 2016. "Bioethanol from microalgae and cyanobacteria: A review and technological outlook." Process Biochemistry 51 (11): 1833-1842. https://doi.org/https://doi.org/10.1016/j.procbio.2016.02.016.
- Ghimire, Anish, Luigi Frunzo, Francesco Pirozzi, Eric Trably, Renaud Escudie, Piet N. L. Lens, and Giovanni Esposito. 2015. "A review on dark fermentative biohydrogen production from organic biomass: Process parameters and use of by-products." Applied Energy 144 (Supplement C): 73-95. https://doi.org/10.1016/j.apenergy.2015.01.045.
- Giacobbe, Simona, Alessandra Piscitelli, Francesca Raganati, Vincenzo Lettera, Giovanni Sannia, Antonio Marzocchella, and Cinzia Pezzella. 2019. "Butanol production from laccasepretreated brewer's spent grain." Biotechnology for Biofuels 12 (1): 47. https://doi.org/10.1186/s13068-019-1383-1.
- Global-Bioenergies, 2015, "Cristal Union and Global Bioenergies have formed a joint venture to build and operate France's first bio-sourced isobutene production plant," http://www.globalbioenergies.com/wp-content/uploads/2015/05/20150521_pr_en.pdf.
- González, M. Isabel, Silvia Alvarez, Francisco A. Riera, and Ricardo Álvarez. 2008. "Lactic acid recovery from whey ultrafiltrate fermentation broths and artificial solutions by nanofiltration." Desalination 228 (1): 84-96. https://doi.org/10.1016/j.desal.2007.08.009.
- GranBio, 2015, "GranBio obtains commercial approval for its first proprietary yeast," http://www.granbio.com.br/en/wp-content/uploads/sites/2/2015/07/GranBio-obtainscommercial-approval-for-its-first-proprietary-yeast-.pdf.
- GranBio, 2018. "Energy cane." Accessed April. http://www.granbio.com.br/en/conteudos/energycane/.
- GranBio, 2020, "GranBio Acquires 100% Equity of the North American Biotechnology Firm American Process Inc. (API)," http://www.granbio.com.br/en/wpcontent/uploads/sites/2/2020/03/GranBio-Acquires-API.pdf.

Di Gregorio, Barry E. 2009. "Biobased Performance Bioplastic: Mirel." Chemistry & Biology 16 (1): 1-2. https://doi.org/10.1016/j.chembiol.2009.01.001.

Guzman, Doris De. 2018. BioAmber files for Bankruptcy. Green Chemicals Blog. Accessed 02/06/2020.

- Hagman, Linda, Alyssa Blumenthal, Mats Eklund, and Niclas Svensson. 2018. "The role of biogas solutions in sustainable biorefineries." Journal of Cleaner Production 172: 3982-3989. https://doi.org/10.1016/j.jclepro.2017.03.180.
- Handojo, Lienda, Harry Triharyogi, and Antonius Indarto. 2019. "Cocoa bean shell waste as potential raw material for dietary fiber powder." International Journal of Recycling of Organic Waste in Agriculture 8 (1): 485-491. https://doi.org/10.1007/s40093-019-0271-9.
- Haruta, S., Cui Z., Huang Z., Li M., Ishii M., Igarashi Y. (2002) "Construction of a stable microbial community with high cellulose-degradation ability." Applied Microbiology and Biotechnology 59 (4): 529-534. https://doi.org/10.1007/s00253-002-1026-4.
- Helal, Christopher J., Alessandra Bartolozzi, Steven D. Goble, Neelakanda S. Mani, Angel Guzman-Perez, Ajay K. Ohri, Zhi-Cai Shi, and Chakrapani Subramanyam. 2018. "Increased building block access through collaboration." Drug Discovery Today. https://doi.org/10.1016/j.drudis.2018.03.001.
- Hoeven, Diederik van der. 2016. BioAmber and Reverdia agreement: strengthening the biobased economy. Bio Based Press.
- IBA, 2020. Irish Beer Market Report 2020. https://www.ibec.ie/drinksireland/-/media/documents/drinks-ireland-new-website/beer/beer-market-report-2020.pdf. Accessed on 29th of May 2021
- Infantes, D., González del Campo A., Villaseñor J., Fernández F. J. (2011) "Influence of pH, temperature and volatile fatty acids on hydrogen production by acidogenic fermentation." International Journal of Hydrogen Energy 36 (24): 15595-15601. https://doi.org/10.1016/j.ijhydene.2011.09.061.
- Institute, Nickel. 2014. "Really Green Ethanol." Accessed 26th April.
- Itoh Y., Tada K., Kanno T., Horiuchi J.-I. (2012) "Selective production of lactic acid in continuous anaerobic acidogenesis by extremely low pH operation." Journal of Bioscience and Bioengineering 114 (5): 537-539. https://doi.org/10.1016/j.jbiosc.2012.05.020.
- Jaeger, Alice, Emanuele Zannini, Aylin W. Sahin, and Elke K. Arendt. 2021. "Barley Protein Properties, Extraction and Applications, with a Focus on Brewers' Spent Grain Protein." Foods 10 (6). https://doi.org/10.3390/foods10061389.
- Jagger, Anna. 2011. Projects: BioAmber, Mitsui plan succinic acid, BDO projects. Independent Commodity Intelligence Services. Accessed 02/06/2020.
- Jankowska E., Chwialkowska J., Stodolny M., Oleskowicz-Popiel P. (2017) "Volatile fatty acids production during mixed culture fermentation – The impact of substrate complexity and pH." Chemical Engineering Journal 326: 901-910. https://doi.org/10.1016/j.cej.2017.06.021.
- Jebali, Ahlem, F. Gabriel Acién, Erika Rodriguez Barradas, Eugenia J. Olguín, Sami Sayadi, and Emilio Molina Grima. 2018. "Pilot-scale outdoor production of Scenedesmus sp. in raceways using

flue gases and centrate from anaerobic digestion as the sole culture medium." Bioresource Technology 262: 1-8. https://doi.org/10.1016/j.biortech.2018.04.057.

- Jones, David S. J. Stan. 2006. "An introduction to crude oil and its processing." In Handbook of Petroleum Processing, edited by David S. J. Stan Jones and Peter R. Pujadó, 1-45. Dordrecht: Springer Netherlands.
- Jones, Rhys Jon, Jaime Massanet-Nicolau, Alan Guwy, Giuliano C. Premier, Richard M. Dinsdale, and Matthew Reilly. 2015. "Removal and recovery of inhibitory volatile fatty acids from mixed acid fermentations by conventional electrodialysis." Bioresource Technology 189 (Supplement C): 279-284. https://doi.org/https://doi.org/10.1016/j.biortech.2015.04.001.
- Karaosmanoglu Gorgec, Firuze, and Ilgi Karapinar. 2019. "Biohydrogen production from hydrolyzed waste wheat by dark fermentation in a continuously operated packed bed reactor: The effect of hydraulic retention time." International Journal of Hydrogen Energy 44 (1): 136-143. https://doi.org/https://doi.org/10.1016/j.ijhydene.2018.08.155.
- Kargi, Fikret, Nur Seza Eren, and Serpil Ozmihci. 2012. "Bio-hydrogen production from cheese whey powder (CWP) solution: Comparison of thermophilic and mesophilic dark fermentations." International Journal of Hydrogen Energy 37 (10): 8338-8342. https://doi.org/10.1016/j.ijhydene.2012.02.162.
- Karimi, Keikhosor, Marzieh Shafiei, and Rajeev Kumar. 2013. "Progress in Physical and Chemical Pretreatment of Lignocellulosic Biomass." In Biofuel Technologies: Recent Developments, edited by Vijai Kumar Gupta and Maria G. Tuohy, 53-96. Berlin, Heidelberg: Springer Berlin Heidelberg.
- Kaur G., Garcia-Gonzalez L., Elst K., Truzzi F., Bertin L., Kaushik A., Balakrishnan M., De Wever H. (2020)
 "Reactive extraction for in-situ carboxylate recovery from mixed culture fermentation."
 Biochemical Engineering Journal 160: 107641.
 https://doi.org/https://doi.org/10.1016/j.bej.2020.107641.
- Kennedy, Helena Tavares. 2017. Breaking the Bottleneck POET-DSM achieves cellulosic biofuel breakthrough. Biofuels Digest. Accessed 29/05/2020.
- Khanra, Saumyakanti, Madhumanti Mondal, Gopinath Halder, O. N. Tiwari, Kalyan Gayen, and Tridib Kumar Bhowmick. 2018. "Downstream processing of microalgae for pigments, protein and carbohydrate in industrial application: A review." Food and Bioproducts Processing. https://doi.org/10.1016/j.fbp.2018.02.002.
- Kim, Nag-Jong, Seong-Jin Lim, and Ho Nam Chang. 2018. "Volatile Fatty Acid Platform: Concept and Application." In Emerging Areas in Bioengineering, 173-190.
- Kleerebezem, Robbert, Bart Joosse, Rene Rozendal, and Mark C. M. Van Loosdrecht. 2015. "Anaerobic digestion without biogas?" Reviews in Environmental Science and Bio/Technology 14 (4): 787-801. https://doi.org/10.1007/s11157-015-9374-6.
- Komesu, Andrea, Maria Regina Wolf Maciel, and Rubens Maciel Filho. 2017. "Central composite experimental design applied to evaluate the lactic acid concentration by short path evaporation." Chemical Engineering and Processing: Process Intensification 117: 89-94. https://doi.org/https://doi.org/10.1016/j.cep.2017.03.020.

- Kotsyurbenko, Oleg R., Michail V. Glagolev, Alla N. Nozhevnikova, and Ralf Conrad. 2001. "Competition between homoacetogenic bacteria and methanogenic archaea for hydrogen at low temperature." FEMS Microbiology Ecology 38 (2-3): 153-159. https://doi.org/10.1111/j.1574-6941.2001.tb00893.x.
- Laird, Karen. 2011. "Chemical recycling closes the LOOPLA for cradle-to-cradle PLA." Plastic Today.
- Laird, Karen. 2012. Telles tale ends.
- Laird, Karen. 2018. "Metabolix signs \$10 million binding LOI with CJ CheilJedang." Accessed April. https://www.plasticstoday.com/business/ metabolix - signs -10- million -binding-loi-cjcheiljedang/127307180225084.
- Lara-Vázquez, Anibal R., Francisco R. Quiroz-Figueroa, Arturo Sánchez, and Idania Valdez-Vazquez. 2014. "Particle size and hydration medium effects on hydration properties and sugar release of wheat straw fibers." Biomass and Bioenergy 68: 67-74. https://doi.org/10.1016/j.biombioe.2014.06.006.
- Lane, Jim. 2015. "Algenol CEO exits, staff cut by 25 percent." Accessed April. https://www.renewableenergyworld.com/articles/2015/10/algenol-ceo-exits-staff-cut-by-25-percent.html.
- Larriba, Marcos, Pablo Navarro, Noemí Delgado-Mellado, Victor Stanisci, Julián García, and Francisco Rodríguez. 2016. "Separation of aromatics from n-alkanes using tricyanomethanide-based ionic liquids: Liquid-liquid extraction, vapor-liquid separation, and thermophysical characterization." Journal of Molecular Liquids 223 (Supplement C): 880-889. https://doi.org/https://doi.org/10.1016/j.molliq.2016.09.017.
- Larson, Andrea. 2010. Natureworks: Green Chemistry'S Contribution to Biotechnology, Innovation, Commercialization, and Strategic Positioning.
- LCYGroup, 2020, "LCY Biosciences to officially resume production of bio-based chemicals in Sarnia, Canada," https://www.lcygroup.com/lcygroup/en/newsinfo.php?dtid=0&cid=1&nid=200&Page=0.
- Li, Shen, Patel Martin K., and Haufe Juliane. 2009. Product overview and market projection of emerging bio-based plastics. In PRO-BIP 2009. Utrecht The Netherlands.
- Li, Lei, Qingming He, Yunmei Wei, Qin He, and Xuya Peng. 2014. "Early warning indicators for monitoring the process failure of anaerobic digestion system of food waste." Bioresource Technology 171: 491-494. https://doi.org/10.1016/j.biortech.2014.08.089.
- Li Y., Zhao J., Achinas S., Zhang Z., Krooneman J., Euverink G. J. W. (2020) "The biomethanation of cow manure in a continuous anaerobic digester can be boosted via a bioaugmentation culture containing Bathyarchaeota." Science of The Total Environment 745: 141042. https://doi.org/10.1016/j.scitotenv.2020.141042.
- Lu, Xueqin, Guangyin Zhen, Adriana Ledezma Estrada, Mo Chen, Jialing Ni, Toshimasa Hojo, Kengo Kubota, and Yu-You Li. 2015. "Operation performance and granule characterization of upflow anaerobic sludge blanket (UASB) reactor treating wastewater with starch as the sole carbon source." Bioresource Technology 180: 264-273. https://doi.org/10.1016/j.biortech.2015.01.010.

- Liang Y. –L., Zhang Z., Wu M., Wu Y., Feng J.-X. (2014) "Isolation, Screening, and Identification of Cellulolytic Bacteria from Natural Reserves in the Subtropical Region of China and Optimization of Cellulase Production by <i>Paenibacillus terrae</i> ME27-1." BioMed Research International 2014: 512497. https://doi.org/10.1155/2014/512497.
- Liguori, Rossana, Carlos R. Soccol, Luciana Porto de Souza Vandenberghe, Adenise L. Woiciechowski, and Vincenza Faraco. 2015. "Second Generation Ethanol Production from Brewers' Spent Grain." Energies 8 (4). https://doi.org/10.3390/en8042575.
- Liu L., Chen H. (2006) "Enzymatic hydrolysis of cellulose materials treated with ionic liquid [BMIM] Cl." Chinese Science Bulletin 51 (20): 2432-2436. https://doi.org/10.1007/s11434-006-2134-9.
- López-Linares, Juan C., M. Teresa García-Cubero, Susana Lucas, and Mónica Coca. 2020. "Integral valorization of cellulosic and hemicellulosic sugars for biobutanol production: ABE fermentation of the whole slurry from microwave pretreated brewer's spent grain." Biomass and Bioenergy 135: 105524. https://doi.org/10.1016/j.biombioe.2020.105524.
- Ma, Hui-Hui, Bi-Xian Zhang, Peng Zhang, Shuang Li, Yun-Fei Gao, and Xiao-Mei Hu. 2016. "An efficient process for lignin extraction and enzymatic hydrolysis of corn stalk by pyrrolidonium ionic liquids." Fuel Processing Technology 148 (Supplement C): 138-145. https://doi.org/10.1016/j.fuproc.2016.02.038.
- Mäkinen, Tuula, and Jukka Leppälahti. 2009. "Review of the Finnish BioRefine—New biomass products program." Environmental Progress & Sustainable Energy 28 (3): 470-474. https://doi.org/10.1002/ep.10395
- Marín, Francisco R., Cristina Soler-Rivas, Obdulio Benavente-García, Julian Castillo, and José A. Pérez-Alvarez. 2007. "By-products from different citrus processes as a source of customized functional fibres." Food Chemistry 100 (2): 736-741. https://doi.org/10.1016/j.foodchem.2005.04.040.
- Marques, Fabrício. 2018. Obstacles in the way. Pesquisa FAPESP (268).
- Martínez, R., P. Torres, M. A. Meneses, J. G. Figueroa, J. A. Pérez-Álvarez, and M. Viuda-Martos. 2012. "Chemical, technological and in vitro antioxidant properties of cocoa (Theobroma cacao L.) co-products." Food Research International 49 (1): 39-45. https://doi.org/10.1016/j.foodres.2012.08.005.
- Martin-Ryals A., Schideman L., Li P., Wilkinson H., Wagner R. (2015) "Improving anaerobic digestion of a cellulosic waste via routine bioaugmentation with cellulolytic microorganisms." Bioresource Technology 189: 62-70. https://doi.org/10.1016/j.biortech.2015.03.069.
- McCarthy A., O'Callaghan Y. C., Connolly A., Piggott C. O., FitzGerald R. J., O'Brien N. M. (2012)
 "Phenolic extracts of brewers' spent grain (BSG) as functional ingredients Assessment of their DNA protective effect against oxidant-induced DNA single strand breaks in U937 cells."
 Food Chemistry 134 (2): 641-646. https://doi.org/10.1016/j.foodchem.2012.02.133.
- Menezes, Camila Aparecida de, and Edson Luiz Silva. 2019. "Hydrogen production from sugarcane juice in expanded granular sludge bed reactors under mesophilic conditions: The role of homoacetogenesis and lactic acid production." Industrial Crops and Products 138: 111586. https://doi.org/10.1016/j.indcrop.2019.111586.
- Metabolix, 2012, "Metabolix announces termination of telles joint venture."

- Monlau, Florian, Abdellatif Barakat, Eric Trably, Claire Dumas, J. P. Steyer, and Hélène Carrere. 2011. "Lignocellulosic Materials Into Biohydrogen and Biomethane: Impact of Structural Features and Pretreatment." Critical Reviews in Environmental Science and Technology 2011. https://doi.org/10.1080/10643389.2011.604258.
- Morão, Ana, and François de Bie. 2019. "Life Cycle Impact Assessment of Polylactic Acid (PLA) Produced from Sugarcane in Thailand." Journal of Polymers and the Environment 27 (11): 2523-2539. https://doi.org/10.1007/s10924-019-01525-9.
- Moreira M. M., Morais S., Carvalho B. D. O., Barros A. A., Delerue-Matos C., Guido L. F. (2013) "Brewer's spent grain from different types of malt: Evaluation of the antioxidant activity and identification of the major phenolic compounds." Food Research International 54 (1): 382-388. https://doi.org/10.1016/j.foodres.2013.07.023.
- Mota, V. T., A. D. N. Ferraz Júnior, E. Trably, and M. Zaiat. 2018. "Biohydrogen production at pH below 3.0: Is it possible?" Water Research 128: 350-361. https://doi.org/10.1016/j.watres.2017.10.060.
- Mosier N., Wyman C., Dale B., Elander R., Lee Y. Y., Holtzapple M., Ladisch M. (2005) "Features of promising technologies for pretreatment of lignocellulosic biomass." Bioresource Technology 96 (6): 673-686. https://doi.org/10.1016/j.biortech.2004.06.025.
- Muñoz-Almagro, Nerea, Lourdes Valadez-Carmona, José A. Mendiola, Elena Ibáñez, and Mar Villamiel. 2019. "Structural characterisation of pectin obtained from cacao pod husk. Comparison of conventional and subcritical water extraction." Carbohydrate Polymers 217: 69-78. https://doi.org/10.1016/j.carbpol.2019.04.040.
- Mussatto S. I., Dragone G, Roberto I. C. (2006) "Brewers' spent grain: generation, characteristics and potential applications." Journal of Cereal Science 43 (1): 1-14. http://dx.doi.org/10.1016/j.jcs.2005.06.001.
- NatureWorks, 2012, "NatureWorks broadens Ingeo product portfolio with Sulzer proprietary production equipment."
- Ngan, Nguyen Vo Chau, Francis Mervin S. Chan, Tran Sy Nam, Huynh Van Thao, Monet Concepcion Maguyon-Detras, Dinh Vuong Hung, Do Minh Cuong, and Nguyen Van Hung. 2020. "Anaerobic Digestion of Rice Straw for Biogas Production." In Sustainable Rice Straw Management, edited by Martin Gummert, Nguyen Van Hung, Pauline Chivenge and Boru Douthwaite, 65-92. Cham: Springer International Publishing.
- Nkemka, Valentine Nkongndem, Brandon Gilroyed, Jay Yanke, Robert Gruninger, Darrell Vedres, Tim McAllister, and Xiying Hao. 2015. "Bioaugmentation with an anaerobic fungus in a two-stage process for biohydrogen and biogas production using corn silage and cattail." Bioresource Technology 185 (Supplement C): 79-88. https://doi.org/10.1016/j.biortech.2015.02.100.
- Novamont, 29/09/2016, 2016, "Opening of the world's first industrial scale plant for the production of butanediol via fermentation of renewable materials," https://www.novamont.com/eng/read-press-release/mater-biotech/.
- Niemi P., Martins D., Buchert J., Faulds C. B. (2013) "Pre-hydrolysis with carbohydrases facilitates the release of protein from brewer's spent grain." Bioresource Technology 136 (Supplement C): 529-534. https://doi.org/10.1016/j.biortech.2013.03.076.

- Niemi P., Tamminen T., Smeds A., Viljanen K., Ohra-aho T., Holopainen-Mantila U., Faulds C. B., Poutanen K., Buchert J. (2012) "Characterization of Lipids and Lignans in Brewer's Spent Grain and Its Enzymatically Extracted Fraction." Journal of Agricultural and Food Chemistry 60 (39): 9910-9917. https://doi.org/10.1021/jf302684x.
- Novozymes, 2014, "BASF, Cargill and Novozymes achieved another milestone in bio-based acrylic acid," https://www.novozymes.com/en/news/news-archive/2014/09/basf-cargill-and-novozymes-achieved-another-milestone-in-bio-based-acrylic-acid.

Novozymes, 2015, "Novozymes to supply enzymes to St1 Biofuels in Finland."

- NREL. 2002. National Reneweble Energy Laboratory: 25 Years of Research Excellence 1977–2002.
- Nzeteu C. O., Trego A. C., Abram F., O'Flaherty V. (2018) "Reproducible, high-yielding, biological caproate production from food waste using a single-phase anaerobic reactor system." Biotechnology for Biofuels 11 (1): 108. https://doi.org/10.1186/s13068-018-1101-4.
- Oliva, A., L. C. Tan, S. Papirio, G. Esposito, and P. N. L. Lens. 2020. "Effect of methanol-organosolv pretreatment on anaerobic digestion of lignocellulosic materials." Renewable Energy. https://doi.org/10.1016/j.renene.2020.12.095.
- Palomo-Briones, Rodolfo, Elías Razo-Flores, Nicolas Bernet, and Eric Trably. 2017. "Dark-fermentative biohydrogen pathways and microbial networks in continuous stirred tank reactors: Novel insights on their control." Applied Energy 198: 77-87. https://doi.org/10.1016/j.apenergy.2017.04.051.
- Patel, Alok, Fabio Mikes, Saskja Bühler, and Leonidas Matsakas. 2018. "Valorization of Brewers' Spent Grain for the Production of Lipids by Oleaginous Yeast." Molecules (Basel, Switzerland) 23 (12): 3052. https://doi.org/10.3390/molecules23123052.
- Patsalou, Maria, Antonios Chrysargyris, Nikolaos Tzortzakis, and Michalis Koutinas. 2020. "A biorefinery for conversion of citrus peel waste into essential oils, pectin, fertilizer and succinic acid via different fermentation strategies." Waste Management 113: 469-477. https://doi.org/10.1016/j.wasman.2020.06.020.
- Pejin, Jelena, Miloš Radosavljević, Milana Pribić, Sunčica Kocić-Tanackov, Dragana Mladenović, Aleksandra Djukić-Vuković, and Ljiljana Mojović. 2018. "Possibility of L-(+)-lactic acid fermentation using malting, brewing, and oil production by-products." Waste Management 79: 153-163. https://doi.org/10.1016/j.wasman.2018.07.035.
- Peng X., Börner R. A., Nges I. A., Liu J. (2014) "Impact of bioaugmentation on biochemical methane potential for wheat straw with addition of Clostridium cellulolyticum." Bioresource Technology 152: 567-571. https://doi.org/10.1016/j.biortech.2013.11.067.
- Plaza P. E., Gallego-Morales L. J., Peñuela-Vásquez M., Lucas S., García-Cubero M. T., Coca M. (2017)
 "Biobutanol production from brewer's spent grain hydrolysates by Clostridium beijerinckii." Bioresource Technology 244: 166-174. https://doi.org/10.1016/j.biortech.2017.07.139.
- POET, 2009, "POET plant produces cellulosic ethanol," https://poet.com/pr/poet-plant-producescellulosic-ethanol.
- Pugazhendhi, Arivalagan, Parthiban Anburajan, Jong-Hun Park, Gopalakrishnan Kumar, Periyasamy Sivagurunathan, and Sang-Hyoun Kim. 2017. "Process performance of biohydrogen

production using glucose at various HRTs and assessment of microbial dynamics variation via q-PCR." International Journal of Hydrogen Energy 42 (45): 27550-27557. https://doi.org/10.1016/j.ijhydene.2017.06.184.

- Qin, Fen, Astrid Z. Johansen, and Solange I. Mussatto. 2018. "Evaluation of different pretreatment strategies for protein extraction from brewer's spent grains." Industrial Crops and Products 125: 443-453. https://doi.org/10.1016/j.indcrop.2018.09.017.
- Raghavendra, S. N., N. K. Rastogi, K. S. M. S. Raghavarao, and R. N. Tharanathan. 2004. "Dietary fiber from coconut residue: effects of different treatments and particle size on the hydration properties." European Food Research and Technology 218 (6): 563-567. https://doi.org/10.1007/s00217-004-0889-2.
- Ramos-Suarez, Maria, Yue Zhang, and Victoria Outram. 2021. "Current perspectives on acidogenic fermentation to produce volatile fatty acids from waste." Reviews in Environmental Science and Bio/Technology 20 (2): 439-478. https://doi.org/10.1007/s11157-021-09566-0.
- Ravindran R., Jaiswal A. K. (2016) "A comprehensive review on pre-treatment strategy for lignocellulosic food industry waste: Challenges and opportunities." Bioresource Technology 199: 92-102. https://doi.org/10.1016/j.biortech.2015.07.106.
- Ravindran R., Jaiswal R., Abu-Ghannam N., Jaiswal A. K. (2018) "A comparative analysis of pretreatment strategies on the properties and hydrolysis of brewers' spent grain." Bioresource Technology 248: 272-279. https://doi.org/10.1016/j.biortech.2017.06.039.
- Regueira, A., R. González-Cabaleiro, I. D. Ofiţeru, J. Rodríguez, and J. M. Lema. 2018. "Electron bifurcation mechanism and homoacetogenesis explain products yields in mixed culture anaerobic fermentations." Water Research 141: 349-356. https://doi.org/https://doi.org/10.1016/j.watres.2018.05.013.
- Reyhanitash, Ehsan, Bart Zaalberg, Sascha R. A. Kersten, and Boelo Schuur. 2016. "Extraction of volatile fatty acids from fermented wastewater." Separation and Purification Technology 161 (Supplement C): 61-68. https://doi.org/10.1016/j.seppur.2016.01.037.
- RFA. 2018. Ethanol Strong. (San Antonio, Texas, USA). http://www.ethanolrfa.org/wpcontent/uploads/2018/02/NECfinalOutlook.pdf.
- Ribau Teixeira, Margarida, Eliana C. Guarda, Elisabete B. Freitas, Claudia F. Galinha, Anouk F. Duque, and Maria A. M. Reis. 2020. "Valorization of raw brewers' spent grain through the production of volatile fatty acids." New Biotechnology 57: 4-10. https://doi.org/10.1016/j.nbt.2020.01.007.
- Robertson J. A., de Monredon F. D., Dysseler P., Guillon F., Amado R., Thibault J.-F. (2000) "Hydration Properties of Dietary Fibre and Resistant Starch: a European Collaborative Study." LWT - Food Science and Technology 33 (2): 72-79. https://doi.org/10.1006/fstl.1999.0595.
- Rocha M. A. A., Raeissi S., Hage P., Weggemans W. M. A., van Spronsen J., Peters C. J., Kroon M. C. (2017) "Recovery of volatile fatty acids from water using medium-chain fatty acids and a cosolvent." Chemical Engineering Science 165: 74-80. https://doi.org/10.1016/j.ces.2017.02.014.
- Rodríguez-Valderrama, Santiago, Carlos Escamilla-Alvarado, Hector J. Amezquita-Garcia, José J. Cano-Gómez, Jean-Pierre Magnin, and Pasiano Rivas-García. 2019. "Evaluation of feeding strategies

in upflow anaerobic sludge bed reactor for hydrogenogenesis at psychrophilic temperature." International Journal of Hydrogen Energy 44 (24): 12346-12355. https://doi.org/10.1016/j.ijhydene.2018.09.215.

- Rojas-Chamorro, José A., Inmaculada Romero, Juan C. López-Linares, and Eulogio Castro. 2020. "Brewer's spent grain as a source of renewable fuel through optimized dilute acid pretreatment." Renewable Energy 148: 81-90. https://doi.org/10.1016/j.renene.2019.12.030.
- Rokem, J. Stefan. 2020. "TCA Cycle Organic Acids Produced by Filamentous Fungi: The Building Blocks of the Future." In Grand Challenges in Fungal Biotechnology, edited by Helena Nevalainen, 439-476. Cham: Springer International Publishing.
- Rommi K., Niemi P., Kemppainen P., Kruus K. (2018) "Impact of thermochemical pre-treatment and carbohydrate and protein hydrolyzing enzyme treatment on fractionation of protein and lignin from brewer's spent grain." Journal of Cereal Science 79: 168-173. https://doi.org/10.1016/j.jcs.2017.10.005.
- Roy R., Rahman Md S., Raynie D. E. (2020) "Recent advances of greener pretreatment technologies of lignocellulose." Current Research in Green and Sustainable Chemistry 3: 100035. https://doi.org/10.1016/j.crgsc.2020.100035.
- Runge, W. 2014. Technology Entrepreneurship : A Treatise on Entrepreneurs and Entrepreneurship for and in Technology Ventures. Vol 2. Vol. v. 2: KIT Scientific Publishing.
- Saady, Noori M. Cata. 2013. "Homoacetogenesis during hydrogen production by mixed cultures dark fermentation: Unresolved challenge." International Journal of Hydrogen Energy 38 (30): 13172-13191. https://doi.org/10.1016/j.ijhydene.2013.07.122.
- Saboe, Patrick O., Lorenz P. Manker, William E. Michener, Darren J. Peterson, David G. Brandner, Stephen P. Deutch, Manish Kumar, Robin M. Cywar, Gregg T. Beckham, and Eric M. Karp. 2018.
 "In situ recovery of bio-based carboxylic acids." Green Chemistry 20 (8): 1791-1804. https://doi.org/10.1039/C7GC03747C.
- Sanchez, A., P. Hernández-Sánchez, and R. Puente. 2019. "Hydration of lignocellulosic biomass. Modelling and experimental validation." Industrial Crops and Products 131: 70-77. https://doi.org/10.1016/j.indcrop.2019.01.029.
- Sánchez, Carmen. 2009. "Lignocellulosic residues: Biodegradation and bioconversion by fungi." Biotechnology Advances 27 (2): 185-194. https://doi.org/10.1016/j.biotechadv.2008.11.001.
- Saw, H. Y., J. Janaun, and D. Subbarao. 2008. "Hydration properties of palm kernel cake." Journal of Food Engineering 89 (2): 227-231. https://doi.org/10.1016/j.jfoodeng.2008.04.027.
- Schieb, Pierre-Alain, Honorine Lescieux-Katir, Maryline Thénot, and Barbara Clément-Larosière. 2015. "Industrial Symbiosis at the Bazancourt-Pomacle Biorefinery." 67-80.
- Schink, B. 1997. "Energetics of syntrophic cooperation in methanogenic degradation." Microbiology and molecular biology reviews : MMBR 61 (2): 262-280. https://doi.org/10.1128/mmbr.61.2.262-280.1997.
- Schmidt, J. E., Ahring B. K. (1991) "Acetate and hydrogen metabolism in intact and disintegrated granules from an acetate-fed, 55° C, UASB reactor." Applied Microbiology and Biotechnology 35 (5): 681-685. https://doi.org/10.1007/BF00169637.

- Scoma, Alberto, Felipe Varela-Corredor, Lorenzo Bertin, Carlo Gostoli, and Serena Bandini. 2016. "Recovery of VFAs from anaerobic digestion of dephenolized Olive Mill Wastewaters by Electrodialysis." Separation and Purification Technology 159 (Supplement C): 81-91. https://doi.org/10.1016/j.seppur.2015.12.029.
- Shen L., Hu H., Ji H., Cai J., He N., Li Q., Wang Y. (2014) "Production of poly(hydroxybutyrate– hydroxyvalerate) from waste organics by the two-stage process: Focus on the intermediate volatile fatty acids." Bioresource Technology 166: 194-200. https://doi.org/10.1016/j.biortech.2014.05.038.
- Shi X., Lin J., Zuo J., Li P., Li X, Guo X. (2017) "Effects of free ammonia on volatile fatty acid accumulation and process performance in the anaerobic digestion of two typical bio-wastes." Journal of Environmental Sciences 55 (Supplement C): 49-57. https://doi.org/10.1016/j.jes.2016.07.006.
- Siegert I., Banks C. (2005) "The effect of volatile fatty acid additions on the anaerobic digestion of cellulose and glucose in batch reactors." Process Biochemistry 40 (11): 3412-3418. https://doi.org/10.1016/j.procbio.2005.01.025.
- Sims, Bryan. 2012. "OPXBio hits key production milestone for bioacrylic acid." Biomass Magazine.
- Singh R., Kumar S. (2019) "A review on biomethane potential of paddy straw and diverse prospects to enhance its biodigestibility." Journal of Cleaner Production 217: 295-307. https://doi.org/10.1016/j.jclepro.2019.01.207.
- Siracusa, Valentina, and Carlo Ingrao. 2016. "The Use of Polylactic Acid in Food Packaging." In Reference Module in Food Science. Elsevier.
- Siriwongrungson, Vilailuck, Raymond J. Zeng, and Irini Angelidaki. 2007. "Homoacetogenesis as the alternative pathway for H2 sink during thermophilic anaerobic degradation of butyrate under suppressed methanogenesis." Water Research 41 (18): 4204-4210. https://doi.org/10.1016/j.watres.2007.05.037.
- Slupska M., Bushong D. (2019) "Lessons from Commercialization of Cellulosic Ethanol A POET Perspective." Biofuels, Bioproducts and Biorefining 13 (4): 857-859. https://doi.org/10.1002/bbb.2033.
- Songer, Dave. 2017. BioAmber buys the bio-based acid plant that helped improve its Q2 financial performance. Biomarket Insights.
- Stadler, Thierry, and Jean-Marie Chauvet. 2018. "New innovative ecosystems in France to develop the Bioeconomy." New Biotechnology 40: 113-118. https://doi.org/10.1016/j.nbt.2017.07.009.
- Stefanello F. S., dos Santos C. O., Bochi V. C., Fruet A. P. B., Soquetta M. B., Dörr A. C., Nörnberg J. L. (2018) "Analysis of polyphenols in brewer's spent grain and its comparison with corn silage and cereal brans commonly used for animal nutrition." Food Chemistry 239: 385-401. https://doi.org/10.1016/j.foodchem.2017.06.130.
- Strazzera, Giuseppe, Federico Battista, Natalia Herrero Garcia, Nicola Frison, and David Bolzonella.2018. "Volatile fatty acids production from food wastes for biorefinery platforms: A review."JournalofEnvironmentalManagementAttps://doi.org/10.1016/j.jenvman.2018.08.039.
- Sulzer. 2019. The one-stop shop for PLA bioplastics. In Sulzer Technical Review 2/2019.

- Talekar, Sachin, Antonio F. Patti, R. Vijayraghavan, and Amit Arora. 2018. "An integrated green biorefinery approach towards simultaneous recovery of pectin and polyphenols coupled with bioethanol production from waste pomegranate peels." Bioresource Technology 266: 322-334. https://doi.org/10.1016/j.biortech.2018.06.072.
- Tao, Bing, Pearl Passanha, Philemon Kumi, Victoria Wilson, Dean Jones, and Sandra Esteves. 2016. "Recovery and concentration of thermally hydrolysed waste activated sludge derived volatile fatty acids and nutrients by microfiltration, electrodialysis and struvite precipitation for polyhydroxyalkanoates production." Chemical Engineering Journal 295 (Supplement C): 11-19. https://doi.org/https://doi.org/10.1016/j.cej.2016.03.036.
- Tizazu B. Z., Moholkar V. S. (2017) "Kinetic and Thermodynamic Analysis of Dilute Acid Hydrolysis of Sugarcane bagasse." Bioresource Technology. https://doi.org/10.1016/j.biortech.2017.11.032. Total-Corbion, 03/12/2018, 2018, "Total Corbion PLA starts-up its 75,000 tons per year bioplastics plant," https://www.totalcorbion.com/news/total-corbion-pla-starts-up-its-75-000-tons-per-year-bioplasticsplant/?p=1&q=2018.
- Tsapekos, P., Kougias P. G., Vasileiou S. A., Treu L., Campanaro S., Lyberatos G., Angelidaki I. (2017) "Bioaugmentation with hydrolytic microbes to improve the anaerobic biodegradability of lignocellulosic agricultural residues." Bioresource Technology 234: 350-359. https://doi.org/10.1016/j.biortech.2017.03.043.
- Uke, Matthew N., and Edward Stentiford. 2013. "Enhancement of the anaerobic hydrolysis and fermentation of municipal solid waste in leachbed reactors by varying flow direction during water addition and leachate recycle." Waste Management 33 (6): 1425-1433. https://doi.org/10.1016/j.wasman.2013.02.020.
- Valadez-Carmona, Lourdes, Alicia Ortiz-Moreno, Guillermo Ceballos-Reyes, Jose A. Mendiola, and Elena Ibáñez. 2018. "Valorization of cacao pod husk through supercritical fluid extraction of phenolic compounds." The Journal of Supercritical Fluids 131: 99-105. https://doi.org/10.1016/j.supflu.2017.09.011.
- van Aarle, Ingrid M., Anastasios Perimenis, Joana Lima-Ramos, Emeline de Hults, Isabelle F. George, and Patrick A. Gerin. 2015. "Mixed inoculum origin and lignocellulosic substrate type both influence the production of volatile fatty acids during acidogenic fermentation." Biochemical Engineering Journal 103: 242-249. https://doi.org/10.1016/j.bej.2015.07.016.
- Vieira E., Rocha M. A. M., Coelho E., Pinho O., Saraiva J. A., Ferreira I. M. P. L. V. O., Coimbra M. A. (2014) "Valuation of brewer's spent grain using a fully recyclable integrated process for extraction of proteins and arabinoxylans." Industrial Crops and Products 52 (Supplement C): 136-143. https://doi.org/10.1016/j.indcrop.2013.10.012.
- Vink, Erwin, Steve Davies, and Jeffrey J. Kolstad. 2010. ORIGINAL RESEARCH: The eco-profile for current Ingeo [®] polylactide production. Vol. 6.
- Vinuesa, J. F., Ph Mirabel, and J. L. Ponche. 2003. "Air quality effects of using reformulated and oxygenated gasoline fuel blends: application to the Strasbourg area (F)." Atmospheric Environment 37 (13): 1757-1774. https://doi.org/10.1016/S1352-2310(03)00067-0.
- Wagner E., Pería M. E., Ortiz G. E., Rojas N. L., Ghiringhelli P. D. (2021) "Valorization of brewer's spent grain by different strategies of structural destabilization and enzymatic saccharification." Industrial Crops and Products 163: 113329. https://doi.org/10.1016/j.indcrop.2021.113329.

- Wang, Kun, Jun Yin, Dongsheng Shen, and Na Li. 2014. "Anaerobic digestion of food waste for volatile fatty acids (VFAs) production with different types of inoculum: Effect of pH." Bioresource Technology 161 (Supplement C): 395-401. https://doi.org/10.1016/j.biortech.2014.03.088.
- Werpy, Todd A., John Holladay, and James White. 2004. Top Value Added Chemicals From Biomass: I. Results of Screening for Potential Candidates from Sugars and Synthesis Gas.

Wesoff, Eric. 2017. Hard Lessons From the Great Algae Biofuel Bubble.

- Wilkinson S., Smart K. A., Cook D. J. (2014) "Optimisation of alkaline reagent based chemical pretreatment of Brewers spent grains for bioethanol production." Industrial Crops and Products 62: 219-227. https://doi.org/https://doi.org/10.1016/j.indcrop.2014.08.036.
- Xu J., Guzman J. J. L., Andersen S. J., Rabaey K., Angenent L. T. (2015) "In-line and selective phase separation of medium-chain carboxylic acids using membrane electrolysis." Chemical Communications 51 (31): 6847-6850. https://doi.org/10.1039/C5CC01897H.
- Xu J., Guzman J. J. L., Angenent A. L. (2021) "Direct Medium-Chain Carboxylic Acid Oil Separation from a Bioreactor by an Electrodialysis/Phase Separation Cell." Environmental Science & Technology 55 (1): 634-644. https://doi.org/10.1021/acs.est.0c04939.
- Yoon, S. Y., S. H. Han, and S. J. Shin. 2014. "The effect of hemicelluloses and lignin on acid hydrolysis of cellulose." Energy 77: 19-24. https://doi.org/10.1016/j.energy.2014.01.104.
- Zacharof, Myrto-Panagiota, and Robert W. Lovitt. 2014. "The filtration characteristics of anaerobic digester effluents employing cross flow ceramic membrane microfiltration for nutrient recovery." Desalination 341: 27-37. https://doi.org/https://doi.org/10.1016/j.desal.2014.02.034.
- Zhang, Y., Angelidaki I. (2015) "Bioelectrochemical recovery of waste-derived volatile fatty acids and production of hydrogen and alkali." Water Research 81 (Supplement C): 188-195.://doi.org/10.1016/j.watres.2015.05.058.
- Zhang Z.-P., Show K.-Y., Tay J.-H., Liang D. T., Lee D.-J. (2008) "Biohydrogen production with anaerobic fluidized bed reactors—A comparison of biofilm-based and granule-based systems." International Journal of Hydrogen Energy 33 (5): 1559-1564. https://doi.org/10.1016/j.ijhydene.2007.09.048.
- Zhang Z.-P., Tay J.-H., Show K.-Y., Yan R., Liang D. T., Lee D. J., Jiang W. J. (2007) "Biohydrogen production in a granular activated carbon anaerobic fluidized bed reactor." International Journal of Hydrogen Energy 32 (2): 185-191. https://doi.org/10.1016/j.ijhydene.2006.08.017.
- Ziolkowska, Jadwiga R., and Leo Simon. 2014. "Recent developments and prospects for algae-based fuels in the US." Renewable and Sustainable Energy Reviews 29: 847-853. https://doi.org/10.1016/j.rser.2013.09.021.
- Zoulikha, Maache-Rezzoug, Maugard Thierry, Zhao Jean-Michel Qiuyu, Armelle Nouviaire, and Rezzoug Sid-Ahmed. 2015. "Combined steam-explosion toward vacuum and dilute-acid spraying of wheat straw. Impact of severity factor on enzymatic hydrolysis." Renewable Energy 78: 516-526. https://doi.org/10.1016/j.renene.2015.01.038.
Chapter 3

Biodegradation of brewery spent grain by mixed microbial cultures from anaerobic sludge and animal dungs

A modified version of this chapter has been submitted as:

Castilla-Archilla J., Thorn C., Pau S., Lens P. N. L. (2022) "Screening for suitable mixed microbial consortia from anaerobic sludge and animal dungs for biodegradation of brewery spent grain" Submitted to Biomass and Bioenergy (major revisions)

Keywords: Lignocellulose; brewery spent grain; mixed microbial culture; hydrolysis; anaerobic digestion; volatile fatty acids

Abstract

Brewery spent grain is a lignocellulosic waste and is a valuable resource from a biorefinery perspective. Bioaugmentation with specialised inocula can be a viable means of improving the degradation of this recalcitrant feedstock. This study used a biodegradation activity test to screen six different sources of microbiota for chemical and energy production from brewery spent grain. The inocula used were anaerobic granular sludge as i) intact granules and ii) crushed granules, as well as animal digestive fluids iii) bovine rumen fluid, and animal manures iv) giraffe manure, v) rhino manure, and vi) tiger manure. Despite the herbivores faeces presenting the largest hydrolytic community, the best solids destruction was achieved using anaerobic granular sludge also resulted in the highest gas production, followed by the giraffe manure. Contrastingly, the highest volatile fatty acid accumulation was achieved using rhino and tiger manure, i.e. $3.86 (\pm 0.55)$ and $3.20 (\pm 0.74)$ gCOD/L, respectively. Interestingly, the poorest solids destruction was with the inocula that accumulated the largest amount of fermented compounds. Further research is required to determine the inhibitors of the hydrolysis and degradation of brewery spent grain.

3.1 Introduction

The finite availability of fossil fuels in conjunction with increasing concern into environmental sustainability and guardianship have led to the development of the biorefinery concept. This is the sustainable process of transforming renewable biomass, through a combination of physical, chemical and biological processes, into a spectrum of products and energy equivalent to fossil fuel-based products (Biorefinery Euroview, 2009; Biopol, 2009). Anaerobic digestion (AD) is one the most extensively applied methods for the treatment of organic solid wastes for sustainable energy production. The process of AD, whereby raw material is finally converted into methane and carbon dioxide, can be defined by four sequential steps: hydrolysis, acidogenesis, acetogenesis and methanogenesis. While historically the desired end product of AD was methane for energy production, more recently it has been adapted as a means of producing higher value intermediary compounds, such as succinate, short chain fatty acids and ethanol (Infantes et al., 2011; Garcia-Aguirre et al., 2017). This can be achieved through the partial inhibition of some of the final steps of AD, leading to the accumulation of intermediate compounds (Shi et al., 2017; Zhang et al., 2018), which are chemical building blocks (Bozell et al., 2010; Castilla-Archilla et al., 2019).

Hemicellulosic and lignocellulosic wastes represent promising feedstock sources for such AD biorefineries due to their abundance and availability (Lynd et al., 2015). Industrial lignocellulosic waste, with low or no value, can provide a constant supply of renewable biomass. A potential source of such a feedstock is from the brewery industry, whose most abundant solid waste is brewery spent grain (BSG). This is a solid stream rich in carbohydrates, lignin, proteins and lipids (Mussatto et al., 2006; Vieira et al., 2014; Rommi et al., 2018) where the average production corresponds to 15-20 kg of BSG produced per hectolitre of beer (Mussatto et al., 2006). In Ireland in 2019, more than 8.2 million cubic meters of beer were produced (IBA, 2020), thus over 170,000 tonnes of this material was produced. However, the use of lignocellulosic material in AD is not well implemented when compared with other more easily biodegradable feedstocks. This is a consequence of the complex structure of lignocellulosic feedstocks, which renders them difficult for microbes to degrade rapidly, therefore resulting in low rates of solid destruction. Additionally, this process can be inhibited by compounds produced during AD, such as humic acids, phenolic compounds or volatile fatty acids (VFAs) (Fernandes et al., 2015; Chapleur et al., 2016; Azman et al., 2017).

The direct AD of some lignocellulosic materials has been studied but the resulting methane production and solid destruction were inefficient (Singh et al., 2019). To improve this, different approaches have been tested, including co-digestion, chemical addition and more recently bioaugmentation (Tsapekos et al., 2017; Wall et al., 2018). The purpose of bioaugmentation is to enhance the native hydrolytic community capable of degrading cellulose and hemicellulose by

introducing more performant hydrolytic microorganisms (Angelidaki et al., 2000; Haruta et al., 2002; Li et al., 2020). This has been applied as a strategy in batch systems where improved overall methane production demonstrated the benefits of bioaugmentation using pure cultures (Peng et al., 2014) and mixed cultures from rumen fluid (Baba et al., 2017). However, when operating systems in continuous mode, microorganisms typically had to be periodically inoculated into the reactor as their positive effect diminished over time (Martin-Ryals et al., 2015; Tsapekos et al., 2017; Jiang et al., 2020). In the work of Li et al. (2020) however, the benefits following bioaugmentation with Bathyarchaeota enriched culture took up to four months to stabilise in a continuous digester, resulting in significant positive effects on the biomethane production from cow manure digester. These remained, although slightly decreased, once bioaugmentation stopped as the supplemented microorganisms became a stable part of the microbial community over time (Li et al., 2020). Indeed, the use of mixed microbial cultures was shown to have a higher robustness to become adapted on process performance and more resilient for lignocellulose degradation (Rahman et al., 2002; Liu et al., 2018; Li et al., 2020). However, the continuous addition of any pure culture, cellulolytic bioculture or mixed microbial culture to promote the bioaugmentation along time would result in additional costs. Therefore, the use of an inoculum that contains a stable hydrolytic activity in its native composition could result in an advantage.

The goal of this chapter was to evaluate and compare different inocula during AD to identify the most suitable consortia for the degradation of BSG. For this purpose, a range of inocula were selected where a different microbial composition was expected as a result of their origin and the content of intestinal track of animals that feed on lignocellulosic materials. These included anaerobic granular sludge from an industrial dairy wastewater treatment facility. Both intact and crushed sludge were tested, where we hypothesised crushed sludge might perform better due to enhanced substrate availability for all microbes compared to those in intact granules. Bovine rumen fluid from cow was used as the microbial consortia present were expected to have strong hydrolytic activity. In addition, faeces from various zoo animals were used, namely giraffe, rhinoceros and tiger. These animals were chosen to represent a ruminant, a monogastric herbivore, and a carnivore as a control with a differentiated microbial community than the herbivore animals. We hypothesised these would serve as candidate sources of microbiota capable of hydrolysing the complex biomass while allowing for the simultaneous production of valuable compounds and/or energy. To assess this, we measured the rate of biomass degradation, the production of fermented compounds (alcohols, VFAs and lactic acid) and biogas. In an effort to elucidate the microbial consortia contributing to the production of these various compounds, the prokaryotic microbial community from the various inocula was characterised using 16S rRNA profiling at the start of the trial and once most biodegradation had occurred (day 21).

3.2 Material and methods

3.2.1 Feedstock

Fresh BSG was supplied by a local lager brewery in Galway (Ireland), and once collected, the BSG was frozen in individual bags. Total solids (TS) and volatile solids (VS) concentrations were determined prior to freezing. Before use, frozen BSG was defrosted overnight at 12°C.

3.2.2 Source of inoculum

Six different sources of inocula were tested. Granular sludge was collected from an up-flow anaerobic sludge bed reactor treating dairy wastewater for methane production at ambient temperature (Kilconnell, Co. Galway, Ireland) and kept at 4°C until further use. This sludge was used for two of the inoculum sources: i) granular (intact) sludge and ii) as crushed sludge (granules were crushed using a Homogenizer Safe-Seal 30 mL, GPE Scientific, UK).

The cow rumen fluid consisted of a composite sample of the rumen contents from the four different stomachs of three different cows removed during meat processing at a slaughterhouse (Ballyhaunis, Co. Mayo, Ireland). The solid fraction of the rumen fluid was retained, TS and VS were calculated, and the sample was stored 15 °C for 24 hours until use.

Three different animal manures (giraffe, rhinoceros and tiger) were obtained from Dublin Zoo (Ireland). The giraffe was selected as a rumen animal, which diet is based on alfalfa, supplemented with leaves from browsing and maintenance pellets. Rhinoceros was chosen as monogastric herbivore, therefore hindgut fermenter fed mainly alfalfa, hay and haylage, as well as a concentrate of proteins. Tiger was chosen as a control with an expected lower lignocellulosic hydrolytic activity as the tiger's diet was carnivorous, fed principally with horse and pig meat. Faeces were collected and returned to the lab where they were blended using a Waring 2-Speed Laboratory blender (USA) at the highest speed. Water was added during the blending of the different inocula to ensure proper mixing, the different amount of water for each of the inoculum was subtracted later to the final water addition during the experimental set up.

3.2.3 Biodegradation activity test

Activity tests were carried out in sacrificial bottles, where each bottle (HDPE plastic) had a total volume of 500 mL and a working volume of 400 mL, adapted from Nzeteu (2016). Bottles were sealed with a rubber stopper, perforated gas tight with a needle to allow attachment of a gas bag (*Figure 3.1*). Sodium bicarbonate was used as a buffer at a concentration of 10 g/L. Bottles were fed with BSG at a concentration of 6 g VS/L with an inoculum to substrate ratio of 1:4 (1.5 g VS/L of inoculum).



Figure 3.1 Experimental bottle and sampling for the biodegradation activity test. A muslin was used to split the liquid and solid fraction.

Experimental bottles were set up using the different inocula and brewery spent grain as feedstock, with the addition of the buffer to the final working volume. Control bottles were also set up consisting of inocula and buffer in the absence of feedstock to determine the contribution of organic compounds from the endogenous material present in the inocula. The inoculum was first added into all the bottles with a fraction of the buffer. The bottles were sparged with nitrogen, and placed under incubation at 37°C for 24 hours prior to feeding for acclimation of the inocula and further degradation of any organic carbon residues therein. Following acclimatization, BSG was added to the experimental bottles, and all bottles were subsequently filled with the buffer to reach the total volume required. All bottles were once again sparged with nitrogen and placed in the incubator (Innova 44) at 37 °C and 100 rpm. This was considered time 0 of the experiment. The gas bags were attached to the bottles on day 30 and were changed every day of sampling.

Destructive sampling was performed on days 0, 1, 2, 4, 7, 10, 14, 21 and 30, where five bottles were taken down at each of these day (three experimental bottles containing inoculum and substrate and two controls containing only inoculum). This was repeated for each inoculum (n=6), thus 45 sacrificial bottles were sampled on each of the 9 days. For the sampling, the content of each bottle was fractionated into leachate and retentate using muslin cloth (*Figure 1*). TS and VS were calculated for both fractions, and a small fraction of each was flash frozen using liquid nitrogen and stored at - 80°C for microbial community analysis. The filtrate was used for quantification of soluble chemical oxygen demand (sCOD), VFAs composition, lactic acid, ethanol, total protein and ammonia.

3.2.4 Chemical analysis

TS and VS were quantified out using standard methods (APHA, 2012). The sCOD was determined using the Reagecon 0 to 1,500 mg/L range kit (Reagecon, Ireland) where 2 mL of sample was added to the vials which were then placed at 150 °C for 2 hours. Vials were cooled and the absorbance was measured using a DR2500 spectrophotometer (Hach, Germany). The biogas was collected in a gas

sampling bags (Tedlar SCV) and gas volumes were determined by water displacement and corrected to standard conditions (0 °C and 1 atm). Biogas composition (H₂, CO₂ and CH₄) was analysed using gas chromatography (GC; AAgilent 7890A, USA) equipped with a glass column and a flame ionisation detector. The carrier gas was argon and the flow rate was 24 mL/min. VFA analysis was performed using a Varian 450-GC equipped with CombiPal autosampler and a flame ionisation detector, following Nzeteu *et al.* (2018). Lactic acid, ethanol, proteins and ammonia were analysed using a discrete nutrient analyser Gallery Plus[®] (ThermoFisher, Finland) based on a colorimetric assay (O'Dell. 1996; ISO 15923-1, C, 2013).

3.2.5 Calculations

The VS fraction, gas production, VFAs and sCOD for each sample day were calculated by subtracting the average value of control bottles from the average in the experimental bottles.

The VS destruction represents the percentage between the amount of BSG consumed at each point since the beginning of the trial and the amount of VS of BSG at the beginning of the trial (*Equation 1*):

$$\% VS_{Destruction} = \frac{VS_0 - VS_P}{VS_0} \cdot 100 \qquad Equation 1$$

Where VS_{Destruction}: amount of BSG consumed, VS: volatile solids amount of BSG, 0: initial point of the trial, P: sample point

The acidification yield was calculated based on the total contribution of the individual VFAs together in terms of g COD/L, divided by the overall sCOD (*Equation 2*):

$$Acidification Yield = \frac{Total VFA concentration}{Soluble COD} \cdot 100 \ [gCOD/100 \ gCOD_{Soluble}] Equation 2$$

The gas production rate was calculated using the difference between the cumulative gas before and after the log phase production per amount of inoculum (g of VS) (*Equation 3*):

$$GPR = \frac{V_f - V_i}{t_f - t_i} \cdot \frac{1}{VS} \quad \left[\frac{mL}{(day \cdot g \, VS_{inoculum})}\right] \quad Equation 3$$

Where V: total volume, t: time in days of the log phase, VS_{inoculum}: amount of inoculum in terms of VS, f: end of the log phase, i: initial of the log phase.

The different rates corresponding to VS destruction, gas and VFA production were calculated using the linear phases from the graphs obtained during the trials, if possible. The number of days corresponding to the different linear phases have been given with the rate and the R-squared value for each of the rates along the experiments. For example, the methane production rate of the giraffe manure was calculated using the linear phase between days 7 to 21 with $R^2 = 0.99$ (*Figure 2D*) or in the case of the crushed granular sludge the initial VS destruction rate was calculated only between the beginning of the trial and day 1 as no trend over time was detected.

3.2.6 Nucleic acid extractions and sequencing analysis

DNA and RNA were co-extracted from 0.75g (fresh weight) of the retentate (*Figure 1*), from sample days 0 and 21 of each of the three biological replicates from the 6 inoculum types (n=36 extractions). Further details can be found in the supplementary material of the thesis (*Supplementary information Chapter 3*).

3.3 Results

These degradation tests aim to evaluate the capability of each one of the different inoculum to degrade the BSG into valuable compounds. The initial pH for all the different conditions was above 8.00, in all the cases the pH decreased achieving the lowest value within the first week and increased after (*Table 3.1*).

Table 3.1 pH value for all the different conditions at the beginning, lowest value achieved and end of the trial

	Day 0	i	Lowest pH	Day 30
		Day	рН	
BMP - A	8.67 ± 0.03	2	7.55 ± 0.05	8.02 ± 0.15
BMP - B	8.36 ± 0.03	4	7.35 ± 0.04	8.02 ± 0.03
BMP - C	8.36 ± 0.01	4	7.01 ± 0.04	8.07 ± 0.03
BMP - D	8.26 ± 0.03	2	7.24 ± 0.04	7.56 ± 0.17
BMP - E	8.36 ± 0.01	2	7.18 ± 0.04	7.65 ± 0.29
BMP - F	8.42 ± 0.04	7	7.16 ± 0.04	8.02 ± 0.03

3.3.1 Biodegradation performance

3.3.1.1 Granular sludge

The VS destruction achieved with granular sludge followed a linear trend until day 4 (*Figure 3.2a*), with a destruction rate of -0.456 g VS_{BSG}/(L·day·g VS_{Inoculum}) (R^2 0.99) (*Table 3.2*). After this, the destruction rate decreased and by the last two sampling points (day 21 and 30), the VS destruction was practically non-existing.

The highest VFAs concentration achieved was on day 2, with 0.32 (± 0.07) g COD/L, while the total sCOD peaked on day 7 at 1.91 g COD/L (*Figure 3.2a*). The acidification yield during this period represented less than 25% of the overall sCOD. For the last two sampling points of the experiment, VFAs were no longer detected and the sCOD was almost depleted by the end of the trial (days 21 and 30).

In the case of the gas production, methane was the major product with a final concentration of 47.7 (\pm 19.9) % (*Figure 3.3a*). Two different production rates were evident: the first from day 0 to day 4, and the second from day 4 to day 14. The first production rate corresponded to 85.43 ml CH₄/(day·g VS_{Inoculum}) (R² 0.99) and the second rate was 26.06 ml CH₄ /(day·gVS_{Inoculum}) (R² 0.98) (*Table 3.2*). After day 21, there was no further gas production.



Figure 3.2 Solid destruction, soluble chemical oxygen demand and concentration of volatile fatty acids for each of the experiments. a) Granular sludge, b) Crushed granular sludge, c) Giraffe manure, d) rumen fluid, e) rhino manure, and f) tiger manure

3.3.1.2 Crushed granular sludge

For the crushed granular sludge, four different rates of VS destruction were observed (*Table 3.2b*). In the first 24 hours, the solid destruction rate was the fastest, -0.889 g VS_{BSG}/(L·day·g VS_{Inoculum}) (*Table 3.2*), which corresponded to a VS destruction of 18.34 (\pm 2.07) %. After the initial day, the VS destruction rate decreased and remained constant until day 7 with a rate of -0.205 g VS_{BSG}/(L·day·g VS_{Inoculum}) (R² 0.99). The third rate from day 7 to day 21 was -0.062 gVS_{BSG}/(L·day·gVS_{Inoculum}) (R² 0.99). Between the last two points of this trial the destruction rate was -0.019 gVS_{BSG}/(L·day·gVS_{Inoculum}), with a final VS destruction of 63.45 (\pm 3.67) %.



Figure 3.3 Gas production and its composition for each of the experiments on BSG degradation during the incubation of BSG with different inocula. a) Granular sludge, b) Crushed granular sludge, c) Giraffe manure, d) rumen fluid, e) rhino manure and f) tiger manure

The highest sCOD concentration achieved with this inoculum was on day 4, with a concentration of 2.60 (\pm 0.08) g COD/L (*Figure 3.2c*). By the next sample point, day 7, the sCOD concentration had halved and continued to decrease slightly over the course of the experiment to a final concentration of 0.38 (\pm 0.07) g COD/L. The VFA content followed a similar trend to the sCOD (*Figure 3.2b*). From day 1 to day 7 the VFA concentration was almost stable, fluctuating around 1.29 (\pm 0.26) g COD/L. After this point the VFA concentration decreased until being completely consumed by day 21.

The initial gas production rate was constant until day 7 at 44.47 ml CH₄/(day·gVS_{Inoculum}) (R² 0.96) (*Table 3.2*). After this day the gas production rate decreased slightly from day 7 to day 21, corresponding to 14.26 ml CH₄/(day·gVS_{Inoculum}) (R² 0.98). In the case of the crushed granular sludge there was production of gas until the last day of the trial, where the production rate was 3.56 ml CH₄/(day·gVS_{Inoculum}).

3.3.1.3 Giraffe manure

Using giraffe manure, an initial VS destruction (day 0 - day 2) of 10.99 (\pm 0.79) % was achieved, with a destruction rate of -0.364 VS _{BSG}/(L·day·g VS_{Inoculum}) (R² 0.94) (*Table 2*). From day 4 until day 21 this rate decreased and was almost constant at -0.114 g VS_{BSG}/(L·day·g VS_{Inoculum}) (R² 0.99). The final VS destruction rate between the two last points corresponded to -0.052 g VS_{BSG}/(L·day·g VS_{Inoculum}), where the final VS destruction at the end of the trial was 55.60 (\pm 6.20) % (*Figure 3.2c*).

	Initial Rate Destruction	Final % VS destruction	sCOD _{final}	Acidification Level, final	Final Accumulative Gas
	gVS _{BSG} /(L·day·g VS _{Inoculum})		g/L	g COD/g sCOD	mL
Granular sludge	-0.456	64.8 ± 11.9	0.11 ± 0.07	0	916.0 ± 241.1
Crushed granular sludge	-0.889	63.4 ± 3.7	0.28 ± 0.02	0	831.4 ± 132.9
Giraffe manure	-0.364	55.6 ± 6.2	1.33 ± 0.16	0.49 ± 0.28	786.0 ± 155.1
Rumen fluid	-1.232	47.6 ± 6.7	3.33 ± 0.55	0.36 ± 0.12	58.7 ± 24.8
Rhino manure	-0.280	33.1 ± 4.23	4.62 ± 0.16	0.84 ± 0.15	50.3 ± 26.2
Tiger manure	-0.090	44.1 ± 4.0	3.28 ± 0.54	0.98 ± 0.38	45.3 ± 18.8

Table 3.2 Main parameters for the solid biodegradation activity test for each of the different conditions

The sCOD and VFAs began increasing immediately, and on days 2 and 4 over 96 % of the sCOD was attributed to VFAs (*Figure 3.2c*). After day 4, the ratio between VFAs and sCOD decreased, and the VFA concentration from day 4 until day 10 fluctuated around $3.34 (\pm 0.63)$ g COD/L. This concentration begun to decrease, reaching a final VFA concentration of 0.65 (\pm 0.30) g COD/L by day 30. The VFA profile consisted mainly of acetic, propionic and butyric acid. Acetic acid was the major compound at the start of the trial and accumulated until day 4, after which it decreased and by day 10, propionic acid was the major compound (*Figure 3.4c*). The butyric acid concentration peaked on day 7, then

decreased and was completely consumed by day 21. On day 30, there were still small concentrations of acetic (0.17 \pm 0.05 g COD/L) and propionic (0.41 \pm 0.15 g COD/L) acid.

Biogas production from giraffe manure was absent until day 7 and the first gas sample could be collected at day 10. Between these two points, CO_2 was the major gas produced. However, its production rate significantly decreased after this point to 2.72 ml $CO_2/(day \cdot g \ VS_{inoculum})$ (R² 0.96) until the end of the trial (*Figure 3.3c*). The methane production was constant from day 7 to day 21, with a rate of 21.88 ml CH₄/(day \cdot g \ VS_{inoculum}) (R² 0.99) (*Figure 3.3c*). From day 21 to 30, the methane production rate halved to just 12.08 ml CH₄/(day \cdot g \ VS_{inoculum}).



Figure 3.4 Volatile fatty acid profile and total soluble chemical oxygen demand. a) Granular sludge, b) Crushed granular sludge, c) Giraffe manure, d) rumen fluid, e) rhino manure, and f) tiger manure

3.3.1.4 Rumen fluid

In the first 24 hours, rumen fluid allowed a VS destruction rate of -1.232 g VS_{BSG}/(L·day·g VS_{Inoculum}), with a VS destruction of 18.69 (\pm 1.38) % (*Table 3.2*). By day 4 the VS destruction reached 43.79 (\pm 2.82) %. Thereon, until the end of the trial, the VS destruction was quite poor, where the VS at the end of the trial (day 30) was 51.52 (\pm 4.89) % (*Figure 3.2d*).

VFA production peaked on day 4 with a total concentration of 1.37 (\pm 0.05) g COD/L (*Figure 3.2d*). This was almost constant throughout the experiment, where the average VFA concentration from day 2 until the end of the trial was 1.21 (\pm 0.21)/g COD L. A similar trend was observed in sCOD concentrations during the early stages of the trial (day 0 – 7). However, the highest concentration of sCOD, 4.91 (\pm 0.82) g COD/L was achieved on day 14 (*Figure 3.2d*). Thereafter, the sCOD concentration decreased continually until day 30 to a final concentration of 2.71 (\pm 0.16) g COD/L. The total gas production was less than 100 mL over the whole trial, and this was produced between days 1 and 4 (*Figure 3.3d*).

3.3.1.5 Rhinoceros manure

By day 4, the VS destruction associated with rhinoceros manure as inoculum was 14.51 (± 1.27) % (*Figure 3.2f*), which corresponded to a rate of -0.280 g VS_{BSG}/(L·day·g VS_{Inoculum}) (R² 0.97) (**Table 3.2**). This rate decreased as the trial proceeded, with two distinct rates: from day 4 to 10 and from day 10 to 21, with a VS destruction rate of 0.122 g VS_{BSG}/(L·day·g VS_{Inoculum}) (R² 0.98) and 0.039 g VS_{BSG}/(L·day·g VS_{Inoculum}) (R² 0.97), respectively. The final VS destruction achieved with this inoculum was 27.80 (± 2.78) % (*Figure 3.2e*).

The VFA profile consisted mainly of acetic, propionic and butyric acid (*Figure 3.4e*). Acetic acid was the major compound comprising around 50 % of the total VFAs after the second day. Propionic acid ranged between 35 - 40 %, and butyric acid between 5 - 10 %. The sCOD increased most rapidly from day 0 to day 4, when it reached a concentration of $3.11 (\pm 0.25)$ g COD/L (*Figure 3.2e*). The sCOD continued increasing, but at a much lower rate, until day 30 when it reached its highest concentration $4.62 (\pm 0.52)$ g COD/L. After day 4 the average contribution of the VFAs to the total sCOD was fairly constant (*Figure 3.2e*), with an acidification level of $84.58 (\pm 5.07)$ %.

The total gas production with this inoculum was 63.0 (\pm 26.6) mL, which was produced between days 4 and 7 in all the replicates (*Figure 3.2e*). This gas composition was mainly CO₂ and a small fraction of CH₄, 34.0 \pm 5.5 mL and 16.5 \pm 5.2 mL, respectively.

3.3.1.6 Tiger manure

Virtually no VS destruction occurred in the first 24 hours when incubating the BSG with tiger manure inoculum (*Figure 3.2f*). From day 1 to day 4 the VS destruction increased from 1.29 (\pm 0.50) % to 28.93 (\pm 1.22) %, where the VS destruction rate was -0.627 g VS_{BSG}/(L·day·g VS_{Inoculum}) (R² 0.98) (*Table 3.2*). From day 7 until the end of the trial the VS destruction slowed down, and by the end of the trial the VS destruction corresponded to 44.08 (\pm 4.02) %.

The VFA concentration increased throughout the experiment, with the highest concentration detected on day 21 at 3.61 (\pm 0.74) g COD/L (*Figure 3.2f*). The VFA profile was similar over the 30 days, where acetic acid was the main compound, followed by propionic acid and then by butyric acid, which represented, respectively, 54.5, 29.7 and 4.6 %. Valeric acid comprised approximately 4.5 % and isovaleric acid, detected after the day 14, close to 5 % (*Figure 3.4f*). A small amount of gas was detected by day 4 (42.2 (\pm 18.8) mL) with CH₄ and CO₂ below the detection limit (*Figure 3.3f*).

3.3.2 Microbial community analysis of the inocula

As the aim of this chapter was to find the best microbial community of hydrolysing BSG, data were mined for prokaryotic genera that are typically associated with enhanced abilities to degrade cellulosic biomass (*Figure 3.5*). We searched for the genera listed by Liang *et al.* (2014) as cellulase producers, in their research isolating and screening cellulolytic bacteria. The common trend to all samples was an increase in these genera with time, where in some samples virtually none were detected on day 0. These included both granular sludge types and counter-intuitively, rumen fluid. Tiger manure, with its large contribution from various *Clostridium*, contained the most members of potentially cellulolytic bacterial genera.

Further details regarding the (S3.2.1 Comparison of the microbial composition of the inocula, (S3.2.2 Presence vs activity of OTUs in each inocula (Day 21) and (S3.2.3 Differences in microbial community structure and relationships with biodegradation data can be found in the supplementary information at the end of the thesis.



Figure 3.5 Relative abundances of bacterial genera associated with cellulose degradation, as detected in each of our inocula at the different sample times. Bars represent the average relative abundance detected in biological replicates (n=3)

The majority of the archaeal community comprised of methanogens (around 87%), with unknown archaea contributing to 10% and *Thermoplasmatales* the final 3%. The sludge samples contained the largest methanogenic consortia, where archaea contributed more than 22% of these communities at the start of the experiment (*Table 3*). Archaea were present in rumen fluid at just 3% relative abundance and in all other samples at less than 1%, where they were virtually absent from tiger manure (0.02% of initial community). While relative numbers of archaea dropped significantly in sludge samples, down to 10% or below by day 21, they contributed to more than a third of the active (cDNA) population. In contrast, archaea increased in number in all samples of animal origin, but most notably in giraffe manure where they increased from 0.8% to 9% relative abundance. In both rumen fluid and tiger manure, archaea were present (DNA) in higher numbers than they were active (cDNA) on day 21, while the giraffe and rhinoceros manure followed the opposite trend (higher abundance at cDNA than DNA level) (*Figure 3.5*).

	Relative abundances of archae (%)					
Inoculum	DNA		cDNA			
	Day 0	Day 21	Day 21			
Granular sludge	21.94	6.14	35.65			
Crushed granular sludge	27.65	10.43	38.85			
Giraffe manure	0.78	9.04	13.05			
Rumen fluid	2.99	4.99	2.22			
Rhino manure	0.17	1.38	2.31			
Tiger manure	0.02	0.28	0.07			

Table 3.3 Relative abundances of archaea in the DNA fraction (days 0 and 21) and in the cDNA fraction (day 21) in the 6 inocula used to degrade brewery spent grain

3.4 Discussion

Six diverse inocula were assessed for their potential to effectively use a relatively recalcitrant (lignocellulosic) material, BSG, as the feedstock. Their performance can be compared firstly based on the destruction of the feedstock, as well as the rate of production of intermediates and end products, where methane will be discussed first as VFA accumulation is dependent on this.

3.4.1 Volatile solids destruction

The highest final VS destructions, of over 60%, were achieved using granular and crushed granular sludge. These were followed (in decreasing efficiency) by giraffe manure (55%), rumen fluid (50%), tiger manure (45%) and rhinoceros manure (33%). The microbial community data were mined for genera associated with cellulolytic activity, the two best performing inocula (sludge) had the fewest of these members, with their total absence detected on day 0. Indeed, it was the tiger manure where

the most abundant of these species were detected, perhaps not surprising as the *Clostridiales* were a dominant clade in this dung type, as is frequently seen in other carnivorous animals (Menke et al., 2014; An et al., 2017). This could explain how this inoculum was able to achieve a VS destruction of 10% better than that of manure from the herbivorous rhinoceros. While the genera searched for were not exhaustive, it appears that the higher VS destruction rates achieved with granular sludge samples could not be explained by the presence of more cellulolytic bacteria.

Rumen fluid resulted in the highest initial VS destruction rates, as expected due to the presence of hydrolytic bacteria and high enzymatic content typical in this sample type (Baba et al., 2017; Deng *et al.*, 2017). However, after the second day the VS fraction remained constant and its destruction only increased at the end of the trial. On the other hand, the sCOD increased from day 10 to day 14 and decreased slowly (*Figure 3.2d*). This could be related to the solubilisation of suspended solids that contributed to the increased sCOD, such as humic acids, which have been shown to inhibit the hydrolytic enzymes responsible for lignocellulosic degradation (Andric et al., 2009).

Another evident trend in the granular sludge samples with the best VS destruction, was their enhanced methane production (*Figure 3.3a*) and lack of VFA (*Figure 3.2a*) accumulation when compared to the other inocula.

Further details for relation of the microbial community and volatile solids destruction can be found in the supplementary material (*S3.3.1 Volatile solids destruction*).

3.4.2 Gas production

The six different inocula used in the present work can be divided in two different main groups based on their biogas production. The group responsible for a high production of gas consisted of granular sludge, crushed granular sludge and giraffe manure. The group presenting poor gas production, all below 100 mL after the 30 days of the experiments, included rumen fluid, rhinoceros manure and tiger manure.

The inocula group capable of high gas production could be subdivided based on two different behaviours: i) inocula that started to produce gas immediately, namely granular and crushed granular sludge; and ii) inocula that produced gas only from day 7, namely giraffe manure. The granular sludge was from an anaerobic reactor for methane production, thus high methanogenic activity was expected and indeed biogas production was quite fast with hydrolysis as the rate limiting step, with VFA accumulation close to zero.

Between the two sludge samples, the only difference was the integrity of the granule structure in the granular sludge inoculum versus its absence in the crushed sludge. The destruction of the granular

structure was performed in an effort to provide the enzymes and hydrolytic bacteria a greater accessibility to colonise and attack the complex cellulose and hemicellulose structures. However, in contrast the granular structure allows soluble compounds to be rapidly converted into methane due to the close proximity of the different microbial groups involved (Trego et al., 2020), where complex molecules released from the hydrolysis of the BSG are converted to VFAs in the external layers of the granules and these are taken up by methanogens in the internal layers by diffusional transport (MacLeod et al., 1990; Gonzalez-Gil et al., 2001). This explains the low VFA and sCOD concentration with this inoculum (Figure 3.2a). Conversely, with crushed granular sludge, the intermediate compounds were produced and excreted directly in the medium, thus promoting the slightly lower gas production rate seen in the crushed sludge variant (Figure 2D). In addition, the structure of granular sludge has been shown to protect methanogens from toxic compounds and hostile environments (Show et al., 2020), a benefit that is lost upon crushing. Similar results were obtained by Schmidt and Aching (1991) when working with intact and disintegrated granules, where the specific methanogenic activity of hydrogenotrophic methanogens (fed with H₂ and CO₂) increased following disintegration, while the activity of acetoclastic methanogens (fed acetate) decreased following granule disruption. They determined this adverse effect was because the methanogens were no longer able to metabolize acetate (Schmidt and Aching, 1991). However, perhaps therein, as well as in our study, lower gas production could be attributed to the fact that flocculants are more sensitive to environmental factors compared to granules with a higher tolerance to toxic compounds in the media (Fang et al., 2000). Indeed in our samples, while relatively more methanogenic archaea were detected on day 21 in samples seeded with crushed than with intact granules, the proportion of active members was notably higher in the granular sludge (DNA: cDNA of 1 : 5.8) than in crushed sludge (DNA:cDNA of 1 : 3.7). A lower proportion of transcriptionally active methanogens in the crushed granules could suggest that conditions were less optimal than those with intact granules. VFA accumulation could have contributed to such a toxic effect on the methanogens, thereby lowering the number of active members and therefore methane production rates (Chapleur et al., 2016). This would explain why biogas production was ongoing in the crushed granular sludge until the end of the trial when compared with the intact granular sludge from which no gas was collected on day 30.

Despite the fact that cumulative gas production from giraffe manure at the end of the trial was lower than in the two sludge samples (*Table 3.2*), gas production in the former was still ongoing and VFAs and sCOD remained (*Figure 3.2c* & *Figure 3.3c*). This suggests further methane production could have occurred upon prolonged incubation. Inoculation with giraffe manure resulted in poor initial rates of gas production (*Figure 3.3c*), which could be a result of the low abundance of archaea in this sample on day 0 (*Table 3*). Methanogens are typically associated with slow doubling times, in the range

of four to fourteen hours with some even taking 12 days (Jabłoński et al., 2015), thus any increase in abundance would likely be slow. Indeed, by day 21 the relative abundance of archaea in this giraffe manure had increased the most, by more than an order of magnitude where, in contrast, numbers in both sludge inocula decreased (**Table 3.3**).

In the case of the other three 'low gas producing' inocula, biogas was detected at only one sample point for both tiger (day 4) and rhinoceros faeces (day 7), the two inocula with the lowest relative abundances of archaea on both days 0 and 21. With rumen fluid as the inoculum, gas production was only measurable on the first two days of the trial, in which both methane and carbon dioxide were detected. The relative abundances of archaea initially detected in rumen fluid samples were the highest of the four animal-origin inocula. While the relative abundance of this clade increased slightly with time (**Table 3.3**), their relative abundance in the active (cDNA) fraction from the same samples was 50% lower than that in the DNA fraction. This suggests some inhibition of transcriptional activity, indicating conditions were sub-optimal for the methanogenic archaea within these samples.

3.4.3 Volatile fatty acid production

As VFAs are typically consumed in AD systems by acetoclastic methanogens, it is no surprise that the three inocula that saw the highest methane production accumulated the lowest VFA concentrations. Contrastingly, the best performing inoculum in terms of total VFAs produced by the end of the trial was rhino manure, which presented the lowest methane production, and the lowest solid destruction. This was closely matched by the accumulation seen in tiger manure samples, where also minimal methanogenesis was detected. Significant concentrations of VFAs also accumulated when using giraffe manure, however this only lasted until day 4, after which VFAs started to be consumed (*Figure 3.4c*). This coincided likely with the increase in methanogenic archaea detected with this inoculum by day 21 (**Table 3.3**).

In samples seeded with rumen fluid, VFA concentrations constituted around 35% of the total sCOD, and the constituents of the remaining fraction of the sCOD are still not clear. It was possible to determine this fraction was not composed of proteins or lactic acid and neither were any alcohol groups detected. More analysis need to be done to identify these soluble compounds, which currently we hypothesise are long chain fatty acids and/or humic acids, produced during the hydrolysis of the BSG (Ueno et al., 2001; Azman et al., 2017).

3.4.4 Potential inhibition

A number of studies have demonstrated the inhibitory effects of different compounds (i.e. humic acids, phenolic compounds or VFAs) and their threshold concentrations at each of the four steps of AD (Siegert et al., 2005; Fernandes et al., 2015; Chapleur et al., 2016). These studies demonstrated

that the most sensitive step is hydrolysis, followed by methanogenesis, and acetogenesis and then acidogenesis. Thus, the accumulation of the VFAs in the present study could have hindered the VS degradation (*Figure 3.2*).

Chapleur *et al.* (2016) studied the inhibitory effect of phenol on the degradation of lignocellulosic materials. One of the first effects, even at a low concentration, was the accumulation of acetic and propionic acid in the system. Acetic acid took 10 days to be consumed and propionic acid 20 days, thus resulting in a lower biogas production rate (Chapleur *et al.*, 2016). This trend was similar to the present study when giraffe manure was used as an inoculum, where until day 7 only hydrolysis was seen, in the absence of both acidogenesis and methanogenesis (*Figure 3.2c*). In our case, however, the toxic compounds would likely have been the VFAs. However, by day 10 there was gas production and VFAs started to be consumed sequentially starting with butyric, then acetic and finally propionic acid. This behaviour would be related to the acclimatization period of the methanogens to the new environmental conditions in the biodegradation activity test and the toxic effect of the VFAs, with concentrations up to 3.86 g/L (*Figure 2*). A similar adaptation period of the inoculum to fermented compounds was observed by Haruta *et al.* (2002) when working with rice straw.

In terms of VFA toxicity during AD, Siegert and Banks (2005) studied their toxic effect on methane production using lignocellulosic material as a feedstock. The increase in VFA concentration initially inhibited cellulolytic activity, then hydrolysis of glucose, methane fermentation and finally, when the VFA concentrations reached 8 times higher than those at the start, acidogenesis and acetogenesis were affected (Siegert and Banks, 2005). The accumulation of fermented compounds perhaps contributed to the lower gas production rates in some of our samples, and to the longer lag phase in gas production from giraffe manure.

Similar observations of inhibition have been reported when studying the effect of fermented compounds and chemicals, such as humic acid, fulvic acid, long chain fatty acids or phenol on the hydrolysis of lignocellulosic and hemicellulosic materials (Hobson et al., 1997: Fernandes et al., 2015; Chapleur et al., 2016). Thus, perhaps it was one of the unidentified intermediates constituting the remaining 65% of the sCOD not attributable to VFAs which was particularly inhibitory in rumen fluid seeded samples. It was assumed that both ruminant samples (giraffe manure and cow bovine rumen fluid) would be well adapted to the BSG feedstock. However, this proved not to be the case for rumen fluid (*Figure 3.2d*). Perhaps this inoculum was more sensitive to changing conditions, whereas microbes originating from manure had already been pre-exposed to sub-optimal conditions (for example low temperature and oxygen) and were able to adjust better. In all other inocula tested except rumen

fluid, VFAs contributed to approximately 90% of the total sCOD for the majority of the experiments (*Figure 3.4*).

The granular sludge, crushed or not, was from an industrial methane production reactor treating dairy wastewater after fat removal, which is an easier biodegradable material compared to BSG. Yet despite this, these two inocula performed well under the conditions tested. Li *et al.* (2020) achieved the highest degradation of a lignocellulosic feedstock with the inocula containing the highest enrichment of cellulolytic bacteria, which differs with the present study as granular sludge presented the lowest cellolytic bacteria content (*Figure 3.5*). The VS destruction by the rumen fluid and manure from herbivorous animals was expected to be faster, as their diet relies on a high capacity to degrade fibres. Among these inocula of herbivore origin, the giraffe manure presented the highest solid destruction and it was the only one capable of consuming VFAs and producing methane in high amounts. This could indicate a relationship between VFA accumulation and a poorer solid destruction rate, thus explaining why the granular sludge was more effective at degrading the lignocellulosic feedstock, due to a higher active methane production which proved advantageous due to the consequence that it lowered the VFA concentrations. Similarly, Jiang *et al.* (2020) obtained optimal AD performance when increasing the organic loading rate using food waste during the bioaugmentation that enhanced the population of acetoclastic methanogens.

A similar relationship between VFAs and hydrolytic activity was determined by Jankowska et al. (2017) during VFA production from maize silage, among other compounds. One of their findings was the association between the increase in VFA concentration and the reduction in hydrolysis yield. Sasaki et al. (2010) optimised methane production from rice straw by submerging carbon fibre textiles, with the aim of retaining and increasing the number of microorganisms. However, they determined that this increase in microorganisms was not enough to fully explain the higher decomposition of the rice straw. In the reactor containing carbon fibre textiles, methane production increased more than 3.5 fold, while the reactor without fibres showed an accumulation of VFAs. This VFA accumulation was not considered by the authors, but could have played a key role in the low biodegradation of rice straw. The toxic effect of accumulated VFAs, could have played a role in decreasing the solid destruction from 57.2 to 30.5 % with and without the carbon fiber textiles (Sasaki et al., 2010). In the present study, the conversion of VFAs into methane when using granular sludge and giraffe manure would have alleviated this possible toxic effect and improved the solid degradation. Contrastingly in samples with little methane production, the hydrolytic activity may have been negatively affected by the accumulation of VFAs. However, further research must be done to validate this hypothesis. Furthermore, the VS degradation of lignocellulosic feedstock by herbivore inocula could potentially be

improved, for example by the continuous recovery of the VFAs using membrane processes that would lower the toxicity in the media (Saboe et al., 2018).

3.5 Conclusions

This study has shown the influence of using different inocula for the production of different chemical compounds from the same feedstock. The best solid destruction was attained using granular sludge, although in relation of the ratio of solid destruction to accumulated fermented compounds, the best performance was from the rhino manure inoculum. An increasing amount of research is focusing on the use of AD to produce intermediate fermented compounds. The hydrolysis efficiency and the accumulation of fermented compounds should be studied to determine if there is any inhibitory effect or deficiency at this level. A deeper understanding of the anaerobic digestion process of lignocellulosic materials for chemical and material production is required as the role of VFAs in the solid degradation is not clear. If methane is not the end product, the relationship between the accumulation of intermediate products, such as VFAs, and a lower hydrolysis yield must be established. A microbial community resistant to high concentrations of fermented compounds should be established, or these fermented compounds should be recovered from the medium to alleviate their toxic effect. Another valuable avenue of research could be into the combination of different inocula that show cellulosic activity and acidogenic activity in order to maximise hydrolysis and VFA accumulation rates.

3.6 References

- An C., Okamoto Y., Xu S., Eo K. Y., Kimura J., Yamamoto N. (2017) "Comparison of fecal microbiota of three captive carnivore species inhabiting Korea" The Journal of veterinary medical science 79 t3, 542-546. https://doi.org/10.1292/jvms.16-0472
- Andrić P., Meyer A. S., Jensen P. A., Dam-Johansen K. (2009) "Effect and Modeling of Glucose Inhibition and In Situ Glucose Removal During Enzymatic Hydrolysis of Pretreated Wheat Straw." Applied Biochemistry and Biotechnology 160 (1): 280. https://doi.org/10.1007/s12010-008-8512-9
- Angelidaki I., Mogensen A. S., Ahring B. K. (2000) "Degradation of organic contaminants found in organic waste", Biodegradation 11(6), pp. 377-383. doi: 10.1023/a:1011643014990
- APHA (2012) "Standard Methods for Examination of Water and Wastewater 2012" American Public Health Association.
- Azman S., Khadem A. F., van Lier J. B., Zeeman G., Plugge C. M. (2015) "Presence and Role of Anaerobic Hydrolytic Microbes in Conversion of Lignocellulosic Biomass for Biogas Production." Critical Reviews in Environmental Science and Technology 45 (23): 2523-2564. https://doi.org/10.1080/10643389.2015.1053727.
- Baba Y., Matsuki Y., Mori Y., Suyama Y., Tada C., Fukuda Y., Saito M., Nakai Y. (2017) "Pretreatment of lignocellulosic biomass by cattle rumen fluid for methane production: Bacterial flora and

enzyme activity analysis." Journal of Bioscience and Bioengineering 123 (4): 489-496. https://doi.org/10.1016/j.jbiosc.2016.11.008.

- Biopol (2009) Assessment of BIOrefinery concepts and the implications for agricultural and forestry POLicy https://edepot.wur.nl/173825
- Biorefinery Euroview (2009) Current situation and potential of the biorefinery concept in the EU: strategic framework and guidelines for its development https://cordis.europa.eu/project/id/44275/reporting/es
- Bozell J., Petersen G.R. (2010) Technology development for the production of biobased products from biorefinery carbohydrates—the US Department of Energy's "Top 10" revisited, Green Chemistry 12(4) 525-728. https://doi.org/10.1039/B922014C
- Castilla-Archilla J., O'Flaherty V., Lens P. N. L. (2019) Biorefineries: Industrial Innovation and Tendencies, In: Bastidas-Oyanedel J.-R., Schmidt J. S., (Eds.), Biorefinery: Integrated Sustainable Processes for Biomass Conversion to Biomaterials, Biofuels, and Fertilizers, Springer, Switzerland, pp 3-35.
- Deng Y., Huang Z., Ruan W., Zhao M., Miao H., Ren H. (2017) "Co-inoculation of cellulolytic rumen bacteria with methanogenic sludge to enhance methanogenesis of rice straw" International Biodeterioration & Biodegradation 117, 224-235. https://doi.org/10.1016/j.ibiod.2017.01.017
- Fang H. H. (2000) "Microbial distribution in UASB granules and its resulting effects" Water Science and Technology 42(12), 201-208. https://doi.org/10.2166/wst.2000.0272
- Fernandes T. V., van Lier J. B., Zeeman G. (2015) Humic Acid-Like and Fulvic Acid-Like Inhibition on the Hydrolysis of Cellulose and Tributyrin. BioEnergy Research 8(2), 821-831. https://doi.org/10.1007/s12155-014-9564-z
- Garcia-Aguirre J., Aymerich E., González-Mtnez. J de G, Esteban-Gutiérrez M. (2017) "Selective VFA production potential from organic waste streams: Assessing temperature and pH influence." *Bioresource Technology* 244 (Part 1): 1081-1088. https://doi.org/https://doi.org/10.1016/j.biortech.2017.07.187.
- Gonzalez-Gil G., Seghezzo L., Lettinga G., Kleerebezem R. (2001) "Kinetics and mass-transfer phenomena in anaerobic granular sludge" Biotechnology and Bioengineering 73(2), 125-134. https://doi.org/10.1002/bit.1044
- Haruta, S., Cui Z., Huang Z., Li M., Ishii M., Igarashi Y. (2002) "Construction of a stable microbial community with high cellulose-degradation ability." Applied Microbiology and Biotechnology 59 (4): 529-534. https://doi.org/10.1007/s00253-002-1026-4.
- Hobson P. N., Stewart C. S. (1997) (Eds.), The rumen microbial ecosystem (2nd ed.) Blackie Academic & Professional, London. DOI: 10.1007/978-94-009-1453-7
- IBA, 2020. Irish Beer Market Report 2020. https://www.ibec.ie/drinksireland/-/media/documents/drinks-ireland-new-website/beer/beer-market-report-2020.pdf. Accessed on 29th of May 2021
- Infantes, D., González del Campo A., Villaseñor J., Fernández F. J. (2011) "Influence of pH, temperature and volatile fatty acids on hydrogen production by acidogenic fermentation." International

Journal of Hydrogen Energy 36 (24): 15595-15601. https://doi.org/10.1016/j.ijhydene.2011.09.061.

- ISO 15923-1, C (2013) "International Organization of Standardization. Ammonium, Nitrate, Nitrite, Orthophosphate", Silicate, Ed. Geneva, Switzerland.
- Jabłoński S., Rodowicz P., Łukaszewicz M. (2015) Methanogenic archaea database containing physiological and biochemical characteristics, International Journal of Systematic and Evolutionary Microbiology 65 (Pt_4), 1360-1368. https://doi.org/https://doi.org/10.1099/ijs.0.000065.
- Jankowska E., Chwialkowska J., Stodolny M., Oleskowicz-Popiel P. (2017) "Volatile fatty acids production during mixed culture fermentation – The impact of substrate complexity and pH." Chemical Engineering Journal 326: 901-910. https://doi.org/10.1016/j.cej.2017.06.021.
- Jiang J., Li L., Li Y., He Y., Wang C., Sun Y. (2020) "Bioaugmentation to enhance anaerobic digestion of food waste: Dosage, frequency and economic analysis", Bioresource Technology 307, 123256. https://doi.org/https://doi.org/10.1016/j.biortech.2020.123256.
- Li Y., Zhao J., Achinas S., Zhang Z., Krooneman J., Euverink G. J. W. (2020) "The biomethanation of cow manure in a continuous anaerobic digester can be boosted via a bioaugmentation culture containing Bathyarchaeota." Science of The Total Environment 745: 141042. https://doi.org/10.1016/j.scitotenv.2020.141042.
- Liang Y. –L., Zhang Z., Wu M., Wu Y., Feng J.-X. (2014) "Isolation, Screening, and Identification of Cellulolytic Bacteria from Natural Reserves in the Subtropical Region of China and Optimization of Cellulase Production by <i>Paenibacillus terrae</i> ME27-1." BioMed Research International 2014: 512497. https://doi.org/10.1155/2014/512497.
- Liu T., Sun L., Nordberg Å., Schnürer A. (2018) "Substrate-Induced Response in Biogas Process Performance and Microbial Community Relates Back to Inoculum Source" Microorganisms 6(3). https://doi.org/10.3390/microorganisms6030080.
- Lynd L. R., van Zyl W. H., McBride J. E., Laser M. (2005) "Consolidated bioprocessing of cellulosic biomass: an update" Current Opinion in Biotechnology 16(5), 577-583. https://doi.org/10.1016/j.copbio.2005.08.009
- MacLeod F. A., Guiot S. R., Costerton J. W. (1990) "Layered structure of bacterial aggregates produced in an upflow anaerobic sludge bed and filter reactor" Applied and Environmental Microbiology 56(6), 1598. doi:10.1128/aem.56.6.1598-1607.1990
- Martin-Ryals A., Schideman L., Li P., Wilkinson H., Wagner R. (2015) "Improving anaerobic digestion of a cellulosic waste via routine bioaugmentation with cellulolytic microorganisms." Bioresource Technology 189: 62-70. https://doi.org/10.1016/j.biortech.2015.03.069.
- Menke S., Wasimuddin, M. Meier, J. Melzheimer, J. K. E. Mfune, S. Heinrich, S., Thalwitzer, S. Wachter, S. Sommer (2014) Oligotyping reveals differences between gut microbiomes of free-ranging sympatric Namibian carnivores (Acinonyx jubatus, Canis mesomelas) on a bacterial specieslike level, Frontiers in Microbiology 5 (526). https://doi.org/10.3389/fmicb.2014.00526.
- Mu Y., Lin X., Wang Z., Hou Q., Wang Y., Hu Z. (2019) High-production dairy cattle exhibit different rumen and fecal bacterial community and rumen metabolite profile than low-production cattle, MicrobiologyOpen 8 (4): e00673. https://doi.org/10.1002/mbo3.673.

- Mussatto S. I., Dragone G, Roberto I. C. (2006) "Brewers' spent grain: generation, characteristics and potential applications." Journal of Cereal Science 43 (1): 1-14. http://dx.doi.org/10.1016/j.jcs.2005.06.001.
- Nzeteu C. O., Anaerobic microbial hydrolysis and fermentation of food waste for volatile fatty acid production, PhD thesis (2016), National University of Ireland, Galway, Ireland.
- Nzeteu C. O., Trego A. C., Abram F., O'Flaherty V. (2018) "Reproducible, high-yielding, biological caproate production from food waste using a single-phase anaerobic reactor system." Biotechnology for Biofuels 11 (1): 108. https://doi.org/10.1186/s13068-018-1101-4.
- O'Dell J. W. (1996) Method 375.2 Determination of Sulfate by Automated Colorimetry, in: W. Andrew (Eds.), Methods for the Determination of Metals in Environmental Samples, Westwood, NJ, pp. 496-508.
- Peng X., Börner R. A., Nges I. A., Liu J. (2014) "Impact of bioaugmentation on biochemical methane potential for wheat straw with addition of Clostridium cellulolyticum." Bioresource Technology 152: 567-571. https://doi.org/10.1016/j.biortech.2013.11.067.
- Rahman K. S. M., Thahira-Rahman J., Lakshmanaperumalsamy P., Banat. I. M. (2002) "Towards efficient crude oil degradation by a mixed bacterial consortium" Bioresource Technology 85(3), 257-261. https://doi.org/https://doi.org/10.1016/S0960-8524(02)00119-0.
- Rommi K., Niemi P., Kemppainen P., Kruus K. (2018) "Impact of thermochemical pre-treatment and carbohydrate and protein hydrolyzing enzyme treatment on fractionation of protein and lignin from brewer's spent grain." Journal of Cereal Science 79: 168-173. https://doi.org/10.1016/j.jcs.2017.10.005.
- Saboe P. O., Manker L. P., Michener W. E., Peterson D. J., Brandner D. G., Deutch S. P., Kumar M., Cywar R. M., Beckham G. T., Karp E. M. (2018) "In situ recovery of bio-based carboxylic acids." *Green Chemistry* 20 (8): 1791-1804. https://doi.org/10.1039/C7GC03747C.
- Sasaki K., Morita M., Hirano S.-i., Sasaki D., Ohmura N., Igarashi Y. (2010) Efficient degradation of rice straw in the reactors packed by carbon fiber textiles, Applied Microbiology and Biotechnology 87(4), pp. 1579-1586. https://doi.org/10.1007/s00253-010-2667-3.
- Schmidt, J. E., Ahring B. K. (1991) "Acetate and hydrogen metabolism in intact and disintegrated granules from an acetate-fed, 55° C, UASB reactor." Applied Microbiology and Biotechnology 35 (5): 681-685. https://doi.org/10.1007/BF00169637.
- Shi X., Lin J., Zuo J., Li P., Li X, Guo X. (2017) "Effects of free ammonia on volatile fatty acid accumulation and process performance in the anaerobic digestion of two typical bio-wastes." Journal of Environmental Sciences 55 (Supplement C): 49-57. https://doi.org/10.1016/j.jes.2016.07.006.
- Show K.-Y., Yan Y., Yao H., Guo H., Li T., Show D.-Y., Chang J.-S., Lee D.-J. (2020) "Anaerobic granulation: A review of granulation hypotheses, bioreactor designs and emerging green applications" Bioresource Technology 300 122751. https://doi.org/10.1016/j.biortech.2020.122751
- Siegert I., Banks C. (2005) "The effect of volatile fatty acid additions on the anaerobic digestion of cellulose and glucose in batch reactors." Process Biochemistry 40 (11): 3412-3418. https://doi.org/10.1016/j.procbio.2005.01.025.

- Singh R., Kumar S. (2019) "A review on biomethane potential of paddy straw and diverse prospects to enhance its biodigestibility." Journal of Cleaner Production 217: 295-307. https://doi.org/10.1016/j.jclepro.2019.01.207.
- Trego A. C., Mills S., Collins G. (2020) "Granular biofilms: Function, application, and new trends as model microbial communities" Critical Reviews in Environmental Science and Technology: 1-24. https://doi.org/10.1080/10643389.2020.1769433.
- Tsapekos, P., Kougias P. G., Vasileiou S. A., Treu L., Campanaro S., Lyberatos G., Angelidaki I. (2017) "Bioaugmentation with hydrolytic microbes to improve the anaerobic biodegradability of lignocellulosic agricultural residues." Bioresource Technology 234: 350-359. https://doi.org/10.1016/j.biortech.2017.03.043.
- Ueno Y., Haruta S., Ishii M., Igarashi Y. (2001) "Changes in product formation and bacterial community by dilution rate on carbohydrate fermentation by methanogenic microflora in continuous flow stirred tank reactor" Applied Microbiology and Biotechnology 57(1), 65-73. https://doi.org/10.1007/s002530100760
- Vieira E., Rocha M. A. M., Coelho E., Pinho O., Saraiva J. A., Ferreira I. M. P. L. V. O., Coimbra M. A. (2014) "Valuation of brewer's spent grain using a fully recyclable integrated process for extraction of proteins and arabinoxylans." Industrial Crops and Products 52 (Supplement C): 136-143. https://doi.org/10.1016/j.indcrop.2013.10.012.
- Wall D. M., Allen E., O'Shea R., O'Kiely P., Murphy J. D. (2016) "Investigating two-phase digestion of grass silage for demand-driven biogas applications: Effect of particle size and rumen fluid addition" Renewable Energy 86 (2016), 1215-1223. https://doi.org/10.1016/j.renene.2015.09.049
- Zhang W., Dai K., Xia X.-Y., Wang H.-J., Chen Y., Lu Y.-Z., Zhang F., Zeng R.J. (2018) Free acetic acid as the key factor for the inhibition of hydrogenotrophic methanogenesis in mesophilic mixed culture fermentation, Bioresource Technology 264, 17-23. DOI: 10.1016/j.biortech.2018.05.049

Chapter 4

Thermal diluted acid hydrolysis of brewery spent grain and downstream hydrolysate acidogenic fermentation

A modified version of this chapter has been submitted as:

Castilla-Archilla, J., Papirio, S., and Lens, P. N. L. (2021) "Two step process for volatile fatty acid production from brewery spent grain: hydrolysis and direct acidogenic fermentation using anaerobic granular sludge" Process Biochemistry 100, 272–283. doi: 10.1016/j.procbio.2020.10.011

Keywords: brewery spent grain; pretreatment; thermal diluted acid hydrolysis; carbohydrates; acidogenic fermentation; volatile fatty acids

Abstract

Brewery spent grain (BSG) is an industrial waste stream with high potential for biorefining purposes due to its high content in carbohydrates, along with proteins and lignin. This work evaluated the production of volatile fatty acids (VFAs) by a two step process using BSG as renewable feedstock. Initially, a thermal (121 °C) diluted acid (sulfuric) hydrolysis of BSG at for 20 minutes was carried out in order to obtain a liquid hydrolysate rich in carbohydrates. Eighteen different combinations in terms of total solids (TS) of BSG (4, 7 and 10 % w/w) and sulfuric acid (0.0, 0.5, 1.0, 1.5, 2.0 and 3.0 % v/v) were used to perform the BSG hydrolysis. The combination using 7.0 % TS of BSG and 1.5 % of sulfuric acid was the most efficient in terms of total carbohydrate recovery (i.e. 0.44 grams of total carbohydrates per gram of TS recovered) without necessarily extra dosing sulfuric acid. Subsequently, an acidogenic batch fermentation using anaerobic granular sludge for 72 hours was performed on the BSG hydrolysate, which was fermented under five different pH conditions (uncontrolled pH from an initial pH 7.0, and constant pH controlled at 4.5, 5.0, 6.0 and 8.0). The highest VFAs concentration was obtained at pH 6.0 and reached 16.89 (± 1.33) g COD/L.

4.1 Introduction

The use of waste as feedstock for biorefineries is increasing with the aim to promote the valorisation of low-value residues and pursue the zero waste perspective (Carmona-Cabello et al., 2018). In the last century, the treatment of organic waste streams has shifted from aerobic to anaerobic bioprocesses, which require a lower energy consumption and allow the simultaneous treatment of waste and production of energy. Anaerobic digestion is one of the sustainable processes driving the transition from fossil fuels to modern, renewable energy sources. This process mainly consists of four different steps: i) hydrolysis, ii) acidogenesis, iii) acetogenesis and iv) methanogenesis (Monlau et al., 2011). The partial or total inhibition of methanogenesis can shift biomethane production to some of the intermediate compounds as final products, such as lactate (Itoh et al., 2012), hydrogen (Rosa et al., 2014; Cisneros-Pérez et al., 2017) or volatile fatty acids (VFAs) (Garcia-Aguirre et al., 2017). Some strategies to prevent methanogenesis and trigger VFA accumulation include the increase of the organic loading rate (Veeravalli et al., 2014) or the ammonium concentration (Shi et al., 2017; Zhao et al., 2018; Bengtsson et al., 2008; Infantes et al., 2011).

VFAs are short chain fatty acids (C₂ to C₆) with a broad range of industrial application and their current production relies on fossil sources (Strazzera et al., 2018). VFAs can be produced from renewable biomass through anaerobic fermentation, but their recovery is still a bottleneck in this technological development. This is especially the case for low strength wastewaters, which yield only diluted VFA concentrations. Consequently, this process is till now mainly used as intermediate step to obtain compounds that are easier to recover, e.g. polyhydroxyalkanoates (PHA) or lipids for biodiesel production (Albuquerque et al., 2011; Patel et al., 2018). Nevertheless, recent improvements in the extraction systems may allow to efficiently extract the VFAs as the final products, especially if high VFA concentrations can be obtained (Wainaina et al., 2019; Saboe et al., 2018). Thus, VFA production depends less on fossil sources and provides a trade opportunity in countries that rely on importing gas and oil. However, some parameters of the acidogenesis phase, such as the VFA concentration and the operating pH must be further investigated.

One of the most important factors affecting the yield and stability of a bioprocess is the consistent composition and availability of the feedstock throughout the year (Ravindram et al., 2016; Fava et al., 2015). Lignocellulosic materials (LMs) represent the most abundant organic feedstocks for bioindustrial purposes, with low or no costs, in comparison to the more expensive energy crops. However, the main issue when using LMs in fermentation processes is associated with their complex structure, where cellulose and hemicellulose are interwoven within lignin (Ayyachamy et al., 2013), being recalcitrant to microbial degradation in the absence of a pretreatment (Mancini et al., 2016). Many studies have focused on different pretreatment methods to make LMs more accessible to

microorganisms and thus release an elevated amount of high-value molecules, such as sugars, proteins or phenolic compounds. Some examples of these pretreatments are enzymatic hydrolysis (Niemi et al., 2012), diluted acid treatment (Carvalheiro et al., 2004) and steam explosion (Silveira et al., 2018). With regard to chemical pretreatments, the hydrolysis efficiency of lignocellulosic wastes depends on the combination of different pretreatment parameters such as temperature, time and acid concentration. This combination of parameters can be captured in a single value, known as the severity factor (Carvalheiro et al., 2004; Zoulikha et al., 2015; Fockink et al., 2018).

Many agro-food industries could provide a constant supply of LMs to be used for either energy production (Mancini et al., 2016; Dishman et al., 2017) or the recovery of high value compounds (Rocha-Martín et al., 2017). One source of sustainable biomass for biorefinery platforms is the brewery sector. The basic ingredients of beer are water, barley, hops and yeast, and the most abundant waste stream deriving from the brewing industry is the malted barley obtained post boiling. This latter mentioned solid waste is known as brewery spent grain (BSG), and approximately 15 – 20 kg is produced per hectolitre of beer. Conventionally, BSG is used for animal feeding or burnt after drying. Rarely, BSG has been used for environmental applications, such as a co-substrate in anaerobic digesters aimed at methane production (28). Recovery of valuable compounds, including lignin (Niemi et al., 2012), polyphenols (Stefanello et al., 2018), prebiotics (Moreira et al., 2013), proteins (Connolly et al., 2013), and total carbohydrates (TC), mainly glucose, xylose and arabinose (Rommi et al., 2018; Ravindran et al., 2019), from BSG has only recently gained interest.

The consistent composition and year-round availability make BSG a suitable renewable feedstock for VFA production. To enable this, a combination of different pretreatment methods and fermentation technologies is required, and the feasibility to combine them must also be evaluated. As pretreatment method for BSG, a thermal diluted sulfuric acid pretreatment was used in this study, as similar processes showed to be reliable methods used at the industrial scale, only requiring a single reaction step and a short time for the hydrolysis of the material. Therefore, we preliminarily focused on a first step on the hydrolysis of BSG using different sulfuric acid and BSG concentrations, in order to obtain a liquid hydrolysate rich in easily fermentable carbohydrates. The pretreatment conditions (i.e. low acid concentration, short reaction time of 20 minutes, and a temperature of 121°C) allowed the solubilisation of complex carbohydrates without the production of toxic compounds (e.g. furfural, hydroxymethylfurfural, formic acid, levulinic acid or total phenolic compounds) that inhibit subsequent fermentation (Carvalheiro et al., 2004; Djioleu and Carrier, 2016). Subsequently, as a second step, two stirred tank reactors inoculated with anaerobic granular sludge were run in batch mode for the production of VFAs from the obtained carbohydrate-rich liquid hydrolysate. It should be noted that between step one and two, only pH neutralization was done after the thermal diluted

sulfuric acid step and no further treatment was carried to remove possible inhibitory compounds coreleased during the BSG hydrolysis. The VFA accumulation during acetogenic fermentation was promoted by the operating parameters initial organic concentration and pH. Different pH values were used to run the batch acetogenic fermenting reactors and evaluate the TC consumption, highest VFA production rate, total VFA concentration and VFA composition.

4.2 Material and methods

4.2.1 Feedstock and inoculum

BSG was supplied by a local brewery in Galway (Ireland) for the production of lager beer. Once collected, the BSG was mixed and frozen in individual bags of 3 kg each at -20 °C. Prior to the hydrolysis step, the BSG was defrosted overnight at 15°C.

As inoculum, an anaerobic granular sludge collected from an UASB reactor treating dairy wastewater for methane production (Kilconnell, Ireland) at ambient temperature was used. The total (TSS) and volatile (VSS) suspended solids concentrations of the inoculum were 72.8 and 59.1 mg/g of wet inoculum, respectively. No pretreatment was applied to the inoculum.

4.2.2 Diluted acid hydrolysis of brewery spent grain

After defrosting, the BSG was mixed with distilled water and blended using a laboratory Waring 2-Speed Laboratory blender (USA) for 1 minute to break down the external structure of the grain and facilitate hydrolysis. Three different total solids (TS) concentrations of BSG (4, 7 and 10 % w/w) were used for the hydrolysis of BSG and the extraction of easily fermentable carbohydrates. The blended mixtures were placed in 500 mL Pyrex bottles with a working volume of 400 mL.

Sulfuric acid (95 % purity and 1.83 g/mL density), supplied from Fisher Chemical (UK), was used to perform diluted acid hydrolysis of BSG at 121 °C for 20 minutes in an autoclave (Sanyo Labo, Japan) as described by Carvalheiro et al. (2004) and Djioleu and Carrier (2016). This cost-efficient method allows the hydrolysis of carbohydrates and proteins, while lignin remains as a precipitate (Ravindran et al., 2018). Sulfuric acid was added at six different concentrations (0.0, 0.5, 1.0, 1.5, 2.0 and 3.0 % v/v), resulting in a total of 18 different combinations in terms of sulfuric acid and TS concentration of BSG. Each combination was run in triplicate. The severity factor was calculated as proposed by Carvalheiro et al. (2004) and Fockink et al. (2018) for the selected combinations to be used as feedstock for the batch acetogenic fermentation. Once hydrolysis was completed, the bottles were placed on ice to stop the reaction, and the hydrolysate was centrifuged at 5000 rpm for 30 minutes (Beckman Coulter, U.S.A.; JS-5.3 Rotor) to recover the supernatant.

4.2.3 Batch fermentation

As substrate for the acidogenic fermentation, the liquid hydrolysate obtained from the diluted acid hydrolysis performed at 7 % (w/v) of TS of BSG and 1.5 % of sulfuric acid (v/v) was used, as this pretreatment condition resulted in the highest extraction efficiency in terms of g TC/g TS (*Table 4.1*). The fermentation system consisted of two duplicate stirred tank reactors running as twin reactors operated at 37 °C. The reactors were made of a section of acrylic tube (inner diameter 15 cm) sealed at both sides with acrylic discs, thus creating a total volume of 2.1 L. A port in the side of the reactors allowed sampling, while a pH probe, gas outlet and tubing for NaOH addition were inserted through the top of the reactor (*Figure 1*). The reactors were operated in batch mode with a working volume of 1.75 L each. In both reactors, the hydrolysate was mixed with distilled water to give a total initial carbohydrate concentration of 16 g COD/L. The reactors were seeded with 3.0 g/L VSS of granular sludge and sparged for 3 minutes with nitrogen gas (N₂) to remove oxygen from the medium. To maintain homogeneous conditions in each reactor, a magnetic stirrer (Stuart UC151, UK) was used and operated at 1,500 rpm.

Table 4.1 Operating conditions used for the hydrolysis of brewery spent grain as well as the composition of carbohydrates extracted and efficiency of the hydrolysis process expressed as percentage of the total carbohydrates recovered per gram of total solids used

H₂SO₄ %	TS (BSG) %	Cellobiose (g/L)	Glucose (g/L)	Xylose (g/L)	Arabinose (g/L)	Efficiency (% gTC/gTS)
0	7	1.13 (± 0.13)	1.23 (± 0.83)	0.00 (± 0.00)	0.00 (± 0.00)	8.18 (± 1.3)
0	10	1.55 (± 0.25)	0.75 (± 0.75)	0.00 (± 0.00)	0.00 (± 0.00)	7.32(± 1.4)
1	7	0.70 (± 0.70)	10.05 (± 0.95)	10.40 (± 1.00)	11.10 (± 0.50)	40.35 (± 1.9)
1	10	2.20 (± 0.35)	12.50 (± 4.15)	14.10 (± 7.40)	14.50 (± 7.25)	34.7 (± 0.5)
1.5	7	0.00 (± 0.00)	12.80 (± 1.70)	12.05 (± 0.85)	11.80 (± 0.10)	43.8 (± 1.1)
1.5	10	2.00 (± 0.10)	16.70 (± 0.70)	15.65 (± 0.05)	12.75 (± 0.15)	36.6 (± 0.4)
3	7	0.00 (± 0.00)	11.32 (± 0.04)	6.58 (± 0.09)	5.96 (± 0.01)	44.6 (± 2.7)
3	10	0.60 (± 0.05)	20.74 (± 1.49)	19.08 (± 2.21)	8.68 (± 1.49)	42.5 (± 0.5)

VFA production was analysed under both uncontrolled and controlled pH conditions over a period of 72 h. The experimental run at uncontrolled pH started from an adjusted pH of 7.0 (\pm 0.01), and no further correction was performed while the process was ongoing, with the aim determining the lowest tolerable pH by the system. In the experiments run at controlled pH, four experimental runs were performed at different pH, where the pH was maintained at either 4.5, 5.0, 6.0 or 8.0 (\pm 0.05 units) throughout. This was achieved using a pH controller (Alpha 100; Thermo Scientific, Singapore) connected to a peristaltic pump (Masterflex 323, UK) for the addition of a 3 M NaOH solution.



Figure 4.1 Schematic experimental set up of the stirred tank reactors operated in batch mode for VFA production usng the hydrolysate of brewery spent grain as feedstock. The stirring plates, pH controller and probes as well as the dosing of the NaOH solution are shown

The reactor operation was monitored by analysing TC, monosaccharides, VFAs, lactic acid, ethanol, soluble chemical oxygen demand (sCOD), TSS and VSS. Sulfate, proteins and ammonium were analysed at the beginning and the end of the batch incubations.

4.2.4 Physico-chemical analysis

The sCOD was analysed using a commercial kit (Reagecon, Ireland) with a range from 0 to 1,500 mg COD/L. 2 mL of sample was added to the vial and digested at 150°C for 2 hours, and the absorbance was measured with a DR2500 spectrophotometer (Hach, Germany). Protein analysis was performed using the Lowry method (1951). TC were analysed by the Dubois method (1956). Glucose, xylose, arabinose and cellobiose were measured using a 1260 Infinity II liquid chromatograph (Agilent, Germany) equipped with a Hi Plex H 7.7 x 300 mm and 8 μ m (p/n PL1170-6830) column (Agilent, UK) kept at 60 °C and an RI detector at 55 °C. The mobile phase was sulfuric acid with a concentration of 0.005 M and a flow rate of 0.7 mL/min. Lactic acid and ethanol were analysed by liquid chromatography using a Prominence LC-20A Series HPLC (Shimadzu, Japan) with a Rezex ROA Organic Acid H+ column (Phenomenex, USA) heated at 40 °C and an SPD-20A UV detector set at 220 nm (Dreschke et al., 2019). As the mobile phase, a 0.0065 mM sulfuric acid solution was used at a flow rate of 0.6 mL/min.

VFA analysis was performed with a 450-GC gas chromatograph (Varian, USA) equipped with a CombiPAL autosampler, a flame ionisation detector and a FFAP BP21 capillary column (SGE Analytical Science, Australia) of 30 m length, 0.25 mm internal diameter and a 0.25 μ m film. Helium was used as carrier gas with a flow rate of 1 mL/min. The oven temperature was risen from 60 (10 s) to 110 °C (20

s) at a rate of 30 °C/min, and from 110 to 200 °C at a rate of 10 °C/min. The temperatures of the injector and detector were 250 and 300 °C, respectively.

Ammonium and sulfate analyses were performed using a Gallery Plus[®] discrete nutrient analyser (ThermoFisher, Finland) based on a colorimetric assay (O'Dell et al., 1996; ISO 15923-1, 2013). TSS and VSS concentrations were measured every 24 hours, following the procedure reported in standard methods (Apha, 2012).

4.2.5 Calculations

The carbohydrates release efficiency during the thermal diluted sulfuric acid hydrolysis was calculated taking into account the total concentration of carbohydrates extracted, the amount of liquid recovered, as well as the TS concentration and volume of sulfuric acid used.

For the acidogenic batch fermentation, to evaluate the effect of the operating pH, TC consumption and VFA production rates were calculated using the exponential phase after the initial acclimatization period from each of the replicate reactors. The rates were determined using 6 – 9 experimental points, always accepting an R-squared value above 0.89.

The calculation of the total carbohydrate consumption rate is reported in *Equation 1*:

$$-r_{TC} = \frac{[Y]_i - [Y]_f}{t_i - t_f} \qquad \left[\frac{gCOD}{(L \cdot h)}\right]$$
 Equation 1

where $-r_{TC}$: total carbohydrate consumption rate, Y: total carbohydrate concentration, t: time of the log phase for TC consumption, i: at the beginning of log phase, f: at the end of the log phase.

The total VFA production rate was determined as reported in *Equation 2*:

$$r_A = \frac{[A]_i - [A]_f}{t_i - t_f} \qquad [\frac{gCOD}{(L \cdot h)}]$$
 Equation 2

where r_A : total VFA production rate, A: total VFA concentration, t: time of the log phase for VFA production, i: at the beginning of the log phase, f: at the end of the log phase.

At each pH, the acidification level was determined (Equation 3) as previously reported by Bengtsson et al. (2008) by calculating the percentage of the total amount of VFAs (in terms of g COD/L) above the total sCOD (which includes the VFAs):

Acidification Level =
$$\frac{Total VFA concentration}{Soluble COD}$$
 $[gCOD/g sCOD]$ Equation 3

This percentage was used to evaluate the optimal pH at the end of each log phase as the highest amount of VFAs in terms of COD divided by the overall sCOD. The TC depletion (Equation 4) and the sCOD consumption (*Equation 5*) were calculated considering the initial and final points of the log phase

for VFA production. The extent of sulfate reduction was calculated using the initial and final points of each experiment (*Equation 6*).

$$Total \ carbohydrates \ depletion = \frac{[Y]_i - [Y]_f}{[Y]_i} \qquad [g \ consumed/g \ beginning] \qquad Equation 4$$
$$sCOD_{consumption} = \frac{[sCOD]_i - [sCOD]_f}{[sCOD]_i} \qquad [g \ consumed/g \ beginning] \qquad Equation 5$$

where Y: total carbohydrate concentration, t: time of log phase for VFAs production, i: at the beginning of the log phase, f: at the end of the log phase.

Sulphate reduction =
$$\frac{[S]_0 - [S]_e}{t_0 - t_e}$$
 [^{g consumed}/_{g beginning}] Equation 6

where S: sulfate concentration, t: time of log phase for VFAs production, 0: beginning of the trial, e: end of the trial.

4.3 Results

4.3.1 Efficiency of diluted acid hydrolysis of brewery spent grain

After acid hydrolysis, for the TC obtained from the 18 different pretreatment combinations; the lowest amount of carbohydrates recovered was in the absence of sulfuric acid, ranging between 2.5 - 8.5 g/L with increasing TS (*Figure 4.2a*). The highest TC concentration was achieved using 10 % TS and 3 % sulfuric acid (v/v). Standardising the extraction efficiencies by the TS concentration used, all the different conditions resulted in a similar ratio of $2.96 (\pm 0.20)$ g TC per gram of TS. The concentration of TC extracted increased linearly from 0.0 to 1.0 % v/v of sulfuric acid. At sulfuric acid concentration concentrations from 1.5 % upwards, the efficiency between the TC extracted and the volume of sulfuric acid used was lower. The composition of the carbohydrates recovered was mainly characterized in terms of glucose, xylose and arabinose (*Table 4.1*). The monosaccharides were analysed in the samples that were considered for the extraction at the fermentation step (7 and 10 % TS of BSG at 1.0 and 1.5 % of sulfuric acid), as well as for lowest and highest sulfuric acid concentration (*Table 4.1*). The presence of cellobiose was detected in all the samples for 10 % TS of BSG, and in the case of 7 % TS for 0.0 and 1.0 % of sulfuric acid.



Figure 4.2 Influence of the sulfuric acid concentration on a) release of total carbohydrates, b) chemical oxygen demand (COD), c) proteins and d) ammonium, during the diluted acid hydrolysis of brewery spent grain at 121°C for 20 minutes. Three different total solids concentrations of brewery spent grain (-10% TS - 7% TS - 4% TS) were used

For protein extraction, increasing the sulfuric acid concentration from 0.0 to 0.5 % (v/v) resulted in the highest release of proteins (*Figure 4.2c*). After this point, the protein extraction efficiency decreased. Similarly, the extraction of ammonium was the most efficient at sulfuric acid concentrations between 0.0 to 1.0 % (v/v) (*Figure 4.2d*).

All combinations of sulfuric acid at 10 % TS of BSG and the combinations of 7 % TS of BSG at 1.0 % of sulfuric acid concentration or lower, resulted in an uncompleted hydrolysis, as indicated by the presence of cellobiose (*Table 4.1*). Taking into account the concentration of all the compounds extracted during the hydrolysis, the conditions operated at 1.5 % of sulfuric acid led to the highest overall efficiency, although this did not result in the highest total carbohydrate concentration. Furthermore, a 7 % TS concentration of BSG was considered the optimal (*Table 4.2*).

% TS BSG (w/w) % H2SO4 (v/v)	4 %	7 %	10 %	
0.0	0.09 (± 0.03)	0.08 (± 0.01)	0.07 (± 0.01)	
0.5	0.25 (± 0.05)	0.23 (± 0.01)	0.21 (± 0.01)	
1.0	0.39 (± 0.03)	0.40 (± 0.02)	0.35 (± 0.01)	
1.5	0.41 (± 0.01)	0.44 (± 0.01)	0.37 (± 0.02)	
2.0	0.43 (± 0.02)	0.45 (± 0.02)	0.40 (± 0.01)	
3.0	0.42 (± 0.05)	0.45 (± 0.03)	0.42 (± 0.01)	

Table 4.2 Total carbohydrates efficiency recovery, gram of total carbohydrates per gram of brewery spent grain used (dried weight), for the thermal diluted acid hydrolysis of brewery spent grain carried out at 121°C for 20 minutes under different sulphuric acid and brewery spent grain concentrations

4.3.2	Acidogenic fermentation of br	rewery spent grain hydrolysate
-------	-------------------------------	--------------------------------

4.3.2.1 Initial conditions

The pH of the liquid fraction obtained after BSG hydrolysis at 7 % TS and 1.5 % sulfuric acid was 0.93 (\pm 0.06). The pH was adjusted by the addition of NaOH before starting acidogenic fermentation in batch reactors. The average initial conditions for the acidogenic fermentation in all experiments were sCOD 23.1 (\pm 2.1) g /L, TC 16.9 (\pm 2.2) g COD/L, proteins 3.35 (\pm 0.4) g/L, ammonium 194.9 (\pm 7.0) mg/L and sulfate 14.7 (\pm 1.8) g/L (Table 2).

4.3.2.2 Acidogenic fermentation at uncontrolled pH

In the experiment operated without pH control, the pH quickly decreased from 7.00 (\pm 0.01) to 6.66 (\pm 0.03) in 2.5 hours, reaching a final value of 4.08 (\pm 0.05) at the end of the experiment (*Figure 4.3*). The total carbohydrate concentration decreased at a constant rate of 0.24 (\pm 0.13) g COD/(L·h) to a concentration of 7.32 (\pm 1.93) g COD/L after 36 hours, when a pH of 4.19 (\pm 0.18) was observed. Subsequently, carbohydrate consumption almost ceased with a final total carbohydrate consumption efficiency of 63.7 (\pm 4.4) % (*Equation 4*). With regards to monosaccharides, glucose was completely consumed, while the arabinose and xylose concentration remained constant (*Table 4.3*).

Table 4.3 Glucose, xylose and arabinose concentration during the acidogenic fermentation along the batch experiment of the brewery spent grain hydrolysate at uncontrolled pH

Time (hour)	0	13	22	30	46	60	71
Glucose (g/L)	5.81 (± 0.19)	4.08 (± 0.63	2.63 (± 0.71)	0 (± 0)	0 (± 0)	0 (± 0)	0 (± 0)
Xylose (g/L)	5.53 (± 0.20)	5.46 (± 0.12)	4.60 (± 0.55)	5.22 (± 0.38)	4.95 (± 0.41	5.03 (± 0.28)	4.86 (± 0.41)
Arabinose (g/L)	2.75 (± 0.10)	2.35 (± 0.36)	2.67 (± 0.08)	2.55 (± 0.12)	2.52 (± 0.01)	2.55 (± 0.05)	2.45 (± 0.03)
рН	6.98 (± 0.01)	5.11 (± 0.23)	4.59 (± 0.18)	4.44 (± 0.17)	4.11 (± 0.07)	4.10 (± 0.06)	4.08 (± 0.05)


Figure 4.3 Profiles of total carbohydrates consumption, acetic and butyric acid production, and pH in the batch reactors fed with the brewery spent grain hydrolysate and operated at uncontrolled pH and 37°C

In terms of VFAs produced, acetic acid production started after approximately 7 hours, with a production rate of 0.06 (\pm 0.02) g COD/(L·h). Butyric acid was produced after 11 hours with a production rate of 0.15 (\pm 0.02) g COD/(L·h), i.e. higher than that observed for acetic acid. The acidification level in terms of g COD of VFAs per 100 g of sCOD was 44.0 (\pm 7.0) %, with acetic and butyric acids as the major constituents representing 95.5 % of the total sCOD (**Figure 4.4**). Lactic acid production started after 11 hours, but its concentration only reached a peak of 1.43 (\pm 0.43) g COD/L, i.e. 36.8 (\pm 12.1) and 24.6 (\pm 8.0) % lower than, respectively, the acetic and butyric acid concentration.



Figure 4.4 Total volatile fatty acids production and percentage of each VFA over total VFAs after the exponential phase during the acidogenic fermentation *under different pH conditions*

4.3.2.3 Acidogenic fermentation at controlled pH (4.50, 5.0, 6.0 and 8.0)

When keeping the pH constant and above 4.5 throughout the experiment, the percentage of TC consumption exceeded 90 % and was similar for each condition (**Figure 4.5***a*). The TC consumption rate increased at higher operating pH (**Figure 4.5***b*). Glucose was completely consumed after approximately 44 hours when working at pH 4.5, and only 24 hours were required to completely remove glucose when increasing the pH up to 8.0. After glucose was consumed, the degradation of xylose and arabinose started (data not shown).

The total amount of VFAs produced at each of the pH conditions tested (4.5, 5.0, 6.0 and 8.0) is shown in *Figure 5*. The acetic and butyric acid content was the most significant fraction of the sCOD (*Figure 5*), ranging between 95.5 and 99.8 %. At pH 4.5 and 5.0, the VFA concentration remained stable from 29 to 35 hours and from 11 to 23 hours, prior to resume production until a final concentration of VFAs of 11.0 (\pm 0.04) and 12.2 (\pm 2.3) g COD/L, respectively (*Figure 4.6a*). In the case of pH 6.0 and 8.0, the intermediate phase with a stable VFA concentration was not observed. Furthermore, a delay of the VFA production compared to carbohydrate consumption was observed (*Figure 4.6a & Figure 4.7*). The acetic acid production rate increased when operating the process at a higher pH. The highest production rate for butyric acid was 0.52 (\pm 0.04) g COD/(L·h) achieved at pH 5.0 (Figure 4.5b).



Figure 4.5 a) Total carbohydrates depleted, reduction of soluble COD and sulphate obtained during the acidogenic fermentation of hydrolysate from brewery spent grain under different pHs in batch reactors as well as b) production rate of the butyric and acetic acid, total carbohydrates consumption rate and acidification level

Lactic acid was produced under all conditions tested. Lactic acid production started earlier than VFA production at pH 4.5 and reached the highest concentration of 3.87 (\pm 0.76) g COD/L at 49 h and decreasing at the end of the trial to 2.07 (\pm 0.72) g COD/L. At pH 6.0, the production of lactic acid started simultaneously with VFAs, with lactic acid reaching the highest concentration of 1.08 (\pm 0.30) g COD/L at 17.5 (\pm 4.7) h, before being completely consumed after 49 h. At pH 8.0, lactic acid reached a peak of 0.62 (\pm 0.02) g COD/L after 18.5 h (*Figure 4.6b*).



Figure 4.6 Fermented compound production during the acidogenic fermentation of brewery spent grain hydrolysate in a batch-operated stirred tank reactor at 37 °C under different pH values. a) Total volatile fatty acids, b) lactic acid, c) acetic acid and d) butyric acid



Figure 4.7 Total carbohydrates consumption at different pH (4.5, 5.0, 6.0 and 8.0) during acidogenic fermentation of brewery spent grain hydrolysate in two batch-operated stirred tank reactors at 37 °C

4.3.2.4 Nitrogen compounds, sulfate and soluble chemical oxygen demand

The average initial proteins and ammonium concentrations for all the combinations corresponded, respectively, to 3.4 (\pm 0.4) g/L and 194.9 (\pm 7.0) mg/L. The final protein concentration was 1.7 (\pm 0.3) and 0.9 (\pm 0.2) g/L in the case of uncontrolled pH and pH 8.0, respectively. This led to a lower increase of the ammonium concentration in the case of uncontrolled pH with 273.1 (\pm 7.7) mg/L, and a higher increment at pH 8.0 up to 468.5 (\pm 6.5) mg/L.

The COD consumption and sulfate reduction percentage at the end of the acidogenic fermentation were investigated for each of the pH conditions tested (**Figure 4.5***a*). The lowest COD consumption of approximately 7.4 (\pm 2.3) % was obtained at uncontrolled pH, while the highest COD consumption of 18.9 (\pm 0.1) and 20.5 (\pm 3.1) % (Equation 5) was observed at pH 6 and 8, respectively (**Figure 4.5***a*). The sCOD/sulfate ratio in the reactors at the beginning of the experiments was 1.41 (\pm 0.10) and 1.28 (\pm 0.10) at the end of the experiments. The average initial and final sulfate concentrations in both reactors were 15.84 (\pm 2.20) g/L and 14.99 (\pm 2.50) g/L, respectively, under all the conditions investigated. This implies an average sulfate reduction efficiency of 5.1 (\pm 1.8) % (**Figure 4.5***a*). The ethanol concentration constantly remained below the detection limit at all pH values investigated.

4.4 Discussion

4.4.1 Operational conditions during the acidogenic fermentation

This study shows for the first time the feasibility of producing high concentrations of VFA from BSG by combining a diluted acid hydrolysis of the BSG in step one with a direct acidogenic fermentation of the BSG hydrolysate using untreated anaerobic granular sludge. Despite the anaerobic granular sludge originated from an industrial reactor aimed at methane production, the operating conditions adopted in this study resulted in VFA accumulation rather than methanogenesis as no VFA consumption was

observed after reaching the maximum VFA concentration (*Figure 4.6*). This supports previous research, which showed that a low pH and a high initial organic loading act as inhibitors of methanogenesis (Shi et al., 2017; Infantes et al., 2011; Görke et al., 2008; Xie et al., 2014). The highest acidification level was achieved at pH 6.0, with VFAs representing 93.0 % (\pm 2.1%) of the soluble COD (17.6 (\pm 0.67) g/L) (*Table 4.4*). This is an improvement of the acidification yield when compared to systems using real wastewater as feedstock under similar optimised operational conditions (*Table 4.4*).

The VFA profile obtained here was similar to those observed in other studies. In the work of Domingos et al. (2014) using lactose as source of carbohydrate at the same pH and temperature, butyric acid was the major VFA produced (Domingos et al., 2014), although a lower TC consumption and VFAs production rate (0.14 and 0.07 g COD/(L·day), respectively) were observed compared to this study (*Figure 4.5b*). Moreover, the sludge used by Domingos et al. (2014) was an acclimated anaerobic acidogenic consortium from a continuous reactor for VFAs production. As in the present work no methane inhibitors or pretreatment of the granular sludge was applied, the observed high TC consumption rate (*Figure 4.5b*) can be attributed to acidogenesis and acetogenesis of the monosaccharides (e.g. glucose, arabinose and xylose; **Table 4.1**) and protein released from the BSG during the acid hydrolysis step. Similarly, suppression of the methanogenesis when operating under similar conditions using sugarcane molasses as feedstock resulted in butyric acid as the major end product, but with a lower acidification level (Silva et al., 2013).

Table 4.4 Average physical-chemical composition of different feedstocks for batch acidogenic fermentation
aimed at volatile fatty acid production. The percentage of the main acids (i.e. Acetic, Propionic and Butyric) as
well as the acidification level in each study is provided

Feedstock	sCOD (g/L)	TC (g COD/L)	рН	Temperature (°C)	VFAs (gCOD/L)	(Ac:Prop:But) (% of VFAs)	Acidification Level	Reference
Cassava alcohol wastewater	32.1	10.1	9.0 – 5.7	35	24.3	(37:20:40)	71.3	Xie et al. (2014)
Cheese whey	-	16.5	6.0	37	19.2	(15:17:55)	N.G.	Domingos et al., (2017)
Sugarcanne molasses	8.1	N.G.	6.0	37	3.0	(53:09:26)	42.1	Silva et al., (2013)
Glycerol	10.0	N.G.	10 (initial)	35	6.7	(08:91:00)	89.9	Shen et al. (2014)
Cheese whey	3.0	N.G.	10.0 → 7.0	35.0	1.1	(07:34:49)	54.7	Atasoy et al. (2020)
BSG hydrolysate	23.1	16.9	6.0	37.0	16.9	(40:01:55)	93.0	This study

*N.G.: data not given

Interestingly, hydrolysates produced by alkaline fermentation (pH 8 in this chapter) give similar acidogenic fermentation end products (*Figure 5*). Xie et al. (2014) operated a (high-strength wastewater fed) system by correcting the pH up to 9.0 every 12 hours obtaining butyric acid as major compound, which can be a result of pH fluctuating between 9.0 and 5.7. Nevertheless, when the system was run at constant pH 8.0 and above, butyric acid production was lower due to the increased pH used increased. Likewise, Shen et al. (2014) used glycerol as feedstock at an initial pH of 10.0, resulting mostly in the production of propionic acid with an acidification level of 89.9 %. This results was similar to that obtained in our work, where operating the system at pH 8.0 decreased the butyric acid and increased the propionic acid concentration (*Figure 4.4*). In contrast, Atasoy et al. (2020) reported the production of solely butyric acid as the major compound at all alkaline pH values investigated, with cheese whey as feedstock and using granular sludge from an UASB reactor treating municipal wastewater.

4.4.2 Total carbohydrates and monosaccharides consumption

Under uncontrolled pH, when the pH achieved the final value of 4.08 (± 0.05), only glucose was completely consumed (*Table 4.4*). On the other hand, more than 90 % of the TC was consumed in all experiments with a controlled pH (*Figure 4.5a*). The lowest pH achieved presumably led to process termination without arabinose and xylose consumption. Casey et al. (2010) similarly investigated the effect of pH and acetic acid on the co-fermentation of glucose and xylose, where low pH and increasing acetic acid concentrations led to an inhibition of the xylose consumption, while glucose consumption was unaffected. These results are in agreement with those obtained in the present study, where at the lowest pH only glucose was consumed. The accumulation of free acids likely led to product inhibition (e.g. acetate inhibition), which hindered the fermentation of xylose and arabinose.

The complete depletion of glucose under all pH values and the absent (at uncontrolled pH) (*Figure 4.7*) or delayed (at pH 4.5 or higher) consumption of xylose and arabinose (data not shown) can also be explained by a mechanism of carbon catabolite repression (Temudo et al., 2009) or the need for an extra ATP that microorganisms require for the uptake of xylose and arabinose (Zhou et al., 2018; Yao and Shimizu, 2013). Indeed, microorganisms start to consume pentoses (e.g. xylose and arabinose) only when the hexoses (e.g. glucose) are depleted (Zhang et al., 2018).

4.4.3 Carbon compounds conversion

The acidification level (i.e. the ratio between the total VFAs and sCOD) increased from 44.0 (\pm 7.0) % at uncontrolled pH to 93.0 (\pm 2.1) % at pH 6.0, whereas it decreased to 76.4 (\pm 0.6) % at pH 8.0 (*Figure 4.5b*). The highest VFA concentration was also obtained at pH 6.0. This increase in the acidification level from the lowest pH up to pH 6.0 can be explained by a minor concentration of the

90

undissociated form of VFAs, which has a strong inhibitory effect on the microorganisms at low pH (Green et al., 2007). The relation of the acids between the dissociated and undissociated form is given by the pH and the dissociation constant (Rumble et al., 2017), which is 4.76 and 4.82 for acetic and butyric acid, respectively (Khan et al., 2019). In this case, the VFA production would result in a selfinhibitory effect at higher free acid concentrations, which can be obtained by either increasing the total VFA concentration or decreasing the pH (Green, 2007). Therefore, to achieve the same inhibitory effect a lower VFA concentration would be required when decreasing the pH. This would explain the lower acidification level at pH 4.5 and 5.0 (Figure 4.5b), the incomplete sugar consumption and the lactic acid accumulation (Figure 4.6b & Figure 4.7). To overcome an excessive concentration of undissociated acids, the pH can be increased inducing a higher demand for alkali. Another strategy could be the use of continuous-flow systems, which withdraw the VFAs from the reactor mixed liquor (Yan et al., 2019; Aydin et al., 2018). For instance, bioreactor configurations with an immersed membrane system (Wainaina et al., 2019), or with a recirculation line coupled with a membrane contactor (Wainaina et al., 2019; Aydin et al., 2018) can allow maintenance of a low free acid concentration and can thus work at a lower feed pH or a higher TC load, meanwhile keeping the same acidification level.

The trend for the VFA production was constant with a final stable concentration until the end of the trial, indicating a low extent of the methanogenesis or even its inhibition at the last stage of the batch experiments when a decrease in VFAs was not observed after reaching the maximum VFA concentration (*Figure 4.6*). This could explain also the higher VFA production rate at pH 5.0 compared to the uncontrolled pH or at pH 6.0 (*Figure 4.5b*). The highest VFA production rate was at pH 5.0, where butyric acid peaked; meanwhile the acetic acid production rate increased along with the operating pH (*Figure 4.5b*). This shows that a lower pH is more favourable for the production of butyric acid.

Under all the acidic pH conditions investigated, butyric acid was the most produced VFA in terms of COD, followed by acetic acid. pH 8.0 resulted in a lower butyric acid production (*Figure 5*). The higher build-up of butyric acid and the absence of ethanol production has been related with the production of hydrogen gas (Zhang et al., 2013). Even though hydrogen was not measured in this study, a high hydrogen pressure likely led to the high butyric acid concentration (Zhang et al., 2013; Rodríguez et al., 2006).

Regarding the production of lactic acid, a similar trend was reported by Domingos et al. (2013) using lactose as the feedstock, which was converted into lactic acid after five days without VFA production. In the present work, at low pH (4.5 and 5.0), lactic acid was produced before the VFAs

91

(*Figure 4.6*), while lactic acid and VFA production occurred simultaneously at higher pH (6.0 and 8.0), although a lower lactic acid concentration was observed (*Figure 4.6b*).

The spectrum of fermentation products obtained can be explained by the use of the anaerobic granular sludge as inoculum. These granules harbour a highly diverse microbial community that results in a complex different combinations of pathways (Temudo et al., 2009; Owusu-Agyeman et al., 2018). These different pathways are influenced by the operational conditions (e.g. different pH and higher organic loading concentration), with pyruvate as the intermediate compound between the monosaccharides and the different spectrum of products obtained during the fermentation process, e.g. lactic acid, acetic acid, butyric acid and propionic acid (Atasoy et al., 2020; Casey et al., 2010; Zhou et al., 2018). The adaptation period of the microorganisms resulted in a delay for the VFA production, which was longer as lowering the pH (Figure 4.6). However, the TC concentration was partially consumed before the lactic acid or VFA production. This could have resulted in the accumulation of pyruvate, which can increase the intracellular NADH. Thus, in combination with a possible decreased intracellular pH, it would have promoted the activation of lactate dehydrogenase resulting in the conversion of pyruvate into lactic acid (Collet et al., 2009; Gheshlaghi et al., 2009). In addition, the longer lag phase for VFA production resulted in a higher accumulation of lactic acid at pH 4.5 and 5.0 (Figure 4.6a). Upon prolonged batch incubation, however, the microorganisms eventually consume the lactic acid for VFA production (Seeliger et al., 2002), as also observed in this study (Figure 4.6a & b)

4.4.4 Conversion of nitrogen and sulfur compounds

The consumption of the proteins during the acidogenic fermentation led to an increment of the ammonium concentration in the fermented liquor. Although the ammonium concentration has a low influence on the anaerobic digestion process, the strongest inhibition effect is due to the free ammonium (FA), which is a function of the total ammonium concentration and the pH through the Anthonisen equation (Anthonisen et al., 1976). FA has been reported to hinder the different steps of the anaerobic digestion when its concentration is increased, the methanogenesis being the most sensitive step. An FA concentration above 30 mg/L starts to inhibit methanogenesis, being almost completely inhibited at a concentration close to 60 mg/L (Shi et al., 2017). In contrast, the acidogenesis is negatively affected only when above 300 mg/L of FA (Zhao et al., 2018). During this work, when operating the system under acidic conditions the FA was much lower (between 0.0 and 0.7 mg/L at pH 4.5 and 6.0, and 65.2 mg/L at pH 8.0), which was below the inhibitory threshold for VFA production.

Sulfate removal was only observed to a limited extent (i.e. 5 %) and was similar at all pH values studied (*Figure 4.5a*). This probably occurred through a partial sulfate reduction towards sulfide production. However, as the hydrogen sulfide concentration was below the detection limit (data not

shown), this suggests that the sulfate reducing bacteria (SRB) were unlikely to be active or were outcompeted by the acidogens populations. Sulfate reducers can use simple organic electron donors such as ethanol and lactate (Papirio et al., 2013). However, during the degradation of carbohydrates to VFAs and lactic acid by acidogens, SRB were likely not favoured kinetically to use these simple compounds at the low pH in the 72 h of the batch experiment.

4.4.5 Hydrolysis efficiency

The thermal diluted acid hydrolysis at 121 °C for 20 minutes allowed the release of glucose, xylose and arabinose from BSG (*Table 4.1*Table 4.2). The use of 1.5 % and 2.0 % sulfuric acid (v/v) with 7 % of TS (w/w) resulted in the most efficient combination in terms of TC per g of TS, 0.44 g and 0.45 TC/g BSG, respectively (*Table 4.2*). As their efficiencies were similar, the lower sulfuric acid concentration was used to progress forward, with a severity factor corresponding to 1.09 under these conditions. In the case of 10 % TS, the presence of cellobiose indicated (*Table 4.1*) the occurrence of an incomplete hydrolysis, while conversely in the case of 7 % no cellobiose was detected.

The extraction efficiency was similar for both TC and sCOD and increased linearly when the sulfuric acid concentration was increased from 0.0 to 1.0 % (v/v) (*Figure 4.2*). After this point, the efficiency in terms of carbohydrates release and sulfuric acid used decreased, resulting in a lower improvement of the carbohydrates extracted from 1 to 3 %. The highest recovery of monosaccharides in the present work was achieved at 1.5 % of sulfuric acid and decreased using 3 % sulfuric acid at 7 % BSG (w/w). For the combinations with 10 % of BSG (w/w), glucose and xylose increased for all the sulfuric acid increments used, meanwhile the arabinose concentration decrease in the case of 3 % sulfuric acid (*Table 4.1*). Even if this combination resulted in the highest TC concentration, the loss of arabinose and the incomplete hydrolysis of the BSG, as indicated by the presence of cellobiose, decreased the efficiency of the TC extracted per amount of BSG used (*Table 4.2*). As a consequence, not only should the severity factor be used to calculate the efficiency of the hydrolysis, but also the TS concentration of the lignocellulosic biomass.

In the last years, studies focusing on the use of BSG as feedstock have proposed different strategies for the storage and initial treatment of the BSG. Few researchers used the BSG as received (Moreira et al., 2013; Kemppainen et al., 2016) and others dried or milled the BSG into different sizes (Patel et al., 2018; Ravindran et al., 2016; Rojas-Chamorro et al., 2020; Plaza et al., 2017). These physical changes are not considered as a pretreatment. However, Niemi et al. (2012) determined their effect during the hydrolysis. In the case of wet milled samples, a more homogeneous slurry was produced. Dried samples have been reported to form aggregations and more quickly sediment after submersion in water, being this related with the hornification of the BSG (Niemi et al., 2012).

93

The hydrolysis of BSG for recovery of carbohydrates has been studied using a single or two stepprocess (

Table 4.5). Carvalheiro et al. (2004) studied the hydrolysis of BSG using a single step, a thermal diluted acid hydrolysis, achieving a high xylose and arabinose recovery at a low severity factor and maximum glucose extraction under a higher severity factor (1.94). These observations are slightly different than those of the present work as a higher glucose extraction was obtained (

Table 4.5) using a lower severity factor (1.09). This could be explained by how Carvalheiro et al. (2004) stored the BSG: washing and drying remove some of the remaining sugars from the brewery process and hornify some of the carbohydrates. Therefore, the use of the raw BSG not only reduces the energy consumption, but also increases the TC recovery.

In studies using a two step hydrolytic pretreatment, some have mainly focused on glucose recovery, using the first step for the enrichment of cellulose in the solid fraction for the next extraction, without quantifying the sugars extracted (mainly xylose and arabinose). Ravindram et al. (2016) carried out different pretreatments followed by enzymatic hydrolysis, with the microwave assisted alkali hydrolysis being the most effective, followed by diluted sulfuric acid. However, the overall efficiency was much lower as only the TC from the second extraction were quantified (

Table **4.5**). Similarly, Patel et al. (2018) followed a method for cellulose concentration using an organosolvent pretreatment achieving a TC concentration of 71.6 g/L, mainly made up of glucose and xylose.

BSG	Characteri	sation									Carbohydrates	Efficiency		
тс	Proteins	Lignin	1 st extraction	ТС	Glu	Xyl	Gal	Ara	2 nd extraction	2 nd extraction TC		recovery	Reference	
(%)	(%)	(%)		g/L	g/L	g/L	g/L	g/L		g/L	g TC/ g BSG	g TC recovered/ 100 g TC raw BSG		
55.1	12.54	13.7	Organosolv pretreatment (10 % BSG w/w; 60 % ethanol; 1 % H ₂ SO ₄ wt dry BSG; 175 °C, 60 minutes)	N.G.	N.G.	N.G.	N.G.	N.G.	Enzymatic hydrolysis (10 % TS; 50 °C; 27 h)	71.6	NG	NG	Patel et al., (2018)	
43.5	23.3	19.4	Wet milling (95 % particle size < 40 μm) + Enzymatic extraction (3 % TS; 50 °C; 5 h)	6.7	2.8	2.1	0.28	1.12	None	-	0.15	0.35	Niemi et al., (2012)	
47.8	23.5	22.8	TDAH (8 % BSG w/w; 3 % H ₂ SO ₄ V/V; 130 °C; 15 min)	43.5	3.99	26.74	-	12.77	None	-	NG	NG	Carvalheiro et al., (2004)	
46.2	-	-	Microwave assisted alkali (1 % BSG w/v, 0.5 % NaOH (w/v); 400 W; 60 s)	N.G.	N.G.	N.G.	N.G.	N.G.	Enzymatic hydrolysis (10 % TS; 50 °C; 120 h)	18.6	0.23	0.50	Ravindram et al., (2018)	
43.5	22.6	4.1#	Steam explosion	-	5.69	8.23	-	3.84	Enzymatic hydrolysis (20 % TS)	N.G.	0.26	0.62	Kemppainen et al., (2016)	
45.6	23.1	12.5	TDAH (11.11 % BSG; 1 % H₂SO₄ wt TS; 130 °C; 26 min)	43.4	9.44	21.68	2.31	11.08	Enzymatic hydrolysis (5 % TS ; 50 °C; 24 h)	14.4 (72.04*)	0.48	0.94	Rojas-Chamorro et al., (2020)	
46.0	NG	NG	TDAH (15 % BSG w/w; 1 % H₂SO₄ wt dry BSG; 121 °C; 30 min)	47.0	20.0	18.4		8.6	Enzymatic hydrolysis (10 % TS; 50 °C; 48 h)	32.9	0.43	0.95	Plaza et al., (2017)	
ND	ND	ND	TDAH (7 % BSG w/w; 1.5 % H2SO4 v/v; 121 °C; 20 min)	32.0	12.8	12.05	-	11.8	None	-	0.44	ND	This study	

Table 4.5 Different studies for the hydrolysis of the brewery spent grain using different process in one or two combined steps for the recovery of monosaccharides

*ND: not determined

*NG: data not given

#: acid insoluble lignin

In the pretreatments where the overall carbohydrates extracted are considered, the obtained efficiency values are seen to be higher. Rojas-Chamorro et al. (2020) and Plaza et al. (2017) used a two step process, with the first step being a thermal diluted acid hydrolysis (similar to the first step applied in this study). Both studies used a lower sulfuric concentration and a higher concentration of BSG than those used in this study, resulting in a lower efficiency recovery and a higher TC concentration compared to this study. However, the final overall efficiency of the two step hydrolysis process recovered almost all the initial carbohydrates from the BSG (Rojas-Chamorro; Plaza et al., 2017).

Besides carbohydrates, the use of sulfuric acid as a pretreatment agent allowed the release of highvalue compounds such as lignin, proteins and soluble fibres from BSG (*Figure 4.2*). Rommi et al. (2018) compared different pretreatment combinations (i.e. raw BSG, steam explosion, alkaline and acid extraction) for lignin and protein recovery, based on a previous work focused on carbohydrates extraction from BSG (Rojas-Chamorro et al., 2020). They observed that the solubilisation of different organic compounds like lignin, proteins and polysaccharides strongly depended on the pretreatment combination. For example, steam explosion and hydrolysis increased lignin solubilisation, while protease was required to improve protein extraction (Rommi et al., 2018). Similarly, the diluted acid hydrolysis and neutralization used in the present study could be developed to recover some of the proteins extracted using the lowest solubility point of proteins (i.e. pH 3 - 5) (Liu et al., 2018; Guimaraes et al., 2012). This was, however, out of the scope of the present study, although the recovery of a broader range of end compounds beyond VFAs would be beneficial for the overall process.

4.5 Conclusions

The feasibility of VFA production from BSG was demonstrated in this study, reinforcing the change in the point of view of BSG from a solid waste into a valuable resource. First, a mixed liquor rich in carbohydrates with 0.44 g of carbohydrates per gram of total solids of BSG was obtained using a single step of a thermal diluted sulfuric acid hydrolysis (1.5 % sulfuric acid v/v). Then, the BSG hydrolysate was used for VFA production during carbohydrate fermentation by an anaerobic granular sludge. At the different pH values tested, the contribution of VFAs to the soluble COD at pH 6.0 was approximately 93.0 (± 2.1) %. Therefore BSG has a great potential as a raw feedstock for VFA production. Further research is required for the development of the process at a larger scale and in a continuous-flow mode.

4.6 References

- Albuquerque, M. G. E., V. Martino, E. Pollet, L. Averous, and M. A. M. Reis (2011) "Mixed culture polyhydroxyalkanoate (PHA) production from volatile fatty acid (VFA)-rich streams: Effect of substrate composition and feeding regime on PHA productivity, composition and properties." *Journal of Biotechnology* 151 (1): 66-76. https://doi.org/10.1016/j.jbiotec.2010.10.070
- Anthonisen A. C., Loehr R. C., Prakasam T. B. S., Srinath E. G. (1976) Inhibition of Nitrification by Ammonia and Nitrous Acid, Journal Water Pollution Control Federation. 48(5) 835-852.
- APHA (2012) "Standard Methods for Examination of Water and Wastewater 2012" American Public Health Association.
- Aydin S., Yesil H., Tugtas A. E. (2018) "Recovery of mixed volatile fatty acids from anaerobically fermented organic wastes by vapor permeation membrane contactors" J. Bioresource Technology. 250 548-555. https://doi.org/10.1016/j.biortech.2017.11.061
- Ayyachamy M., Cliffe F. E., Coyne J. M., Collier J., Tuohy M. G. (2013) "Lignin: untapped biopolymers in biomass conversion technologies." Biomass Conversion and Biorefinery 3 (3): 255-269. https://doi.org/10.1007/s13399-013-0084-4.
- Atasoy M., Eyice O., Schnürer A., Cetecioglu Z. (2019) "Volatile fatty acids production via mixed culture fermentation: Revealing the link between pH, inoculum type and bacterial composition." Bioresource Technology 292: 121889. https://doi.org/10.1016/j.biortech.2019.121889.
- Bengtsson, Simon, Jakob Hallquist, Alan Werker, and Thomas Welander. 2008. "Acidogenic fermentation of industrial wastewaters: Effects of chemostat retention time and pH on volatile fatty acids production." Biochemical Engineering Journal 40 (3): 492-499. https://doi.org/http://dx.doi.org/10.1016/j.bej.2008.02.004.
- Carmona-Cabello M., Garcia I. L., Leiva-Candia D., Dorado M. P. (2018) "Valorization of food waste based on its composition through the concept of biorefinery" J. Current Opinion in Green and Sustainable Chemistry. 14 67-79. https://doi.org/10.1016/j.cogsc.2018.06.011
- Carvalheiro F., Duarte L. C., Medeiros R., Gírio F. M. (2004) "Optimization of Brewery's spent grain dilute-acid hydrolysis for the production of pentose-rich culture media." Applied Biochemistry and Biotechnology 115 (1-3): 1059-1072. https://doi.org/10.1385/ABAB:115:1-3:1059.
- Casey E., Sedlak M., Ho N. W. Y., Mosier N. S. (2010) "Effect of acetic acid and pH on the cofermentation of glucose and xylose to ethanol by a genetically engineered strain of Saccharomyces cerevisiae" FEMS Yeast Research. 10 385-393. DOI: 10.1111/j.1567-1364.2010.00623.x
- Cisneros-Pérez C., Etchebehere C., Celis L. B., Carrillo-Reyes J., Alatriste-Mondragón F., Razo-Flores E. (2017) "Effect of inoculum pretreatment on the microbial community structure and its performance during dark fermentation using anaerobic fluidized-bed reactors." International Journal of Hydrogen Energy 42 (15): 9589-9599. https://doi.org/10.1016/j.ijhydene.2017.03.157.
- Collet C., Girbal L., Péringer P., Schwitzguébel J.-P., Soucaille P. (2006) "Metabolism of lactose by Clostridium thermolacticum growing in continuous culture" Archives of Microbiology. 185 331-339. DOI 10.1007/s00203-006-0098-4

- Connolly A., Piggott C., FitzGerald R. (2013) "Characterisation of protein-rich isolates and antioxidative phenolic extracts from pale and black brewers' spent grain". Vol. 48.
- Dishman M. (2017) U.S. Department of Energy (DOE) Bioenergy Technologies Office (BETO) Project Peer Review POET-DSM Project Liberty.
- Djioleu A., Carrier D. J. (2016) "Effects of Dilute Acid Pretreatment Parameters on Sugar Production during Biochemical Conversion of Switchgrass Using a Full Factorial Design" J. ACS Sustainable Chemistry & Engineering. 4 (8) 4124-4130. https://doi.org/10.1021/acssuschemeng.6b00441
- DuBois M., Gilles K. A., Hamilton J. K., Rebers P. A., Smith F. (1956) "Colorimetric Method for Determination of Sugars and Related Substances" J. Analytical Chemistry. 28 (1956) 350-356.
- Domingos, Joana M. B., Gonzalo A. Martinez, Alberto Scoma, Serena Fraraccio, Frederiek-Maarten Kerckhof, Nico Boon, Maria A. M. Reis, Fabio Fava, and Lorenzo Bertin. 2017. "Effect of Operational Parameters in the Continuous Anaerobic Fermentation of Cheese Whey on Titers, Yields, Productivities, and Microbial Community Structures." ACS Sustainable Chemistry & Engineering 5 (2): 1400-1407. https://doi.org/10.1021/acssuschemeng.6b01901.
- Dreschke G., Papirio S., Sisinni D. M. G., Lens P. N. L., Esposito G. (2019) "Effect of feed glucose and acetic acid on continuous biohydrogen production by Thermotoga neapolitana" J. Bioresource Technology. 273 (2019) 416-424. DOI: 10.1016/j.biortech.2018.11.040
- Fava F., Totaro G., Diels D., Reis M., Duarte J., Carioca O. B., Poggi-Varaldo H. M., Ferreira B. S. (2015)
 "Biowaste biorefinery in Europe: opportunities and research & development needs." *New Biotechnology* 32 (1): 100-108. http://dx.doi.org/10.1016/j.nbt.2013.11.003.
- Fockink D. H., Sánchez J. H., Ramos L. P. (2018) "Comprehensive analysis of sugarcane bagasse steam explosion using autocatalysis and dilute acid hydrolysis (H₃PO₄ and H₂SO₄) at equivalent combined severity factors" J. Industrial Crops and Products. 123 563-572. https://doi.org/10.1016/j.indcrop.2018.07.017
- Garcia-Aguirre J., Aymerich E., González-Mtnez. J de G, Esteban-Gutiérrez M. (2017) "Selective VFA production potential from organic waste streams: Assessing temperature and pH influence." *Bioresource Technology* 244 (Part 1): 1081-1088. https://doi.org/https://doi.org/10.1016/j.biortech.2017.07.187.
- Gheshlaghi R., Scharer J. M., Moo-Young M., Chou C. P. (2009) "Metabolic pathways of clostridia for
producing butanol" Biotechnology Advances 27 764-781.
doi:10.1016/j.biotechadv.2009.06.002
- Görke B., Stülke J. (2008) "Carbon catabolite repression in bacteria: many ways to make the most out of nutrients" J. Nature Reviews Microbiology. 6 613-624. doi:10.1038/nrmicro1932
- Green D. W., Perry R. H. (2007) Perry's Chemical Engineers' Handbook, McGraw-Hill Education, 8th ed.
- Guimarães R. C. A., Favaro S. P., Viana A. C. A., Neto J. A. B., Neves V. A., Honer M. R. (2012) "Study of the proteins in the defatted flour and protein concentrate of baru nuts (*Dipteryx alata* Vog)" J. Ciênc. Tecnol. Aliment. 32 (3) 464-470. DOI: 10.1590/S0101-20612012005000065
- Infantes, D., González del Campo A., Villaseñor J., Fernández F. J. (2011) "Influence of pH, temperature and volatile fatty acids on hydrogen production by acidogenic fermentation." International

Journal of Hydrogen Energy 36 (24): 15595-15601. https://doi.org/10.1016/j.ijhydene.2011.09.061.

- Itoh Y., Tada K., Kanno T., Horiuchi J.-I. (2012) "Selective production of lactic acid in continuous anaerobic acidogenesis by extremely low pH operation." Journal of Bioscience and Bioengineering 114 (5): 537-539. https://doi.org/10.1016/j.jbiosc.2012.05.020.
- Kemppainen K., Rommi K., Holopainen U., Kruus K. (2016) "Steam explosion of Brewer's spent grain improves enzymatic digestibility of carbohydrates and affects solubility and stability of proteins" Applied Biochemistry and Biotechnology. 180(1) 94-108. DOI 10.1007/s12010-016-2085-9
- Khan M. A., Ngo H. H., Guo W., Liu Y., Nghiem L. D., Chang S. W., Nguyen D. D., Zhangd S., Luo G., Jia H. (2019) "Optimization of hydraulic retention time and organic loading rate for volatile fatty acid production from low strength wastewater in an anaerobic membrane bioreactor" J. Bioresource Technology. 271 100-108. https://doi.org/10.1016/j.biortech.2018.09.075
- Mancini G., Papirio S., Lens P. N. L., Esposito G. (2016) "Effect of N -methylmorpholine- N -oxide Pretreatment on Biogas Production from Rice Straw, Cocoa Shell, and Hazelnut Skin" J. Environmental Engineering Science. 33 (11) 843-850. http://doi.org/10.1089/ees.2016.0138
- Monlau, Florian, Abdellatif Barakat, Eric Trably, Claire Dumas, J. P. Steyer, and Hélène Carrere. 2011. "Lignocellulosic Materials Into Biohydrogen and Biomethane: Impact of Structural Features and Pretreatment." Critical Reviews in Environmental Science and Technology 2011. https://doi.org/10.1080/10643389.2011.604258.
- Niemi P., Tamminen T., Smeds A., Viljanen K., Ohra-aho T., Holopainen-Mantila U., Faulds C. B., Poutanen K., Buchert J. (2012) "Characterization of Lipids and Lignans in Brewer's Spent Grain and Its Enzymatically Extracted Fraction." Journal of Agricultural and Food Chemistry 60 (39): 9910-9917. https://doi.org/10.1021/jf302684x.
- Liu N., Chen Q., Li G., Zhu Z., Yi J., Li C., Chen X., Wang Y. (2018) "Properties and Stability of Perilla Seed Protein-Stabilized Oil-in-Water Emulsions: Influence of Protein Concentration, pH, NaCl Concentration and Thermal Treatment" J. Molecules. 23 (7) 1533. doi: 10.3390/molecules23071533
- Lowry O. H., Rosebrough N. J., Farr A. L., Randall R. J. (1951) "Protein measurement with the Folin phenol reagent" J Biol Chem. 193 (1) 265-275.
- O'Dell J. W. (1996) Method 375.2 Determination of Sulfate by Automated Colorimetry, in: W. Andrew (Eds.), Methods for the Determination of Metals in Environmental Samples, Westwood, NJ, pp. 496-508.
- Outeiriño D., Costa-Trigoa I., Paz A., Deive F. J., Rodríguez A., Domínguez J. M. (2019) "Biorefining brewery spent grain polysaccharides through biotuning of ionic liquids" J. Carbohydrate Polymers. 203 (2019) 265-274. https://doi.org/10.1016/j.carbpol.2018.09.042
- Owusu-Agyeman I., Eyice Ö., Cetecioglu Z., Plaza E. (2019) "The study of structure of anaerobic granules and methane producing pathways of pilot-scale UASB reactors treating municipal wastewater under sub-mesophilic conditions" Bioresource Technology. 290 121733. https://doi.org/10.1016/j.biortech.2019.121733
- Papirio S., Esposito G., Pirozzi F. (2013) Biological inverse fluidized-bed reactors for the treatment of low pH- and sulphate-containing wastewaters under different COD/SO42- conditions, J. Environmental Technology, 34 1141-1149. https://doi.org/10.1080/09593330.2012.737864

- Patel A., Mikes F., Bühler S., Matsakas L. (2018) "Valorization of Brewers' Spent Grain for the Production of Lipids by Oleaginous Yeast." *Molecules (Basel, Switzerland)* 23 (12): 3052. https://doi.org/10.3390/molecules23123052. https://pubmed.ncbi.nlm.nih.gov/30469531.
- Plaza P. E., Gallego-Morales L. J., Peñuela-Vásquez M., Lucas S., García-Cubero M. T., Coca M. (2017)
 "Biobutanol production from brewer's spent grain hydrolysates by Clostridium beijerinckii." Bioresource Technology 244: 166-174. https://doi.org/10.1016/j.biortech.2017.07.139.
- Ravindran R., Jaiswal A. K.. (2016) "A comprehensive review on pre-treatment strategy for lignocellulosic food industry waste: Challenges and opportunities." Bioresource Technology 199: 92-102. https://doi.org/10.1016/j.biortech.2015.07.106.
- Ravindran, Rajeev, Swarna Jaiswal, Nissreen Abu-Ghannam, and Amit K. Jaiswal. 2018. "A comparative analysis of pretreatment strategies on the properties and hydrolysis of brewers' spent grain." Bioresource Technology 248: 272-279. https://doi.org/10.1016/j.biortech.2017.06.039.
- Rocha-Martín J., Martinez-Bernal C., Pérez-Cobas Y., Reyes-Sosa F. M., García B. D. (2017) Additives enhancing enzymatic hydrolysis of lignocellulosic biomass, J. Bioresource Technology. 244 (1) 48-56. https://doi.org/10.1016/j.biortech.2017.06.132
- Rodríguez J., Kleerebezem R., Lema J. M., van Loosdrecht M. C. M. (2006) "Modeling product formation in anaerobic mixed culture fermentations" J. Biotechnology and Bioengineering. 93 592-606. https://doi.org/10.1002/bit.20765
- Rojas-Chamorro J. A., Romero I., López-Linares J. C., Castro E. (2020) "Brewer's spent grain as a source of renewable fuel through optimized dilute acid pretreatment." Renewable Energy 148: 81-90. https://doi.org/10.1016/j.renene.2019.12.030.
- Rommi K., Niemi P., Kemppainen P., Kruus K. (2018) "Impact of thermochemical pre-treatment and carbohydrate and protein hydrolyzing enzyme treatment on fractionation of protein and lignin from brewer's spent grain." Journal of Cereal Science 79: 168-173. https://doi.org/10.1016/j.jcs.2017.10.005.
- Rosa F., Rúbia P., Santos S. C., Silva E. L. (2014) "Different ratios of carbon sources in the fermentation of cheese whey and glucose as substrates for hydrogen and ethanol production in continuous reactors." *International Journal of Hydrogen Energy* 39 (3): 1288-1296. https://doi.org/10.1016/j.ijhydene.2013.11.011.
- Rumble J. (2017) CRC Handbook of Chemistry and Physics, CRC Press LLC, 98th Ed.
- Saboe P. O., Manker L. P., Michener W. E., Peterson D. J., Brandner D. G., Deutch S. P., Kumar M., Cywar R. M., Beckham G. T., Karp E. M. (2018) "In situ recovery of bio-based carboxylic acids." *Green Chemistry* 20 (8): 1791-1804. https://doi.org/10.1039/C7GC03747C.
- Seeliger S., Janssen P. H., Schink B. (2002) "Energetics and kinetics of lactate fermentation to acetate and propionate via methylmalonyl-CoA or acrylyl-CoA" FEMS Microbiology Letters. 211(1) 65-70. https://doi.org/10.1016/S0378-1097(02)00651-1
- Shen L., Hu H., Ji H., Cai J., He N., Li Q., Wang Y. (2014) "Production of poly(hydroxybutyrate– hydroxyvalerate) from waste organics by the two-stage process: Focus on the intermediate volatile fatty acids." Bioresource Technology 166: 194-200. https://doi.org/10.1016/j.biortech.2014.05.038.

- Shi X., Lin J., Zuo J., Li P., Li X, Guo X. (2017) "Effects of free ammonia on volatile fatty acid accumulation and process performance in the anaerobic digestion of two typical bio-wastes." Journal of Environmental Sciences 55 (Supplement C): 49-57. https://doi.org/10.1016/j.jes.2016.07.006.
- Silva F., Serafim L., Nadais M., Arroja L., Capela I. (2013) "Acidogenic Fermentation Towards Valorisation of Organic Waste Streams into Volatile Fatty Acids" Chemical and Biochemical Engineering Quarterly 27 467-476.
- Silveira M. H. L., Chandel A. K., Vanelli B. A., Sacilotto K. S., Cardoso E. B. (2018) "Production of hemicellulosic sugars from sugarcane bagasse via steam explosion employing industrially feasible conditions: Pilot scale study" J. Bioresource Technology Reports. 3 138-146. https://doi.org/10.1016/j.biteb.2018.07.011
- Strazzera G., Battista F., Garcia N. H., Frison N., Bolzonella D. (2018) "Volatile fatty acids production from food wastes for biorefinery platforms: A review." *Journal of Environmental Management* 226: 278-288. https://doi.org/https://doi.org/10.1016/j.jenvman.2018.08.039.
- Temudo M. F., Mato T., Kleerebezem R., van Loosdrecht M. C. M. (2009) "Xylose anaerobic conversion by open-mixed cultures" J. Applied Microbiology and Biotechnology. 82 (2009) 231-239. doi: 10.1007/s00253-008-1749-y
- Veeravalli S. S., Chaganti S. R., Lalman J. A., Heath D. D. (2014) "Optimizing hydrogen production from a switchgrass steam exploded liquor using a mixed anaerobic culture in an upflow anaerobic sludge blanket reactor" International Journal of Hydrogen Energy. 39 (7) 3160-3175. http://dx.doi.org/10.1016/j.ijhydene.2013.12.057
- Wainaina S., Parchami M., Mahboubi A., Horváth I. S., Taherzadeh M. J. (2019) "Food waste-derived volatile fatty acids platform using an immersed membrane bioreactor." *Bioresource Technology* 274: 329-334. https://doi.org/10.1016/j.biortech.2018.11.104.
- Xie L., Liu H., Chen Y.-g., Zhou Q. (2014) "pH-adjustment strategy for volatile fatty acid production from high-strength wastewater for biological nutrient removal" Water Science and Technology. 69(10) 2043-2051. doi: 10.2166/wst.2014.120
- Yan M., et al. (2019) Acclimatization contributes to stable anaerobic digestion of organic fraction of municipal solid waste under extreme ammonia levels: Focusing on microbial community dynamics. Bioresource Technology. 286, 121376. https://doi.org/10.1016/j.biortech.2019.121376
- Yao R., Shimizu K. (2013) "Recent progress in metabolic engineering for the production of biofuels and biochemicals from renewable sources with particular emphasis on catabolite regulation and its modulation" J. Process Biochemistry. 48 (9) 1409-1417. https://doi.org/10.1016/j.procbio.2013.02.032
- Zhang F., Zhang Y., Chen M., van Loosdrecht M. C. M., Zeng R. J. (2013) "A modified metabolic model for mixed culture fermentation with energy conserving electron bifurcation reaction and metabolite transport energy" J. Biotechnology and Bioengineering. 110 1884-1894. DOI: 10.1002/bit.24855
- Zhang W., Dai K., Xia X., Wang H., Chen Y., Lu Y., Zhang F., Zeng R. J. (2018) "Free acetic acid as the key factor for the inhibition of hydrogenotrophic methanogenesis in mesophilic mixed culture fermentation" J. Bioresource Technology. 264 17-23. DOI: 10.1016/j.biortech.2018.05.049

- Zhao J., Liu Y., Wang Y., Lian Y., Wang Q., Yang Q., Wang D., Xie G.-J., Zeng G., Sun Y., Li X., Ni B.-J. (2018) "Clarifying the Role of Free Ammonia in the Production of Short-Chain Fatty Acids from Waste Activated Sludge Anaerobic Fermentation" ACS Sustainable Chemistry & Engineering. 6 (11) 14104-14113. DOI: 10.1021/acssuschemeng.8b02670
- Zhou M., Yan B., Wong J. W. C., Zhang Y. (2018) "Enhanced volatile fatty acids production from anaerobic fermentation of food waste: A mini-review focusing on acidogenic metabolic pathways" J. Bioresource Technology. 248 68-78. https://doi.org/10.1016/j.biortech.2017.06.121
- Zoulikha M.-R., Thierry M., Qiuyu Z J.-M., Nouviaire A., Sid-Ahmed R. (2015) "Combined steamexplosion toward vacuum and dilute-acid spraying of wheat straw. Impact of severity factor on enzymatic hydrolysis." Renewable Energy 78: 516-526. https://doi.org/10.1016/j.renene.2015.01.038.

Chapter 5

Continuous volatile fatty acid production from brewery spent grain hydrolysate using an expanded granular sludge bed reactor

A modified version of this chapter has been submitted as:

Castilla-Archilla J., Heiberger J., Mills S., Hilbig J., Collins G., Lens P. N. L. (2021) "Continuous Volatile Fatty Acid Production From Acid Brewery Spent Grain Leachate in Expanded Granular Sludge Bed Reactors" Frontiers in Sustainable Food Systems 5, 129. DOI=10.3389/fsufs.2021.664944

Keywords: brewery spent grain; pretreatment; thermal diluted acid hydrolysis; carbohydrates; acidogenic fermentation; volatile fatty acids

Abstract

The production of volatile fatty acids (VFAs) in expanded granular sludge bed (EGSB) reactors using liquid hydrolysate from thermal diluted acid hydrolysis of brewery spent grain was evaluated. Partial inhibition of the anaerobic digestion process to induce VFA accumulation was achieved by applying a high organic loading rate (from 15.3 to 46.0 gCOD/(L·day)), and using a feed with an inlet concentration of 15 g/L total carbohydrates. Two EGSB reactors were operated under identical conditions, both inoculated with the same granular sludge. However, granular sludge in one reactor (R1) was subsequently disaggregated to flocculent sludge by a pH shock, whereas granules remained intact in the other reactor (R2). The hydraulic retention time (HRT) of both reactors was decreased from 36 to 24, 18 and 12 hours. The main fermented compounds were acetic acid, butyric acid, propionic acid and ethanol. Despite fluctuations between these products, their total concentration was quite stable throughout the trial at about 134.2 (± 27.8) and 141.1 (± 21.7) mmol/L respectively, for R1 and R2. Methane was detected at the beginning of the trial, and following some periods of instability in the granular sludge reactor (R2). The hydrogen yield increased as the HRT decreased. The highest VFA production was achieved in the granular sludge reactor at a 24 h HRT, corresponding to 120.4 (± 15.0) mmol/L of VFAs. This corresponded to an acidification level of 83.4 (± 5.9) g COD of VFA per total gram of soluble COD.

5.1 Introduction

Anaerobic digestion (AD) has been widely implemented as a method of biological wastewater treatment, where mixed microbial communities convert organic matter to biogas (CH₄ and CO₂). However, in the last decade there has been a growing interest in hydrogen production via dark fermentation, facilitated by the partial inhibition of the anaerobic digestion process (Cisneros-Pérez *et al.*, 2015). During the acidogenesis step of AD, hydrogen and carbon dioxide are produced, along with fermented products such as ethanol and/or volatile fatty acids (VFAs). To improve this process and maximize hydrogen production, different pre-treatments have been applied to the inoculum and during reactor start up, including heat treatment (Cisneros-Pérez *et al.*, 2015; Carrillo-Reyes *et al.*, 2016), wash-out of methanogens (Cisneros-Pérez *et al.*, 2015; Carrillo-Reyes *et al.*, 2016), wash-out of different operational conditions, such as pH or organic loading rate, can however, change the metabolic activity microorganisms. Different reactor configurations can also influence microbial community composition by altering selection pressures such as solids retention time, resulting in the wash out of microorganisms with slower growth rates, e.g. methanogens, or enriching faster growing organisms, e.g. acidogens (Lim *et al.*, 2020; Etchebehere *et al.*, 2016).

The VFAs produced simultaneously with hydrogen are valuable chemicals, currently obtained from fossil sources. Their potential production from renewable sources has not yet been developed at industrial scale. The development of hydrogen production from wastewater into more sophisticated technologies for VFA production could allow VFAs to be produced independently of fossil sources. One of the difficulties is, however, to maintain consistent VFA production, despite shifts in microbial community composition and changes in bioreactor performance (Cisneros-Pérez *et al.*, 2017). The accumulation of organic compounds in the reactor mixed liquor can alter reaction kinetics in the microbial metabolic pathways of acidogenesis and methanogenesis (Ciranna *et al.*, 2014; Pugazhendhi *et al.*, 2017; Menezes and Silva, 2019). This can activate or inactivate the consumption of hydrogen by different microorganisms, resulting in unstable hydrogen and VFA production (Saady, 2013; Regueira *et al.*, 2020). In addition, the main problem with VFA production from wastewater is that their recovery is difficult and energy demanding, due to low VFA concentrations and complex composition of the bioreactor mixed liquor (Reyhanitash *et al.*, 2016). The current state of-the-art for VFA extraction enables their recovery when operating at a low pH and with VFA concentrations exceeding 200 – 500 mmol/L (Rocha *et al.*, 2017; Reyhanitash *et al.*, 2017; Saboe *et al.*, 2018).

Many wastewaters are low or medium strength and do not contain high carbohydrate concentrations, making them unsuitable for VFA production and recovery. Various reactor configurations have been optimised to use such wastewaters for hydrogen production via dark fermentation, without focusing on VFA production. Some of these reactors were operated with a

105

synthetic feed concentration of about 60 g/L total carbohydrates (TC) (Zhang *et al.*, 2008), and an organic loading rate (OLR) of up to 240 g TC/(L·d) (Zhang *et al.*, 2008; Menezes and Silva, 2019). However, waste streams with a higher TC concentration would be required for targeted VFA production, such as those originating from the agroindustry (Menezes and Silva, 2019). The agriculture and food industry could, therefore, become a supplier of renewable feedstock for VFA production. These wastes, mostly lignocellulosic materials, are abundant and can be obtained with low or no cost. For example, the brewing industry produces high amounts of brewery spent grain (BSG), which is a solid, lignocellulosic material rich in carbohydrates, proteins and lignin. After the hydrolysis process, most of these carbohydrates are present in the liquid hydrolysate at concentrations up to 30 - 70 g/L of TC (Patel et al., 2017). This liquid hydrolysate can be used for VFA production without any further treatment for production of VFAs as demonstrated in *Chapter 4*.

The aim of this work was to study the production of VFAs from BSG by two different sludge types, in a continuous high rate EGSB system with a long HRT (36 to 12 hours) and high OLR (10 to 30 gTC/(L·d)). Two continuous EGSB systems were used, seeded with the same granular sludge. After seeding, the granular sludge in one reactor was disaggregated by providing a pH shock. This allowed to study the effects of the microbial selection pressure on the resulting flocculent sludge. The effectiveness of this strategy for targeted VFA production was determined by monitoring subsequent sludge development and measuring VFA and hydrogen production in both reactors.

5.2 Material and methods

5.2.1 Feedstock and inoculum

The hydrolysate from the hydrolysis pretreatment developed in *Chapter 4 w*as used for the continuous VFA production. Briefly, BSG at 7% TS concentration (w/w) was mixed with 1.5 % sulfuric acid (v/v), and autoclaved at 121 °C for 20 minutes. Two modifications were carried out in the present work, the remaining solids following hydrolysis were washed with water of an equal volume to the extracted hydrolysate, obtaining a second liquid with a TC concentration between 6 – 8 g/L. This extract was used to dilute the initial feedstock to a final concentration of 15 g/L of TC. In addition, the hydrolysate was centrifuged after neutralization to remove any further precipitate in the liquid fraction. Thus, the feed was mainly composed of 6.5 (\pm 1.4) g/L glucose, 5.5 (\pm 0.4) g/L xylose, 2.8 (\pm 1.0) g/L arabinose and 2.1 (\pm 0.4) g/L proteins.

Once the feedstock was prepared, it was placed in a fridge at 4 °C and connected to the inlet of the EGSB reactors. The feedstock was continuously stirred using a magnetic stirrer (Stuart UC151, UK) operated at 1,500 rpm. Initially, the feedstock was replaced every 3 days. However, after the first two weeks of operation it was noticed that the TC concentration in the feedstock tank decreased slightly

three days after preparation. The feedstock was, subsequently, replaced every two days and the bottle was closed with a rubber bung to prevent contamination. A needle with a 25 μ m filter connected was used to pierce the bung and prevent vacuum formation. Consequently, no further degradation of TC was noticed in the feedstock.

Same granular sludge was used to seed the reactors. To study the selection pressure of the EGSB reactor system, two biomass types derived from the same UASB inoculum were studied: flocculent and granular sludge. Both reactors were seeded with the same granular sludge at an inoculum concentration of 1.5 g/L of volatile suspend solids (VSS). Granules in Reactor 1 (R1) were disaggregated into flocculent sludge by increasing the pH from 5.9 to 10.5 for 12 hours on day 21, similarly to Kan (2013). In contrast, Reactor 2 (R2) was operated with intact granular sludge during the whole experiment.

Due to a decrease in VFA production in R1 beginning on day 67, the reactor was reseeded with fresh granular sludge on day 81, with the same initial concentration of 1.5 g/L VSS. The fresh granular sludge was added without emptying the reactor, the flocculent sludge already present inside of the reactor was not removed.

5.2.2 Experimental set-up

Two EGSB reactors (total volume of 2.27 L; working volume of 1.78 L) were operated under the same conditions (*Figure 5.1*). The feedstock was introduced from the bottom of the reactor using a pump (Master-Flex L/S, USA). To keep the up-flow velocity constant, a recycling line was connected from the fourth port of the reactor (70 cm from the bottom) to a connection tube placed below the feed inlet. Reactor mixed liquor was circulated through this line by a pump (Watson-Marlow 323, UK). A gas meter (Omega, USA) was connected to the outlet gas line, with a port for sampling purposes. A Cole-Parmer 300 (China) pH controller, connected to a Master-Flex pump, was used for sodium hydroxide addition (3 M), maintaining the pH at 5.90 (± 0.05). The pH probe of the pH controller was placed in the third port from the bottom of the reactor (40 cm) and the NaOH addition tube was connected to the recycling line of the reactor. The temperature was kept at 37 °C using a water bath (Grant Instruments TC120).



Figure 5.1 Schematic set up of the EGSB reactor used for the continuous production of VFAs. Pump for the influent supply, recycling pump and base addition (3M NaOH) connected to a pH controller are shown

5.2.3 Experimental design

Initially, during the first 12 hours of operation, the reactors were operated in batch mode, with a feedstock concentration of 10 g/L TC to allow the acclimatization of the anaerobic granular sludge. After the acclimatization period both reactors were fed with a constant TC concentration of approximately 15 g/L. R1 was started in advance, and 15 days later, Reactor 2 (R2) was started-up in a similar way. R1 was operated for 108 days and R2 was operated for 93 days. The trial was divided into four different phases in which the HRT was gradually decreased from 36 hours (Phase IV), as indicated in Table 1. All the changes in the reactor, HRT and up-flow velocity were carried out simultaneously in both reactors.

The initial up-flow velocity was 6 m/h, which was decreased to 3.0 m/h, at day 29 for R1 and day 14 for R2. A further reduction from 3.0 to 2.0 m/h in R1 (flocculent sludge) and from 3.0 to 2.5 m/h in R2 (granular sludge) was done on days 81 and 66, respectively.

Table 5.1 Hydraulic retention time and organic loading rate for each of the different phases during the continuous acidogenic fermentation of brewery spent grain hydrolysate using in an expanded granular sludge bed reactor

		Phase I	Phase II	Phase III	Phase IV
HRT	Г (h)	36	24	18	12
OLR (g1	「C/(L·d))	10	15	20	30
Time	R1	$0 \rightarrow 37$	$37 \rightarrow 67$	$67 \rightarrow 94$	$94 \rightarrow 108$
(days)	R2	$0 \rightarrow 22$	$22 \rightarrow 52$	$52 \rightarrow 79$	$79 \rightarrow 93$

*Note that R1 was started-up 15 days prior to R2. Changes in HRT and OLR were done on the same operational day

5.2.4 Analytical methods

Total suspend solids (TSS) and VSS were sampled twice weekly from the effluent and the recycling ports at 70 and 75 cm height, respectively, in order to determine any possible wash-out and biomass flotation within the reactor due to the settleability of the biomass. Samples for chemical analysis were centrifuged at 9,000 rpm for 5 minutes and the supernatant was filtered using a syringe filter with 0.22 µm pore size. The samples were then frozen for further analysis. TC analysis was performed by the Dubois method (Dubois et al., 1956). Ammonia, sulfate and total proteins were analysed by using a discrete nutrient analyser (Gallery Plus, ThermoFisher) as described by Castilla-Archilla *et al.* (2020). The sCOD was analysed using a continuous Seal Analytical AA3 flow autoanalyser (Norderstedt, Germany) (SEAL, 2019).

VFAs, glucose, xylose, arabinose, lactose and ethanol were measured using a 1260 Infinity II liquid chromatograph (Agilent, Germany) equipped with a Hi Plex H 7.7 x 300 mm and 8 μ m (p/n PL1170-6830) column (Agilent, UK) kept at 60 °C and an RI detector at 55 °C. The mobile phase was sulfuric acid with a concentration of 5 mM and a flow rate of 0.7 mL/min (Castilla-Archilla et al., 2020). The composition of the gas in terms of H₂, CH₄ and CO₂ was analysed by gas chromatography (7890B, Agilant, USA) as reported by Oliva *et al.* (2020).

5.2.5 Calculations

All gas was converted to standard conditions (1 atm, 273 K). The gas production rate was calculated using the fraction for each of the gases multiplied by the total gas accumulated between two points and divided by the time elapsed (*Equation 1*) (Mota *et al.*, 2018):

$$r_{G} = \frac{(V_{T,n} \cdot Y_{G,n}) - (V_{T,n-1} \cdot Y_{G,n-1})}{(t_{n} - t_{n-1})} \qquad ({}^{mL}/day) \qquad Equation 1$$

where: $V_{T,n}$, $V_{T,n-1}$ are total gas produced at time n and n-1; $Y_{G,n}$, $Y_{G,n-1}$: fraction of the gas at time n and n-1; t_n , t_{n-1} : time at point n and previous point.

The volumetric gas production rate (VGPR) was calculated by dividing the gas rate production by the working volume of the reactor (*Equation 2*):

$$VGPR = \frac{r_G}{v_W} \qquad \qquad ({^{mL}}/{day \cdot L}) \qquad \qquad Equation 2$$

where $V_{\boldsymbol{w}}$ is the working volume of the reactor

The hydrogen yield (HY) was considered as the amount of hydrogen (mmol) per amount of the TC consumed (mmol) (Fernández, Villaseñor and Infantes, 2011). The molecular weight of the TC was calculated based on the contribution of each of the monosaccharides (*Equation 3*):

$$HY = \frac{\frac{r_{H_2}}{22.4}}{\frac{([TC]_i - [TC]_e) \cdot Q_f}{P.m.(TC)}} \qquad (\frac{mmol \ H_2}{mmol \ TC_{consumed}}) \ Equation 3$$

Where, r_{H2} is the hydrogen production rate (mL/day), [TC] the total carbohydrates concentration, Q_f the feeding flow, P.m. the molecular weight for the total carbohydrates, i: the inlet of the reactor and e: effluent of the reactor. This was used to compare how much of the TC was used for hydrogen production in each of the phases.

5.3 Results

5.3.1 Total carbohydrates fermentation during the acidogenic fermentation

The main fermented compounds obtained in both reactors were acetic acid, butyric acid, propionic acid and ethanol, their concentrations fluctuated in both reactors during the different phases (**Figure 5.2***a*). In both EGSB systems, propionic acid concentrations decreased when a lower HRT was applied. In R1, acetic acid was the major compound, followed by ethanol, propionic and butyric acid during Phase 1. After disaggregation of the granular sludge by increasing the pH, the butyric acid concentration increased and the ethanol and propionic acid concentrations decreased. A similar trend was observed during Phase II until day 50, when butyric acid was the major fermentation product and propionic acid concentrations were much lower than in the previous phase (**Figure 5.2***a*).



Lactic acid was produced in R1 during Phase III, and during periods of instability in R2. In R1, when the OLR was increased on day 66 up to 20 g TC/(L·d) (Phase III), lactic acid became the major product replacing all the other compounds (**Figure 5.2***a*).). On day 80, R1 was reseeded with the same fresh granular sludge and lactic acid was still the major compound until day 86. Subsequently, lactic acid concentrations decreased and were depleted by day 92. At the same time, the concentration of acetic acid, butyric acid and ethanol started to increase 5 days after the reseeding of the reactor. Ethanol achieved its maximum concentration during this phase. By phase IV, butyric and acetic acid became the major fermentation products (**Figure 5.2***a*).

In R2, the trend was quite similar during each phase, with acetic acid as the major compound throughout its operation (Figure 5.2b). During Phase III, the pH of R2 dropped to 4.85 on day 52 due

to a problem with the NaOH addition pump. As a result of this instability, the lactic acid concentration increased, while the butyric acid and ethanol concentrations decreased (Figure 5.2b). Lactic acid was detected on some other days, however no pH instabilities or operational problems corresponded with these peaks. During Phase-III the acetic acid concentration increased until the upflow velocity was reduced from 3.0 to 2.0 m/h on day 66, resulting in a decrease in acetic concentration from 76.5 to 39.2 mmol/L (Figure 5.2b). Acetic acid then fluctuated with butyric acid and ethanol as the major compound, throughout Phase -III. Acetic acid was once again major compound in Phase IV (Figure 5.2b).

Both reactors performed similarly in terms of the TC reduction, which exceeded 98 % during all phases. The total amount of ethanol, acetic acid, butyric acid and propionic acid remained almost constant throughout the trial (Figure 5.3), except when lactic acid became the major product, during periods of instability such as the beginning of Phase III in R1 until five days after the reactor was reseded (days 63 to 85) (Figure 5.2a). During this period, when lactic acid was the major product, the combined concentration of ethanol, acetic acid, butyric acid and propionic acid decreased from day 67 to day 87 (Figure 5.3). In addition, both reactors presented similar fluctuations between the main fermentation products.



Figure 5.3 Total concentration of the fermentation products (acetic acid, propionic, butyric acid and ethanol) for R1 and R2 during the continuous acidogenic fermentation of brewery spent grain hydrolysate in expanded granular sludge bed reactors. Note that R2 started 15 days after R1. The changes for the operational conditions were applied simultaneously for both reactors (see **Table 5.1**)

5.3.2 Biomass behaviour

5.3.2.1 Flocculent sludge (R1)

R1 was initially operated for three weeks with intact granular sludge before granules were disaggregated by increasing the pH on day 21 for 12 hours. During these initial three weeks, some changes in the biomass characteristics were observed. The granules, which were initially black, compact and spherical, began to peel (*Figure 5.4a*), resulting in the formation of grey flocculent sludge. By day 9, the sludge bed had three different stratified layers despite the upflow velocity in the reactor

being 6 m/h (*Figure 5.4b*). The bottom layer was composed of black granular sludge similar in appearance to the seed sludge. The top layer was mostly grey-brown flocculent sludge, with some aggregates. The intermediate layer consisted of black granules coated with flocculent biomass, similar to that of the top layer. By day 15, further disaggregation of the new grey-brown aggregates in the top layer occurred and the black granules in the middle layer were no longer covered by the grey flocculent sludge (*Figure 5.4c*), resulting in the wash-out of some of this biomass. Initially, the VSS concentration in the recycling and effluent lines was lower than 2 g VSS/L. After natural disaggregation of the new aggregates, this increased to above 6.0 g VSS/L on day 18 (*Figure 5.4*).



Figure 5.4 Granular sludge development in an EGSB reactor aimed at acidogenesis of brewery spent grain hydrolysate. Fig. 4A – Fig. 4D corresponds to Reactor 1 (R1) and Fig. 4E – Fig. 4H corresponds to Reactor 2 (R2). **a)** R1 day 0, the fresh granular sludge (inoculum); **b)** R1 day 9, three different layers in the sludge bed; **c)** R1 day 19, the granular sludge and the new flocculent sludge prior pH disaggregation; **d)** R1 day 30, flocculent sludge after the pH disaggregation. **e)** R2 day 4, granular sludge and new flocs; **f)** R2 day 13, three layers within the sludge bed; **G)** R2 day 16, granular sludge and flocs after natural disaggregation; and **H)** R2 day 40, two different layers in the sludge bed with granular sludge at the bottom and flocculent sludge on top

To fully disaggregate the remaining granules in R1, the pH was increased to 10.5 for 12 hours on day 21. Subsequently, all granules were disintegrated and only flocculent sludge was present in R1 (*Figure 5.4d*). On day 29, the up-flow velocity was decreased from 6 to 3 m/h, which resulted in a reduction in the VSS content in the recycling and effluent lines (*Figure 5.5a*). When the HRT was further shortened from 36 to 24 h (Phase II), the VSS concentration in the effluent ports increased, to reach its highest concentration of over8 g VSS/L. Beyond this point the VSS concentration decreased, with the lowest values in Phase III-a (0.87 (\pm 0.03) and 0.86 (\pm 0.11) g VSS/L for the recycling and effluent ports, respectively). On day 80 due to low VFA production, R1 was reseeded with granular sludge. The up-flow velocity was decreased on day 81 to 2 m/h. The VSS during the end of this Phase III-b was around 4.0 and 3.6 g VSS/L in the recycling and effluent lines. Finally, during Phase IV the VSS in both sampling ports decreased to a similar value of 2.6 g VSS/L (*Figure 5.5a*).

5.3.2.2 Intact granular sludge

R2 was seeded with the same granular sludge, which developed in a similar manner to R1 for the first 15 days of operation, i.e. the granular sludge peeled and flocculent sludge of a lighter colour appeared in the upper part of the sludge bed (*Figure 5.4e*). This flocculent sludge started to form aggregates and cover some of the original black granular sludge, which resulted in three layers in the bed of the reactor: i) black granules at the bottom, ii) black granular sludge covered by the flocculent sludge in the middle section, and iii) new aggregates and granules of grey-brown colour formed at the top section of the bed sludge (*Figure 5.4f*). The VSS concentration in the recycling line increased from 0.67 (\pm 0.01) g VSS/L on day 0 to 2.44 (\pm 0.01) g VSS/L on day 13.5 (*Figure 5.5*). In order to prevent disaggregation of the newly formed aggregates due to the shear forces, the up-flow velocity was changed from 6 to 3 m/h. Nevertheless, the biomass followed a similar trend to R1, and on day 15 the new grey flocculent sludge coating the black granules, as well as the new aggregates and granules with a lighter colour were naturally disaggregated (*Figure 5.4g*). This resulted in an increase in the VSS concentration in the effluent and to a greater extent, the recycle line (*Figure 5.5b*).

For the remainder of the trial, the bottom of the sludge bed in R2 was composed of mainly black granules. The top section was composed of flocculent sludge and the middle section of a mixture of both (*Figure 4-H*). The flocculent sludge bed section expanded through Phase I and by Phase II reached the recycle line. However, the VSS content in the effluent was higher during Phase I, with the highest concentration corresponding to 9.30 (\pm 0.85) g VSS/L. When the HRT was decreased to 18 h the VSS content in both the effluent and recycle lines decreased until the end of the trial (*Figure 5.5*).



Figure 5.5 Volatile suspended solids in the recycling and effluent sampling ports during the continuous acidogenic fermentation using brewery spent grain liquid hydrolysate in an expanded granular sludge bed reactor throughout the trial for A) Reactor 1 (flocculent sludge), and B) Reactor 2 (granular sludge) (===== natural disaggregation of the new grey biomass; == flocculation of the granular biomass by pH increase; up-flow changes; == HRT changes; == = biomass addition). Note that R2 started 15 days after R1. The changes for the operational conditions were applied simultaneously for both reactors

5.3.3 Gas production

H₂ and CO₂ were the major components of the biogas (Figure 5.6). At the beginning of the trial, methane was detected in both reactors until day 21 in R1 and day 19 in R2 (Figure 5.6). Methane was also detected in R2 from day 55 to 57, albeit below quantification level as the signal during GC analysis was below the lower standard value (5%). Following the reseeding of R1 on day 81, methane was detected between days 88 and 90 at a concentration of 7%. Prior to this, H₂ was the major compound of the total biogas and no methane was detected. For all other gas phase samples in both reactors methane was below than the detection limit.

Generally, in R1, the VGPR increased when operating at a lower HRT (*Figure 5.5a*). Although during Phase III the VGPR decreased until the reactor was reseeded, after which, the VGPR increased. In R2, the gas production was more stable for Phase I than Phase III, when the VGPR increased with decreasing HRT (*Figure 5.5b*). The highest VGPR and HY were achieved during Phase III before reducing the up-flow velocity from 3. 0 to 2.5 m/h. By Phase IV, the VGPR and HY decreased further until the end of the trial (*Figure 5.5b*).



Figure 5.6 Volumetric gas production and hydrogen yield during the continuous acidogenic fermentation of brewery spent grain liquid hydrolysate in an expanded granular sludge bed reactor for A) Reactor 1 (flocculent sludge) and B) Reactor 2 (granular sludge). VGPR: Volumetric Gas Production Rate, HY: Hydrogen Yield. Note that R2 started 15 days after R1. The changes for the operational conditions were applied simultaneously for both reactors

5.3.4 Sulfate and nitrogen compounds

The highest sulfate effluent concentration occurred during Phase I in both reactors (*Figure 5.7*). In R1, the sulfate concentration decreased during Phase II and the sulfate removal efficiency was 50.9 %. When the HRT was decreased to 18 h, the sulfate concentration increased. The final sulfate removal efficiency was 29.5 % at the end of the trial (HRT = 12). For R2, the sulfate concentration decreased at

the end of Phase I and during Phase II. The lowest sulfate concentration was achieved during Phase III, with a sulfate removal efficiency corresponding to 29.1 %.

The protein concentration in R1 during Phase I fluctuated between 0.4 and 0.6 g/L, increasing up to 1.8 g/L during the disaggregation of the granular sludge (*Figure 5.7b*). During Phase II, the protein concentration slightly increased, with a protein removal efficiency of 72.5 (\pm 4.7) %. During Phase III-a, the protein removal efficiency decreased to between 36.5 and 17.6 %, before increasing again after reseeding. On the other hand, the ammonium concentration during this phase decreased and did not increase again for the rest of the trial. In R2, the protein concentrations of 0.51 (\pm 0.06) g/L. On day 43 the protein removal efficiency was lower with a higher effluent concentration (*Figure 5.7b*), along with the production of lactic acid as a result of a period of instability. The ammonium concentration was only stable until day 30 (*Figure 5.7b*). On day 43 (*Figure 5.7b*).



5.4 Discussion

5.4.1 Conversion of carbon compounds

This study demonstrated the production of VFAs using granular and flocculent anaerobic sludge, without pre-treatment, from thermal acid hydrolysate of brewery spent grain. The use of granular sludge (R2) resulted in stable VFA production, with the highest production, 120.4 (± 15.0) mmol/L (total VFAs), at a HRT of 24 hours. On the other hand, the flocculent sludge reactor (R1) had poorer performance, as operating the system at a HRT of 18 h suppressed VFA production.

Many studies have focused on the production of hydrogen as the main product of dark fermentation, without optimisation for VFA production. In addition, many report the average VFA value and not the continuous VFA production (Pugazhendhi et al., 2017; Rodriguez-Valderrama et al., 2019; De Menezes and Silva, 2019). In all these studies, and the present study, a decrease in HRT resulted in higher butyric acid production and lower total VFA production. Pugazhendhi et al. (2017) obtained the highest VFA production, a mixture of acetic, butyric and propionic acid, at a HRT of 3 hours using glucose as feedstock in a fixed bed reactor (**Table 5.2**). On the other hand, the highest HY occurred at a 1.5 hours HRT with an increase in the butyric acid concentration and a decrease in overall VFA production (Pugazhendhi *et al.*, 2017). Furthermore, in the present study, operating the system at longer HRT (36 hours) resulted in a higher yield in terms of VFA production as a function of TC consumed: 1.28 (± 0.13) and 1.38 (± 0.13) mmol VFA/mmol TC, respectively, R1 and R2.

Type of reactor	HRT (h)	рН	Temperature (°C)	Feedstock	TC (g/L)	HAc (mmol/L)	HBu (mmol/L)	HProp (mmol/L)	HVa (mmol/L)	HCa (mmol/L)	EtOH (mmol/L)	Lactic Acid (mmol/L)	Sugar Reduction (%)	Total VFAs (mmol/L)	References
UASB	24.0	5 - 6	21.0	Sucrose	31.0	27.5	55.5	0.3	-	-	10.2	-	93.5	83.3	Rodriguez- Valderrama et al., 2019
UASB	4.8	2.7	30.0	Sucrose	4.5	7.8	1.1	1.1	0.4		-	3.6	80.3	10.0	Mota et al., 2018
UASB	4.0	4	37.0	Glucose	10.0	25.0	11.8	0.0	-	5.5	5.8	-	99.5	36.8	Zhang et al, 2007
EGSB	2.0	3.76 - 4.51	30.0	Sucrose	15.0	25.0	46.5	5.4	0.0	0.0	0.0	0.0	79.0	76.9	de Menezes and Silva, 2019
Fixed bed reactor	3.0	5.5	37.0	Glucose	15.0	20.0	43.1	27.0	-	-	-	15.6	87.3	90.1	Pugazhendhi et al., 2017
EGSB	24.0	5.85 - 6.0	37.0	Glucose, Xylose and Arabinose	15.0	74.4	36.6	9.5	0.0	0.0	31.1	3.4	98.7	120.5	This study

Table 5.2 Optimal conditions for volatile fatty acids production in different high rate reactors

- Not specified

Similar results to the present study were obtained by De Menezes and Silva (2019), when working at the same TC concentration (15 g/L) (**Table 5.2**). Furthermore, in that study, a lower TC inlet concentration led to a larger range of VFAs (De Menezes and Silva, 2019). Moreover, in the present study, the range of the VFAs increased as the propionic acid concentration increased at a longer HRT (*Figure 2*). Zhang et al. (2007) obtained a more complex mixture of fermented compounds (e.g. ethanol, acetic, propionic, butyric and caproic acid) when using a 4 hour HRT versus 1 hour. At the lower HRT, the total amount of VFAs decreased and the butyric acid concentration increased, meanwhile the other compounds diminished (Zhang et al., 2007).

Rodriguez-Valderrama et al. (2019) operated a reactor at 21 °C under different HRTs (4, 12 and 24 hours), but kept the OLR constant by varying the COD inlet concentration. The authors pre-treated the biomass by rinsing with sterilized isotonic solution and heating at 90 °C for 1 h to inhibit methanogens and improve the hydrogen production. The major fermented compounds produced were acetic acid and butyric acid, representing more than 93 % of the total fermented compounds under all conditions investigated. Nevertheless, the total amount of VFAs reached by Rodriguez-Valderrama et al. (2019) was lower than in the present work (Table 5.2).

Regardless of the biomass type, either flocculent or granular, the total fermented compounds were similar for both reactors, in contrast to previous studies using granular and flocculent sludge (Atasoy *et al.*, 2019). Atasoy et al. (2019) used granular and flocculent sludge from different processing stages of a municipal wastewater treatment plant. The similar total VFA concentrations for both biomass types in the present study could be as a result of using the same inoculum for both reactors.

Kan (2013) used different biomass pretreatment methods to enhance hydrogen production, including, heat treatment which suppressed the activity of some homoacetogens and base pretreatment (pH 10.0), which led to a larger range of VFAs produced. Many studies pre-treated the inoculum to enhance the hydrogen production (Zhang *et al.*, 2007; Larriba *et al.*, 2016; Rodríguez-Valderrama *et al.*, 2019; Menezes and Silva, 2019; Pugazhendhi *et al.*, 2017), however the highest VFA production they achieved was lower than in the present study (**Table 5.2**). In addition, any studies that reported VFA production resulted in similar fluctuations for the individual fermented compounds (Cisneros-Pérez *et al.*, 2017; Muñoz-Páez *et al.*, 2020; Mota *et al.*, 2018). These fluctuations could be the result of metabolic fluxes caused by stress responses of fermentative organisms to the continuous accumulation of fermented compounds (Ciranna *et al.*, 2014; Regueira *et al.*, 2018; Saady, 2013). Therefore, to avoid this fluctuation and allow targeted production of desired products, further research on the metabolic pathways of fermentation is required, e.g. through metabolic flux modelling (Regueira *et al.*, 2018; Regueira *et al.*, 2020).

120
Almost all previous studies (**Table 5.2**) created a high OLR by decreasing the HRT, whereas only a few have tried to operate the system at high OLR by increasing the total carbohydrate. This high TC approach was followed by Zhang *et al.* (2008), who operated a reactor with HRTs ranging from 0.25 to 3 hours and glucose concentrations from 5 to 120 g/L. Combinations of HRTs of 1.5 hours with 60 g/L and 2 h with 80 g/L led to a glucose consumption exceeding 80 and 90%, respectively. Meanwhile, the optimal conditions for gas production were at HRTs of 0.25, 0.5 and 0.75 h. The longer HRT could have been beneficial for the accumulation of a high concentration of fermented compounds rather than hydrogen production, however the production of the VFAs was not reported by Zhang *et al.* (2008).

Aside from the periods of instability, with lactic acid production, the VGPR throughout the trial increased as the HRT was reduced. This is in line with other studies that demonstrated increased hydrogen concentrations and HY at low HRT (Menezes and Silva, 2019; Cisneros-Pérez *et al.*, 2017; Zhang *et al.*, 2007). In addition, in these and the present study, propionic acid concentrations decreased. The decrease in HRT resulted in metabolites remaining inside of the system for a shorter period, which contributed to an increase in HY but a decrease in the concentration of compounds originating from hydrogen scavengers, such as propionic acid (Cord-Ruwisch *et al.*, 1988; Saady, 2013). Thus, the uptake of hydrogen or intermediate metabolites by the scavengers was slower than the hydrogen production rate, resulting in a higher hydrogen release. Therefore, the use of a longer HRT is beneficial for the VFA production. Likewise, the decrease of the sulfate concentration would be related to a higher activity of the sulfate reducing bacteria, which convert sulfate into hydrogen sulfide (Papirio et al., 2013). The presence of propionic acid at the largest HRT in both systems could have resulted in a low sulfate reduction efficiency (Bertolino *et al.*, 2012). Moreover, as sulfate reducing bacteria are hydrogen scavengers, the increase of the hydrogen availability in the media could have contributed to their development.

5.4.2 Performance and selection pressure of the flocculent (R1) and whole granular (R2) sludge reactors

The anaerobic granular sludge used as inoculum was from an industrial reactor operated for methane production (Castilla-Archilla *et al.*, 2020). Thus, some methane production was expected. The operational conditions showed, nevertheless, to be adequate to suppress methane production with no need for biomass pre-treatment. Methane was detected in both reactors from the beginning of the trial until day 21 and 29 in R1 and R2, respectively (*Figure 6*). On the other hand, the highest VFA concentrations were obtained on day 3 and 4, in R1 and R2 respectively. High concentrations of VFAs with an acidic pH in the liquor can trigger methane inhibition, resulting in VFA accumulation (Agler *et al.*, 2013; Lu *et al.*, 2015). This could explain the decrease in volumetric methane production rate after

72 hours (Figure 5.6), when total VFA concentrations reached 115.56 and 108.5 mmol/L, in R1 and R2, respectively. The remaining methane production could be a result of accumulation and subsequent release of gas inside of the granules (Wang and Liang, 2016).

During Phase III-a, R1 started to produce lactic acid and the total VFA and ethanol concentrations decreased from day 67 to day 80, when the average total VFA concentration was 23.04 (± 4.60) mmol/L (Figure 5.2). Despite the low VFA concentration no methane was detected during this period, which could have been a result of the wash-out of methanogens from the reactor due to the flocculent nature of the sludge and the low growth rates of the methanogens (Lim et al., 2020). A similar outcome was reported by Palomo-Briones et al. (2017) when using flocculent sludge in a continuous stirred tank reactor operated at a long HRT (24 h). This enriched the biomass for lactic acid production by causing washout of acidogenic microorganisms (Palomo-Briones et al., 2017). Indeed, the washout strategy is effective in enriching hydrogen producing bacteria, when using flocculent sludge without additional treatment (Carrillo-Reyes et al., 2016). de Menezes and Silva (2019) used thermally treated blended granular sludge and obtained lactic acid as the major product, which was favoured by increasing the inlet carbohydrates concentration coupled to longer HRTs (Menezes and Silva, 2019). Approximately one week after reseeding, methane was detected on day 88 with a simultaneous increase in the VFA concentration (Figure 5.2a & Figure 5.6a). This could be explained by a lag phase for the fresh granular sludge to acclimatise to the conditions in the reactor, with a pH close to 6.0 and high concentrations of lactic acid. In addition, as the flocculent sludge prior reseeding was left in the reactor, this sludge was adapted to the medium. Thus, initially the flocculent sludge would have had a higher carbohydrate uptake, favouring lactic acid production, whereas the granular sludge would have produced VFAs and methane. Once the VFA concentration increased, methane production ceased again, probably because the inhibitory threshold concentration for methanogens was reached again (Agler *et al.*, 2013).

During periods of instability in R2, methane was detected at low concentrations in the biogas (< 5 %). This could be as a result of the reduction of the VFA concentrations in the reactor mixed liquor during the production of lactic acid. Therefore, the decrease of the VFA concentration observed on day 51 would have allowed the methanogens to become active again, accounting for the methane production observed on day 55. The role of the selection pressure in the present study would have been different for each of the reactors based on the sludge type, flocculent (R1) and granular (R2). The development of flocculent sludge in R1 led to wash-out, not only of methanogens but also of other bacteria beneficial for hydrogen and VFA production present at Phase I and II. However, during Phase III the operational conditions promoted lactic acid production over VFAs (*Figure 2-A*). Granulation protects the microbial community within the granules from inhibitors present in the media and

improves biomass retention in reactors (Lu *et al.*, 2015). Therefore, the granular sludge in R2 may have protected some of the methanogens, which became active again when the conditions of the reactor mixed liquor were favourable. This could explain the methane production inside of the reactor when the VFA concentrations were low enough to promote methanogenesis. Additionally, it may be possible that some bacteria within the granules were still growing, and were continually replacing other bacteria which were sloughed off the granule surface and washed out. Therefore, the granular sludge could have acted as a carrier material (Menezes and Silva, 2019) and source of bacteria to replace the washed-out fraction. The use of the anaerobic granular sludge without pretreatment can be more beneficial for the production of VFAs. Especially as the operational conditions demonstrated to prevent the production of methane (*Figure 5.6*).

5.4.3 Aggregation and disaggregation of the biomass

In both systems the biomass initially had a similar behaviour, where new biomass with a grey colour developed (*Figure 5.4*). Similar changes of colour have been related with the development of granular biomass for hydrogen production (Fang *et al.*, 2002). Thus, this may have been the result of the growth and development of acidogenic and hydrogen producing organisms favoured by the operational conditions in the reactor, e.g. lower pH, higher temperature and higher organic content. Despite reducing the upflow velocity in R2 to prevent disaggregation, the newly formed grey aggregates in the stratified sludge bed (*Figure 5.4f*), and the grey coating on the original black granules broke apart. A similar effect was reported by Muñoz-Páez *et al.* (2020) when using agave bagasse hydrolysate for hydrogen production. Initially, their reactor was fed using glucose and xylose, once agave hydrolysate was used as feedstock, the granules started to break down (Muñoz-Páez *et al.*, 2020). The authors attributed this to the spatial distribution of microbes in the granules and the change of the substrate.

In the present study, between the formation and disaggregation of the new granules there was no change in any of the operational conditions (i.e. HRT, pH or change of substrate). Additionally, no significant changes in the appearance of the black granular sludge in the bottom layer were observed. Therefore, this could be a result of the development of a new, predominantly acidogenic microbial community on the surface of some of the seed biomass in the middle of the sludge bed and in the new, flocculent biomass on top of the sludge bed. Acidogenic organisms often grow as filamentous outgrowths on the surface of methanogenic granules and as flocculent biomass in the reactor liquor (Arcand et al., 1994; Batstone et al., 2004; Guiot et al., 1992; McHugh et al., 2003; McAteer et al., 2020). It may be the case that the flocculent biomass in the present study consisted of acidogenic organisms, which used the outer layers of the methanogenic granules for attachment until some conditions changed and the flocculent sludge became loose. For example, the loss or change in extracellular polymeric substances production which contribute to the adhesion of the biomass (Lu *et*

al., 2015). However, more studies on the microbial community composition, EPS type and role of the acidifiers in aggregation are required to determine the changes in biomass composition and their role in stability improvement.

5.5 Conclusion

This study demonstrated the feasibility for the production of VFAs using a high rate system with BSG liquid hydrolysate as feedstock. No pretreatment to the biomass was required as the operational conditions minimized or suppressed methanogenesis, with the production and accumulation of VFAs. The use of granular sludge resulted in more stable VFA production compared to flocculent sludge, which produced lactic acid when the HRT was 18 hours. Operating the system at 24 hours HRT allowed a maximum VFA production of 120.4 (± 15.0) mmol/L, which contributed approximately 83.4% (± 5.9) of the soluble COD. Further research is required to optimise this process and convert it from simply a wastewater treatment processes into a combined wastewater treatment-chemical production bioprocesses.

5.6 References

- Agler, M. T., Spirito, C. M., Usack, J. G., Werner, J. J. and Angenent, L. T. (2013) 'Development of a highly specific and productive process for n-caproic acid production: applying lessons from methanogenic microbiomes', Water Science and Technology, 69(1), pp. 62-68.
- Atasoy, M., Eyice, O., Schnürer, A. and Cetecioglu, Z. (2019) 'Volatile fatty acids production via mixed culture fermentation: Revealing the link between pH, inoculum type and bacterial composition', Bioresource Technology, 292, pp. 121889.
- Bertolino, S. M., Rodrigues, I. C. B., Guerra-Sá, R., Aquino, S. F. and Leão, V. A. (2012) 'Implications of volatile fatty acid profile on the metabolic pathway during continuous sulfate reduction', Journal of Environmental Management, 103, pp. 15-23.
- Carrillo-Reyes, J., Cortés-Carmona, M. A., Bárcenas-Ruiz, C. D. and Razo-Flores, E. (2016) 'Cell washout enrichment increases the stability and performance of biohydrogen producing packedbed reactors and the community transition along the operation time', Renewable Energy, 97(Supplement C), pp. 266-273.
- Castilla-Archilla, J., Papirio, S. and Lens, P. N. L. (2020) 'Two step process for volatile fatty acid production from brewery spent grain: hydrolysis and direct acidogenic fermentation using anaerobic granular sludge', Process Biochemistry.
- Ciranna, A., Ferrari, R., Santala, V. and Karp, M. (2014) 'Inhibitory effects of substrate and soluble end products on biohydrogen production of the alkalithermophile Caloramator celer: Kinetic, metabolic and transcription analyses', International Journal of Hydrogen Energy, 39(12), pp. 6391-6401.

- Cisneros-Pérez, C., Carrillo-Reyes, J., Celis, L. B., Alatriste-Mondragón, F., Etchebehere, C. and Razo-Flores, E. (2015) 'Inoculum pretreatment promotes differences in hydrogen production performance in EGSB reactors', International Journal of Hydrogen Energy, 40(19), pp. 6329-6339.
- Cisneros-Pérez, C., Etchebehere, C., Celis, L. B., Carrillo-Reyes, J., Alatriste-Mondragón, F. and Razo-Flores, E. (2017) 'Effect of inoculum pretreatment on the microbial community structure and its performance during dark fermentation using anaerobic fluidized-bed reactors', International Journal of Hydrogen Energy, 42(15), pp. 9589-9599.
- Cord-Ruwisch, R., Seitz, H.-J. and Conrad, R. (1988) 'The capacity of hydrogenotrophic anaerobic bacteria to compete for traces of hydrogen depends on the redox potential of the terminal electron acceptor', Archives of Microbiology, 149(4), pp. 350-357.
- Etchebehere, C., Castelló, E., Wenzel, J., del Pilar Anzola-Rojas, M., Borzacconi, L., Buitrón, G., Cabrol, L., Carminato, V. M., Carrillo-Reyes, J., Cisneros-Pérez, C., Fuentes, L., Moreno-Andrade, I., Razo-Flores, E., Filippi, G. R., Tapia-Venegas, E., Toledo-Alarcón, J. and Zaiat, M. (2016)
 'Microbial communities from 20 different hydrogen-producing reactors studied by 454 pyrosequencing', Applied Microbiology and Biotechnology, 100(7), pp. 3371-3384.
- Fang, H. H. P., Liu, H. and Zhang, T. (2002) 'Characterization of a hydrogen-producing granular sludge', Biotechnology and Bioengineering, 78(1), pp. 44-52.
- Feng, Y., Wang, Q., Duan, J.-L., Li, X.-Y., Ma, J.-Y., Wu, L., Han, Y., Liu, X.-Y., Zhang, Y.-B. and Yuan, X.-Z. (2020) 'Attachment and adhesion force between biogas bubbles and anaerobic granular sludge in the up-flow anaerobic sludge blanket', Water Research, 171, pp. 115458.
- Fernández, F. J., Villaseñor, J. and Infantes, D. (2011) 'Kinetic and stoichiometric modelling of acidogenic fermentation of glucose and fructose', Biomass and Bioenergy, 35(9), pp. 3877-3883.
- Kan, E. (2013) 'Effects of pretreatments of anaerobic sludge and culture conditions on hydrogen productivity in dark anaerobic fermentation', Renewable Energy, 49, pp. 227-231.
- Larriba, M., Navarro, P., Delgado-Mellado, N., Stanisci, V., García, J. and Rodríguez, F. (2016) 'Separation of aromatics from n-alkanes using tricyanomethanide-based ionic liquids: Liquidliquid extraction, vapor-liquid separation, and thermophysical characterization', Journal of Molecular Liquids, 223(Supplement C), pp. 880-889.
- Lens, P. N., De Beer, D., Cronenberg, C. C., Houwen, F. P., Ottengraf, S. P. and Verstraete, W. H. (1993) 'Heterogeneous Distribution of Microbial Activity in Methanogenic Aggregates: pH and Glucose Microprofiles', Applied and environmental microbiology, 59(11), pp. 3803-3815.
- Lim, J. X., Zhou, Y. and Vadivelu, V. M. (2020) 'Enhanced volatile fatty acid production and microbial population analysis in anaerobic treatment of high strength wastewater', Journal of Water Process Engineering, 33, pp. 101058.
- Lu, X., Zhen, G., Estrada, A. L., Chen, M., Ni, J., Hojo, T., Kubota, K. and Li, Y.-Y. (2015) 'Operation performance and granule characterization of upflow anaerobic sludge blanket (UASB) reactor treating wastewater with starch as the sole carbon source', Bioresource Technology, 180, pp. 264-273.

- McAteer, Paul G., Anna Christine Trego, Camilla Thorn, Therese Mahony, Florence Abram, and Vincent O'Flaherty. 2020. "Reactor configuration influences microbial community structure during high-rate, low-temperature anaerobic treatment of dairy wastewater." Bioresource Technology 307: 123221. https://doi.org/https://doi.org/10.1016/j.biortech.2020.123221.
- de Menezes, C. A. d. and Silva, E. L. (2019) 'Hydrogen production from sugarcane juice in expanded granular sludge bed reactors under mesophilic conditions: The role of homoacetogenesis and lactic acid production', Industrial Crops and Products, 138, pp. 111586.
- Mota, V. T., Ferraz Júnior, A. D. N., Trably, E. and Zaiat, M. (2018) 'Biohydrogen production at pH below 3.0: Is it possible?', Water Research, 128, pp. 350-361.
- Muñoz-Páez, K. M., Alvarado-Michi, E. L., Moreno-Andrade, I., Buitrón, G. and Valdez-Vazquez, I. (2020) 'Comparison of suspended and granular cell anaerobic bioreactors for hydrogen production from acid agave bagasse hydrolyzates', International Journal of Hydrogen Energy, 45(1), pp. 275-285.
- Oliva, A., Tan, L. C., Papirio, S., Esposito, G. and Lens, P. N. L. (2020) 'Effect of methanol-organosolv pretreatment on anaerobic digestion of lignocellulosic materials', Renewable Energy.
- Palomo-Briones, R., Razo-Flores, E., Bernet, N. and Trably, E. (2017) 'Dark-fermentative biohydrogen pathways and microbial networks in continuous stirred tank reactors: Novel insights on their control', Applied Energy, 198, pp. 77-87.
- Papirio S., Esposito G., Pirozzi F., Biological inverse fluidized-bed reactors for the treatment of low pHand sulphate-containing wastewaters under different COD/SO₄²⁻ conditions. (2013), vol. 34, pp. 1141-1149
- Pugazhendhi, A., Anburajan, P., Park, J.-H., Kumar, G., Sivagurunathan, P. and Kim, S.-H. (2017) 'Process performance of biohydrogen production using glucose at various HRTs and assessment of microbial dynamics variation via q-PCR', International Journal of Hydrogen Energy, 42(45), pp. 27550-27557.
- Regueira, A., Bevilacqua, R., Lema, J. M., Carballa, M. and Mauricio-Iglesias, M. (2020) 'A metabolic model for targeted volatile fatty acids production by cofermentation of carbohydrates and proteins', Bioresource Technology, 298, pp. 122535.
- Regueira, A., González-Cabaleiro, R., Ofiţeru, I. D., Rodríguez, J. and Lema, J. M. (2018) 'Electron bifurcation mechanism and homoacetogenesis explain products yields in mixed culture anaerobic fermentations', Water Research, 141, pp. 349-356.
- Reyhanitash, E., Kersten, S. R. A. and Schuur, B. (2017) 'Recovery of volatile fatty acids from fermented wastewater by adsorption', ACS sustainable chemistry & engineering, 5(10), pp. 9176-9184.
- Reyhanitash, E., Zaalberg, B., Kersten, S. R. A. and Schuur, B. (2016) 'Extraction of volatile fatty acids from fermented wastewater', Separation and Purification Technology, 161(Supplement C), pp. 61-68.
- Rocha, M. A. A., Raeissi, S., Hage, P., Weggemans, W. M. A., van Spronsen, J., Peters, C. J. and Kroon,
 M. C. (2017) 'Recovery of volatile fatty acids from water using medium-chain fatty acids and a cosolvent', Chemical Engineering Science, 165, pp. 74-80.

- Rodríguez-Valderrama, S., Escamilla-Alvarado, C., Amezquita-Garcia, H. J., Cano-Gómez, J. J., Magnin, J.-P. and Rivas-García, P. (2019) 'Evaluation of feeding strategies in upflow anaerobic sludge bed reactor for hydrogenogenesis at psychrophilic temperature', International Journal of Hydrogen Energy, 44(24), pp. 12346-12355.
- Saady, N. M. C. (2013) 'Homoacetogenesis during hydrogen production by mixed cultures dark fermentation: Unresolved challenge', International Journal of Hydrogen Energy, 38(30), pp. 13172-13191.
- Saboe, P. O., Manker, L. P., Michener, W. E., Peterson, D. J., Brandner, D. G., Deutch, S. P., Kumar, M., Cywar, R. M., Beckham, G. T. and Karp, E. M. (2018) 'In situ recovery of bio-based carboxylic acids', Green Chemistry, 20(8), pp. 1791-1804.
- SEAL (2019) SEAL AutoAnalyzer 3 HR. Available at: https://www.sealanalytical.com/Products/SegmentedFlowAnalyzers/AA3HRAutoAnalyzer/tabid/59/language/ en-US/Default.aspx (Accessed: 15/01/2021
- Zhang, Z.-P., Show, K.-Y., Tay, J.-H., Liang, D. T. and Lee, D.-J. (2008) 'Biohydrogen production with anaerobic fluidized bed reactors—A comparison of biofilm-based and granule-based systems', International Journal of Hydrogen Energy, 33(5), pp. 1559-1564.
- Zhang, Z.-P., Tay, J.-H., Show, K.-Y., Yan, R., Tee Liang, D., Lee, D.-J. and Jiang, W.-J. (2007) 'Biohydrogen production in a granular activated carbon anaerobic fluidized bed reactor', International Journal of Hydrogen Energy, 32(2), pp. 185-191.

Chapter 6

Brewery spent grain cascade process for recovery of carbohydrates, proteins and lignin

Keywords: lignocellulose; brewery spent grain; cascade process; pretreatment; thermal diluted acid hydrolysis; severity factor; enzymatic hydrolysis; water retention capacity

Abstract

Brewery spent grain (BSG) contains a large fraction of proteins, lignin and carbohydrates. Some of these compounds have a high value in the food and pharmaceutical industry. This chapter investigated a cascade process by combining a chemical pretreatment step followed by enzymatic hydrolysis. First, a response surface methodology was used to optimise diluted hydrochloric acid hydrolysis with different combinations of temperature, time and acid concentration. Three different combinations were identified based on i) the highest protein content in the solid fraction after the pretreatment (C1: 0.49 % acid at 87.7 °C for 92.7 min), ii) the highest liquid recovery (C2: 0.80 % acid at 121.0 °C for 142 min), and iii) the lowest acid concentration (C3: 0.10 % of acid, 104.0 °C for 70 min). In addition, a fourth condition was added based on the lowest water retention capacity (C4: 0.20 % of acid, 121.0 °C for 20 min). The enzymatic hydrolysis for the four different conditions was carried out for 24 hours to further enhance the release of the carbohydrates. The efficiency of the enzymatic hydrolysis was evaluated in presence and absence of a large concentration of reducing carbohydrates, this was carried out centrifuging the slurry after pretreatment, recovering and washing the solid prior to adding the enzyme; in the second case the enzymatic cocktail was added straight after the chemical pretreatment. No effect was observed, and even hydrolysing the whole slurry resulted in a higher release of carbohydrates. The overall carbohydrate recovery efficiency in the liquid fraction for C1, C2, C3 and C4 corresponded to 399.1 ± 26.1 , 535.8 ± 28.7 , 257.0 ± 11.5 and 446.3 ± 81.1 mg carbohydrate per dry gram of BSG. The solid fractions for C1, C2 and C4 were evaluated using a larger volume. For each of them two different solid fractions were obtained, a fine solid and a coarse solid fraction. The highest protein content in the fine solid was for C4, meanwhile in the case of the coarse solid fraction the highest proteins content corresponded to C1. Further analysis is required to determine the potential of these solid fractions as food supplement.

6.1 Introduction

Growing populations have resulted in increased food demand. Thus, alternative sources of food are required to mitigate this demand. For example, lignocellulose is the largest solid biomass that could be valorised as an alternative source of fibre, proteins and nutraceutical compounds (Contreras et al., 2019). The recovery of high value compounds from existing lignocellulosic waste streams represent an opportunity for revalorisation through decomposition and fractionation of dietary fiber, protein or phenolic acids (Martinez et al., 2012; Laguna et al., 2018). This process could be coupled with carbohydrate extraction for chemical and fuel production (Celiktas et al., 2014).

Brewery spent grain (BSG) is a lignocellulosic by-product of beer production with a large amount of protein, phenolic compounds and fibre that have shown prebiotic and nutraceutical activities (Vieira *et al.* 2014; Rommi *et al.* 2018; Cermeño et al., 2019). In the beer production process, barley is germinated to activate natural enzymes and initiate degradation of the grain. Then, this malted barley is milled and suspended in water. To promote further hydrolysis the temperature is slowly increased until the denaturalization of the enzymes present (Mussatto et al., 2006). As a result, the easiest biodegradable compounds are solubilised and the most recalcitrant compounds are concentrated in the remaining solid. For example, more than 98% and 92% reductions in the starch and glucan content, respectively, meanwhile arabinoxylan is concentrated almost 3.5 times and the protein fraction above 2.5 times (Celus et al., 2006). The recovery and use of these complex compounds has been investigated previously. For example, recovery of the carbohydrate fraction for chemical and energy production (Wilkinson et al., 2014; Plaza et al., 2017; Rojas-Chamorro et al., 2020), or the recovery of bioactive compounds such as protein or phenolic compounds (McCarthy et al., 2012; Cermeño et al., 2019; Connolly et al., 2019). However, few studies evaluated the simultaneous recovery of the carbohydrates or other fractions such as lignin or proteins (Niemi et al., 2013; Rommi et al., 2018).

BSG consists of the husk, the pericarp and the seed coat which are mainly composed of interwoven cellulose, hemicellulose and lignin, with a small fraction of proteins and lipids (Mussatto et al., 2006; Lynch et al., 2016; Jaeger et al., 2021). These fractions act as a barrier for enzymatic and mild chemical hydrolysis during protein extraction (Niemi et al., 2013; Ibbet et al., 2019). The most common pretreatment for the release of these proteins is alkaline pretreatment, which solubilises proteins with low degradation of the lignocellulose fraction (Connoly et al., 2019; Wilkinson et al., 2014). Many studies have focused on the optimization of carbohydrate solubilisation from BSG using enzymatic hydrolysis (Niemi et al. 2012). These optimisations usually include an initial chemical hydrolysis with a combination of different temperature, time and acid concentration that degrade not only cellulose and hemicellulose but also proteins (Plaza et al. 2017, Patel et al. 2018; Ravindran et al. 2018; Rojas-Chamorro et al. 2020). This combination of time, temperature and acid concentration during the

pretreatment can be integrated into a single value known as severity factor (Carvalheiro et al. 2004). Increasing this factor results in better carbohydrate solubilisation, but also partially degrades solubilised proteins into peptides (Kemppainen et al., 2016).

The current chapter aims to investigate the minimum severity factor for the breakdown of the lignocellulose wall that protects the protein fraction in BSG using different combinations of i) temperature, ii) acid concentration, and iii) time as pretreatment step prior to enzymatic hydrolysis. The effect of these combinations is evaluated in terms of final carbohydrate efficiency release after the enzymatic hydrolysis, as well as the recovery and quality of the protein and lignin solid fractions. First, a diluted acid hydrolysis was optimised before enzymatic hydrolysis. After acid pretreatment and enzymatic hydrolysis, the carbohydrate and protein release were evaluated using acid and alkaline conditions in short volume. The best hydrolysis results were carried out using a larger volume for the fractionation of the liquid and the solid streams. The carbohydrate, protein and ammonium concentration were analysed in the liquid fraction; the total solid, water retention capacity (WRC), liquid/solid ratio and protein fraction were analysed in the solid fractions.

6.2 Material and methods

6.2.1 Feedstock

BSG used for the production of Indian Pale Ale was kindly supplied by a local brewery in Galway (Ireland), which differed from the BSG used in *Chapter 4* and *Chapter 5*, which was from lager production. The BSG was collected from the plant after the wort production. This was mixed and frozen in individual bags of 500 g each at -20 °C within 3 hours. Prior to the hydrolysis step, the BSG bag was defrosted overnight at 15 °C.

6.2.2 Hydrolysis process

6.2.2.1 Sulfuric and hydrochloric acid comparison

The BSG was blended using distilled water in a Waring 2-Speed laboratory blender (USA) at the highest speed for 1 minute. The first set of experiments tested the effectiveness of thermal diluted acid hydrolysis using hydrochloric or sulfuric acid at different concentrations (0.00, 0.05, 0.10, 0.25, 0.5, 1.00 and 1.50 % v/v). Sulfuric acid (95 % purity and 1.83 g/mL density) and hydrochloric acid (37 %) were supplied from Fisher Chemical (UK). A solid concentration of 10 % (w/w) BSG was used and hydrolysis was done in an autoclave (Sanyo Labo, Japan) at 121 °C for 20 minutes as described previously (Carvalheiro et al., 2004; Djioleu and Carrier, 2016). Liquid samples obtained after hydrolysis were analysed for monosaccharides and disaccharides (cellobiose, maltose, glucose, xylose and arabinose), total protein and ammonia content. Based on these experiments, hydrochloric acid was chosen as a pretreatment for the subsequent steps of this work.

6.2.2.2 Pretreatment optimization

A Response Surface Methodology (RSM) was developed using Minitab v18. This statistical tool was used to optimise different combinations of temperature, acid concentration and time in order to optimise the maximum carbohydrate solubilisation, liquid recovery and protein content in the solid residue. The maximum and minimum value for the optimization were introduced to the software to generate the different combinations (45 combinations), based on three different values for each of the three different parameters (*Table 6.1*). This optimization was done prior to enzyme addition using Duran bottles (250 mL) with a working volume of 100 mL. Incubations at temperatures higher than 100 °C were carried out in an autoclave (Sanyo Labo, Japan) and incubations below 100 °C were carried out in a thermostatic incubator (Grant, UK).

Table 6.1 Parameters used in the Response Surface Methodology for the optimization of the thermal diluted hydrochloric acid hydrolysis as pretreatment prior to enzyme addition

Acid concentration	on Temperature	Time
(% v/v)	(°C)	(min)
0.10	60.0	20
0.55	90.5	100
1.00	121.0	180

6.2.2.3 Enzymatic hydrolysis

The three different combinations obtained during the RSM trial (**Table 6.2**: *Condition 1, 2 and 3*) plus an additional condition based on the results of the lowest water retention capacity of the initial experiment (**Table 6.2**: *Condition 4*) were used as pretreatments to test enzymatic hydrolysis. The enzymatic hydrolysis for each of the conditions was carried out using Depol 40L (kindly provided by Biocatalysts, UK) with an enzymatic load of 15 cellulose units/g of BSG (TS). To evaluate the possible inhibition of the enzymes in the presence of reducing sugars and inhibitor compounds released during the pretreatment, two different approaches were tested for the efficiency of the enzymatic hydrolysis. The first approach involved adding the enzymes to the whole slurry (WS) obtained after pretreatment. The second was by centrifuging the WS, recovering the pellet and re-suspending it in the same amount of liquid that was recovered during the centrifugation. The enzymatic hydrolysis was carried out in 250 mL Duran bottles in an incubator at 50 °C. To ensure the correct homogenization the bottles were mixed agitated using 100 rpm using the thermostatic incubator (Grant, UK). The pH was initially corrected to optimal conditions for the enzymatic hydrolysis as suggested by the manufacturer (pH 5.0). The total length of the enzymatic hydrolysis was 24 hours. All tests were performed in replicates (n = 6). Samples were taken at 0, 3, 6, 12, 18 and 24 hours by removing an aliquot using aseptic techniques and sterilised implements. Samples were then boiled for 5 minutes. After the enzymatic hydrolysis was completed, the solubilisation of carbohydrates and proteins was evaluated under acidic (pH 1.0) and alkaline (pH 8.0) conditions. This was done by subdividing the six replicates in half (n = 3 for each pH). The pH was corrected and bottles were left overnight at 4 °C. The following morning, the liquid fraction was recovered by centrifugation (Beckman Coulter, Allegra X-30R Centrifuge (Germany)) and filtered using filter paper (Sigma-Aldrich, Grade 542, 150 mm diameter) to remove any remaining solid particles. The efficiency of the process was based on the carbohydrates recovered by gram of BSG used (TS), also the ratio between the liquid and solid fraction was calculated to determine the efficiency of the hydrolysis in terms of liquid recovered.

Condition	Acid concentration	Temperature	Time
	(% v/v)	(°C)	(min)
C1	0.49	87.7	92
C2	0.80	121.0	142
С3	0.10	104.0	70
C4	0.20	121.0	20

Table 6.2 Parameters used for pretreatment prior the enzymatic hydrolysis of the brewery spent grain

6.2.2.4 Solid and liquid fractionation

Conditions 1, 2 and 4 (*Table 6.2*) were selected for further experimentation on solid and liquid fractionation using a larger volume. Enzymatic hydrolysis was carried out for 24, 6 and 24 hours for C1, C2 and C4, respectively, using a working volume of 800 mL in a 1000 mL Duran bottle under the same conditions. Following hydrolysis, the solid fraction and an aliquot of the liquid fraction were stored for further analysis. The process used for the fractionation of these hydrolysis experiments is summarised in *Figure 6.1*. Briefly, a fraction of the WS was taken before and after the enzymatic hydrolysis. After the enzymatic hydrolysis the pH was corrected to 1.0, and left overnight at 4 °C. The following day the slurry was filtered using a 100 μ m sieve, which was selected after sieves with different pore size (250, 100 and 25 μ m) were tested. Two different fractions were obtained: a coarse solid (CS) fraction with particles larger than 100 μ m and a liquid fraction containing fine solids smaller than 100 μ m. The liquid fraction was centrifuged and the fine solid (FS) fraction was recovered and frozen for further analysis. The acid liquid fraction was neutralized and filtered using paper filter (Sigma-Aldrich, Grade 542, 150 mm diameter). The CS were re-suspended with distilled water using the same volume as the liquid with fine particles. Then, a sample was taken and the pH was increased to 8.0. The bottle was left overnight at 4 °C, and the slurry was sieved using a 25 μ m sieve.



Figure 6.1 Process diagram for the hydrolysis of brewery spent grain and the fractionation and treatment of the different streams. Three different pretreatment conditions were used before enzymatic hydrolysis. The blue dots represent sampling of liquid (\bullet), and the yellow dots represent the samples taken from the solid fraction (\bullet). There was no solid fraction after neutralization of the acid liquid, which has been represented with the yellow shape circle (\bigcirc)

6.2.3 Physico-chemical analysis

The protein content in the solid fraction was determined by the Kjeldahl method with a correction factor of 5.50 as a more accurate approximation for BSG (Mariotti et al., 2019). The characterisation of the solid samples in terms of hemicellulose, cellulose and lignin content was carried out by Celignis Limited (Limerick, Ireland). In addition, for the raw BSG the starch content was analysed as well.

Samples for chemical analysis were centrifuged at 9,000 rpm for 5 minutes and the supernatant was filtered using a syringe filter with 0.22 μ m pore size. The samples were then frozen for further analysis. Protein and ammonia were analysed using a discrete nutrient analyser (Gallery Plus, ThermoFisher) as described in *Chapter 4*. Total suspend solids (TSS) and volatile suspend solids (VSS) were measured following the standard method (APHA, 2012).

Maltose, cellobiose, glucose, xylose and arabinose were measured using a 1260 Infinity II liquid chromatograph (Agilent, Germany) equipped with a Hi Plex H 7.7 x 300 mm and 8 μ m (p/n PL1170-6830) column (Agilent, UK) kept at 60 °C and an RI detector at 55 °C. The mobile phase was sulfuric acid at a concentration of 5 mM and a flow rate of 0.7 mL/min.

6.2.4 Calculation

The efficiency of hydrolysis was calculated based on the total carbohydrates in the liquid fraction after the neutralization of the acid liquid as explained in *Chapter 4*. The severity factor for each of the combinations during the pretreatment step was calculated as described in *Chapter 4*.

The WRC was calculated following Robertson et al. (2000). This represents the amount of water retained by the solid fraction after centrifugation as it is reported in *Equation 6.1*:

 $WRC = \frac{\text{Residue fresh weight-residue dry weight}}{\text{Residue dry weight}} (g/g)$

Equation 6.1

6.3 Results

6.3.1 Comparison using hydrochloric and sulfuric acid

The hydrochloric acid performed better in terms of carbohydrates and proteins release from BSG (Figure 6.2 1*a* & 1*b*). The increase of the pentoses (xylose and arabinose) and hexose (glucose) were similar using 0.25 % hydrochloric acid. Almost the maximum xylose and arabinose concentration was achieved at a 1.0 % acid concentration. On the other hand, the glucose concentration was still increasing from 1.0 to 1.5 % (Figure 6.2 1*a*). The release of carbohydrates using sulfuric acid was slower than with hydrochloric acid (Figure 6.2 2*a*). For solubilised proteins and ammonia, the highest concentrations and rate were also achieved with hydrochloric acid (Figure 6.2 2*b*).



Figure 6.2 Compound characterisation in the filtered liquid fraction (20 μ L) of the hydrolysis of brewery spent grain 10 % TS (w/w) under different acid concentrations, hydrochloric (**1**) or sulfuric (**2**), in the autoclave at 121 °C for 20 min. **a**) Carbohydrates characterisation, and **b**) proteins and ammonium concentration

The WRC in the sulfuric acid pretreatment increased from 0 to 0.10 % v/v and achieved the lowest WRC with an acid concentration of 0.50 %. For the hydrochloric acid, this increase of the WRC was not as clear as with sulfuric acid, achieving the lowest WRC at 0.25 % hydrochloric acid (Figure 6.3*a*). In both treatments, the WRC followed a similar trend where the WRC increased with a higher rate for the sulfuric acid. In terms of the water recovery, *Figure 6.3b* shows the amount of liquid fraction recovered from the initial 100 mL. In both cases the highest increase of the liquid fraction recovered was from 0.05 to 0.25 % acid. The liquid fraction recovered was higher with hydrochloric than with sulfuric acid (Figure 6.3*b*). However, the liquid from the hydrochloric acid hydrolysis was lower when increasing the acid concentration from 0.25 to 1.5 %, whereas the liquid recovered from sulfuric acid was almost constant with concentrations ranging from 0.5 to 1.5 % (Figure 6.3*b*).



Figure 6.3 a) Water retention capacity (gram of water per gram of total solids), and **b)** liquor recovered from 100 mL during the hydrolysis of brewery spent grain with 10 % TS (w/w) carried in the autoclave at 121 °C for 20 min with sulfuric and hydrochloric acid under different concentrations

Some changes were observed in the colour and some properties of the BSG structure when the acid concentration was increased, such as the WRC and the liquid recovered (*Figure 6.3*). Hydrochloric acid

concentrations of 0.10 %, or less, resulted in a liquid fraction with a milky colour (*Figure 6.4*) and a white layer was observed on the top of these samples. When the hydrochloric acid concentration was increased from 0.10 to 0.25 %, the liquid fraction turned to a golden/brown colour. The white layer then changed to a darker colour. There were some floating particles after centrifugation, with a larger amount observed upon increasing the acid concentration.



Figure 6.4 Tubes after centrifugation of the whole slurry for each of the different thermal diluted hydrochloric acid pretreatments of brewery spent grain at different acid concentration carried at 121 °C for 20 minutes

6.3.2 Response surface methodology

The RSM was used to evaluate the effect of combinations of acid concentration, temperature and time on the diluted acid hydrolysis pretreatment. To generate these combinations, the highest and lowest parameters were fixed, and the software used three different values for each parameter (*Table 6.1*) to generate 45 different combinations. The main parameters evaluated by the response surface methodology included the total protein content in the solid fraction, total carbohydrates, monosaccharide concentration, pentose concentration (arabinose and xylose) and solid weight.

As lignocellulose is a protective barrier for proteins (Ibbet et al., 2019), the conditions with the maximum protein content in the pellet was selected as an indicator for when this barrier started to break down. This corresponded to a combination of 0.49 % acid at 87.7 °C for 92.7 min, which was used to generate Condition 1 (C1), and which had a severity factor of 0.70. The second condition was related with the WRC of the BSG, which was based on the lowest amount of solid obtained. The resulting combination corresponding to Condition 2 (C2) was 0.80 % acid at 121.0 °C for 142 min with a severity factor of 1.87. The third condition was designed by fixing the acid concentration at the

lowest level tested, giving Condition 3 (C3) a combination of 0.10 % of acid, 104.0 °C for 70 min. Condition 4 (C4) was added based on the WRC during the initial experiment when comparing hydrochloric acid with sulfuric acid (*Figure 6.3a*). Thus, Condition 4 was carried out using 0.20 % of acid, 121.0 °C for 20 min.

6.3.3 Small enzymatic hydrolysis

The enzymatic hydrolysis for the four different conditions was carried out for 24 hours to test additional degradation of the complex carbohydrates by the enzymes, using the WS and the resuspended pellet (RP). *Figure 6.5* shows the carbohydrates concentration during the enzymatic hydrolysis. The biggest increases in total carbohydrates were in condition C1 and C4 for the WS, corresponding to 23.10 (\pm 5.11) and 24.91 (\pm 4.64) g/L, respectively. In addition, in the case of the WS for C1 and C4 there was maltose present at the beginning of the enzymatic hydrolysis. The maltose was fully consumed in C1 after 12 hours, in C4 the maltose concentration increased during the beginning of the enzymatic hydrolysis, with the highest concentration achieved after 9 hours, and the concentration slightly decreased at the end of the hydrolysis (*Table 6.3*). Maltose was not present in any of the other samples. In C2, the WS and the RP hydrolysis resulted in a similar carbohydrate increment, 14.31 (\pm 1.24) g/L and 13.56 (\pm 2.33) g/L. C2 differed from the rest of the conditions as more carbohydrates were released from the RP with an increase of 5.17 (\pm 1.69) g/L, whereas the increase of the WS was 3.74 (\pm 2.86) g/L. This increase in carbohydrate concentrations was mainly by cellobiose, as all the other type of carbohydrates kept a similar concentration until the end of the hydrolysis step (*Table 6.3*).

		۷	Whole Slurr	y	Resuspended Pellet					
Cond.	Cellobiose	Maltose	Glucose	Xylose	Arabinose	Cellobiose	Maltose	Glucose	Xylose	Arabinose
	g/L	g/L	g/L	g/L	g/L	g/L	g/L	g/L	g/L	g/L
1	5.05 ± 0.81	-2.45 ± 0.49	12.56 ± 1.34	7.45 ± 0.71	0.50 ± 1.75	1.77 ± 0.37	0.00 ± 0.00	8.93 ± 1.29	5.50 ± 0.65	0.07 ± 0.14
2	2.67 ± 0.63	0.00 ± 0.00	0.71 ± 0.90	0.36 ± 0.81	0.00 ± 0.52	3.47 ± 0.20	0.00 ± 0.00	1.18 ± 0.67	0.31 ± 0.52	0.22 ± 0.30
3	-1.90 ± 0.28	0.00 ± 0.00	11.96 ± 0.66	3.32 ± 0.21	0.95 ± 0.09	-2.63 ± 1.13	0.00 ± 0.00	11.96 ± 0.64	3.28 ± 29	0.95 ± 0.28
4	3.65 ± 0.83	-0.41 ± 0.42	11.71 ± 1.28	9.03 ± 0.92	0.93 ± 1.20	1.55 ± 0.93	-0.02 ± 0.03	6.68 ± 1.40	5.30 ± 1.64	0.77 ± 1.98

Table 6.3 Increase in carbohydrate concentration after 24 hours of enzymatic hydrolysis for four different pretreatment conditions



Figure 6.5 Carbohydrate release during the enzymatic hydrolysis for the four different pretreatments. *Left*) Whole slurry and *right*) resuspended solid

For conditions C1, C2 and C3, the cellobiose concentration increased along the enzymatic hydrolysis, whereas the cellobiose concentration only decreased in the case of C3 (*Table 6.3*). For all of the different enzymatic hydrolysis conditions tested, the lowest monosaccharide released during the enzymatic hydrolysis corresponded to arabinose (*Table 6.3*). The largest increase in xylose concentration was in C2 and C4, slightly lower compered to when using RP and the highest release for the WS at C4 (*Table 6.3*). C1 and C4 had a similar efficiency during the enzymatic hydrolysis, although the overall efficiency was higher for C4 as the initial concentration of the carbohydrates was higher at time 0 for C4 than for C1 (*Figure 6.5*).

For each of the hydrolysates, the liquid recovered as supernatant during the centrifugation and the total amount of solid in the pellet were used to calculate the ratio between the liquid and solid fraction. This was used to understand the extent of the hydrolysis for the lignocellulose degradation. The lowest ratio between the liquid and solid fraction was obtained for C3, which had the lowest severity factor (*Table 6.4*). As well, the influence of the pH above this liquid ratio was studied at acid and alkaline pH, respectively, 1.0 and 8.0. In all the cases where the enzymatic hydrolysis was carried out using the RP, the ratio liquid/solid was higher for the lowest pH. On the other hand, only C2 had a higher ratio under acid conditions after the WS enzymatic hydrolysis, as conditions C1, C3 and C4 presented a larger liquid fraction at alkaline conditions. For C2 centrifuging the slurry under acidic conditions resulted in an increase in the volume of supernatant recovered, meanwhile the alkaline conditions resulted in a lower ratio between the liquid and solid phase.

Table 6.4 pH after thermal diluted hydrochloric acid pretreatment and ratio between liquid and solid fraction of the slurry after the acid hydrolysis, whole slurry left overnight at pH 1.0 and pH 8.0, and resuspended solid left overnight at pH 1.0 or pH 8.0

	рН	Condition 1	Condition 2	Condition 3	Condition 4
After acid	pH:	1.00 ± 0.03	0.81 ± 0.04	2.94 ± 0.10	1.71 ± 0.02
11901019313	Ratio:	1.38 ± 0.30	1.64 ± 0.08	0.33 ± 0.07	1.66 ± 0.19
Whole	1.0	1.33 ± 0.30	2.24 ± 0.07	0.48 ± 0.10	2.31 ± 0.09
slurry	8.0	1.63 ± 0.25	1.36 ± 0.25	0.51 ± 0.06	2.42 ± 0.27
Resuspended	1.0	2.24 ± 0.07	1.76 ± 0.15	0.20 ± 0.05	2.61 ± 0.43
penet	8.0	1.36 ± 0.25	1.13 ± 0.03	0.16 ± 0.02	2.03 ± 0.10

Despite of using the same pretreatment and enzymatic hydrolysis condition, the change of the pH overnight before centrifuging resulted in changes in colour. The acid pH corresponded to a golden colour, meanwhile the alkaline conditions resulted in a dark brown colour (*Figure 6.6*).



Figure 6.6 Recovered supernatant after enzymatic hydrolysis of pretreated brewery spent grain using 0.80 % of HCl (ν/ν) at 121.0 °C for 142 minutes with an initial volume of 150 mL of water. The slurry before centrifuging was left overnight at 4.0 °C a pH: 1.0 for the left bottle (golden colour), and 8.0 for the right bottle (dark brown colour)

6.3.4 Large enzymatic hydrolysis

Larger enzymatic hydrolysis experiments (800 mL) were carried out in triplicate to evaluate the amount and composition for the liquid and the solid fraction for C1, C2 and C4. The process depicted in *Figure 6.1* was used for these hydrolysis experiments. The process was initially designed following the previous step by the centrifugation of the slurry instead of using the 100 μ m sieve to recover the liquid and solid fraction. However, during this process there were two main issues: i) some of the solids were floating in the liquid and when using the paper filter this was clogged and no liquid was passing through the paper; and ii) when centrifuging it was possible to observe two different layers of solid material in the centrifuge bottle (Figure 6.7). The texture of the upper layer was similar to mud made by fine particles, and in the bottom layer it was possible to observe the remaining fractions from the grain such as the husk (Figure 6.7e & f). First, for C2 and C4 the final slurry was passed through the sieves from the biggest to the smallest size (250, 100 and 25 μ m). Some of the biggest particles passed through the 250 μ m sieve, and the main fraction with large solid particles was retained in the 100 μ m sieve, in the case of the 25 μ m sieve nothing was retained. Thus, it was decided to use a 100 μ m sieve before centrifuging to split the two layers and recover the two fractions independently. Once the acid liquid with FS was centrifuged, it was possible to recover a pellet with those FS (Figure 6.7h). Additionally, the colour of the fine particles in the upper layer changed based on the diluted acid pretreatment applied. Fine particles were white for the lowest severity factor in C2 and C3 (Figure 6.7c), but got darker as the severity factor increased (Figure 6.7b & h).



Figure 6.7 Centrifuged slurry after the diluted hydrochloric acid pretreatment followed by the enzymatic hydrolysis, corresponding to **a**) C1, **b**) C2, **c**) C3 and **d**) C4. Picture **e**) shows the centrifuged bottle with the mixed slurry, **f**) shows on the left part of the picture the upper layer with the fine solid and on the right the bigger particles with the husk and other fraction of the grains. Picture **g**) is the solid precipitated after the centrifugation of the acid solid with fine particles. Picture **h**) shows the collected fraction of the fine particles after centrifugation

The pH of the recovered acid liquid fraction after centrifugation was then increased up to 4.5 to precipitate some of the proteins and any other compounds that was solubilised under extreme acid conditions. However, in the case of C2, with the highest severity factor, after neutralization and filtration of 559.7 (\pm 23.6) mL, only 4.6 (\pm 1.7) g of solid were collected. When neutralizing the acid liquid obtained using other conditions with a lower severity factor C1 and C4, no solids were precipitated.

As the protein recovery is done mainly in the literature under alkaline conditions (Connolly et al., 2019; He et al., 2021), the CS fraction was re-suspended and the pH was increased to 8.0. The liquid and solid fractions were analysed before and after the pH increase. In C1, the resulting protein content in the liquid fraction increased over 50 % from the re-suspended to the alkaline liquid. In C2, the protein concentration in the liquid fraction almost doubled before and after changing the pH to alkaline conditions (*Table 6.5*). For C4, no major changes in the liquid were found before and after this pH increase. Under the alkaline conditions, the WRC was higher in the case of C1 and C2, decreasing in the case of C4.

When the CS fraction for C2 was re-suspended and the pH increased to 8.0, this resulted in a loss of more than half of the previous weight before alkaline resuspension. The initial solid amount used for this alkaline wash corresponded to 128.8 (\pm 28.2) g, after increasing the pH to 8.0, leaving it



overnight at 4 °C, only 52.3 (\pm 3.0) g of solids were retained in a 25 μ m sieve. The liquid ratio for this case was greater than for C1 and C4 (*Figure 6.8*).

Figure 6.8 a) Water retention capacity for the different solid fractions after the thermal acid diluted and enzymatic hydrolysis of brewery spent grain (10 % w/w) under different conditions (C1, C2 an C3). **b)** Ratio between the liquor and solid fraction after each of the separation steps along the process

Table 6.5 and *Table 6.6* show the summary of the main parameters measured during the large hydrolysis for the liquid and solid fraction, respectively. The largest amount of proteins and ammonium in the liquid fraction corresponded to C2, with the highest severity factor. Correspondingly, all the solid fractions of C2 presented the lowest protein content. The protein concentration in the liquid fraction in C4 was slightly higher than in C1. The WRC for C1 was slightly different between the FSs and CSs, with the highest value for the alkaline solids. C2 had the lowest WRC for all the different solid fractions, meanwhile C4 presented the highest value among the conditions for the fine and course solids with similar values (*Figure 6.8a*). Additionally, due to the different conditions. For example, in C4 extra samples were taken for the solids. To compare the different conditions between them, the ratio between the amount of solids and liquid was calculated for each of the separation steps, i.e. sieve at 100 μm, centrifugation, sieve at 25 μm, as well the overall solid extracted from the slurry (coarse plus fine solids) and the acid liquid fraction (*Table 6.5b*).

Liquid	Pre	e-treated liqu	uid	Hydrolysed slurry			Acid liquid		
fraction	C1	C2	C4	C4	C4	C4	C1	C2	С3
Amount (mL)	NA	NA	NA	NA	NA	NA	496.3 ± 25.1	560.0 ± 23.8	486.7 ± 20.8
Total Protein (g/L)	7.6 ± 0.5	19.0 ± 0.2	9.2 ± 0.3	8.1 ± 0.4	17.5 ± 0.3	9.6 ± 0.6	8.1 ± 0.5		10.3 ± 0.5
Ammonium (mg/L)	124.7 ± 5.9	190.1 ± 1.3	130.3 ± 9.2	115.8 ± 6.2	148.1 ± 4.5	124.4 ± 8.0	115.8 ± 7.7		127.6 ± 5.6
Liquid		Final		Re-s	uspended lic	quid	A	Alkaline liqui	d
Liquid fraction	C1	Final C2	C4	Re-s C1	uspended lic C2	quid C4	۹ C1	Alkaline liquio C2	d C3
Liquid fraction Amount (mL)	C1	Final C2	C4	Re-s C1 680.33	uspended lic C2 636.50	Quid C4 659.72	۹ C1 510.3 ± 30.1	C2 542.0 ± 45.3	d C3 501.7 ± 27.5
Liquid fraction Amount (mL) Total Protein (g/L)	C1 7.7±0.5	Final C2 16.4 ± 0.3	C4 9.0±0.1	Re-s C1 680.33 1.4 ± 0.2	2.4 ± 0.6	2.6 ± 0.3	C1 510.3 ± 30.1 2.0 ± 0.4	Alkaline liquid C2 542.0 ± 45.3 4.6 ± 0.5	C3 501.7 ± 27.5 2.6 ± 0.4

Table 6.5 Liquid characterisation for each of the fractions produced during the fractionation and different streams for brewery spent grain cascade process

Table 6.6 Solid fraction characterisation during the fractionation and different streams for brewery spent grain cascade process

	Fine solids			Coarse solids			Alkaline solids		
	C1	C2	C4	C1	C2	C4	C1	C2	C4
Amount (g)	103.1 ± 17.0	91.6 ± 7.4	59.4 ± 10.6	205.7 ± 4.1 *181 ± 2.6	128.8 ± 28.2 *105.7 ± 18.4	208.1 ± 8.4 *185.1 ± 6.2	154.5 ± 10.9	52.2 ± 3.0	119.6 ± 10.9
TS (%)	16.5 ± 1.0	18.1 ± 0.8	7.1 ± 0.9	13.0 ± 1.7	20.5 ± 3.4	7.0 ± 0.6	10.6 ± 2.34	14.2 ± 2.7	12.6 ± 0.5
Proteins (%)	26.2 ± 2.2	18.1 ± 0.8	27.3 ± 0.5	27.7 ± 3.0	11.5 ± 0.8	15.7 ± 1.2	14.0 ± 0.3	8.7 ± 0.3	12.6 ± 0.6
WRC (Gh ₂ O/ Gts)	5.1 ± 0.4	4.5 ± 0.2	13.3 ± 1.8	6.8 ± 1.0	4.0 ± 0.8	13.4 ± 1.2	8.8 ± 2.4	6.2 ± 1.4	7.0 ± 0.4
Ratio liquid/solid (MI/g)	4.9 ± 1.0	6.1 ± 0.5	8.3 ± 1.1	2.9 ± 0.1	5.5 ± 1.5	2.6 ± 0.25	3.3 ± 0.4	10.4 ± 0.3	4.2 ± 0.5

6.4 Discussion

6.4.1 Effect of pretreatment

The first part of this work compared the use of sulfuric or hydrochloric acid for the thermal diluted acid hydrolysis of the BSG. Sulfuric acid is the most commonly used acid for pretreatment of lignocellulosic material (Yoon et al., 2014; Zoulikha et al., 2015; Plaza et al., 2017; Tizazu and Moholkar, 2017; Rojas-Chamorro et al., 2020). However, hydrochloric acid was tested here as it is a food grade quality chemical and target chemical compounds for this study are potentially for use in the food or cosmetics industry. The use of hydrochloric acid resulted in a more efficient release of carbohydrate and protein, which is in agreement with previous studies (White et al., 2008). The hydrochloric ions have been suggested to be responsible for disrupting hydrogen bonds in lignocellulose (Wilkinson et al., 2014).

When the acid concentration was increased from 0.10 to 0.25 % (v/v), the colour of the solids and the liquid phase changed (Figure 6.4). The darker colour in the solid fraction could be a result of the production of humins (Patil and Lund, 2011). However, aside from C2 the severity factor was not high enough to promote the kinetic reaction when using 1.0 % hydrochloric acid (v/v) or less. A similar colour change was observed during the fractionation of the hydrolysates after enzymatic hydrolysis using larger volumes (*Figure 6.7*). In all these cases, the colour change in the liquid could be a result of the release of polyphenols or other compounds from the matrix of the BSG. In addition, the release of these compounds could be related to the darkening of the solid fraction due to the Maillard reaction resulting in formation of melanoids, a coloured polymeric fraction with important antioxidant and probiotic activities (Patrignani and Gonzalez-Forte, 2021). This causes a reduction of the crystallinity and a loss of the proteinic structure, which facilitates the binding of phenolic compounds to the carbohydrate oligomers (Hellebois et al., 2021; Patrignani and Gonzalez-Forte, 2021). Alternatively, the phenolic compounds and the proteins could interact to develop phenolic-protein conjugates with antioxidant and antimicrobial activity (Liu et al., 2019), making them a good resource for the pharmaceutical or food industry (Buitimea-Cantua et al., 2018). Some of these reactions also take place during the malting and mashing processes in the brewery (Celus et al., 2006). Thus, the use of mild thermal diluted acid pretreatment could create a similar relative structure to that which is formed during the roasting of barley. For example, at higher severity factors the colour of the liquid and FS turned from white to light-brownish, and finally dark brown (Figure 6.4 and Figure 6.7).

6.4.2 Enzymatic hydrolysis

The combination of parameters for pretreatment was optimised using RSM. This was to identify different conditions that could lead to the most efficient enzymatic hydrolysis. To do this, different parameters were chosen as indicators for the efficiency of the hydrolysis. For example, the reduction of the hemicellulose was followed by the release of the xylose and arabinose. However, as no degradation of pentose was detected under any of the conditions, the resulting combination was with the maximum parameters given to the software. In the same way, the optimal combination for glucose or total carbohydrate release corresponded to the combination with the highest acid concentration, temperature and time used during this part of the trial (1.00 % v/v of acid, 121 °C and 180 min). These conditions had a severity factor of 2.07, higher than when using the 1.5 % hydrochloric acid at 121.0 °C for 20 min. Furthermore, increasing then the acid concentration from 1.0 to 1.5 % resulted in a very small increase of the amount of carbohydrates released (**Figure 6.2**). Thus, these parameters were discarded as pretreatment to evaluate the degradation of the lignocellulose structure during the enzymatic hydrolysis.

For all the pretreatment combinations, the enzymatic hydrolysis was carried using the WS or the RP. The highest carbohydrate release during the enzymatic hydrolysis was achieved using the WS for C1 and C4, which was approximately 40 and 80 % higher than RS. This could be explained by the washout of some solubilised or suspended carbohydrates in the liquid fraction, such as colloids of small fractions of hemicellulose or starch (Hubbe and Rojas, 2008; Wilkinson et al., 2016; He et al., 2019; Wagner et al., 2021). The results of the present work differ from some previous publications that attribute the lower efficiency of the enzymatic hydrolysis to the release of carbohydrates and inhibitory compounds in the broth when using thermal acid hydrolysis (Yoon et al., 2014; Plaza et al., 2017; Wagner et al., 2021). However, these publications carried out the hydrolysis using a higher acid concentration. For example, using 2% sulfuric acid (v/v) for 15 min at 121 °C as a pretreatment showed the lowest glucose extraction during enzymatic hydrolysis, extracting only 37.26 mg of glucose per dry gram of BSG (Wagner et al., 2021). For WS in the present work, this efficiency in C1 and C4 corresponded to 125.61 and 106.87 mg glucose per dry gram of BSG, respectively. Plaza et al. (2017) carried out a pretreatment of BSG by decreasing the pH to 1.0 using sulfuric acid at 121 °C for 20 min, tested after the enzymatic hydrolysis using the WS and RP. They obtained a higher extraction efficiency for glucose using the RP, meanwhile the WS resulted in a higher release of pentose (xylose and arabinose) (Plaza et al., 2017).

In C2 the enzymatic hydrolysis efficiency decreased for the WS and RP. As C2 had the highest SF, the poor enzymatic hydrolysis performance may have been related with the presence of inhibitory compounds such as 5-hydroxymethilfurfural, levulonic acid or humins (Patil and Lund, 2011; Yoon et al., 2014; Plaza et al., 2017). Tizazu and Moholkar (2017) studied the degradation of monosaccharides and production of inhibitory compounds using 33 % (w/v) sugarcane bagasse at 2 % sulfuric acid (v/v) with different temperatures (100 to 130 °C) and time (0 to 120 min). In their work, even without degradation of xylose, glucose or arabinose, there was a small production of inhibitory compounds (Tizazu and Moholkar, 2017). In the case of C2 during the present work, the solid concentration and acid concentration were much lower and as acid has the most significant effect during the pretreatment hydrolysis (Zoulikha et al., 2015), the production of inhibitory compounds would have been lower. Besides, the efficiency of the enzymatic hydrolysis did not improve for RS which would have had some of these compounds washed away and the total carbohydrates concentration at the beginning of the enzymatic hydrolysis was smaller than the WS for C1 and C2 (Figure 6.5). The other possibility for the limited enzymatic hydrolysis in C2 could be related with the complete degradation of hemicellulose, resulting in an increase of the cellulose content giving rise to interactions between microfibrils and lignin, thus limiting the enzyme accessibility (Wagner et al., 2021). Even more, if the solid contains a larger lignin fraction, this could be adsorbed by the enzymes in non-productive binding

having a higher adsorption capacity when operating at lower pH (Pareek et al., 2013; Rahikainen et al., 2013). This effect increased when the enzymatic hydrolysis is carried out at 50 °C (Mou et al., 2021) with the higher reduction of the efficiency activity for xylanase and cellulase, whereas ß-glucosidase was the less affected enzyme (Berlin et al., 2006). This could explain the limited enzymatic efficiency for C2 with mainly only cellobiose released.

Based on the results for the xylose and monosaccharides release during the enzymatic hydrolysis for the different conditions, the pretreatments C1, C3 and C4 were effective (*Table 6.3*). However, the most efficient hydrolysis performance was under C1 and C4 with a release of 234.4 (± 39.3) and 225.04 (± 81.7) mg carbohydrates per dry gram of BSG. Only C2 showed a poor enzymatic hydrolysis in WS and RS. Although the efficiency of C2 pretreatment corresponded to 486.9 (± 24.2) mg carbohydrate per dry gram of BSG, the slight release of glucose and cellobiose after 6 hours of enzymatic hydrolysis increased the efficiency up to 535.8 (± 28.7) mg carbohydrate per dry gram of BSG. For all the other conditions, the enzymatic hydrolysis of WS performed better than RP (*Figure 6.5*). C3 was considered non-effective as the hydrolysis efficiency (*Table 6.3*), carbohydrate concentration and the liquid recovered were lower than the other conditions. Only between 13.95 and 33.6 % of the total water content in the slurry was recovered as supernatant after centrifugation. It has been reported that the chemical hydrolysis is largely increased when the pH is below 2.0 (Wilkinson and Cook, 2014), which could explain the low efficiency of C3. For these reasons, and in view of the main objective to produce a liquid fraction rich in carbohydrates, C3 was not tested at larger volume.

6.4.3 Solid fraction

For all the different pretreatment conditions it was possible to observe two layers when centrifuging, the FS on top and the CS on the bottom fraction (*Figure 6.4* and *Figure 6.7*). These two different layers with fine and course solids were also observed previously by lbbett et al. (2019), who were able to obtain a FS with a protein concentration up to 44.9 (\pm 1.4) % by an improved milling process during the absence of chemical or enzymatic hydrolysis (lbbett et al., 2019) (this concentration was corrected to the factor 5.50 used for the Kjeldahl method (Mariotti et al., 2008)).

Despite the higher concentration of the acid used during the pretreatment step in C1 than in C4, the use of a lower temperature could have led in the uncomplete degradation of the hemicellulose. Thus, the proteins can interact with the arabynoxylans, or some other hydrolysed oligomer residuals, which led in the aggregation of the proteins in the CS fraction (Hellebois et al., 2021), which would explain the largest protein content in the CS for C2 (*Figure 6.8*). Similarly, the higher temperature of the pretreatment in C2 than in C4 would have resulted in a larger degradation of the pectin-rich fractions, and set the crystallinity in a new stable conformation where proteinic structures are

significantly reduced (Ouhida et al., 2002; Patrignani and Gonzalez-Forte, 2021). As well, this would have had a great impact on the degradation of hemicellulose which protects proteins inside of the BSG. Thus, increasing the severity factor resulted in a higher glucose and protein concentration in the liquid fraction, as [C2] > [C4] > [C1] (*Figure 6.8*).

The lower acid concentration and time in C4 than C1 could have resulted in a higher release of the proteins linked with the lignocellulose fraction, but the reduction in hydrolysis time would have prevented the aggregation of the proteins with the CSs. For example, in C1 a lower pectin degradation could have bound a larger amount of proteins (Ouhida et al., 2002), increasing the protein content in the CS. Meanwhile in the case of C2 as the severity factor was higher, the hemicellulose was fully degraded, explaining the lower content of proteins and limited enzymatic hydrolysis (Hellebois et al., 2021; Wagner et al., 2021). In the case of C2, not only the hemicellulose would have been degraded, but also some fraction of the proteins into peptides. When operating with a higher acid concentration, this degradation of the proteins to peptide is much larger (Kemppainen et al., 2016). In *Chapter 4*, the diluted acid hydrolysis was carried out using a lower concentration of BSG (7 % instead of 10 % w/w), and 1.5 % instead of 0.80 % sulfuric acid (v/v). Therefore, in *Chapter 4* it would have been possible that a larger fraction of the proteins was degraded to peptides and became soluble at a pH lower than 1.0. When the acid liquid hydrolysate was extracted and then neutralised, the protein content corresponded to 32.6 % of proteins above the total solids, although the non-volatile solid fraction corresponded the 47 % of the total solids due to the large amount of acid and base used.

The resuspension of the CS in alkaline liquid was intended to recover further compounds that could be solubilised under alkaline pH. This is the most common approach for protein recovery. The largest solubilisation of solids when CS were re-suspended at alkaline pH corresponded to C2, which enhanced the protein concentration in the liquid fraction 2 fold. Hellebois et al. (2021) studied this effect determining that the heat pretreatment led to significantly higher active glutelin network chains. Similarly, the higher SF in C2 compared with C4 would have resulted in a higher cellulose degradation and lignin enrichment. The hydrophobicity of proteins increases when the pH decreases (Berlin et al., 2006; Mou et al., 2021), which, with the electrostatic force would have contributed to the protein absorption on the lignin by hydrogen bonds and other non-covalent forces (Pareek et al., 2013; Rahikainen et al., 2013). Therefore, it would have been possible that at pH 1.0, the proteins and other oligomers interacted to form aggregates with a size larger than 100 µm, which would have been disintegrated by increasing the pH from 1.0 to 8.0. As well, this loss of hydrophobicity by increasing the pH could explain the higher WRC in C2 for the alkaline solid fraction.

The WRC has a key role in the access of microorganisms and enzymes to degrade lignocellulose (Raghavendra et al., 2004; Jeihanipour et al., 2010; Guo et al., 2018; Oliva et al., 2020). However, the absorbed water in C3 was higher than for the other conditions (Figure 6.4), whereas the enzymatic hydrolysis efficiency was limited with a lower amount of carbohydrate released (*Table 6.3*). This effect is related with the initial breaking down of the crystallinity structure of the cellulose (Mosier et al., 2005; López-Linares et al., 2020). Thus, a low degradation of this structure would have resulted in the higher WRC, as more water was absorbed. Further degradation of the cellulose and hemicellulose structure would have resulted in the solubilisation of these structures decreasing the WRC and increasing the liquor recovered from the hydrolysate. This was easily observed in the initial thermal diluted acid hydrolysis when comparing hydrochloric and sulfuric acid (Figure 6.3). For the sulfuric acid, the increase of the acid concentration up to 0.10 % (v/v) resulted in a higher WRC but without carbohydrate degradation. When the acid concentration was increased to 0.25 and 0.50 % the WRC decreased and the recovered liquid was higher (Figure 6.2). This behaviour corresponded as well with the degradation of glucose and the highest degradation of arabinose and xylose. The different solid fractions obtained could be valorised as food supplement. For example, the WRC has been related with nutritional properties (Raghavendra et al., 2004; MartÍnez et al., 2012). In the case of the FS for C4, the WRC observed (Figure 6.8) was higher than many of the current sources of fibres from the food industry (Raghavendra et al., 2004).

6.5 Conclusion

A cascade process approach using BSG was followed to obtain a liquid hydrolysate rich in carbohydrates and the production of two different solid streams (fine and coarse solids). The use of a mild pretreatment and the enzymatic hydrolysis of the WS resulted in a higher carbohydrate efficiency recovery for C1, C2 and C4, corresponding to 399.1 (± 26.1), 535.8 (± 28.7), 257.0 (± 11.5) and 446.3 (± 81.1) mg carbohydrate per dry gram of BSG respectively. The WRC of the solids is equal to or higher than some of the current sources of fibres for the food industry. The quality of proteins enriched in the FSs should be evaluated. As well, the possible enrichment of the solid fractions with phenolic compounds or lignin was related with the change of colour by production of melanoids or phenolic-protein conjugates. The combination of these products could result in a food supplement with a high potential in antioxidant and prebiotic activity. Finally, the liquid hydrolysate rich in carbohydrates is an ideal candidate for the production of the VFAs.

6.6 References

APHA (2012) "Standard Methods for Examination of Water and Wastewater 2012" American Public Health Association.

- Berlin A., Balakshin M, Gilkes N., Kadla J., Maximenko V., Kubo S., Saddler J. (2006) "Inhibition of
cellulase, xylanase and β-glucosidase activities by softwood lignin preparations." Journal of
BiotechnologyBiotechnology125https://doi.org/https://doi.org/10.1016/j.jbiotec.2006.02.021.
- Buitimea-Cantúa, Nydia E., Janet A. Gutiérrez-Uribe, and Sergio O. Serna-Saldívar. 2017. "Phenolic– Protein Interactions: Effects on Food Properties and Health Benefits." *Journal of Medicinal Food* 21 (2): 188-198. https://doi.org/10.1089/jmf.2017.0057.
- Carvalheiro F., Duarte L. C., Medeiros R., Gírio F. M. (2004) "Optimization of Brewery's spent grain dilute-acid hydrolysis for the production of pentose-rich culture media." Applied Biochemistry and Biotechnology 115 (1-3): 1059-1072. https://doi.org/10.1385/ABAB:115:1-3:1059.
- Celiktas, Melih Soner, Christian Kirsch, and Irina Smirnova. 2014. "Cascade processing of wheat bran through a biorefinery approach." *Energy Conversion and Management* 84: 633-639. https://doi.org/https://doi.org/10.1016/j.enconman.2014.04.039.
- Celus I., Brijs K., Delcour J. (2006) "The effect of malting and mashing on barley protein extractability." Journal of Cereal Science 44: 203-211. https://doi.org/10.1016/j.jcs.2006.06.003.
- Cermeño M., Connolly A., O'Keeffe M. B., Flynn C., Alashi A. M., Aluko R. E., FitzGerald R. J. (2019) "Identification of bioactive peptides from brewers' spent grain and contribution of Leu/Ile to bioactive potency." Journal of Functional Foods 60: 103455. https://doi.org/10.1016/j.jff.2019.103455.
- Contreras M. M., Lama-Muñoz A., Gutiérrez-Pérez J. M., Espínola F., Moya M., Castro E. (2019) "Protein extraction from agri-food residues for integration in biorefinery: Potential techniques and current status." Bioresource Technology 280: 459-477. https://doi.org/https://doi.org/10.1016/j.biortech.2019.02.040.
- Connolly A., Cermeño M., Crowley D., O'Callaghan Y., O'Brien N. M., FitzGerald R. J. (2019) "Characterisation of the in vitro bioactive properties of alkaline and enzyme extracted brewers' spent grain protein hydrolysates." Food Research International 121: 524-532. https://doi.org/10.1016/j.foodres.2018.12.008.
- Djioleu A., Carrier D. J. (2016) "Effects of Dilute Acid Pretreatment Parameters on Sugar Production during Biochemical Conversion of Switchgrass Using a Full Factorial Design" J. ACS Sustainable Chemistry & Engineering. 4 (8) 4124-4130. https://doi.org/10.1021/acssuschemeng.6b00441
- Guo Z., Ling Z., Wang C., Zhang X., Xu F. (2018) "Integration of facile deep eutectic solvents pretreatment for enhanced enzymatic hydrolysis and lignin valorization from industrial xylose residue." *Bioresource Technology* 265: 334-339. https://doi.org/https://doi.org/10.1016/j.biortech.2018.06.027.
- He Y., Kuhn D. D., Ogejo J. A., O'Keefe S. F., Fraguas C. F., Wiersema B. D., Jin Q., Yu D., Huang H. (2019)
 "Wet fractionation process to produce high protein and high fiber products from brewer's spent grain." *Food and Bioproducts Processing* 117: 266-274. https://doi.org/https://doi.org/10.1016/j.fbp.2019.07.011.
- He Y., Kuhn D. D., O'Keefe S. F., Ogejo J. A., Fraguas C. F., Wang H., Huang H. (2021) "Protein production from brewer's spent grain via wet fractionation: process optimization and techno-economic analysis." *Food and Bioproducts Processing* 126: 234-244. https://doi.org/https://doi.org/10.1016/j.fbp.2021.01.005.

- Hellebois T., Gaiani C., Planchon S., Renaut J., Soukoulis C. (2021) "Impact of heat treatment on the acid induced gelation of brewers' spent grain protein isolate." *Food Hydrocolloids* 113: 106531. https://doi.org/https://doi.org/10.1016/j.foodhyd.2020.106531.
- Hubbe, Martin, and Orlando Rojas. 2008. "Colloidal stability and aggregation of Lignocellulosic materials in aqueous suspension: A review." BioResources 3.
- Ibbett, Roger, Roderick White, Greg Tucker, and Tim Foster. 2019. "Hydro-mechanical processing of brewer's spent grain as a novel route for separation of protein products with differentiated techno-functional properties." *Innovative Food Science & Emerging Technologies* 56: 102184. https://doi.org/https://doi.org/10.1016/j.ifset.2019.102184.
- Jaeger, Alice, Emanuele Zannini, Aylin W. Sahin, and Elke K. Arendt. 2021. "Barley Protein Properties, Extraction and Applications, with a Focus on Brewers' Spent Grain Protein." Foods 10 (6). https://doi.org/10.3390/foods10061389.
- Jeihanipour, Azam, Keikhosro Karimi, and Mohammad J. Taherzadeh. 2010. "Enhancement of ethanol and biogas production from high-crystalline cellulose by different modes of NMO pretreatment." *Biotechnology and Bioengineering* 105 (3): 469-476. https://doi.org/https://doi.org/10.1002/bit.22558.
- Kemppainen K., Rommi K., Holopainen U., Kruus K. (2016) "Steam explosion of Brewer's spent grain improves enzymatic digestibility of carbohydrates and affects solubility and stability of proteins" Applied Biochemistry and Biotechnology. 180(1) 94-108. DOI 10.1007/s12010-016-2085-9
- Laguna, Oscar, Abdellatif Barakat, Hadil Alhamada, Erwann Durand, Bruno Baréa, Frédéric Fine, Pierre Villeneuve, Morgane Citeau, Sylvie Dauguet, and Jérôme Lecomte. 2018. "Production of proteins and phenolic compounds enriched fractions from rapeseed and sunflower meals by dry fractionation processes." Industrial Crops and Products 118: 160-172. https://doi.org/https://doi.org/10.1016/j.indcrop.2018.03.045.
- Liu N., Chen Q., Li G., Zhu Z., Yi J., Li C., Chen X., Wang Y. (2018) "Properties and Stability of Perilla Seed Protein-Stabilized Oil-in-Water Emulsions: Influence of Protein Concentration, pH, NaCl Concentration and Thermal Treatment" J. Molecules. 23 (7) 1533. doi: 10.3390/molecules23071533
- López-Linares, Juan C., M. Teresa García-Cubero, Susana Lucas, and Mónica Coca. 2020. "Integral valorization of cellulosic and hemicellulosic sugars for biobutanol production: ABE fermentation of the whole slurry from microwave pretreated brewer's spent grain." Biomass and Bioenergy 135: 105524. https://doi.org/10.1016/j.biombioe.2020.105524.
- Lynch K. M., Steffen E. J., Arendt E. K. (2016) "Brewers' spent grain: a review with an emphasis on food and health." Journal of the Institute of Brewing 122 (4): 553-568. https://doi.org/https://doi.org/10.1002/jib.363. https://doi.org/10.1002/jib.363.
- Mariotti, François, Daniel Tomé, and Philippe Patureau Mirand. 2008. "Converting Nitrogen into Protein—Beyond 6.25 and Jones' Factors." *Critical Reviews in Food Science and Nutrition* 48 (2): 177-184. https://doi.org/10.1080/10408390701279749.
- Martínez, R., P. Torres, M. A. Meneses, J. G. Figueroa, J. A. Pérez-Álvarez, and M. Viuda-Martos. 2012. "Chemical, technological and in vitro antioxidant properties of cocoa (Theobroma cacao L.)

co-products." Food Research International 49 (1): 39-45. https://doi.org/https://doi.org/10.1016/j.foodres.2012.08.005.

- McCarthy A., O'Callaghan Y. C., Connolly A., Piggott C. O., FitzGerald R. J., O'Brien N. M. (2012)
 "Phenolic extracts of brewers' spent grain (BSG) as functional ingredients Assessment of their DNA protective effect against oxidant-induced DNA single strand breaks in U937 cells."
 Food Chemistry 134 (2): 641-646. https://doi.org/10.1016/j.foodchem.2012.02.133.
- Mosier N., Wyman C., Dale B., Elander R., Lee Y. Y., Holtzapple M., Ladisch M. (2005) "Features of promising technologies for pretreatment of lignocellulosic biomass." Bioresource Technology 96 (6): 673-686. https://doi.org/10.1016/j.biortech.2004.06.025.
- Mou, Hongyan, Xiao Wu, Jin Huang, Yibei Liu, and Huiming Fan. 2021. "Eucalyptus lignin modification
for dynamic adsorption with lignocellulose-degradation enzymes dependent on pH values."
Industrial Crops and Products 169: 113650.
https://doi.org/https://doi.org/10.1016/j.indcrop.2021.113650.
- Mussatto S. I., Dragone G, Roberto I. C. (2006) "Brewers' spent grain: generation, characteristics and potential applications." Journal of Cereal Science 43 (1): 1-14. http://dx.doi.org/10.1016/j.jcs.2005.06.001.
- Niemi P., Martins D., Buchert J., Faulds C. B. (2013) "Pre-hydrolysis with carbohydrases facilitates the release of protein from brewer's spent grain." Bioresource Technology 136 (Supplement C): 529-534. https://doi.org/10.1016/j.biortech.2013.03.076.
- Niemi P., Tamminen T., Smeds A., Viljanen K., Ohra-aho T., Holopainen-Mantila U., Faulds C. B., Poutanen K., Buchert J. (2012) "Characterization of Lipids and Lignans in Brewer's Spent Grain and Its Enzymatically Extracted Fraction." Journal of Agricultural and Food Chemistry 60 (39): 9910-9917. https://doi.org/10.1021/jf302684x.
- Oliva, A., L. C. Tan, S. Papirio, G. Esposito, and P. N. L. Lens. 2020. "Effect of methanol-organosolv pretreatment on anaerobic digestion of lignocellulosic materials." Renewable Energy. https://doi.org/10.1016/j.renene.2020.12.095.
- Ouhida I., Pérez J. F., Gasa J. (2002) "Soybean (Glycine max) Cell Wall Composition and Availability to Feed Enzymes." *Journal of Agricultural and Food Chemistry* 50 (7): 1933-1938. https://doi.org/10.1021/jf010686u.
- Pareek, Nidhi, Thomas Gillgren, and Leif J. Jönsson. 2013. "Adsorption of proteins involved in hydrolysis of lignocellulose on lignins and hemicelluloses." *Bioresource Technology* 148: 70-77. https://doi.org/https://doi.org/10.1016/j.biortech.2013.08.121.
- Patel A., Mikes F., Bühler S., Matsakas L. (2018) "Valorization of Brewers' Spent Grain for the Production of Lipids by Oleaginous Yeast." *Molecules (Basel, Switzerland)* 23 (12): 3052. https://doi.org/10.3390/molecules23123052. https://pubmed.ncbi.nlm.nih.gov/30469531.
- Patil S. K. R., Lund C. R. F. (2011) "Formation and Growth of Humins via Aldol Addition and Condensation during Acid-Catalyzed Conversion of 5-Hydroxymethylfurfural." *Energy & Fuels* 25 (10): 4745-4755. https://doi.org/10.1021/ef2010157.
- Patrignani M., González-Forte L. S. (2021) "Characterisation of melanoidins derived from Brewers' spent grain: new insights into their structure and antioxidant activity." International Journal

of Food Science & Technology 56 (1): 384-391. https://doi.org/https://doi.org/10.1111/ijfs.14653. https://doi.org/10.1111/ijfs.14653.

- Plaza P. E., Gallego-Morales L. J., Peñuela-Vásquez M., Lucas S., García-Cubero M. T., Coca M. (2017)
 "Biobutanol production from brewer's spent grain hydrolysates by Clostridium beijerinckii." Bioresource Technology 244: 166-174. https://doi.org/10.1016/j.biortech.2017.07.139.
- Raghavendra, S. N., N. K. Rastogi, K. S. M. S. Raghavarao, and R. N. Tharanathan. 2004. "Dietary fiber from coconut residue: effects of different treatments and particle size on the hydration properties." European Food Research and Technology 218 (6): 563-567. https://doi.org/10.1007/s00217-004-0889-2.
- Ravindran R., Jaiswal R., Abu-Ghannam N., Jaiswal A. K. (2018) "A comparative analysis of pretreatment strategies on the properties and hydrolysis of brewers' spent grain." Bioresource Technology 248: 272-279. https://doi.org/10.1016/j.biortech.2017.06.039.
- Rahikainen J. L., Evans J. D., Mikander S., Kalliola A., Puranen T., Tamminen T., Marjamaa K., Kruus K. (2013) "Cellulase–lignin interactions—The role of carbohydrate-binding module and pH in non-productive binding." *Enzyme and Microbial Technology* 53 (5): 315-321. https://doi.org/https://doi.org/10.1016/j.enzmictec.2013.07.003.
- Robertson J. A., de Monredon F. D., Dysseler P., Guillon F., Amado R., Thibault J.-F. (2000) "Hydration Properties of Dietary Fibre and Resistant Starch: a European Collaborative Study." LWT - Food Science and Technology 33 (2): 72-79. https://doi.org/10.1006/fstl.1999.0595.
- Rommi K., Niemi P., Kemppainen P., Kruus K. (2018) "Impact of thermochemical pre-treatment and carbohydrate and protein hydrolyzing enzyme treatment on fractionation of protein and lignin from brewer's spent grain." Journal of Cereal Science 79: 168-173. https://doi.org/10.1016/j.jcs.2017.10.005.
- Tizazu B. Z., Moholkar V. S. (2017) "Kinetic and Thermodynamic Analysis of Dilute Acid Hydrolysis of Sugarcane bagasse." Bioresource Technology. https://doi.org/10.1016/j.biortech.2017.11.032. Total-Corbion, 03/12/2018, 2018, "Total Corbion PLA starts-up its 75,000 tons per year bioplastics plant," https://www.totalcorbion.com/news/total-corbion-pla-starts-up-its-75-000-tons-per-year-bioplasticsplant/?p=1&q=2018.
- Yoon, S. Y., S. H. Han, and S. J. Shin. 2014. "The effect of hemicelluloses and lignin on acid hydrolysis of cellulose." Energy 77: 19-24. https://doi.org/10.1016/j.energy.2014.01.104.
- Vieira, Elsa, M. Angélica M. Rocha, Elisabete Coelho, Olívia Pinho, Jorge A. Saraiva, Isabel M. P. L. V. O. Ferreira, and Manuel A. Coimbra. 2014. "Valuation of brewer's spent grain using a fully recyclable integrated process for extraction of proteins and arabinoxylans." Industrial Crops and Products 52 (Supplement C): 136-143. https://doi.org/https://doi.org/10.1016/j.indcrop.2013.10.012.
- Wilkinson S., Smart K. A., Cook D. J. (2014) "Optimisation of alkaline reagent based chemical pretreatment of Brewers spent grains for bioethanol production." Industrial Crops and Products 62: 219-227. https://doi.org/https://doi.org/10.1016/j.indcrop.2014.08.036.
- Wilkinson S., Smart K. A., James S., Cook D. J. (2016) "Maximising high solid loading enzymatic saccharification yield from acid-catalysed hydrothermally-pretreated brewers spent grain." *Biofuel Research Journal* 3 (2): 417-429. https://doi.org/10.18331/BRJ2016.3.2.7.

- White, Jane S., Biju K. Yohannan, and Graeme M. Walker. 2008. "Bioconversion of brewer's spent
grains to bioethanol." *FEMS Yeast Research* 8 (7): 1175-1184.
https://doi.org/https://doi.org/10.1111/j.1567-1364.2008.00390.x.
- Zoulikha M.-R., Thierry M., Qiuyu Z J.-M., Nouviaire A., Sid-Ahmed R. (2015) "Combined steamexplosion toward vacuum and dilute-acid spraying of wheat straw. Impact of severity factor on enzymatic hydrolysis." Renewable Energy 78: 516-526. https://doi.org/10.1016/j.renene.2015.01.038.

Chapter 7

Continuous volatile fatty acid production and recovery at high organic concentration using an upflow granular sludge bed reactor

Keywords: glucose; expanded granular sludge bed; high strength feedstock; organic loading rate; acidogenesis; volatile fatty acids; electrochemical cell

Abstract

A novel approach for volatile fatty acid production and recovery was developed by coupling an electrochemical cell to the recycling line of an expanded granular sludge bed reactor. The reactor was fed with glucose (from 15 g/L up to 60 g/L, pH 5.7 at 37 °C) to improve the VFA production as no hydrolysis step is required. A synthetic broth was initially used to characterise the VFA rate transfer in the electrochemical cell. Different voltages were applied (1.5, 2.0, 2.5, 3.0 and 3.5 V) between the cathode and the anode chamber. The lowest voltages (1.5 and 2.0) were ineffective as the current dropped to 0 by day 2. In the case of the higher voltages, the best recovery rate was at 3.5 V. In the first phase, 15 g/L of glucose was used to feed the UASB reactor with a 36 h hydraulic retention time (HRT). For the next phase the HRT was reduced to 24 h, and this was kept constant throughout the trial. The coupling of the electrochemical cell to the UASB reactor was done when the feed glucose concentration corresponded to 15 of g/L. This resulted in the removal of the VFAs from a total concentration of 6.72 (± 1.16) g/L to 3.65 (± 0.45) g/L. As well, the sodium hydroxide consumption decreased by about 49 %. The concentration of VFAs in the anode chamber was enhanced up to 4 fold in comparison with the VFA concentration in the fermentation broth. The electrochemical cell was under dimensioned for the reactor size, thus when the OLR was increased above 30 g/(L·day) the flow of fresh water continuously added to the anode chamber was increased in order to avoid any possible saturation of the VFAs. This led to dilution of the VFA recovery solution. The highest concentration of VFAs in the reactor corresponded to 32 g/L and was achieved when using 60 g/L of glucose as feedstock. Interestingly, the largest caproic acid production achieved was 6.01 g/L when the system operated with a glucose feeding of 30 g/L.
7.1 Introduction

In the last decades the production of hydrogen from organic wastes is gaining more attention, often coupled to the simultaneous production of volatile fatty acids (VFAs) as by-products. These are valuable chemicals, currently obtained from fossil fuel sources but which could be produced from renewable feedstock. However, renewable VFA production has not yet been developed at industrial level (Strazzera et al., 2018). The main bottleneck is the rather low concentration of VFAs obtained in the acidogenic bioreactor fermentation broth, which makes their recovery difficult. Currently, extraction systems are capable of recovering VFAs in systems that operate at a pH lower than 5.0 or with VFA concentrations in the range of 20 – 50 gl/L (Saboe et al., 2018; Wainaina et al., 2019). However, such high concentrations in the reactor could lead to the accumulation of the VFAs in undissociated forms, which are toxic for the microorganisms (Kaur et al., 2020). Thus, some research focuses on the chain elongation of VFAs for production of e.g. caproic acid, which is easier to recover (Spirito et al., 2014; Xu et al., 2020). Or on the use of VFAs as intermediate compound for production of other chemicals, such a biopolymers, e.g. polyhydroxyalkanoates (Albuquerque et al., 2011).

An alternative process to recover VFAs as final product is the electrodialysis system. This combines a membrane separation system for the VFA recovery using an electrochemical cell. The recovery of VFAs is driven by the different voltages applied between the cathode and the anode (Bak et al., 2019). The protons are reduced at the cathode, and the VFAs migrate to the anode through the anion exchange membrane (AEM) (Zhang and Angelidaki, 2015). Despite obtaining a higher VFA concentration, the downstream processing for the purification step constitutes up to 35 % of the total costs (Baroi et al., 2017). One additional advantage of this approach is the production of OH⁻, which contributes to neutralizing the pH and decreasing the chemical cost for maintaining a stable pH (Zhang and Angelidaki, 2015; Xu et al., 2015).

The aim of this chapter is to achieve high levels of VFA production by using a synthetic high strength feedstock with carbohydrate concentrations from 15 to 60 g/L. The feedstock was fermented at low retention time with anaerobic granular sludge to produce VFAs under acidogenic conditions using an upflow anaerobic sludge bed reactor (UASB). In *Chapter 5*, operating the same system at pH 6.0 and with lower carbohydrate concentrations (15 g/L) prevented methane production achieving a high acidification yield. The initial carbohydrate concentration was increased stepwise, while keeping a constant hydraulic retention time. Thus, increasing the organic content and the VFA concentration within the reactor. Simultaneously, an electrochemical cell (EC) was coupled to the recycling line for the recovery and concentration of VFAs from the fermentation broth, simultaneously alleviating their possible toxic effect.

7.2 Material and methods

7.2.1 Feedstock

7.2.1.1 Electrochemical cell characterisation with synthetic broth

The electrochemical cells were tested using a synthetic feedstock, consisting of acetate, propionate, butyrate and ethanol, using the optimal VFA production obtained in *Chapter 4* (6.79 g/L of sodium acetate, 1.34 g/L of sodium propionate, 5.14 g/L of sodium butyrate and 1.67 g/L of ethanol). As well 300 mg/L of ammonium chloride, 300 mg/L of potassium phosphate and 7.5 g/L of sodium bicarbonate were used to increase the conductivity of the synthetic broth. The pH of this solution was corrected to 6.0. For the anode, a solution of 7.15 g/L (50 mM) sodium sulfate was used for the recovery of the VFAs.

7.2.1.2 Acidogenic fermentation

The continuous reactor was fed with a synthetic feedstock with glucose as main carbon source based on the recipe of Zhang et al. (2008), the main difference was that 10 g/L of sodium bicarbonate was added and a higher concentration of proteins was used. Peptone from wheat was used as proteic substrate (Sigma Aldrich), with a ratio of 208 mg of peptone per g of glucose, similar to the ratio obtained in the hydrolysate in Condition 4, *Chapter 6*. The glucose concentration in the feedstock was increased during each phase from 15 to 30, 45 and 60 g/L. Every time the feedstock was prepared, it was sparged for 5 minutes with nitrogen gas to remove oxygen. The bottle was closed and a gas bag with nitrogen was attached to prevent the formation of vacuum inside of the bottle. The bottle was place in a fridge at 4 °C. The bottle was continuously stirred to homogenize the feedstock.

7.2.2 Inoculum

The acidogenic fermentation reactor was inoculated with granular sludge from an up-flow sludge blanket reactor at the Faxe Wastewater Treatment Plant (Lindegårdsvej, Denmark). This system treats wastewater from a brewery and sweet confectionary plant at 35 °C, with an average COD concentration of 3.9 g COD/L. The inoculum total suspend solids (TSS) and the volatile suspended solids (VSS) corresponded to 8.69 (\pm 0.09) and 7.71 (\pm 0.07) g per 100 g of wet granular sludge, respectively. The reactors were seeded with 1.5 g/L of VSS.

7.2.3 Experimental design and operation

7.2.3.1 Electrochemical cell characterisation with synthetic broth

Each of the electrochemical cells were constructed as previously described by Yang et al. (2022), with the only difference being the use of an anion exchange membrane (AEM). Briefly, two pieces of polycarbonate plastic chambers (5 x 5 x 8 cm) were bolted together, separated by an AEM (AMI 7001, Membrane International, Ringwood, NJ, USA). The AEM membrane was submerged for 24 hours in

sodium hydroxide with a concentration of 50 g/L. Each of the chambers had a total volume of 200 mL, and a working volume of 180 mL. The cathode electrode was an alloy mesh (4 x 4 cm) made of titanium coated with IrO_2 (4 x 4 x 0.1 cm) (Magneto Special Anodes B.V., Netherlands). The anode electrode was a titanium woven wire mesh (4 x 4 x 0.015 cm) (William Gregor Limited, UK), this electrode was coated with 0.5 mg Pt/cm² (20 % Pt/C, JM) and Nafion (5%, Dupont). The electrical connections were done as described by Zhang and Angelidaki (2015). An external resistance of 10 Ω was used in the external circuit to record the current.

Different voltage potentials (Vellemean, DC lab switching mode power supply) were applied between the cathode and the anode to study the VFA recovery rate. These voltages corresponded to 1.5 V, 2.5 V, 3.0 V and 3.5 V. The conductivity, pH and VFA concentration were determined for the cathode and the anode chambers. The cathode section was continuously stirred using a magnetic stirrer to improve the homogenization of VFAs inside of the chamber.

7.2.3.2 Acidogenic fermentation and volatile acid recovery

Two twin up-flow anaerobic sludge BED (UASB) reactors were used for the acidogenic fermentation (total volume 1.69 L, working volume 1.06 L). A recycling line was used to keep upflow velocity constant (2.5 m/h) using a peristaltic pump (Watson-Marlow 323, UK) (*Figure 7.1*). The feedstock was introduced from the bottom of the reactor using a pump (Longer Pump BT100-2J, China). The outlet gas line was connected to an automated water displacement gas meter with a 100 mL compartment (Angelidaki et al., 1991). A loop in the effluent line was introduced to prevent the gas escape through the effluent line due to the backpressure generated by the water displacement gas meter. The pH of the reactor was controlled by a pH regulator and dispenser (ProSystem Aqua, Spain) for sodium hydroxide addition, maintaining the pH at 5.70. The temperature was kept at 37 °C using a water bath. To avoid the wash-out of the biomass, a filter was added after the recycling line and before the effluent. The filter was made by adding a titanium mesh with 1 mm pore size in a sectional tube with an internal diameter of 4.8 cm (*Figure 7.2*).



Figure 7.1 Schematic set up of the volatile fatty acid production and recovery system. The recycling line of the upflow anaerobic sludge bed reactor is coupled to the cathode of an electrochemical cell for the volatile fatty acid recovery



Figure 7.2 Sectional tube with a titanium mesh with a pore size of 1 mm, and placement inside of the upflow anaerobic sludge bed reactor

Once the granular sludge was placed in the reactors, these were filled up with feedstock at a concentration of 10 g/L glucose. Then, the reactors were sparged with nitrogen gas for 5 minutes to remove oxygen from the medium. The system was then run for 24 hours in batch mode to allow sludge acclimatization. After this period, for Phase I, the feedstock was spupplied at an initial hydraulic

retention time (HRT) of 36 hours and a glucose concentration of 15 g/L. For the second phase, the HRT was decreased and kept for the rest of the trial to 24 h. At the third phase, the electrochemical cell was connected to the fermentation reactor by connecting the recycling line to the cathode. During the initial 9 days the tap water added in the anode was not removed. After this period, a pump was connected to continuously add fresh tap water to the system with a HRT of 5 days in the anode chamber. A further decrease of the HRT from 5 to 2 days on day 67, corresponding to Phase IV. The glucose concentration was increased stepwise from 15 g/L in Phase III up to 60 g/L in the final phase of the trial (*Table 7.1*).

Table 7.1 Hydraulic retention time and glucose feeding concentration for each of the phases, with and without the electrochemical cell coupled to the recycling line of the reactor. HRT: hydraulic retention time; EC: electrochemical cell

Phase	HRT (h)	[Glucose] (g/L)	EC	Time (days)
I	36	15	Off	0 → 19
II	24	15	Off	19 → 47
	24	15	On	47→ 59
IV	24	30	On	59 → 83
V	24	45	On	83 → 101
VI	24	60	On	$101 \rightarrow 114$

In addition, during the development of the trial, further adaptions were carried in the set-up. To avoid the accumulation of the biomass inside the anode chamber, a 250 mL Duran bottle was added in the recycling line before the EC as a settling bottle during Phase III. The sludge was recovered from the bottle once per week and reintroduced to the reactor. During Phase VI, a Duran bottle (250 mL) was added in the gas outlet, before the gas counter to avoid any possible clogging of the gas meter.

7.2.4 Analytical methods

Total suspended solids (TSS) and volatile suspend solids (VSS) were determined using the standard method (APHA, 2012). The gas composition (H₂, CO₂ and CH₄) was determined by gas chromatography (Trace 1310 GC-TCD, Thermo Fisher, Denmark) as described by Yang et al. (2022). Ethanol, acetic acid, propionic acid, butyric acid, valeric acid and caproic acid were measured by gas chromatography (HP 5890 series II) with nitrogen as carrier gas, equipped with a fused silica capillary column (Yang et al., 2022). Lactic acid and glucose were measured using High Performance Liquid Chromatography (Agilent) as described by Tsapekos et al. (2020).

7.2.5 Calculations

The recovery rate for each of the VFAs was calculated by the slope obtained in the increase concentration of the cathode chamber during the EC characterisation. The current density was calculated as previously described by Zhang and Angelidaki (2012). Briefly, the current intensity was obtained by dividing the measured voltage across the external resistor by the resistance. Then, the current intensity was devided by the surface area of the anode electrode to obtain the current density. The yield gas production was calculated as described in *Chapter 5*.

7.3 Results

7.3.1 Electrochemical cell characterisation with synthetic broth

Different voltages were applied between the anode and cathode chambers for approximately one week at each of the voltages. The lowest voltages (1.5 and 2.0 V) started with a similar current. However, this current dropped to almost 0 by day 2 (*Figure 7.3*). At 2.5 V the current slightly decreased initially, becoming more constant after 1.2 days with an average current of 6.54 (\pm 1.17) mA/m². A similar trend was observed at 3.0 V when initially the current decreased, although the current for 3.0 V was almost double than that for 2.5 V. Application of 3.5 V resulted in a larger increase in the current, more than five times higher than 2.5 V, and almost three times higher than at 3.0 V (Figure 7.3).



Figure 7.3 Current corresponding to each of the different voltages applied between the cathode and the anode chamber using the same volatile fatty acid synthetic broth

When the voltage applied was 3.5 V, the liquid level in the anode started to increase and decrease in the cathode (*Figure 7.2*). In EC1, the cathode chamber was completely empty by day 7 and in EC2 the cathode chamber was completely empty by day 8.



Figure 7.4 Electrochemical cells after 6 days applying a voltage between the cathode and the anode of 3.5 V, a) EC1 and b) EC2

Table 7.2 summarises the recovery rate and final recovery for the VFAs in the anode chamber for the synthetic broth. *Figure 7.5* shows the profile for the different parameters measured along the trial. Two different rates in terms of current and acid recovery were detected when operating the system at 2.5 V (*Table 7.2*). Under this voltage, the acetic acid concentration increased during the first day up to 582.9 (\pm 24.9) mg/L, with the higher concentration on day 7 corresponding to 1,419.56 (\pm 24.9) mg/L (*Figure 7.5c*). As well, this behaviour was detected for propionic and butyric acid (*Figure 7.5 g & i*). The concentration of acetic acid was similar for 3.0 and 3.5 V, although at the last point the acid concentration decreased. This decrease at the end of the trial using 3.0 V took also place for the butyric and propionic acid concentrations.

	Current Acetic acid		Propionic acid		Butyric acid		
		Rate	Recovery	Rate	Recovery	Rate	Recovery
	(A/m²)	(mg/L/day)	(100g /g)	(mg/L/day)	(100g /g)	(mg/L/day)	(100g /g)
1) 2.5 V	10.38 ± 2.0	489.7 ± 33.0	22.0 ± 2.4	58.2 ± 38.1	20.0 ± 2.3	313.0 ± 52.6	18.8 ± 2.7
2)	6.5 ± 1.2	142.0 ± 32.5	67.9 ± 14.1	23.1 ± 6.6	60.2 ± 13.3	117.9 ± 43.4	53.5 ± 13.1
3.0 V	12.84 ± 3.52	590.2 ± 80.9	62.7 ± 15.1	73.4 ± 7.0	54.1 ± 11.7	277.4 ± 24.0	53.5 ± 12.3
3.5 V	37.83 ± 3.32	1034.5 ± 25.8	83.5 ± 6.0	96.7	77.7 ± 43.2	407.3 ± 62.5	74.0 ± 13.2

Table 7.2 Average current, recovery rate and final acid recovery in the anode chamber of a synthetic mixture broth for recovery of acetic, propionic and butyric acid under different voltages

The conductivity in the anode chamber was similar for all conditions tested, regardless of the voltage applied (*Figure 7.5*). Whereas in the cathode this largely increased as when using a higher voltage. For example, by day 7 the difference in conductivity between the cathode and the anode chamber when applying 3.5 V corresponded to 47.5 (\pm 2.0) mS/cm (*Figure 7.5 a&b*). Similarly, using 3.5 V reached quickly the lowest pH in the anode chamber, and the highest pH for the cathode (*Figure 7.5 c&d*). Although at the end of this condition the pH increased.



Figure 7.5 Conductivity, pH and acid concentration along the recovery of acetic, propionic and butyric acid under different voltages applied between the anode and cathode chamber of the electrochemical cell

The increase of the current and the recovery rate during the VFA recovery from the synthetic broth did not follow a linear trend (*Figure 7.6*). The increase of the current by almost three times, from 13.0 (\pm 3.6) to 38.5 (\pm 2.7) A/m², resulted in a lower increase for the recovery rate of the VFAs (*Figure 7.6*).



Figure 7.6 Recovery rate of a synthetic broth (acetic, 4.89 g/L; propionic, 1.02 g/L; and butyric acid, 4.11) using an electrochemical cell and different voltages. The different currents were calculated based on the stability periods for the different voltages (2.5, 3.0 and 3.5 V)

7.3.2 Continuous volatile fatty acid production and recovery

7.3.2.1 Reactor performance

A short batch period (1 day) was used to acclimatize the inoculum to the new operational conditions. In both reactors, butyric acid was the major acid produced during this period, followed by acetic acid. Once the reactors were operated as continuous system, acetic surpassed the butyric acid concentration and there was production of low concentrations of ethanol, propionic acid, valeric acid and caproic acid. Between days 9 - 10, the caproic acid concentration started to be produced, becoming even the major compound in R1 at the end of Phase I. Ethanol was produced during the batch adaptation period of R1, being fully consumed by day 2. After day 4, ethanol was again detected in the fermentation broth, with a continuous low production along the trial. In R2, the caproic acid concentrations were similar to the butyric acid concentration, whereas acetic acid was the major fermented compound. The ethanol production in R2 took place on day 5. As well, caproic acid was produced during Phase I for both systems, with caproic acid being produced along the trial (*Table 7.3*).

Each of the ECs were connected to each UASB reactor on day 47, starting then Phase III. This resulted in a decrease of the total VFA concentration in both systems and in the sodium hydroxide consumption (*Table 7.3*). Both reactors had a lower acetic and butyric acid concentrations, whereas caproic acid was the major compound in the fermentation broth in R1. The major different between the two UASB reactors was the concentration of these compounds, in R1 the concentration of the three acids was similar, in R2 acetic acid was the major fermented compound followed by butyric and caproic acid. Nevertheless, the total VFA concentration was similar for both reactors. During Phase III,

there was some turbidity in the anode chamber of EC1, which was a result of a tear in the membrane. EC1 was disconnected for 12 hours and the membrane was replaced on day 55. At Phase IV, the glucose concentration in the feedstock was increased to 30 g/L. Similar to Phase III, the major compounds in R1 were butyric and caproic acids. By day 78 the total VFA concentration in R1 had decreased by half, from 20.14 to 9.14 g/L. Acetic acid was the major product in R2 at the beginning of the phase, although by day 67.0 butyric acid became the major fermentation product and by the end of phase IV, acetic and caproic acid had a similar concentration. The EC2 membrane was broken on day 73, thus the EC2 was disconnected for 12 hours and connected again on day 74.



Figure 7.7 Reactor performance for the fermentation of the glucose for the volatile fatty acid production using an expanded granular sludge bed reactor with an electrochemical cell coupled to the recycling line since Phase III. a) System 1 and b) System 2

In Phase V, there were two instability periods in R1 due to a decrease of pH. The first time, the pH decrease to 5.5, resulting in a lower production of VFAs with lactic acid production, which was not detected previously. After this short period, butyric acid became the major fermentation compound.

However, on day 97 the pH dropped to 4.8. This resulted in incomplete consumption of the glucose, as its concentration reached 13.68 g/L. On day 99, the total VFA concentration was 3.41 g/L, and the lactic acid concentration increased to 4.96 g/L. The profile of the fermented compounds in R2 was similar to the previous phases, acetic acid was the major compound followed by butyric and caproic acid. Also, the pH dropped in R2 on day 97, with some lactic acid production (2.32 g/L) and a small accumulation of glucose (0.34 g/L). After this decrease of pH, the profile of VFAs was similar and no further lactic acid was detected during this phase. In Phase VI, the glucose concentration in the feedstock was increased up to 60 g/L, meanwhile the HRT was kept at 24 h. In both systems butyric acid peaked at the beginning of this phase, whereas acetic acid increased along the phase (*Figure 7.7*).

Table 7.3 Main average parameters during the simultaneous production and recovery of the volatile fatty acids using an expanded granular sludge bed reactor coupled with an electrochemical cell

		Acetic acid (g/L)	Butyric acid (g/L)	Caproic acid (g/L)	Total VFAs (g/L)	NaOH Consumption (g/day)	CO2 production yield (mMol / mmol glucose consumed)	CH₄ production yield (mMol / mmol glucose consumed)
	Phase I	1.93 ± 0.46	1.60 ± 0.45	0.77 ± 0.77	4.72 ± 1.10	1.60 ± 1.02	0.07 ± 0.02	0.03 ± 0.01
	Phase II	2.59 ± 0.75	2.19 ± 0.35	1.24 ± 0.36	6.72 ± 1.16	3.95 ± 1.46	0.87 ± 0.31	0.33 ± 0.12
tor 1	Phase III	1.00 ± 0.26	1.19 ± 0.14	0.98 ± 0.07	3.65 ± 0.45	1.79 ± 0.79	0.95 ± 0.11	0.29 ± 0.05
React	Phase IV	4.02 ± 2.08	3.97 ± 2.06	3.17 ± 1.76	12.22 ± 5.62	6.00 ± 2.68	0.64 ± 0.10	0.25 ± 0.07
	Phase V	3.41 ± 2.02	5.29 ± 3.83	1.74 ± 1.18	11.38 ± 6.49	14.51 ± 3.36	0.92 ± 0.29	0.16 ± 0.06
	Phase VI	8.37 ± 1.75	15.70 ± 1.40	2.59 ± 1.15	26.35 ± 12.63	39.89 ± 6.92	0.52 ± 0.39	0.05 ± 0.04
	Phase I	2.52 ± 0.76	1.66 ± 0.37	0.74 ± 0.48	5.40 ± 1.07	1.63 ± 0.61	0.42 ± 0.22	0.17 ± 0.11
Reactor 2	Phase II	3.13 ± 0.64	2.17 ± 0.43	0.83 ± 0.22	6.96 ± 0.92	3.91 ± 1.01	0.85 ± 0.32	0.39 ± 0.15
	Phase III	1.69 ± 0.22	1.04 ± 0.07	0.55 ± 0.32	3.54 ± 0.40	2.10 ± 0.45	0.50 ± 0.40	0.30 ± 0.18
	Phase IV	3.85 ± 1.01	3.24 ± 1.00	2.58 ± 1.23	10.64 ± 3.23	7.41 ± 0.37	1.11 ± 0.35	0.44 ± 0.20
	Phase V	7.21 ± 1.07	5.13 ± 1.37	2.87 ± 0.77	15.96 ± 2.65	12.56 ± 1.46	1.03 ± 0.53	0.29 ± 0.12
	Phase VI	7.41 ± 2.38	12.25 ± 4.81	2.17 ± 1.10	21.52 ± 6.28	30.55 ± 4.56	1.01 ± 0.16	0.23 ± 0.12

Hydrogen was detected under the quantification limit throughout the trial (< 5 % hydrogen). In R1, the carbon dioxide yield was higher than the methane yield throughout the trial (*Figure 7.8*)). During Phase I, almost no gas was produced. Phase II and III resulted in a stable gas production where the carbon dioxide yield fluctuated between 1.51 and 0.49 mM carbon dioxide/mM glucose. This carbon dioxide yield was stable in R1 for Phases III, IV and V, with an average concentration of 0.85 (± 0.28) mM carbon dioxide produced per mM of glucose consumed. The carbon dioxide yield decreased in

Phase VI to 0.12 mM carbon dioxide/mM glucose. The methane yield in R1 was constant from Phase II to Phase IV, with an average concentration of 0.30 (\pm 0.11) mM methane/mM glucose. The methane yield of R1 during Phase VI was to 0.14 (\pm 0.04) mM methane/mM glucose.

The gas production in R2 was larger and with more variability as carbon dioxide and methane fluctuated throughout the trial (*Figure 7.8*). The average carbon dioxide yield in R2 from Phase II to V corresponded to 0.90 (\pm 0.28) mM carbon dioxide/mM glucose, which decreased in Phase VI. In Phase III methane became the major gas produced in R2 instead of carbon dioxide for a short period of time. In the case of the methane yield, a similar trend was followed in R2 as in R1, with an average yield for Phases II, III and IV about 0.39 (\pm 0.18) mM methane/mM glucose, corresponding to the yield during Phase VI of 0.29 (\pm 0.12) mM methane/mM glucose.



Figure 7.8 Carbon dioxide and methane production yield (mM of gas per mM of glucose consumed) of the glucose fed UASB reactors. a) R1 and b) R2

7.3.2.2 Electrochemical cell

Each of the cathode chambers of the EC was connected to the recycling line of the UASB reactor on day 47.8. The anode chamber was filled with tap water. The conductivity in EC1 increased after 3 days from 0.95 to 19.60 mS/cm. The maximum current achieved was 19.72 A/m², which decreased to 11.00 and 8.20 A/m² before the AEM was replaced on day 55. Acetic, butyric and caproic acid rapidly increased, this concentration six days after connecting EC1 amounted to 4.07, 5.01 and 2.42 g/L for, respectively, acetic, butyric and caproic acid. Propionic and valeric acid were also detected at 0.48 and 1.03 g/L, respectively. The AEM was replaced and the EC1 connected again on day 55. After fresh water was continuously added to the anode chamber the VFA concentration increased slower (*Figure 7.9a*). The conductivity in the anode chamber in EC2 achieved quickly 11.81 mS/cm on day 50.8, from this day until the fresh water was continuously added to the anode chamber in EC2 achieved quickly 11.81 mS/cm on day 56.8), the conductivity was almost constant corresponding to 10.35 (± 0.68) A/m². The maximum VFA concentration was achieved on day 56.8, with the primary compound being acetic acid, followed by butyric and caproic acid at concentrations of 10.98 g/L, 5.10 g/L and 3.04 g/L, respectively. Small amounts of valeric and propionic acid were produced at 0.68 and 0.47 g/L, respectively.



Figure 7.9 Electrochemical cell performance for the recovery of the volatile fatty acids from the fermentation broth a) *EC1 and b*) *EC2*

During Phase IV, in R1, the acetic acid concentration decreased for the first five days of Phase IV to 1.8 g/L, after which, the concentration increased to 5.7 g/L on day 67. Butyric and caproic acid achieved the highest concentration before decreasing the HRT at 4.39 and 2.07 g/L respectively. On day 78, caproic acid became the major acid in the anode chamber, and the current almost ceased. After this day all the VFAs, and acetic and butyric acid concentrations slightly increased by the end of

Phase IV. In the case of EC2, the highest concentration was also achieved before reducing the HRT. Acetic acid was the major compound with a concentration of 6.53 g/L, followed by butyric and propionic acid that reached 4.73 and 2.45 g/L, respectively. Then, the AEM was replaced in EC2 and fresh tap water was added to the anode chamber. The concentration of the acetic, butyric and propionic acid slightly increased in the anode chamber by the end of Phase IV.

Phase V resulted in the lowest VFA accumulation in the anode chamber. The VFA concentration in EC1 was almost constant until day 94.8 when the butyric acid concentration started to increase, followed by acetic acid (*Figure 7.9a*). In EC2 acetic acid was the major acid, and the butyric acid concentration was lower. In both EC units the current largely increased at the same time that the conductivity almost dropped to 2.92 for EC1, and 1.11 mS/cm for EC2 on day 97. This corresponded with the lowest acidification period in both reactors. In the final Phase VI, the current was almost constant for EC1 and EC2, with EC2 higher than EC1 (*Figure 7.9*). On the other hand, the conductivity was similar and decreasing slightly for both systems.

7.4 Discussion

7.4.1 Fermentation performance

The use of a simple carbohydrate (glucose) as the substrate for the fermentation step was proposed in order to obtain a fast conversion of the carbohydrates into VFAs. The glucose fed UASB system gave high concentrations of VFAs as a hydrolytic step was not required. This represents an advantage over more complex carbohydrates such as cellulose, where hydrolysis is the limiting step for VFA production (*Sieger and Bank, 2005*). Other systems to achieve a similar VFA concentration required a longer HRT, from 10 to 49 days (Li et al., 2014; Nzeteu et al., 2018; Xu et al., 2020). On the other hand, some systems work with a high volumetric production by using a lower concentration of easy biodegradable carbohydrates (up to 15 g/L) and lower HRT than the present study, achieving high volumetric production of total VFAs up to 72.0 and 56.0 g/(L·day), with a lower concentration (6 - 7 g/L) than in the present study (Pugazhendhi et al., 2017; de Menezes and Silva, 2019). The main VFA produced during those studies were acetic and butyric acid in low concentration, i.e. 1.50 g acetic acid/L and 4.10 g butyric acid/L (de Menezes and Silva, 2019); and, 1.20 g acetic acid/L and 3.80 g butyric acid/L (Pugazhendhi et al., 2017). These concentrations are lower than the VFA concentration achieved in the present work. Moreover, there was no production of caproic acid in these studies.

The loop in the effluent line was added as the gas started to leave the reactor through the effluent line instead of the outlet gas line when the water column reached 40 mL. Thus, the loop has slightly increased the gas pressure inside of the reactor. Caproic acid is produced by chain elongation that is favoured by high hydrogen pressures, using lactic acid or ethanol as electron donors and butyric acid

as electron acceptors (Spirito et al., 2014; Nzeteu et al., 2018; Baleeiro et al., 2021). At the end of Phase I, caproic acid started to increase in both systems, however during Phase II its concentration decreased, indicating that the increase in OLR may have resulted in a decrease in the caproic acid concentration, with larger amounts of acetic and butyric acid produced. This is in agreement with previous studies where increased OLRs resulted in a higher production of acetic and butyric acid at the expense of caproic acid (Pugazhendhi et al., 2017; de Menezes and Silva, 2019). In Phase III, the concentration of acetic and butyric acid decreased in the fermentation broth due to the connection of the EC. In Phase IV, the OLR was increased from 15 to 30 g glucose/L/day and the caproic acid concentration increased in both systems. The highest caproic acid concentration was achieved during this phase in R1 (6.04 \pm 0.02 g/L). In the subsequent phases, increasing the OLR resulted in a lower production of caproic acid and higher production of butyric and acetic acid. It is possible that the increase in OLR inhibited the elongation process for the caproic production, or caproic acid production was restricted by the absence of hydrogen. The increase of the OLR resulted in a larger amount of carbon dioxide produced, the increase of the carbon dioxide in the total gas fraction would have resulted in a higher activity of the methanogens (Xu et al., 2021). As well, the carbon dioxide could drag the hydrogen in the liquid phase decreasing the hydrogen partial pressure in the liquid. The EC would have alleviated this effect by increasing the partial hydrogen pressure in the liquid fraction, favouring the chain elongation and preventing the methanogenesis.

In both systems, there was methane present throughout the trial despite the rather low pH (5.7). Acetoclastic and hydrogenotrophic methanogens can be active at pH's as low as 5.5, or even, in the case of hydrogenotrophs, as low as 4.8 (Baleeiro et al., 2021). Hydrogen levels in both systems were minimal, even though hydrogen was produced in the cathode chamber. Thus, all hydrogen was consumed for methane production or chain elongation. Chain elongation could explain the higher caproic acid production and lower gas yield in R1 than R2. A higher hydrogen pressure may have triggered the production of butyric acid rather than acetic acid (Saady, 2013). At the same time, if this partial pressure of hydrogen cannot be reduced by methanogens, acetic acid production would be inhibited (Siriwongrungson et al., 2007). The higher conversion yield of the glucose to hydrogen and carbon dioxide in R2 (Table 7.3) could be explained by the total VFA production and its profile. The total VFA concentration in the fermentation broth for R1 and R2 was similar at the beginning of the trial, but R1 presented a higher VFA concentration from Phase III onwards (Table 7.3). In the case of Phase V, the decrease of VFA during this phase can be attributed to an inefficient glucose consumption and lactic acid production. For the butyric and caproic acid concentration, these achieved a higher average concentration in R1 than R2. Meanwhile acetic acid was the major compound in R2. Acetic acid production is related with a larger gas production (Saady et al., 2013), meanwhile caproic acid

production is made via gas scavenging (Spirito et al., 2014). Therefore, the higher acetic acid and lower caproic production in R2 resulted in a larger amount of gas with a higher conversion of glucose to carbon dioxide and methane (*Figure 7.8*).

Further optimisation during the fermentation could be done by decreasing the pH to reduce methanogenesis. As well, lowering the temperature or increase the pressure of the headspace can be considered to increase the hydrogen solubility. Finally, using a real liquid hydrolysate as the liquid fraction rich in carbohydrates obtained in Chapter 6 should be carried out to evaluate the stability and efficiency of the process.

7.4.2 Volatile fatty acids

The increase in the current during the VFA recovery from the synthetic broth did not follow a linear trend (*Figure 7.6*). The increase of the recovery rate was less effective when increasing the current after 10 A/m². The highest voltage (3.5 V) led to the highest conductivity difference between the cathode and the anode chamber (*Figure 7.5a&b*). This could have led to a high osmotic pressure which triggered the migration of water from the cathode to the anode (*Figure 7.4*). Acetic acid presented the highest recovery rate at all voltages applied. Furthermore, the increase of the current resulted in a larger increase of the recovery rate for acetic acid than in the case of the butyric or propionic acid (*Figure 7.6*). This could be as a result of the shorter chain of acetic acid and a higher concentration in the synthetic broth. In the case of butyric acid, the higher recovery rate would have been as a result of the larger butyric concentration than that of propionic acid. However, further optimization of the EC is required to determine the optimal current and recovery rate for combinations of different currents with different VFA concentrations.

It was expected that acetic acid would be the main fermentation product in the anode side as this presented the highest recovery rate during the EC characterisation. However, during some periods when the butyric acid concentration was slightly higher in the fermentation broth, butyric acid was the major extracted compound in the anode chamber, i.e. in R1 on days 48 to 54, from days 60 to 69 and from days 87 to day 110.7 (*Figure 7.9a*). In the case of the anode chamber in EC2, butyric acid only became the highest extracted compound during Phase VI, when the difference between butyric and acetic acid in the broth was above 13 g/L. The main difference between the two EC systems was the current, which was higher in EC2. Working with similar concentrations of acetic and butyric acid and a low current ($6.1 \pm 1.4 \text{ A/m}^2$) during the EC characterisation resulted in similar recovery rates of 141.95 (\pm 32.45) g acetic acid/L/day and 117.9 (\pm 43.4) g butyric acid/L/day. This difference increased when using a higher current (*Figure 7.6*). Thus, in the case of EC1 the lower current in the system led to a higher effect on the recovery rate driven by the acid concentration in the fermentation broth.

The largest caproic acid concentration in the anode chamber for EC1 was in Phase IV, where caproic acid was the major fermented compound in R1 with a concentration of about 3.96 g/L on day 77.8 (Figure 7.9a). On the previous days in R1, caproic acid achieved the highest concentration in the fermentation broth (6.04 \pm 0.02 g/L). During this period in Phase IV the current amounted to 7.1 (\pm 0.6) A/m^2 and the HRT of the anode was 3 days. However, a higher recovery of caproic acid took place in EC1 and EC2 during Phase III. The caproic acid concentration in the anode chamber of EC1 was 2.5 times higher than the concentration inside the fermentation broth, with the EC current corresponding to 9.6 (\pm 1.1) A/m² and the HRT of the anode was 5 days. During this period, the current in EC2 was similar (9.8 \pm 0.5 A/m²). However, even with a lower caproic acid concentration in the fermentation broth $(0.7 \pm 0.1 \text{ g/L})$, the concentration in the anode chamber was higher in EC2 than EC1 (3.8 times higher than in the fermentation broth). This corresponded with the period where EC2 achieved the highest VFA recovery. This represents a higher recovery using EC, than a similar system used by Xu et al. (2020), who were able to increase the acetic, butyric and caproic acid concentration by 1.76, 1.26 and 1.51 times. The only difference between their system and the present work was the cathode: they used a titanium mesh (Xu et al., 2020), whereas in the present study the titanium mesh was coated with IrO_2 . In addition, Xu et al. (2020) tested an electrodialysis/phase separator cell and a pertraction system (membrane-based liquid-liquid extraction). The main aim of their work was to reach a concentration of undissociated caproic acid higher than 10.9 g/L that resulted in a second liquid phase (Xu et al., 2020). A lot of work has been done for chain elongation and the recovery of caproic acid, with petraction systems as the most common recovery system, where the driving force for the extraction is mainly due to the pH gradient (Xu et al., 2021). The advantage of the system in the present chapter is the high VFA concentration obtained in the UASB reactor, enhancing the driving force for the VFA extraction.

The efficieny of VFA recovery using liquid-liquid extraction systems increases exponentially when the pH is lower than the pKa (Saboe et al., 2018; Kaur et al., 2020). The EC coupled to the recycling line of the EGSB reactor allowed the concentration of VFAs in the anode chamber and consequent reduction of the pH. This prevents the contamination of the chamber, but the major advantage is the use of an organic phase for the recovery and fractionation of VFAs by liquid-liquid extraction. Indeed, this technique has been broadly considered for the recovery of VFAs (Spirito et al., 2014; Saboe et al., 2018; Kaur et al., 2020). However, as the last approach relies on the undissociated form of the acids and the hydrophobicity of the molecules, the VFA with lower numbers of carbons result in a low recovery efficiency as the alkyl length of the VFA is the shortest, i.e. acetic acid (Rocha et al., 2017). In the present study the anode chamber had a pH ranging between 1.2 and 1.8, thus, VFAs would be in their undissociated form largely enhancing the efficiency for the VFA extraction using an organic phase

(Kaur et al., 2020). For example, using the highest VFA concentration achieved in the anode chamber during the present study, with the graphs for the equilibrium concentration of VFAs in the organic phase at pH 3.0 from Saboe et al. (2018), the estimated concentration of VFAs in the organic fraction would be about 40.4, 50.98 and 39.10 g/L for acetic, butyric and caproic acid, respectively.

The coupled UASB-EC system showed promising results, as good recovery of the VFAs from the fermentation broth, decrease of the alkaline demand and the production of hydrogen was beneficial for chain elongation. However, the EC was not able to recover the largest fraction of VFAs from the fermentation broth, which is likely related with an under dimensioning of the unit. Thus, further optimisation is required to optimise the VFA recovery process. For this, a deeper characterisation by combining different currents and acid concentrations is required to determine the optimal combination for the recovery. As well, a deeper understanding of the relation between the current and the conductivity to determine the possible saturation of the system. For example, if the conductivity is still constant but voltage decreases, as happened at the beginning of the trial using the lowest voltages (1.5 and 2.0 V). Also, this effect was observed during the simultaneous production and recovery of the VFAs, when on day 88 the current decreased to 0.04 A/m² when the highest caproic acid was achieved. Then, the current increased in the subsequent days by the continuous addition of fresh water in the anode chamber that lowered the VFA concentration. Thus, it is necessary to determine any possible saturation effect related to the VFA concentration in the anode chamber and the required HRT to avoid such a problem.

7.5 Conclusion

This study was designed as a proof of concept to demonstrate the potential for VFA recovery by increasing their concentration in the fermentation broth with a larger volumetric production rate and recovery by coupling an EC to the recycling line of a UASB reactor. This was possible by using a high rate reactor with good biomass retention and a high feedstock concentration of a simple carbohydrate (glucose). The major fermented compounds were acetic acid, butyric acid and caproic acid. The EC was able to concentrate the VFAs up to 4.6 times in the anode chamber. The system shows some promising advantages as it allows VFA production under optimal conditions and the recovery of VFAs in their easier to recover (undissociated) form. At the same time, the EC contributed to increases in the hydrogen partial pressure and decreases in alkaline requirement.

7.6 References

- Albuquerque, M. G. E., V. Martino, E. Pollet, L. Averous, and M. A. M. Reis (2011) "Mixed culture polyhydroxyalkanoate (PHA) production from volatile fatty acid (VFA)-rich streams: Effect of substrate composition and feeding regime on PHA productivity, composition and properties." *Journal of Biotechnology* 151 (1): 66-76. https://doi.org/10.1016/j.jbiotec.2010.10.070
- APHA (2012) "Standard Methods for Examination of Water and Wastewater 2012" American Public Health Association.
- Bak C., Yun Y.-M., Kim J.-H., Kang S. (2019) "Electrodialytic separation of volatile fatty acids from hydrogen fermented food wastes." International Journal of Hydrogen Energy 44 (6): 3356-3362. https://doi.org/10.1016/j.ijhydene.2018.07.134.
- Baleeiro F. C. F., Kleinsteuber S., Sträuber H. (2021) "Hydrogen as a Co-electron Donor for Chain Elongation With Complex Communities." *Frontiers in Bioengineering and Biotechnology* 9 (251). https://doi.org/10.3389/fbioe.2021.650631.
- Baroi G. N., Gavala H. N., Westermann P., Skiadas I. V. (2017) "Fermentative production of butyric acid from wheat straw: Economic evaluation." Industrial Crops and Products 104 (Supplement C): 68-80. https://doi.org/https://doi.org/10.1016/j.indcrop.2017.04.008.
- Guzman J. J. L, Angenent L. T. (2021) "Direct Medium-Chain Carboxylic Acid Oil Separation from a Bioreactor by an Electrodialysis/Phase Separation Cell." *Environmental Science & Technology* 55 (1): 634-644. https://doi.org/10.1021/acs.est.0c04939.
- Kaur G., Garcia-Gonzalez L., Elst K., Truzzi F., Bertin L., Kaushik A., Balakrishnan M., De Wever H. (2020) "Reactive extraction for in-situ carboxylate recovery from mixed culture fermentation." Biochemical Engineering Journal 160: 107641. https://doi.org/https://doi.org/10.1016/j.bej.2020.107641.
- Li, Lei, Qingming He, Yunmei Wei, Qin He, and Xuya Peng. 2014. "Early warning indicators for monitoring the process failure of anaerobic digestion system of food waste." Bioresource Technology 171: 491-494. https://doi.org/10.1016/j.biortech.2014.08.089.
- de Menezes, C. A. d. and Silva, E. L. (2019) 'Hydrogen production from sugarcane juice in expanded granular sludge bed reactors under mesophilic conditions: The role of homoacetogenesis and lactic acid production', Industrial Crops and Products, 138, pp. 111586.
- Nzeteu C. O., Trego A. C., Abram F., O'Flaherty V. (2018) "Reproducible, high-yielding, biological caproate production from food waste using a single-phase anaerobic reactor system." Biotechnology for Biofuels 11 (1): 108. https://doi.org/10.1186/s13068-018-1101-4.
- Pugazhendhi, A., Anburajan, P., Park, J.-H., Kumar, G., Sivagurunathan, P. and Kim, S.-H. (2017) 'Process performance of biohydrogen production using glucose at various HRTs and assessment of microbial dynamics variation via q-PCR', International Journal of Hydrogen Energy, 42(45), pp. 27550-27557.
- Rocha M. A. A., Raeissi S., Hage P., Weggemans W. M. A., van Spronsen J., Peters C. J., Kroon M. C. (2017) "Recovery of volatile fatty acids from water using medium-chain fatty acids and a cosolvent." Chemical Engineering Science 165: 74-80. https://doi.org/10.1016/j.ces.2017.02.014.

- Saady, N. M. C. (2013) 'Homoacetogenesis during hydrogen production by mixed cultures dark fermentation: Unresolved challenge', International Journal of Hydrogen Energy, 38(30), pp. 13172-13191.
- Saboe P. O., Manker L. P., Michener W. E., Peterson D. J., Brandner D. G., Deutch S. P., Kumar M., Cywar R. M., Beckham G. T., Karp E. M. (2018) 'In situ recovery of bio-based carboxylic acids', Green Chemistry, 20(8), pp. 1791-1804.
- Siegert I., Banks C. (2005) "The effect of volatile fatty acid additions on the anaerobic digestion of cellulose and glucose in batch reactors." Process Biochemistry 40 (11): 3412-3418. https://doi.org/10.1016/j.procbio.2005.01.025.
- Siriwongrungson, Vilailuck, Raymond J. Zeng, and Irini Angelidaki. 2007. "Homoacetogenesis as the alternative pathway for H2 sink during thermophilic anaerobic degradation of butyrate under suppressed methanogenesis." Water Research 41 (18): 4204-4210. https://doi.org/10.1016/j.watres.2007.05.037.
- Spirito C. M., Richter H., Rabaey K., Stams A. J. M., Angenent L. T. (2014) "Chain elongation in anaerobic reactor microbiomes to recover resources from waste." Current Opinion in Biotechnology 27: 115-122. https://doi.org/https://doi.org/10.1016/j.copbio.2014.01.003.
- Strazzera G., Battista F., Garcia N. H., Frison N., Bolzonella D. (2018) "Volatile fatty acids production from food wastes for biorefinery platforms: A review." *Journal of Environmental Management* 226: 278-288. https://doi.org/https://doi.org/10.1016/j.jenvman.2018.08.039.
- Tsapekos P., Alvarado-Morales M., Baladi S., Bosma E. F., Angelidaki I. (2020) "Fermentative Production of Lactic Acid as a Sustainable Approach to Valorize Household Bio-Waste." Frontiers in Sustainability 1 (4). https://doi.org/10.3389/frsus.2020.00004.
- Wainaina S., Parchami M., Mahboubi A., Horváth I. S., Taherzadeh M. J. (2019) "Food waste-derived volatile fatty acids platform using an immersed membrane bioreactor." *Bioresource Technology* 274: 329-334. https://doi.org/10.1016/j.biortech.2018.11.104.
- Xu J., Guzman J. J. L., Andersen S. J., Rabaey K., Angenent L. T. (2015) "In-line and selective phase separation of medium-chain carboxylic acids using membrane electrolysis." Chemical Communications 51 (31): 6847-6850. https://doi.org/10.1039/C5CC01897H.
- Xu J., Guzman J. J. L, Andersen S. J., Rabaey K., Angenent L. T. (2015) "In-line and selective phase separation of medium-chain carboxylic acids using membrane electrolysis." Chemical Communications 51 (31): 6847-6850. https://doi.org/10.1039/C5CC01897H.
- Xu J., Guzman J. J. L., Angenent A. L. (2021) "Direct Medium-Chain Carboxylic Acid Oil Separation from a Bioreactor by an Electrodialysis/Phase Separation Cell." Environmental Science & Technology 55 (1): 634-644. https://doi.org/10.1021/acs.est.0c04939.
- Yang X., Jiang Y., Zou R., Xu M., Su Y., Angelidaki I., Zhang Y. (2022) "Green electricity-driven simultaneous ammonia recovery and in-situ upcycling for microbial protein production." *Chemical Engineering Journal* 430: 132890. https://doi.org/https://doi.org/10.1016/j.cej.2021.132890.
- Zhang Z.-P., Show K.-Y., Tay J.-H., Liang D. T., Lee D.-J. (2008) "Biohydrogen production with anaerobic fluidized bed reactors—A comparison of biofilm-based and granule-based systems."

International Journal of Hydrogen Energy 33 (5): 1559-1564. https://doi.org/10.1016/j.ijhydene.2007.09.048.

- Zhang Y., Angelidaki I. (2012) "A simple and rapid method for monitoring dissolved oxygen in water with a submersible microbial fuel cell (SBMFC)." *Biosensors and Bioelectronics* 38 (1): 189-194. https://doi.org/https://doi.org/10.1016/j.bios.2012.05.032.
- Zhang, Y., Angelidaki I. (2015) "Bioelectrochemical recovery of waste-derived volatile fatty acids and production of hydrogen and alkali." Water Research 81 (Supplement C): 188-195.://doi.org/10.1016/j.watres.2015.05.058.

Chapter 8

General discussion and future perspectives

8.1 Key findings

The pillar for the design of this thesis was the development of a biorefinery in Ireland using a common industrial by-product. Brewery spent grain was selected due to its composition and as an abundant feedstock in Ireland with low value, with a production above 170,000 tons in 2019 (*Chapter 2*). The main objective was the production of the volatile fatty acids using brewery spent grain as a feedstock. The initial approach for simultaneous hydrolysis and acidogenic fermentation for volatile fatty acid production evolved to a more complex process in view of the high valuable compounds that are contained in the brewery spent grain, including the carbohydrates, proteins and lignin. *Figure 8.1* gives the main findings of this PhD thesis and the connections between the different chapters of the thesis.



Figure 8.1 Key findings from the different chapters, development and relation between chapters of this PhD thesis

8.2 Core findings and future recommendations

During the introduction of this thesis (*Chapter 1*) it is was mentioned that the current definition of a biorefinery should exclude the classification "Crop biorefinery" and "Residue biorefinery". The residues should be characterised by their composition and the required pretreatment for their valorisation. Then, the biorefinery should be classified by the process technology and the platform. For example, in the case of the BSG the process technology would involve chemical and biochemical processing to obtain three different streams, each of them aiming for different platforms: i) carbohydrate platform, ii) protein platform and iii) lignin platform. On the other hand, some of the current biorefineries for production of biogas or ethanol rely on the fees that companies pay for the disposal of the wastes. However, as the valorisation of wastes and the closing of the circular economy is improving, these wastes will be regarded as valuable resources instead of nuisance material for disposal.

The strength of the biorefinery should be focused on the production of a different range of final products during the processing of the source used as feedstock. This could give extra capacities to the biorefinery plant for diverting the process based on the market demand to increase the profits of the plant. Such biorefineries should operate as a hub centre to treat products with similar initial characteristics, increasing their production capacity and achieving the minimum volume required for the feasibility of the plant. For example, the approach for the protein recovery as applied in *Chapter 6* would have led in a liquid fraction rich in carbohydrates. These proteins could be used in the food industry, meanwhile the liquid could be used inside of a carbohydrate platform in combination with other liquors with similar composition. However, more work is required at lab scale to test the combination of different waste streams. The different technologies should be carefully evaluated for each of the different feedstock and wastes, as highlighted in *Chapter 2*. Some of the unsuccessful industrial cases were related with using a technology developed for a similar feedstock that did not perform as well as expected. For instance, this was the case of unexpected difficulties when treating corn stove instead of wood chips (Slupska and Bushong, 2019).

The first approach followed during the experimental work of this PhD thesis was the simultaneous hydrolysis and fermentation of the brewery spent grain in *Chapter 3*. As BSG is a lignocellulosic material, not all microorganisms are able to degrade it. Thus, the chapter focused on evaluating six different inocula with a microbial consortium potentially capable to degrade and accumulate fermented compounds. The solid degradation by the rumen fluid and manure from herbivore animals were expected to achieve the higher performance. Surprisingly, rumen fluid presented the highest initial destruction rate, but it almost ceased after 2 days. The highest solid degradation and gas

production was achieved by granular sludge, which presented the lowest hydrolytic activity. In addition, giraffe manure presented an unusual trend in comparison with the other sludge, as the solid degradation was still ongoing by the end of the trial after 30 days. In this case, the giraffe manure initially had VFA accumulation, however by day 10 these were being consumed. This improvement for the solid degradation during the middle of the trial could be related to an improvement in the efficiency for the methanogenization of the fermented compounds. Thus, one of the finding of this chapter was the association between the increase in the VFA concentration and the decrease of the solids degradation. The hypothesis developed during *Chapter 3* is that VFAs affect or even completely inhibit the hydrolytic degradation of the lignocellulosic feedstock. Therefore, further studies to determine the type of inhibition is important. Biodegradation activity tests should be carried out using granular sludge, with and without chemical inhibition of the methanogenesis, under different initial VFA concentrations. Further research is required the obtained the results to calculate the kinetic and stoichiometric parameters of the lignocellulose feedstock degradation. For example, the Lineweaver-Burk plot can be used to obtain the maximum rate and the Michaelis constant to evaluate any possible inhibition of the lignocellulose by VFAs.

The second approach was based on a two-step process developed during *Chapter 4*. First, the hydrolysis of the BSG, and the recovery of the liquid fraction for the acidogenic fermentation was investigated using batch stirring tank reactors. The efficiency of the hydrolysis was determined by the ratio of the total amount of the carbohydrates recovered in the liquid fraction per gram of BSG dry matter. The most efficient was using 7 % of BSG (w/w) with a 1.5 % of sulfuric acid concentration (v/v). Increasing the concentration of BSG resulted in a lower carbohydrate efficiency recovery. Similarly, a higher amount of sulfuric acid did not result in a better extraction as well. The remaining solid was optimised in a later part of the PhD thesis, which was carried out with some changes during *Chapter 6*. The recovered liquid from the hydrolysis was then fermented at different pH values using batch reactors. The fastest production rate was achieved at pH 5.0, although the highest acidification yield achieved at pH 6.0. Operating the system at a pH lower than 4.5 resulted in an inefficient consumption of the carbohydrates as the pentose sugars were not consumed (arabinose and xylose).

The thermal diluted sulfuric acid hydrolysis and downstream acidogenic fermentation of the brewery spent grain showed a high efficiency during the batch experiments. Therefore, the next step was the continuous fermentation of the liquid hydrolysate. This was carried out in *Chapter 5* using a continuous high rate system. During this work and the previous trial (*Chapter 5* and *Chapter 4*) no pretreatment of the biomass was carried out as the operational conditions were expected to inhibit the methanogenesis. In the previous batch experiments (*Chapter 4*), gas production and gas composition were not analysed. In the case of the continuous production, two twin reactors were

running, one of them with the intact granular sludge, and in the other, the granules were disaggregated to compare the operational pressure conditions. The best performance was using granular sludge, in this case methane production was reductant, and was only detected during some instability periods when VFA concentrations decreased. Even, the produced liquid hydrolysate in *Chapter 4* and *Chapter 6* contains a mixture of hexose (glucose) and pentose (xylose and arabinose), which was fully consumed as demonstrated in *Chapter 4* and *Chapter 5*.

During the preparation of the feedstock, when using the batch reactors in *Chapter 4* the liquid fraction after the thermal diluted acid hydrolysis was neutralised inside the reactors. However, in the continuous VFA production, the liquid hydrolysate was neutralised in a bottle. This neutralisation led into the precipitation of a solid fraction. It was hypothesised that this fraction could be the precipitation of some organic compounds, such as proteins and salts formed by the sulfuric acid during the hydrolysis and the sodium hydroxide during the neutralisation step. The sample had a high content of inorganic matter, $52.1 (\pm 5.6)$ % of non-volatile solids, which was a result of using the sulfuric acid and sodium hydroxide during the hydrolysis and neutralisation steps (*Chapter 5*). In addition, a sample of this solid was used to determine its protein content, which corresponded to 37 %. Thus, above 75 % of the volatile solids corresponded to proteins. A deeper study of the hydrolysis process was subsequently developed during *Chapter 6* to enhance the protein recovery at the same time of the carbohydrate release.

The holistic processing of the BSG was tackled by a cascade process in *Chapter 6*. The main goal of this chapter, beside the optimal carbohydrate release, was to understand how the different pretreatment combinations affect the composition of the different extracted fractions. In this case, enzymes were used for further improve the hydrolysis. To target higher value markets, such as food, cosmetics or pharmaceuticals, hydrochloric acid was used instead of sulfuric acid and the enzymatic cocktail was food quality grade. Different combinations of time, acid and BSG concentrations were applied. To determine the extent of the hydrolysis, additional techniques could be used instead of the lignocellulosic characterisation by the common degradation of the material by subsequent chemical hydrolysis, for example nuclear magnetic resonance spectroscopy (NMR) or fourier-transform infrared spectroscopy (FTIR) (Bevilacqua et al., 2017; Guo et al., 2018). This could be used to determine the structure and the crystallinity of the BSG in a faster and easier way. Additionally, it was not possible to try a BSG concentration higher than 10 % as the blender was not able to manage it. The physical disruption of the BSG resulted in a higher water capacity adsorption, thus the liquid in the mixture decreased, making it impossible to blend the BSG. This happened during Chapter 4 and Chapter 6. Some other physical treatments for wet material should be considered in order to increase the BSG fraction, which would be as well evaluated for the release of the protein, lignin and carbohydrate fractions.

During this work, three different conditions achieved a large release of the carbohydrates. Although in one of the conditions, the carbohydrates were mainly released during the thermal diluted hydrochloric acid pretreatment (Condition 2). In the case of the conditions where enzymatic hydrolysis enhanced the release of the carbohydrates (Conditions 1 and 4), only one concentration of enzymes was evaluated. Further work should be done to determine the optimal dose of enzymes in order to avoid an overuse that would lead in an unnecessary extra cost.

The final stage of this PhD thesis was the continuous production and recovery of the VFAs. A novel process using an electrochemical cell directly coupled to the recirculation of the expanded granular sludge bed reactor was tested in *Chapter 7*. This was a proof of concept to demonstrate the feasibility of the process. The electrochemical cell was able to increase the VFA concentration in the anode chamber more than 4 times than the current concentration in the fermentation broth. In addition, the electrolysis process resulted in a lower sodium hydroxide consumption and the production of hydrogen. Even, the hydrogen was completely consumed for chain elongation and methane production. Nevertheless, further studies are required as the electrochemical cell was under dimensioned, which would not only enhance the VFA recovery but also the chain elongation. Further optimisation of the process could be done by evaluating the decrease of the pH that would reduce the alkaline consumption and could also reduce the methane production. Using a lower temperature in the reactor, i.e. 30, 25 or 20 °C, could increase the solubility of the hydrogen.

8.3 Future developments

8.3.1 Brewery spent grain biorefinery

The obtained results are encouraging as foundation for the scaling up process and practical implication of a biorefinery using BSG as feedstock. The BSG could be used for the production of two solid fractions of food grade quality, VFAs and syngas production (*Figure 8.2*), and the remaining liquid fraction as potential feedstock for biogas and fertilizer production. This process is be divided in **i**) upstream process and **ii**) downstream process:

i) The upstream process would be focus on the production of the three different streams: the coarse and fine solids, and the liquid fraction rich in carbohydrates. The fine solids have a texture similar to dough, their content on lignin and possible antioxidant and probiotic activity must be analysed. As well, the organoleptic properties should be determined as food additive, or even for use in the pharmaceutical or cosmetic industry. The coarse solids are composed by the remaining husk and other larger fractions of the BSG. In the same way, the characterisation of the coarse solids must be carried out to target some of the different industrial markets, with a lower market value, for instance, as animal food supplement.



Figure 8.2 Proposed diagram for a brewery spent grain biorefinery developed under the current PhD thesis. The brewery spent grain is chemically pretreated prior enzymatic hydrolysis, after two different solid fractions are obtained. The liquid fraction is used in a downstream process for volatile fatty acid production, with biogas as by-product

ii) The downstream process is focused on the use of the liquid fraction for production of high valuable products. This liquid is rich in carbohydrates consisting in a mix of pentoses and hexoses. Therefore, this liquid could be integrated inside of a biorefinery carbohydrate platform, even combining with other liquids with similar characteristics. The current PhD thesis proposed the use of this easily biodegradable carbohydrates for the production of the VFAs. Thus, during the downstream fermentation process the main final product would be the VFAs, gas (CO₂ and H₂) as by-product and remaining liquid for biogas (CH₄ and CO₂) production or/and as a fertilizer.

8.3.2 Simultaneous volatile fatty acid production and recovery

The next step for the simultaneous volatile fatty acid production and recovery is the fractionation of each of the individual VFAs as a final product, i.e. acetic acid, butyric acid and caproic acid. The system developed during this PhD thesis using a higher rate reactor (expanded granular sludge bed reactor) coupled with the electrochemical cell to increase the VFA concentration could be integrated with a recovery system. For example, the liquid-liquid recovery and distillation system proposed by Saboe et al. (2018) could achieve high extraction yields of the VFAs from the cathode chamber due to the low pH and the quality of the phase, only anions can cross the anion exchange membrane (*Figure 8.3*). The organic phase used would be recovered by distillation and the individual VFAs recovered as final products. The systems should be evaluated in terms of energy consumption, for example the electricity consumed by the electrochemical cell and the energy in the distillation column. On the other hand, there are some parameters that could counterbalance and underpin the feasibility of the process, for example the lower required alkaline chemical addition, the production of a higher value compound (caproic favoured above acetic), and the remaining organic fraction for biogas production that could power the distillation unit. All these parameters would be used to evaluate the economic feasibility of a pilot plant for the production and fractionation of each of the VFAs as final products.



Figure 8.3 Continuous system for simultaneous production, recovery and fractionation of volatile fatty acids. The easily biodegradable carbohydrates from brewery spent grain (**Chapter 6**) are fed into the expanded granular sludge bed reactor (**Chapter 5**). The recycling line is connected to a settler to decrease and return the solid particles and biomass into the reactor, the clearer liquid is coupled within an electrochemical cell for recovery of the volatile fatty acids at the anode chamber (**Chapter 5**). The anode, with low pH, is connected in a closed loop with a liquid-liquid recovery system using an organic phase to extract the volatile fatty acids. Finally, the organic phase is boiled in a distillation column for the fractionation of each of the individual volatile fatty acids and the organic phase is recovered. The unit for simultaneous volatile fatty acid production and recovery, bioreactor and electrochemical cell, has been developed during this PhD thesis. The liquid-liquid separation and the distillation column has been adapted from Saboe et al. (2018)

8.4 References

- Bevilacqua M., Praticò G., Plesner J., Molloy M., Skov T., Larsen F. H. (2017) "Solubilization of industrial grade plant protein by enzymatic hydrolysis monitored by vibrational and nuclear magnetic resonance spectroscopy: A feasibility study." *Food Research International* 102: 256-264. https://doi.org/https://doi.org/10.1016/j.foodres.2017.09.069.
- Guo Z., Ling Z., Wang C., Zhang X., Xu F. (2018) "Integration of facile deep eutectic solvents pretreatment for enhanced enzymatic hydrolysis and lignin valorization from industrial xylose residue." *Bioresource Technology* 265: 334-339. https://doi.org/https://doi.org/10.1016/j.biortech.2018.06.027.
- Saboe P. O., Manker L. P., Michener W. E., Peterson D. J., Brandner D. G., Deutch S. P., Kumar M., Cywar R. M., Beckham G. T., Karp E. M. (2018) 'In situ recovery of bio-based carboxylic acids', Green Chemistry, 20(8), pp. 1791-1804.

Slupska M., Bushong D. (2019) "Lessons from Commercialization of Cellulosic Ethanol – A POET Perspective." Biofuels, Bioproducts and Biorefining 13 (4): 857-859. https://doi.org/10.1002/bbb.2033.

Supplementary information Chapter 3

S3.1 Methods

S3.1.1 Nucleic acid extractions

DNA and RNA were co-extracted from 0.75g (fresh weight) of the retentate (*Figure 1*), from sample days 0 and 21 of each of the three biological replicates from the 6 inoculum types (n=36 extractions) using the CTAB-phenol chloroform method of Griffiths *et al.* (2000). The resulting nucleic acid pellet was eluted in nuclease free water. To investigate the active fraction present on day 21, an aliquot of eluted pellet was DNase treated with Turbo DNase (Ambion) following the manufacturer's instructions to isolate the RNA. PCR with universal prokaryotic primers 515F/806R (Caporaso et al., 2012) was used to verify total removal of genomic DNA, checking for the absence of PCR product (at a series of dilutions). The generation of cDNA was performed using SuperScript[®] III Reverse Transcriptase (Invitrogen), following the manufacturer's recommended guidelines, on a normalised concentration of RNA (10 ng μ L⁻¹). DNA was normalised to 50 ng μ L⁻¹ and samples were shipped on dry ice to the RTL sequencing facility (Lubbock, Texas, US). Amplicon library preparation was performed using the Parada *et al.* (2016) version of primers 515f/806r targeting the archaeal/bacterial V4 16S rRNA region, followed by addition of sequencing adapters and dual-index barcodes, using a limited number of PCR cycles as per the Illumina protocol (Illumina, 2013). Paired-end sequencing was performed using the Illumina MiSeq platform with 2 x 300bp V3 kit chemistry to a minimum depth of 10K.

S3.1.2 Sequence analysis

Mothur (version v.1.44.3) was used to process of paired-end 16S rRNA reads (Schloss et al., 2009) following the pipeline detailed in Kozich *et al.* (2013). The SILVA database (v132), trimmed to the V4 region, was used for alignment of sequences and the Ribosomal Database (v.9) was used for classification. Samples contained an average of 19,200 reads after quality filtering. The dataset was normalised for subsequent analyses by subsampling to the lowest number of sequences. Distance matrices were created in Mothur using the Theta Yue Clayton metric for the three subsets of data (DNA day 0; DNA day 21 and cDNA day 21) which were subsequently ordinated using nMDS. Clustering of samples as a function of inoculum source was tested for statistical significance using analysis of molecular variance, and Spearman correlations of experimental variables with nMDS axes were tested using the corr.axes function in Mothur. Differentially abundant OTUs were detected with LEfSe analysis (Segata et al., 2011). Data visualisation was performed in R (R Core Team, 2019) using the ggplot2 package (Wickham, 2011). Statistically significant differences in means as a function of inoculum were tested for using the non-parametric Kruskal-Wallis test. Where statistical differences were seen (p < 0.01), a post-hoc test was performed using Fisher's Least Significant Difference (LSD)

and these statistical tests were performed using the agricolae package (de Mendiburu and Yaseen, 2011).

S3.2 Microbial community analyses of the inocula

S3.2.1 Comparison of the microbial composition of the inocula

Initial prokaryotic microbial populations within the manure samples (DNA day 0) were dominated by relatively few phyla, namely the *Firmicutes, Proteobacteria* and *Bacteroidetes*, and this was particularly the case for the tiger manure (*Supplementary Figure 1*). The same phyla were present in rumen fluid samples, in addition to a significant contribution from the *Fusobacteria*. As expected, no large differences were seen between crushed and un-crushed granular sludge, where both differed considerably from the inocula of animal origin, with dominant phyla from the *Synergistetes, Euryarchaea* and *Bacteroidetes*. Additionally, a higher proportion of unclassified bacteria were present within these sludge samples.



Supplementary Figure 1 Relative abundances of OTUs assigned to the level of phylum, in each of the 6 inocula used for BSG degradation. Nine bars are shown for each inocula, where the first three are DNA day 0 samples, and subsequent three from DNA day 21 and the final three from cDNA day 21 samples

By day 21 in most samples, except tiger manure, a similar trend of increased numbers of dominant phyla at the DNA level was observed, alongside more variability between triplicate samples. Rumen

fluid, giraffe manure and rhinoceros manure experienced an increase in abundance of *Synergistetes* and *Euryarchaeota* which initially (day 0) were present in very low numbers in these samples. The dominant phyla present at the cDNA level followed similar trends to those in their DNA counterparts on day 21.

Alpha diversity analysis included sample coverage, which indicated sufficient coverage (> 97%) for all samples except giraffe manure on day 0 (96%) and rumen fluid at the DNA level on both days 0 (93%) and day 21 (96%) (*Supplementary Table 1*). The number of OTUs observed in these samples clearly explained this trend, where rumen fluid contained almost triple, and giraffe manure double, the amount of OTUs detected in other samples (*Supplementary Figure 2*). The rhinoceros manure samples were significantly richer (p < 0.05) than the sludge and tiger manure samples. The tiger manure that was the least rich (p < 0.05) of all inocula with only 300 OTUs observed.

Supplementary Table 4 Alpha diversity indices for the six inocula degrading BSG on days 0 (DNA level) and 21 (DNA and cDNA level) data shown is the average of three biological replicates

Day	Fractio n	Inocula	Percent Coverage (Good's)	Observe d OTUs	Estimate d OTUs (Chao)	Simpson' s Evenness	Simpson' s Diversity (Inverse)
		GS	97.7	340.8	533.7	0.03	9.0
		GS crushed	97.6	363.1	555.4	0.05	16.6
		Rumen fluid	92.6	948.2	1581.7	0.02	23.3
0	DNA	Giraffe manur					
		e	95.7	602.9	968.4	0.03	20.3
		Rhino manure	97.2	384.3	636.1	0.02	6.6
		Tiger manure	98.2	299.2	437.4	0.09	25.7
21 -	DNA	GS	97.8	312.1	469.8	0.01	4.5
		GS crushed	97.4	346.7	582.6	0.02	7.7
		Rumen fluid	95.9	569.0	927.1	0.05	30.4
		Giraffe manur					
		e	98.1	298.8	471.4	0.04	11.3
		Rhino manure	97.8	306.3	500.2	0.03	8.2
		Tiger manure	98.2	276.1	437.9	0.05	15.0
	cDNA	GS	98.3	272.9	410.7	0.04	10.2
		GS crushed	98.5	231.8	367.5	0.03	8.1
		Rumen fluid	98.1	264.3	435.7	0.03	9.7
		Giraffe manur					
		e	98.8	184.1	291.0	0.03	5.8
		Rhino manure	98.3	250.9	391.8	0.04	9.3
		Tiger manure	97.7	283.6	525.1	0.02	6.3

While a similar trend in richness was still present on day 21, differences between inocula were no longer statistically significant. From day 0 to 21, a reduction in the number of OTUs detected was seen in the rumen fluid and giraffe manures samples, both of which fell by more than 40%. Rhinoceros manure also saw a reduction, albeit much smaller while granular sludge and manure samples

remained similar or fell very slightly. There were no statistically significant differences between the number of OTUs detected in the active (cDNA) fraction of the six inocula on day 21, and with the exception of tiger manure, fewer OTUs were detected in the cDNA than the DNA fraction. Patterns in richness between inocula generally translated into similar patterns in diversity, except for tiger manure where a higher evenness (more equitably distributed OTUs) resulted in the highest diversity index for these samples on day 0.



Supplementary Figure 2 Alpha diversity indices of observed OTU richness and Simpson's diversity of BSG degradation by different inocula. Points represent the mean of each inoculum in the three datasets (DNA day 0 and 21 and cDNA day 21) and error bars show standard error of the mean (n=3). Letters within the points illustrate statistical differences as a function of inoculant for each day or fraction, where shared letters denote no difference (p > 0.05), and unshared letters denote a statistical difference within a group (p < 0.05)

S3.2.2 Presence vs activity of OTUs in each inocula (Day 21)

A differential abundance analysis was performed to detect OTUs that were present (DNA) in significantly higher numbers than they were active (cDNA) with the intention of determining OTUs were perhaps under non-optimum conditions. Rumen fluid samples had more than double the number of lower activity OTUs than the other inocula on day 21, with 100 OTUs present in higher abundance at the DNA level. Rhinoceros manure meanwhile had the fewest number of lower activity OTUs (n=18), while giraffe (38 OTUs) and tiger (51 OTUs) were similar. Lower activity OTUs were
present in crushed granular sludge (40 OTUs) than in its uncrushed counterpart (28 OTUs). Upon investigating the more abundant of these OTUs (> 0.1% at DNA level) (*Supplementary Figure 3*), both rumen fluid and tiger manure samples showed a trend of OTUs from the *Firmicutes* that were most often under-represented in the active fraction, a number of which were unclassified *Ruminococcaceae* and *Lachnospiraceae*. A number of OTUs from the genera *Bacteroides* were identified as lower in activity, particularly so in the giraffe and tiger manure samples. Only one OTU from the methanogenic archaea was identified as significantly lower in abundance in the active (cDNA) fraction: *Methanomicrobium* in the rumen fluid sample.



Supplementary Figure 3 OTUs that were significantly more abundant in the DNA (present) and cDNA (active) fractions from the same samples on day 21 of the trial on BSG degradation with different inocula, as determined using LEfSe analysis. OTUs are labelled by their genera level assignment and coloured by their phylum level assignments. Points depict the average relative abundance in the three biological replicates, with abundance at the DNA level denoted with diamonds and at the cDNA level with circles

S3.2.3 Differences in microbial community structure and relationships with

biodegradation data

Ordination of samples was performed using nMDS of distance matrices made from the three data sub-sets (DNA day 0; DNA day 21 and cDNA day 21), where the stress index was acceptable for all three sub-sets (stress < 0.17). In all 3 datasets, grouping of samples was seen as function of inoculum origin (p < 0.01; AMOVA). As expected, over time the variability between biological replicates increased (more spread on DNA day 21 than day 0) except for tiger manure, and the active community

(cDNA) was more variable than the community in the DNA fraction on the same day 21. On day 0, the experimental conditions had the most effect upon the structuring of the granular sludge samples, namely pH and VFA concentrations, except acetate which had a strong effect on the rumen fluid microbial structure (*Supplementary Figure 4*). On day 21, the structure of the microbial community in both the present (DNA) and active (cDNA) fractions of the tiger manure samples were correlated with the concentration of intermediates, namely VFAs and sCOD. Meanwhile methane volume and pH correlated with both granular sludge type communities, in both DNA and cDNA fractions, as well as with select giraffe manure samples in the DNA fraction.



Supplementary Figure 4 nMDS of DNA sample dissimilarity as calculated with the Yue & Clayton metric. Clustering of samples was tested for statistical significance by analysis of molecular variance (AMOVA). Arrows represent Spearman's rank correlations of experimental variables with a significant impact (p,0.05) on structuring the communities in each of the three datasets

S3.3 Discussion

S3.3.1 Volatile solids destruction

An evident difference in the microbial community between the sludge samples and animal dungs was the presence of highly abundant OTUs assigned to the phylum of *Synergistetes*. While a relatively recent characterised phylum, members of this clade are often detected in anaerobic environments, and indeed comprise a member of the core microbiome of sludge AD systems proposed by Rivière *et al.* (2009). The *Synergistetes* are most often found in anaerobic systems that are rich in amino acids [46], such as in digesters treating protein-rich dairy wastewater from which our granular sludge was sourced. The majority of *Synergistetes* are unable to use carbohydrates (Jumas-Bilak and Marchanding, 2014), and instead obtain energy by their efficient degradation of amino acids, from which they produce short chain fatty acids and sulfate, which can then be used by methanogens and sulfate reducers (Vartoukian et al., 2007). The high protein content typically seen with this feedstock,

around 27% dry weight (Mathias et al., 2015), would therefore make it an ideal substrate for this phylum and could explain the increase in its relative abundance seen in all inocula, except tiger manure from which it was absent.

S3.3 References

- Caporaso J. G., Lauber C. L., Walters W. A., Berg-Lyons D., Huntley J., Fierer N., Owens S. M., Betley J., Fraser L., Bauer M., Gormley N., Gilbert J. A., Smith G., Knight R. (2012) "Ultra-high-throughput microbial community analysis on the Illumina HiSeq and MiSeq platforms" ISME Journal 6(8) 1621-1624. https://doi.org/10.1038/ismej.2012.8
- Griffiths R. I., Whiteley A. S., O'Donnell A. G., Bailey M. J. (2000) "Rapid Method for Coextraction of DNA and RNA from Natural Environments for Analysis of Ribosomal DNA- and rRNA-Based Microbial Community Composition" Applied and Environmental Microbiology 66(12), 5488-5491
- Jumas-Bilak E., Marchandin H. (2014) "The Phylum Synergistetes. In: E. Rosenberg, E. F. DeLong, L. Lory, E. Stackebrandt" F. Thompson, (Eds.), The Prokaryotes: Other Major Lineages of Bacteria and The Archaea (4th ed.), Springer-Verlag, Berlin, pp. 931-954. https://doi.org/10.1007/978-3-642-38954-2_384
- Kozich J. J., Westcott S. L., Baxter N. T., Highlander S. K., Schloss P. D. "Development of a Dual-Index Sequencing Strategy and Curation Pipeline for Analyzing Amplicon Sequence Data on the MiSeq Illumina Sequencing Platform" (2013) Applied and Environmental Microbiology 79 (17), 5112. https://doi.org/10.1128/AEM.01043-13.
- Mathias T.R. d. S., Alexandre V. M. F., Cammarota M. C., de Mello P. P. M., Sérvulo E. F. C. (2015) "Characterization and determination of brewer's solid wastes composition" Journal of the Institute of Brewing 121 (3), 400-404. https://doi.org/https://doi.org/10.1002/jib.229.
- de Mendiburu F., Yaseen M. (2020) "agricolae: Statistical Procedures for Agricultural Research.R package version 1.4.0" https://myaseen208.github.io/agricolae/https://cran.rproject.org/package=agricolae.
- Parada A. E., Needham D. M., Fuhrman J. A. (2016) "Every base matters: assessing small subunit rRNA primers for marine microbiomes with mock communities, time series and global field samples" Environmental Microbiology 18 (5), 1403-1414. https://doi.org/https://doi.org/10.1111/1462-2920.13023.
- R Core Team, 2019. R: A Language and Environment for Statistical Computing (Version 3.5. 2, R Foundation for Statistical Computing, Vienna, Austria).
- Rivière D., Desvignes V., Pelletier E., Chaussonnerie S., Guermazi S., Weissenbach J., Li T., Camacho P., Sghir A. (2009) "Towards the definition of a core of microorganisms involved in anaerobic digestion of sludge" The ISME Journal 3(6), 700-714. https://doi.org/10.1038/ismej.2009.2.
- Schloss P.D., Westcott S. L., Ryabin T., Hall J. R., Hartmann M., Hollister E. B., Lesniewski R. A., Oakley B. B., Parks D. H., Robinson C. J., Sahl J. W., Stres B., Thallinger G. G., Van Horn D. J., Weber C. F. (2009) "Introducing mothur: Open-Source, Platform-Independent, Community-Supported Software for Describing and Comparing Microbial Communities" Applied and Environmental Microbiology 75(23), 7537. https://doi.org/10.1128/AEM.01541-09.

- Segata N., Izard J., Waldron L., Gevers D., Miropolsky L., Garrett W. S., Huttenhower C. (2011) "Metagenomic biomarker discovery and explanation" Genome Biology 12(6) R60. https://doi.org/10.1186/gb-2011-12-6-r60.
- Vartoukian S. R., Palmer R. M., Wade W. G. (2007) "The division "Synergistes"", Anaerobe 13(3), 99-106. https://doi.org/https://doi.org/10.1016/j.anaerobe.2007.05.004.
- Wickham H. (2011) "ggplot2. WIREs Computational Statistics" 3(2), 180-185. https://doi.org/10.1002/wics.147